

# External Development Partner and International Monetary Fund Influence on Domestic Health Financing Sources in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal

Frederik Federspiel, MD, MPH

Supervisors: Josephine Borghi MSc, PhD, Melisa Martinez-Alvarez MSc, PhD, Henning Tarp Jensen MSc, PhD

Thesis submitted in accordance with the requirements for the degree of Doctor of Philosophy of the University of London December 2024

# Department of Global Health and Development Faculty of Public Health and Policy LONDON SCHOOL OF HYGIENE & TROPICAL MEDICINE

Partially funded by the Economic and Social Research Council (ESRC)

# **Declaration by candidate**

I, Frederik Federspiel, confirm that the work presented in this thesis is my own. No artificial intelligence has been used in the work, other than the unavoidable use through using online search engines for information. Where information has been derived from other sources, I confirm that this has been indicated in the thesis. I have read and understood the London School of Hygiene & Tropical Medicine (LSHTM) research practice policies. I have done my utmost to follow these and all other LSHTM academic and ethical policies and regulations. To the extent of my knowledge, the work does not infringe upon any copyright or intellectual property rights. I have endeavoured to represent my data and interpret my results as accurately as possible, while acknowledging and discussing any remaining limitations and bias in an honest manner.

Frederik Federspiel



# Abstract

#### Introduction

Equitably contributed domestic health financing is essential for making progress toward Universal Health Coverage (UHC) in Low- and Middle-Income Countries (LMICs). This PhD thesis explores how External Development Partners (EDPs) and the International Monetary Fund (IMF) influence the mix of domestic health financing sources in recipient/borrower countries through their financial and non-financial activities, including the debt and loan conditionalities that follow from loans.

#### Methods

The research applies a mixed-methods approach to study the research questions. This consists of an econometric cross-country panel data study and a case study in Senegal using key informant interviews, purposive document review and descriptive statistical analysis.

#### Results

At the global level, official development assistance for the health sector channelled via the government reduced both Government Health Expenditure as a Source (GHE-S) and Out-Of-Pocket payments (OOP) per Gross Domestic Product (GDP), while public external debt servicing per GDP was associated with a shift from GHE-S toward OOP out of all current health expenditure.

In Senegal, EDPs, including the World Bank (WB) after year 2000, and the IMF promoted GHE-S and community-based health insurance while seeking to reduce OOP. However, past WB promotion of user fees, ongoing IMF fiscal austerity measures, debt obligations and inadequate political priority may have limited GHE-S expansion and led to a continued high reliance on OOP.

## Conclusions

In countries experiencing similar or worse debt burdens than Senegal, debt relief may be a necessary policy option to help generate needed fiscal space for health. Debt-for-health-swaps should be considered to ensure funds freed up also benefit the health sector, along with dedicated health spending floors in IMF programmes.

Building on the work in this thesis, more country case studies should be performed of EDP+IMF influence on domestic health financing sources as well as the country-level impacts of public debt on domestic health financing.

# **Table of contents**

Abstract		3
Table of	contents	4
List of fig	gures	7
List of ta	ables	9
Acknow	ledgements	10
Preface.		12
List of a	bbreviations	14
1. Cha	pter 1: Introduction	16
1.1.	Background	16
1.2.	Research aims and objectives	19
1.3.	Thesis structure	19
Refere	nces	20
2. Cha	pter 2: Key concepts, terminology, levels and trends	23
2.1.	Health financing concepts, characteristics, levels and trends	23
2.2.	Development assistance definitions, terminology, levels and trends	31
2.3.	Debt terminology, levels and trends	33
2.4.	IMF/WB mandates, programmes and loan conditionality terminology	34
Refere	nces	36
3. Cha	pter 3: Literature review, hypotheses and conceptual framework	40
3.1.	Theoretical policy influence analysis frameworks	40
3.2.	EDP influence on LMIC health financing	41
3.3. financi	Development assistance for the health sector and sources of domestic ng	health 44
3.4.	Non-health development assistance and government health spending	46
3.5.	Public External Debt and Domestic Health Financing	47
3.6.	IMF programmes, loan conditionalities and domestic health financing	49
3.7.	Literature gaps	51
3.8.	Hypotheses and conceptual framework	54
3.9.	Further delineation of research topic	57
Refere	nces	58
4. Cha	pter 4: Methods	64
4.1.	Study approach	64
4.2.	Global panel data study	64
4.3.	In-depth case study in Senegal	78
4.4.	Epistemology and ontology	88
Refere	nces	89
5. Cha Covid-19	pter 5: Growing Debt Burden in Low- and Middle-Income Countries D 9 May Constrain Health Financing	)uring 97

5.1.	Abstract	
5.2.	Introduction	
5.3.	Covid-19 has led to more development lending which may increase	debt levels 101
5.4.	What do we know about the relationship between debt and health 102	n expenditure?
5.5.	Recommendations	103
Refere	nces	104
6. Cha Domesti	pter 6: Development Assistance, Public External Debt, IMF condition in the condition of the	tionalities and 108
6.1.	Chapter overview:	108
6.2.	Abstract	109
6.3.	Introduction	110
6.4.	Methods	114
6.5.	Results	122
6.6.	Discussion	125
6.7.	Conclusions	130
Refere	ences	132
7. Cha Sources	pter 7: External Development Partner Influence on Domestic Hea in Senegal: A Mixed Methods Case Study	alth Financing 139
7.1.	Chapter overview	139
7.2.	Abstract	140
7.3.	Introduction	141
7.4.	Methods	142
7.5.	Results	152
7.6.	Discussion	159
7.7.	Conclusions	164
Refere	ences	164
8. Cha Sources	pter 8: Influence of Public External Borrowing on Domestic Hea : A Case Study of Senegal	alth Financing 172
8.1.	Chapter overview	172
8.2.	Abstract	173
8.3.	Introduction	174
8.4.	Methods	176
8.5.	Results	182
8.6.	Discussion	197
8.7.	Conclusions and future research	
Refere	ences	
9. Cha	pter 9: Discussion	211
9.1.	Discussion of main findings	212
9.2.	Main thesis contributions	218

9.3.	Methodological limitations and reflections	221
9.4.	Recommendations	229
9.5.	Conclusion	237
Refere	ences	
10. A	Appendices	243
10.1.	Appendix 1: Literature review methods and detailed literature summaries.	243
Refere	ences	251
10.2.	Appendix 2: Key health indicators in Senegal	253
Refere	ences	253
10.3.	Appendix 3: Evidence of retention of copyright (Chapter 5)	254
10.4. 6	Appendix 4: Supplementary methods description, results and metadata for 257	r Chapter
Refere	ences	307
10.5.	Appendix 5: Interview supplementary materials	310
10.6.	Appendix 6: Abbreviations and detailed results table for Chapter 8	321
Refere	ences	334

# List of figures

Figure 2.1: The Universal Health Coverage Cube	28
Figure 2.2: Main health financing sources in Low- and Middle-Income Countries (LMICs),	
2000-2021	29
Figure 2.3: Government health spending per general government budget, LMICs, 2000-20	)21
Figure 2.4: Official Development Assistance (ODA) and ODA for health, 2002-2022	33
Figure 2.5: Public and Publicly Guaranteed (PPG) external debt stock and service in LMIC	s,
2000-2022	34
Figure 3.1: Main hypothesized relationships in thesis	54
Figure 4.1: PPG external debt servicing per government revenue in Senegal, 2000-2022 .	80
Figure 4.2: Regional map of Senegal	81
Figure 4.3: Main sources of health financing in Senegal, 2000-2021	82
Figure 5.1: PPG external debt stock as a proportion of Gross Domestic Product (GDP) in	
LMICs, 2000-2020	99
Figure 5.2: PPG external debt service and government health expenditure from domestic	
revenue (GHE-S) per GDP among LMICs in 20191	00
Figure 5.3: Number of LMICs that spent more on public external debt servicing than health	٦,
2000-2019	00
Figure 5.4: PPG external debt service and government health expenditure from domestic	
revenue (GHE-S) per GDP among LMICs in 20191	01
Figure 6.1: PPG external debt service out of total government expenditure in 83 LMICs	
separated by region, 2000-20211	11
Figure 7.1: Framework for analysing external development partner influence on domestic	
health financing contributions1	43
Figure 7.2: GHE-S per General Government Expenditure (GGE), per GDP, and per capita 145	÷
Figure 7.3: Sources of health financing in Senegal per current health expenditure, 2000-	
20211	46
Figure 7.4: Total ODA for health in Senegal, 2002-20221	47
Figure 7.5: Flowchart for purposive document review1	49
Figure 7.6: Main official development partners in the health sector in Senegal1	50
Figure 7.7: Government spending for sectors with available data and external health	
financing in Senegal, 2000-20211	58
Figure 8.1: PPG external debt stocks and service per GDP in Senegal compared to LMICs	3
and Sub-Saharan Africa (SSA), 2000-20221	74

Figure 8.2: Framework illustrating how lending activities from external official creditors may	y
affect domestic health financing 1	76
Figure 8.3: Health financing per capita by source in Senegal, 2000-2021 1	78
Figure 8.4: PPG borrowing, debt stock, GGE and GHE-S in Senegal, 1970-2022 1	83
Figure 8.5: Total official development loans received by sector, Senegal, 2002-2022 1	85
Figure 8.6: Share of ODA provided as loans in Senegal, 2002-2022 1	85
Figure 8.7: GGE, government revenue and GHE-S in Senegal, 2000-2022 1	87
Figure 8.8: Selected government sectoral spending and PPG external debt service in	
Senegal, 2000-2022 1	89
Figure 8.9: Total health financing sources (capital + current) in Senegal, 1981-1989 1	93
Figure 9.1: Integrated results from chapters 6-82	212
Figure 9.2: Photo of the IMF resident representative office and Ministry of Finance in	
Senegal, December, 2019	224

# List of tables

Table 4.1: Selected statistics for Senegal compared to SSA, Low-Income Countries (LICs),	
and Lower-Middle income Countries (LMCs)	)
Table 4.2: Selected socioeconomic and health system indicators for the Senegalese regions	
of Dakar and Tambacounda81	I
Table 4.3: Interview participants in case study of Senegal         83	3
Table 4.4: Example interview data framework matrix .         85	5
Table 6.1: Descriptive statistics for variables used in models.         114	ļ
Table 6.2: Two-step system GMM results for reduced form model specifications	3
Table 7.1: Overview of health financing schemes in Senegal         148	3
Table 7.2: Interview participants (Chapter 7)	I
Table 8.1: Interview participants (Chapter 8)	)
Table 10.1: Search strategies applied at the design stage of the research in 2018	1
Table 10.2: Summary of multi-country panel data studies of associations between	
development assistance for the health sector and government health expenditure	3
Table 10.3: Explanation of IMF/WB terms and abbreviations used in Chapter 8	I
Table 10.4: Overview of IMF and World Bank instruments and arrangements in Senegal	
322	

# **Acknowledgements**

The work in this thesis relies on the efforts and support of a number of people.

First of all, I would like to thank my supervisors: Dr. Melisa Martinez-Alvarez, Dr. Josephine Borghi and Dr. Henning Tarp Jensen, for staying with me through this journey, and for always helping me to do better. Your approach of challenging my thoughts and raising the bar, while always making sure there was a path forward I could follow, has helped greatly develop my thinking and my writing. I came as a doctor with an MPH, and I feel I leave as a social scientist, and that is truly your credit.

I would like to thank Dr. Elhadji Mamadou Mbaye for sharing his expertise on Senegalese health policy and its key stakeholders with me, and Dr. Timothy Powell-Jackson and Dr. Gorik Ooms for providing helpful feedback on my ideas during early stages of the research.

A heartfelt thank you to all the study participants for opening their doors to me, a Toubab with rusty French – I poorly deserved yet sincerely appreciated the kindness and patience you showed me. Your insights form the backbone of this thesis and have been deeply inspiring and enlightening to me.

I owe a debt of gratitude to Ms. Aminata Diop for her extraordinary interpretation of interviews in Dakar, and to Ms. Aminta Gueye for precise and professional transcription of interviews.

I sincerely thank the funders that believed in me and supported me to carry out this research: The Economic and Social Research Council (ESRC) Doctoral Training Partnership (DTP), Reinholdt W. Jorck og Hustrus Fond, Torben og Alice Frimodts Fond, Christian Larsen og Dommer Ellen Larsens Legat, The Anglo-Danish Society Scholarship, Den Raben-Levetzauske Fond, Christian og Ottilia Brorsons Rejselegat for yngre videnskabsmænd- og kvinder, and Carl og Ellen Hertz' legat til Dansk Læge- og Naturvidenskab.

The work in this thesis relies on the work of countless other researchers in my field, and I would like to thank all of those whose shoulders I stand on - I genuinely could not have done it without you.

To my parents: You taught me the importance of hard work and determination, and it is thanks to your unconditional love, support and guidance that I am where I am today. Thank you.

To my children: Thank you Viggo, Indigo and Leo, for pulling me out of my own head and reminding me of what is most important in life.

Dear Penny: You are my rock, and a beacon in my life. You have supported me day in and day out, lifted me up in times of despair, and shined on the horizon as a moral, empathetic and intellectual guiding star. Thank you for being you, and for always being there for me.

Finally, dear Matteo: We began this journey together, but only one of us gets to continue. The world has lost a brilliant mind, a warm heart, and a loyal friend. Nearly three years on, I still

cannot understand it. Your name brings tears to my eyes, but my heart is filled with appreciation for having known you. I dedicate this thesis to you and to my family.

# **Preface**

This thesis consists of four introductory and methods chapters, a published analytical commentary, three main results chapters, and a discussion chapter. The published commentary does not directly address the research objectives but brings attention to the topic of the growing debt burdens in Low- and Middle-Income Countries (LMICs) and how this could impact domestic government health spending, presenting a brief research agenda pursued in Chapters 6 and 8, and making some preliminary policy recommendations.

The analytical commentary (Chapter 5) and the main results chapters (Chapters 6-8) were written in 1<sup>st</sup> person plural in anticipation of journal submission and honouring the input received from supervisors and our Senegalese co-author, however they are my own work, supported by their intellectual guidance.

The work carried out for this thesis has also allowed me to make contributions to related research projects as outlined below. These do not contribute to the thesis but address related research questions.

## Published

*"Equity of resource flows for reproductive, maternal, newborn, and child health: are those most in need being left behind?"* Martinez-Alvarez M, **Federspiel F**, Singh N S, Schäferhoff M, Lewis Sabin M, Onoka C et al. BMJ 2020; 368 :m305 doi:10.1136/bmj.m305

*"The Global Financing Facility at five: time for a change?"* Seidelmann L, Koutsoumpa M, **Federspiel F**, Philips M. Sex Reprod Health Matters. 2020 Dec;28(2):1795446. doi: 10.1080/26410397.2020.1795446. PMID: 32772645; PMCID: PMC7887919.

## Forthcoming

"Tracking progress in reproductive, maternal, newborn, child and adolescent health and nutrition: the Countdown to 2030 for Women's, Children's and Adolescents' Health report by Professor Ties Boerma". The Lancet. Prof Agbessi Amouzou; Prof Aluisio J D Barros; Jennifer Requejo; Cheikh Faye; Nadia Akseer; Prof Eran Bendavid; Cauane Blumenberg; Josephine Borghi; Sama El Baz; **Frederik Federspiel**; Leonardo Z Ferreira; Elizabeth Hazel; Sam Heft-Neal; Franciele Hellwig; Li Liu; Melinda Munos; Catherine Pitt; Yushra Ribhi Shawar; Prof Jeremy Shiffman; Yvonne Tam; Neff Walker; Pierre Akilimali; Prof Leontine Alkema; Paoli Benhanzin; Peter Binyaruku; Prof Zulfiqar Bhutta; Andrea Blanchard; Hannah Blencowe; Ellen Bradley; Nouria Brikci; Beatriz Caicedo-Velásquez; Anthony Costello; Winfred Dotse-

Gborgbortsi; Prof Shams El Arifeen; Prof Majid Ezzati; Lynn P Freedman; Prof Asha George; Prof Michel Guillot; Prof Claudia Hanson; Rebecca Heidkamp; Prof Luis Huicho; Yoko Inagaki; Chimaraoke Izugbara; Safia Jiwani; Caroline Kabiru; Helen Kiarie; Mary Kinney; Fati Kirakoya-Samadoulougou; Prof Joy Lawn; Amnesty Lefevre; Nyovani Madise; Roland Gouda Mady; Bruno Masquelier; Dessalegn Melesse; Kristine Nilsen; Jamie Perin; Prof Usha Ram; Marina Romanello; Ghada E Saad; Sudha Sharma; Estelle Sidze; Prof Paul Spiegel; Hannah Tappis; Andrew J Tatem; Prof Marleen Temmerman; Prof Cesar G Victora; Francisco Villavicencio; Yohannes Wado; Peter Waiswa; Prof Jon Wakefield; Shelley Walton; Danzhen You; Mickey Chopra; Prof Robert E Black; Ties Boerma

# List of abbreviations

2SLS	Two-stage least squares
ABBB	Arellano-Bover/ Blundell-Bond
AFDB	The African Development Bank
ANSD	National Agency of Statistics and Demography
BLUE	Best Linear Unbiased Estimator
BMGF	Bill and Melinda Gates Foundation
BRD	Battle-Related Death
СВНІ	Community-Based Health Insurance
CHE	Current Health Expenditure
CPF	Country Partnership Framework
CRS	Creditor Reporting System
DAC	Development Assistance Committee
DAH	Development Assistance for Health
DAHG	Development Assistance for Health to Government
DAHNG	Development Assistance for Health to Non-Government
DHS	Demographic and Health Surveys
DRM	Domestic Resource Mobilisation
EDP	External Development Partner
FE	Fixed-Effects
GHED	Global Health Expenditure Database
GFATM	Global Fund to Fight AIDS, Tuberculosis and Malaria
GDP	Gross Domestic Product
GGD	General Government Gross Debt
GGHE	General Government Health Expenditure
GGHE-D	Domestic General Government Health Expenditure
GGE	General Government Expenditure
GHE	Government Health Expenditure
GHE-A	Government Health Expenditure as Agent
GHE-S	Government Health Expenditure as Source
GLS	Generalized Least Squares
GMM	Generalized Method of Moments
GNI	Gross National Income
HIC	High-Income Country
HIPC	Heavily Indebted Poor Countries
HSS	Health Systems Strengthening
IBRD	International Bank for Reconstruction and Development
IDA	International Development Association
IFI	International Financial Institution
IHME	Institute for Health Metrics and Evaluation
IMF	International Monetary Fund
LDC	Least Developed Country
LMC	Lower-Middle income Country
LMICs	Low- and Middle-Income Countries

LSHTM	London School of Hygiene & Tropical Medicine
LTEF	Long-Term Expenditure Framework
MDG	Millennium Development Goal
MDRI	Multilateral Debt Relief Initiative
MLE	Maximum Likelihood Estimation
MoF	Ministry of Economy and Finance (Senegal)
MHSA	Ministry of Health and Social Action (Senegal)
MNCH	Maternal and Child Health
MTEF	Medium-Term Expenditure Framework
NGO	Non-Government Organization
NHA	National Health Accounts
NPISH	Non-Profit Institutions Serving Households
ODA	Official Development Assistance
ODA+ for health	Official Development Assistance+ for health
OECD	Organisation for Economic Co-operation and Development
PCI	Policy Coordination Instrument
PPG	Public and Publicly Guaranteed
PRGF	Poverty Reduction and Growth Facility
PRSP PSI	Poverty Reduction Strategy Paper Policy Support Instrument
RMNCH	Reproductive, Maternal, Neonatal and Child Health
RCF	Rapid Credit Facility
RFI	Rapid Financing Instrument
RSF	Resilience and Sustainability Facility
SRMNCAH	Sexual, Reproductive, Maternal, Neonatal, Child and Adolescent Health
SAP	Structural Adjustment Program
SBA	Stand-By Arrangement
SCF	Standby Credit Facility
SDR	Special Drawing Rights
SHA	System of Health Accounts
SHI	Social Health Insurance
SSI	Semi-Structured Interview
UHC	Universal Health Coverage
UN	United Nations
VAT	Value-Added Tax
VHI	Voluntary Health Insurance
VHPS	Voluntary Health care Payment Schemes
WHO	World Health Organisation
WB	World Bank
UHC	Universal Health Coverage
UN	United Nations

# 1. Chapter 1: Introduction

This chapter presents relevant backgrund for the thesis, positions the research in relation to recent developments in global health and development concepts, to then describe the aims, objectives and structure of the thesis.

# 1.1. Background

1.1.1. Universal Health Coverage through equitable domestic health financing Universal Health Coverage (UHC) is characterized by all members of a country population having access to quality health services where and when they need them, in a manner that protects them from having to individually carry the burden of payment for their own care (1). UHC, including financial risk protection, i.e. protection from the risk of suffering financial hardship when seeking health services (2), forms a cornerstone of the health Sustainable Development Goals (SDGs, goal 3.8) (3), and has become increasingly recognized as an essential development priority (4). A necessary step in making progress toward UHC is that countries finance their health systems in an equitable manner, improving health service access in an affordable way, including for poor and vulnerable groups, and achieving lasting public health improvements without causing financial hardship to individuals and households (5, 6). This latter dimension necessitates a degree of progressivity of financing, whereby contributions made to the health system align with people's ability to pay (4, 7-15). Tax-based contributions from the government tend to be more progressive than e.g. user fees at the point of care, as it is most often raised as a proportion of income whereas user fees do not take into account a person's income or other financial means (4). The reliance on different types of health financing sources in a country therefore matters in its pursuit of UHC. Do countries pursue a path with a high reliance on user fees at the point of care, or Out Of Pocket Payments (OOP), do they manage to increase the more progressive source of health financing from domestic government revenue (Government Health Expenditure as a Source, GHE-S), or do they pursue other paths such as Social Health Insurance (SHI) or Voluntary Health Insurance (VHI)? For Low- and Middle-Income Countries (LMICs), these paths sit in a complex political, macroeconomic and fiscal context, where among other factors, External Development Partners (EDPs)<sup>1</sup> and International Financial Institutions (IFIs) (e.g. the International Monetary

<sup>&</sup>lt;sup>1</sup> For clarity, the term EDP used in this thesis covers all official international organisations with a development mandate, both bilateral e.g. the UK Department for International Development, and multilateral e.g. UNICEF. NGOs and private philanthropies are generally excluded, except for the case of the Bill and Melinda Gates Foundation (BMGF) due to its shere size and impact in the health sector. The WB thus falls under both the categories of being an EDP and an IFI, while the IMF technically does not have a development mandate and thus cannot be classified as an EDP, although it arguably works at a central, macroeconomic level that strongly co-determines the economic

Fund (IMF), the World Bank (WB), the African Development Bank (AfDB) and others) disburse development assistance and non-concessional loans and provide technical advice to governments. It is at this intersection, between key external official bilateral and multilateral institutions and the domestic government and country health system that this research is situated.

#### 1.1.2. External partners and domestic health financing

A major part of the financial transactions made with EDPs + IMF are as concessional and nonconcessional loans that all need repayment, leading to a financial outflow from LMICs that exceeds the original inflow due to interest accumulation (and a net real-term outflow if interests exceed inflation). Debt servicing levels in particular have expanded substantially over the past decade (16). This has led to growing concerns over governments' ability to finance public sectors including the health sector, i.e. overall fiscal space, defined by the World Health Organisation (WHO) as "the budgetary room allowing a government to provide resources for public purposes without impacting fiscal sustainability", and the fiscal space for health, threatening progress toward UHC (17-22). In 2023, 55% of Least Developed Countries (LDCs) and other LICs were assessed as being at high risk of, or in, debt distress (20). While a significant proportion of the Public and Publicly Guaranteed (PPG) external debt in the world's poorest countries, i.e. debt owed by the government and its public bodies as well as private debt guaranteed by public bodies to external creditors<sup>2</sup> (16, 23), derives from private creditors (21% in 2022 among WB International Development Association (IDA)-eligible countries), most of it comes from external official creditors (79% in 2022) (18). With their financial assistance including loans that require repayment from government budgets, are EDPs indeed supporting LMICs to pursue development paths characterized by growing GHE-S, supporting progress on UHC?

As mentioned, assistance from the EDPs + IMF is also provided in the form of technical advice, policy advice, capacity building and more. As a distinguishing feature from other external official creditors, the IMF and WB provide their loans under programs that come with extensive policy recommendations and loan conditionalities relating to the fiscal, macroeconomic, and sectoral level (WB being the key sectoral policy advisor of the two) (24, 25). Developed in collaboration with borrower governments, these programs constitute an overall macroeconomic framework setting the boundaries and trajectories for participating LMIC

development trajectory of its debitor countries. The term EDPs + IMF is therefore used to group the main external official partners of LMICs in the development and financial sphere.

<sup>&</sup>lt;sup>2</sup> WB data does not count debt to the IMF as part of PPG external debt, but refers to this as the separate category: "Use of IMF credit and special drawing right allocations" (18).

economies. This gives the IMF and WB a privileged and powerful position in the centre of domestic policy making in borrower LMICs, making them core candidates for scrutiny of external influences on domestic health financing policy, influence that has been both alleged and observed in many contexts (26-40).

To understand both the financial and non-financial influence of EDPs + IMF on domestic health system financing in this thesis, the question is therefore further asked whether the non-financial interactions, policy recommendations and conditionalities that follow with external financing, affect how countries finance their health systems, the fiscal space available for health, and the mix of health financing sources, with ensuing implications for the progress toward UHC.

A dedicated exploration of the relationships between external financial assistance from EDPs + IMF, their policy influence, and domestic health sector financing, may be important to identify which components of their activities support or hinder progress toward health financing being more equitably contributed with adequate financial risk protection in pursuit of UHC. In particular, the growing debt burdens experienced by many LMICs renders the exploration and understanding of these relationships ever more topical.

#### 1.1.3. Universal health coverage and domestic resource mobilisation

The global development community has increasingly focused its attention on achieving sustainable development and UHC through developing countries' own resources. This is mainly referred to as Domestic Resource Mobilization (DRM), defined by the WB as "increasing the flow of taxes and other income into government treasuries" i.e. increasing public revenue (41). This expands overall fiscal space and hence fiscal space for health, provided the health sector is prioritised and benefits from increases in domestic revenue. If increased domestic revenue is allocated to the health sector, this should help improve the degree of progressivity of health financing, enhancing equity of health financing contributions at the macro-level in terms of to what degree financial contributions are made in proportion to ability to pay (42), ceteris paribus. A 2017 WB report outlining a series of strategies for strengthening DRM stated that: "Domestic resources are central to achieving development objectives" (43). The United Nations (UN) has convened in a series of meetings on international development financing, first in Monterrey, Mexico in 2002, then in Doha, Qatar in 2008, and then in Addis Ababa, Ethiopia in 2015 (44-46). Through these meetings, the concept of DRM has received increasing mention and has become a central component to achieving sustainable development (44-46). To help translate this agenda into action, the Addis Ababa Action Agenda was followed up by the Addis Tax Initiative, supported by more

than 50 country governments, aiming to increase DRM and the effective use of these resources through taxation mechanisms (47-50). In a similar vein, the Organization for Economic Co-Operation and Development (OECD) with its Development Assistance Committee (DAC) made up by 31 of the world's major donor countries plus the European Union (51), has established a "Global Partnership for Effective Development Co-operation" (2014) (52). This succeeded a series of High Level Fora on Aid Effectiveness between 2003-2011 (53-55), and it arrived at the conclusion that "*no country should be dependent on others*" *resources for its own development*" (48, 52, 56), thus highlighting the need for mobilization of domestic resources and envisaging a global transition beyond development cooperation mentioned that development co-ordination should catalyse DRM and effective DRM being crucial (57). In light of the growing emphasis on DRM, an improved understanding of whether and how EDPs+IMF are indeed supporting countries to make the proposed transition and fund health services from tax-based domestic revenue without patients suffering financial hardship at the point of care, in the pursuit of UHC, seems warranted.

The aims and objectives for this research are as follows.

# **1.2.** Research aims and objectives

This research aims to investigate how EDPs + IMF affect the mix of domestic health financing sources in recipient/borrower countries. It uses a cross-country panel data study and a case study of Senegal to pursue the below objectives.

- To investigate the relationships between development assistance, public external debt and IMF loan conditionalities and the levels and balance of Government Health Expenditure as a Source (GHE-S) and Out-Of-Pocket payments (OOP) among 105 LMICs from 2005-2019 (Chapter 6).
- 2. To explore mechanisms of development partner influence on the mix of domestic health financing sources present in Senegal (Chapter 7).
- To explore how external official lending, public external debt obligations and IMF/WB policy recommendations and loan conditionalities affect the mix of domestic health financing sources in Senegal (Chapter 8).
- 4. To develop recommendations for policy and for future research based on the study findings (Chapters 6, 7, 8 and 9)

# **1.3.** Thesis structure

The remainder of the thesis is structured as follows:

- Chapter 2 introduces key concepts, terminology and variables used in this thesis, including their levels and trends.

- Chapter 3 provides an overview of the literature underlying the research topic, identifies research gaps, and explains the economic hypotheses to be explored in this thesis and the conceptual framework guiding this exploration as informed by the literature.
- Chapter 4 describes the methods used in this thesis and the epistemological position of the research.
- Chapter 5 presents a comparative analysis of debt servicing and government health expenditure in LMICs.
- Chapter 6 presents the results of a multi-country panel data analysis of the associations between development assistance, public external debt, IMF loan conditionalities, and GHE-S and OOP in 105 LMICs from 2005-2019.
- Chapter 7 presents the results of a mixed-methods case study investigating development partner mechanisms of influence on the mix of health financing sources in Senegal.
- Chapter 8 presents the results of a mixed-methods case study investigating how development loans, public external debt and IMF/WB policy recommendations and conditionalities have affected the different domestic health financing sources present in Senegal.
- Chapter 9 brings together and discusses the key findings in chapters 5-8, discusses limitations of the research, and makes recommendations for policy and future research based on the study findings.

# References

1. WHO. Universal health coverage. 2023 [cited 2023 August 2]. Available from:

https://www.who.int/health-topics/universal-health-coverage#tab=tab\_1.

2. Saksena P, Hsu J, Evans DB. Financial risk protection and universal health coverage: evidence and measurement challenges. PLoS Med. 2014;11(9):e1001701.

3. United Nations. Goal 3: Ensure healthy lives and promote well-being for all at all ages. 2023 [cited 2023 December 18]. Available from: <u>https://www.un.org/sustainabledevelopment/health/</u>.

4. UHC2030. A History of Universal Health Coverage in the UN. 2019 [cited 2024 May 21]. Available from: https://www.uhc2030.org/un-hlm-2019/a-history-of-universal-health-coverage-in-the-un/.

6. WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available from: <a href="https://www.who.int/publications/i/item/9789241564021">https://www.who.int/publications/i/item/9789241564021</a>.

8. Asante A, Price J, Hayen A, Jan S, Wiseman V. Equity in Health Care Financing in Low- and Middle-Income Countries: A Systematic Review of Evidence from Studies Using Benefit and Financing Incidence Analyses. PLoS One. 2016;11(4):e0152866.

9. Ataguba JE, Asante AD, Limwattananon S, Wiseman V. How to do (or not to do) ... a health financing incidence analysis. Health Policy Plan. 2018;33(3):436-44.

11. Martinez-Alvarez M, Federspiel F, Singh NS, Schaferhoff M, Lewis Sabin M, Onoka C, et al. Equity of resource flows for reproductive, maternal, newborn, and child health: are those most in need being left behind? BMJ. 2020;368:m305.

<sup>5.</sup> McIntyre D, Meheus F, Rottingen JA. What level of domestic government health expenditure should we aspire to for universal health coverage? Health Econ Policy Law. 2017;12(2):125-37.

<sup>7.</sup> Kwesiga B, Ataguba JE, Abewe C, Kizza P, Zikusooka CM. Who pays for and who benefits from health care services in Uganda? BMC Health Serv Res. 2015;15:44.

<sup>10.</sup> Kakwani NC. Measurement of Tax Progressivity: An International Comparison. The Economic Journal. 1977;87(345).

12. Gini C. On the Measure of Concentration with Special Reference to Income and Statistics. Colorado College Publication, General Series. 1936;No. 208:73–9.

13. Cheng Q, Asante A, Susilo D, Satrya A, Man N, Fattah RA, et al. Equity of health financing in Indonesia: A 5-year financing incidence analysis (2015-2019). Lancet Reg Health West Pac. 2022;21:100400.

14. Yu CP, Whynes DK, Sach TH. Equity in health care financing: The case of Malaysia. Int J Equity Health. 2008;7:15.

15. Binyaruka P, Martinez-Alvarez M, Pitt C, Borghi J. Assessing equity and efficiency of health financing towards universal health coverage between regions in Tanzania. Social science & medicine (1982). 2024;340:116457.

16. World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: https://data.worldbank.org/.

17. The Lancet. Economic storms threaten to cast health adrift. Lancet. 2024;403(10424):325.

18. Bank W. International Debt Report 2023. 2023 [cited 2024 February 22]. Available from: https://www.worldbank.org/en/programs/debt-statistics/idr/products.

19. Fan V, Gupta S. What's Rising Debt Got to Do With Health Spending? 2023 [cited 2023 September 27]. Available from: https://www.cgdev.org/blog/whats-rising-debt-got-do-health-spending.

20. United Nations DESA FSDO. Financing for Sustainable Development Report 2024. 2024 [cited 2024 April 17]. Available from: <u>https://desapublications.un.org/publications/financing-sustainable-development-report-2024</u>.

21. Ahazie J, Riaz S, Fan V. Universal Health Coverage in Time of Crises: Don't Abandon the Ship to Avoid the Storm. 2023 [cited 2024 May 21]. Available from: <u>https://www.cgdev.org/blog/universal-health-coverage-time-crises-dont-abandon-ship-avoid-storm</u>.

22. WHO. Fiscal space for health. 2024 [cited 2024 February 15]. Available from:

https://www.who.int/teams/health-systems-governance-and-financing/health-financing/policy/fiscal-space-for-health.

23. World Bank. International Debt Statistics. 2024 [cited 2024 April 24]. Available from:

https://databank.worldbank.org/source/international-debt-statistics#

24. IMF. IMF Conditionality. 2023 [cited 2023 December 18]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/IMF-Conditionality.

25. World Bank. Development Policy Retrospective 2021. 2023 [cited 2023 December 18]. Available from: https://documents1.worldbank.org/curated/en/099623509132210285/pdf/IDU0249804670b2fc0466f083850d1aad 1818915.pdf.

26. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

27. Akazili J, McIntyre D, Kanmiki EW, Gyapong J, Oduro A, Sankoh O, et al. Assessing the catastrophic effects of out-of-pocket healthcare payments prior to the uptake of a nationwide health insurance scheme in Ghana. Glob Health Action. 2017;10(1):1289735.

28. Mhazo AT, Maponga CC. The political economy of health financing reforms in Zimbabwe: a scoping review. Int J Equity Health. 2022;21(1):42.

29. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

30. Ooms G, Schrecker T. Expenditure ceilings, multilateral financial institutions, and the health of poor populations. The Lancet. 2005;365(9473):1821-3.

31. Center for Global Development. Does the IMF Constrain Health Spending in Poor Countries? . 2007 [cited 2024 March 19]. Available from: https://www.cgdev.org/sites/default/files/14103\_file\_IMF\_report.pdf.

32. Brunswijck G. Unhealthy Conditions. Eurodad; 2018.

33. Stuckler D, Basu S. The International Monetary Fund's effects on global health: before and after the 2008 financial crisis. Int J Health Serv. 2009;39(4):771-81.

34. Stuckler D, Basu S. The Body Economic: Why Austerity Kills: Basic Books; 2013. ISBN: 978-0465063987

35. Abocejo F. The Impact of International Monetary Fund (IMF) Structural Adjustment Policies (SAP) on the Philippines. Recoletos Multidisciplinary Research Journal. 2014;2(1):19-24.

36. Isiani MC, Anthonia Obi-Ani N, Obi-Ani P, Chidume CG, Okoye-Ugwu S. Interrogating the International Monetary Fund (IMF) Policies in Nigeria, 1986–2018. Cogent Arts & Humanities. 2021;8(1).

37. Meurs M, Seidelmann L, Koutsoumpa M. How healthy is a 'healthy economy'? Incompatibility between current pathways towards SDG3 and SDG8. Global Health. 2019;15(1):83.

38. Birn AE, Nervi L, Siqueira E. Neoliberalism Redux: The Global Health Policy Agenda and the Politics of Cooptation in Latin America and Beyond. Development and Change. 2016;47(4):734-59.

39. Stubbs T, Kentikelenis A. Targeted social safeguards in the age of universal social protection: the IMF and health systems of low-income countries. Critical Public Health. 2017;28(2):132-9.

40. Foley EE. Your pocket is what cures you: The Politics of health in Senegal: Rutgers University Press; 2010. ISBN: 978-0-8135-4667-4

41. World Bank. Domestic Resource Mobilization. 2016 [cited 2018 September 2]. Available from: http://www.worldbank.org/en/topic/governance/brief/domestic-resource-mobilization. 42. Culyer A. Efficiency, equity and equality in health and health care. CHE Research Paper 120: University of York; 2015. ISBN:

43. World Bank. Strengthening Domestic Resource Mobilization: Moving from Theory to Practice in Lowand Middle-Income Countries. Washington DC, USA: World Bank; 2017.

44. United Nations. Monterrey Consensus of the International Conference on Financing for Development. Monterrey, Mexico: United Nations; 2002.

45. United Nations. Doha Declaration on Financing for Development. Doha, Qatar: United Nations; 2008.
 46. United Nations. Addis Ababa Action Agenda of the Third International Conference on Financing for

46. United Nations. Addis Ababa Action Agenda of the Third International Conference on Financing for Development (Addis Ababa Action Agenda). New York, USA: United Nations; 2015.

47. International Tax Compact. Addis Tax Initiative: ATI Monitoring Brief 2016. Bonn, Germany; 2018.

48. United Nations. Financing for Development Conference: The Addis Tax Initiative - Declaration. 2016.
49. Addis Tax Initiative. Members. 2018 [cited 2018 July 23]. Available from:

https://www.addistaxinitiative.net/#slider-4.

50. Addis Tax Initiative. ATI Declaration 2025. 2020 [cited 2023 December 18]. Available from:

https://www.addistaxinitiative.net/resource/ati-declaration-2025

51. OECD. DAC members. 2018 [cited 2018 September 2]. Available from:

http://www.oecd.org/dac/dacmembers.htm.

52. Global Partnership for Effective Development Co-operation. First high-level meeting, Mexico City, 15 - 16 April 2014. 2014 [cited 2018 June 26]. Available from: <u>http://effectivecooperation.org/events/1st-high-level-meeting/</u>.

53. OECD. Rome Declaration on Harmonisation. 2003. Available from:

https://www.oecd.org/dac/effectiveness/31451637.pdf.

54. OECD. The Paris Declaration on Aid Effectiveness and the Accra Agenda for Action. 2005, 2008 [cited 2024 July 29]. Available from: <u>https://web-archive.oecd.org/temp/2021-08-02/73869-</u>

parisdeclarationandaccraagendaforaction.htm.

55. OECD. Busan Partnership for Effective Development Co-operation. 2011 [cited 2024 July 29]. Available from: <u>https://web-archive.oecd.org/temp/2023-10-09/57958-busanpartnership.htm</u>.

56. OECD. DACnews July 2014. 2014 [cited 2018 June 26]. Available from:

http://www.oecd.org/dac/dacnewsjuly2014.htm.

57. Global Partnership for Effective Development Co-operation. 2022 Effective Development Co-operation Summit Declaration. 2022 [cited 2023 December 18]. Available from:

https://www.effectivecooperation.org/system/files/2022-12/Final%20Outcome%20Document.pdf.

# 2. Chapter 2: Key concepts, terminology, levels and trends

Some of the terminology used in this thesis has been introduced in the previous chapter. This chapter introduces further relevant key concepts, variables and associated terminology for health financing, development assistance, and International Monetary Fund (IMF) and World Bank (WB) programmes and loan conditionalities, including their levels and trends.

# 2.1. Health financing concepts, characteristics, levels and trends

## 2.1.1. Main health financing functions

The three main functions of health financing have been conceptualised as A: revenue collection, B: pooling, and C: purchasing (1-3). Revenue collection refers to the way in which money is raised for the health system, e.g. from tax contributions via the government, to insurance schemes as premiums, or directly at the point of care (2, 3), as explained in the following sections. Pooling refers to the collection of contributed funds in a shared pool (2, 3). This serves functions of sharing financial risk across individuals should a member of the pool fall ill, and if funds are contributed progressively i.e. according to ability to pay, a funding pool may also serve to redistribute funds from wealthier to poorer pool members, referred to as so-called cross-subsidy (2, 4, 5). Purchasing refers to the payment or funding allocation to health service providers, which in turn determines which health services are available to whom (2, 3).

This thesis relates to the first two main functions of health financing: First and foremost from which sources revenue is collected but also to what extent it is pooled, as the different main sources of financing span from no pooling to national-level pooling (Section 2.1.3). These functions in turn relate to concepts of equity and progressivity as explained below.

## 2.1.2. Conceptualisation of health financing equity and progressivity

Equity has traditionally been used to describe the concepts of "fairness", "justice" or "due proportion", as opposed to the concepts described by equality of "sameness" or "equal distribution" (6). The concept of due proportion implies an unequal treatment in proportion to some quality that is thought to hold moral value. In 1875, Karl Marx popularised the slogan: *"From each according to his ability, to each according to his needs*" (7). Scottish philosopher DD Raphael wrote in 1946: *"Unequal treatment, then, is justified when the general claim to equality is overridden by an additional special claim*", and "*special need gives rise to a special moral claim*" (8). Health economists Culyer and Wagstaff concurred with the notion of unequal treatment for unequal moral claims or needs in relation to health care in 1994: *"the principle*"

of distribution according to need calls for unequal expenditures" in favour of those with higher needs (9). For contributions, Culyer wrote in 2015 that equity implies: *"lower/higher contributions from households with a lower/higher ability to pay*" (10). This is captured in Marx' slogan: From each according to ability (7, 10). Culyer and Wagstaff further define this as *"vertical equity": "unlike treatment of unlike individuals*", as opposed to "horizontal equity": *"like treatment of like individuals*" (9, 10). These concepts originate from Aristotle's Nicomachean Ethics (11). This thesis follows these concepts and defines vertical equity as unequal contribution according to ability to pay. Such contributions can be made either proportionally or progressively, meaning either in direct proportion to ability to pay, or as a growing proportion of ability to pay with higher ability to pay (12, 13). Horizontal equity is relevant for health financing allocations - equal financing allocations for people with equal need - which this thesis does not consider.

Financing progressivity is usually assessed using Financing Incidence Analysis (FIA), where vertical equity is directly measured at the household level by comparing burden of payment with people's ability to pay (e.g. (13-21)). This thesis does not undertake FIA, but instead explores the influence of development partners on the mix of health financing sources at the macroeconomic level and uses that to consider the possible implications for vertical equity, given our understanding of the degree of progressivity of different key health financing sources. Tax-based contributions for instance tend to have a degree of proportionality with ability to pay or may even be contributed progressively e.g. as tiered income tax, while user fees charged at the point of care when one has fallen ill do not take into consideration a person's ability to pay for the health services needed (22).

Prepayment describes the payment of funds before they are needed. It is an essential component of equitable health financing contributions (22): If funds are not prepaid, the burden of payment will fall on the patient at the point of care, i.e. user fees. This implies payment irrespective of ability to pay, which also implies a disproportionate placement of the burden of payment for health services on the poor due to the positive link between poverty and need for health care (23).

Prepayment allows for pooling of contributions. The more members of a pool, the larger the degree of financial risk sharing, meaning when one person falls ill, there are more people to carry the burden of payment (24, 25). Pools can vary in the number of members, the amount of funds in a pool, and the scale of pooling (from community to national level). Cross-subsidy, i.e. the redistribution of funds from rich to poor, occurs when individuals across the income and wealth spectra are part of the same pool and contribute according to ability to pay (4).

Due to the positive links between income/wealth and health, this usually also implies risk sharing via the healthy paying for health services for the sick (23).

In summary, equitable health financing contributions are in this thesis defined as being according to ability to pay, funds being prepaid and pooled (the larger the pool, the better). This entails cross-subsidy and financial risk-sharing between wealthier and poorer pool members. A higher degree of financing progressivity is considered more equitable.

#### 2.1.3. Main health financing sources

Health systems are financed by the following five main sources of funding, as categorised by the World Health Organisation (WHO) System of Health Accounts (SHA) (26) and instrumentalised in the Global Health Expenditure Database (GHED) (27). A: Government domestic revenue, which refers to all government transfers made to the health sector that are domestic in origin, i.e. from taxes, fees, levies, revenue from state enterprises, etc. B: Social Health Insurance (SHI) contributions from employers and employees. C: Voluntary prepayments to Voluntary Health Insurance (VHI) schemes, which broadly correspond to Voluntary Healthcare Payment Schemes (VHPS). D: Out Of Pocket payments (OOP), e.g. user fees and other patient-borne costs at the point of care. E: External health financing for aid-recipient countries, which may be given in the form of development assistance for the health sector, and can go either through the government or be channeled directly to a health institution or programme. Governments may co-finance social health insurance schemes and voluntary health insurance schemes, which the WHO SHA and GHED take into account in the joint indicator: Domestic General Government Health Expenditure (GGHE-D). This is the standard measure for a government's own financing of the health sector, and can also be referred to as Government Health Expenditure as Source (GHE-S).

Domestic government financing has the highest degree of pooling of funds and thus the highest level of financial risk sharing, and it is considered the most equitable and progressive of these main sources (28). While the composition of tax revenue sources varies greatly across Low- and Middle-Income Countries (LMICs), the principal sources of taxation - personal and corporate income and wealth taxes, Value-Added Tax (VAT), and trade taxes (29) – tend to be levied from the formal sector, with the poorest members of society, the unemployed, smallhold farmers and informal sector workers, mainly contributing in the form of VAT on part of their consumption<sup>3</sup>. This means that in terms of contributions, the burden of

<sup>&</sup>lt;sup>3</sup> Key consumption goods such as staples, bread, milk etc. are often exempt from VAT, and in subsistence contexts, consumption may be lower among poorer people, which leads to some progressivity of VAT.

payment for domestic government health financing is larger for wealthier individuals (progressivity), and that money is redistributed from those with more means to those with less (cross-subsidy) (4, 5).

SHI schemes usually start with formal sector workers with various degrees of expansion beyond this group, with different models of co-contribution from employer, employee and the government. The blend of financing sources will determine the overall progressivity of this source, but the pools of these schemes are limited to different groups of employees, that will generally tend to be financially better off than informal sector workers, unemployed, children, or retirees (if not covered based on past employment). This means that the level of cross-subsidy from rich to poor and from healthy to sick may be limited, while the level of financial risk-sharing extends to the given size of the pool. These varying characteristics lead to variations in how equitable different SHI schemes can be considered.

VHI or VHPS may, in relatively wealthier middle-income countries, take the form of private health insurance mainly purchased by growing upper- and middle classes that seek higher quality or faster services than what the public system has to offer (30). In the poorest settings, VHI/VHPS are constituted mainly by Community-Based Health Insurance (CBHI) schemes to help cover user fees for health care (30). Funds are thus pooled, but pools are typically smaller as they are locally held with lower financial risk sharing than for domestic government health financing. As participants in such schemes tend to be from comparable socioeconomic strata, and sometimes within a given geographical area, there is a relatively lower level of cross-subsidy both between socioeconomic strata and geographical areas in VHI/VHPS as compared to GHE-S. The degree of progressivity of financing is limited by the variation in ability to pay between members, and other considerations of relevance to equity of health financing contributions such as pooling, prepayment and associated financial risk protection are also variable.

OOP is the most regressive and least equitable form of financing, where the same costs are borne by any health system user irrespective of that user's ability to pay (13), funds are not prepaid and financial risk is not pooled (28). OOP has been shown to cause financial catastrophe for households, push people into poverty, exacerbate financial inequality and lead to inequitable health service access and lower service use among poorer populations (22, 28, 31-34). OOP has been associated with adverse health outcomes when making up a larger proportion of total health spending (35, 36). Reducing user charges has been associated with improved health outcomes, particularly among children and low-income groups (37), while expansion of government health spending has been associated with mortality reductions, especially among children under 5 (35, 36).

Development assistance for the health sector can arguably be seen as a highly progressive transfer of funds from wealthier tax-payers in donor-countries to LMICs (17). However, in line with the global move toward supporting countries to be able to finance their health systems from domestic funds and eventually become aid-independent (Chapter 1), this research focuses on what happens to the mix of *domestic* health financing sources when external health financing is received.

#### 2.1.4. Universal Health Coverage

Universal Health Coverage (UHC) has three main dimensions, as illustrated in Figure 2.1: Providing more health services of better quality; to a greater proportion of the population in need; while increasing the level of financial risk protection (38). From a health financing perspective, aside from efficiency gains and health service price decreases, providing more services of higher quality to more people generally requires more funds, which due to the critical component of increased financial risk protection have to be prepaid, contributed according to ability to pay, and pooled (39). Pooled health financing has been found to be a strong driver of progress on UHC in LMICs, along with efficiency gains and contextual factors (40). In terms of the mix of domestic health financing sources in LMICs, these changes correspond to lowering reliance on OOP while expanding more equitable health financing sources such as GHE-S, SHI, or to some extent VHI, although the latter two may entail varying degrees of cost sharing. This thesis does not examine service or population coverage, but rather it focuses on the financial foundation for expanding these in an equitable manner whereby prepaid, pooled and progressively contributed funding is increased and cost sharing is reduced.



Figure 2.1: The Universal Health Coverage (UHC) Cube (38). Figure reproduced with permission from the World Health Organization (WHO) from the webpage: "What are the Overall Principles of HBP Design?", available at <u>https://www.who.int/teams/health-financing-and-economics/economic-</u> <u>analysis/health-technology-assessment-and-benefit-package-design/resource-guide-for-the-use-ofhta-and-hbp-design-processes/what-are-the-overall-principles-of-hbp-design, accessed on June 21, 2024. No modifications made. Website copyright 2024.</u>

## 2.1.5. LMIC health financing levels and trends

There is profound inequality in Current Health Expenditure (CHE) between country income groups, both in absolute terms and per capita (constant 2021 US\$) (Figure 2.2). In 2021, CHE was \$23 billion in Low-Income Countries (LICs), \$371 billion in Lower-Middle income Countries (LMCs), and \$1631 billion in Upper-Middle Income Countries (UMICs). Per capita, CHE was \$38 in LICs, \$109 in LMCs, and \$646 in UMICs. Relative to Gross Domestic Product (GDP), CHE corresponded to 5.3% in LICs, 4.1% in LMCs, and 6.0% in UMICs.

In all country income groups, CHE has expanded roughly four-fold between 2000-2021. Per capita, this has corresponded to an approximate doubling in LICs and LMCs and a tripling in UMICs. Relative to GDP, growth has been more limited. Over the same time period, CHE in LICs has grown from 4% to 5.3% of GDP, increased slightly in LMCs from 3.7% to 4.1%, and grown in UMICs from 5.3% to 6% in UMICs.





There are clear differences in the composition of CHE between country income groups. In LICs, both Out Of Pocket payments (OOP) and Government Health Expenditure as a Source (GHE-S) have expanded slightly per capita and per GDP, but the main addition of health financing has come from external sources. GHE-S has decreased as a share of CHE from 33% in 2000 to 23% in 2021, and OOP/CHE has decreased from 52% 2000 to 39% in 2021,

mainly due to expansion of external health financing. Voluntary Health Insurance (VHI) and Social Health Insurance (SHI) are still marginal.

In LMCs, expansion of GHE-S has been stronger than that for OOP, resulting in a growing share of CHE from 32% in 2000 to 41% in 2021, while OOP/CHE has decreased from 58% in 2000 to 45% in 2021. VHI and SHI is more prominent, and external health financing less so. In UMICs, both GHE-S and SHI have expanded strongly, and more so than the expansion of

OOP, resulting in GHE-S reaching 56% of CHE in 2021 from 40% in 2000, and a contraction of OOP as a share of CHE from 43% in 2000 to 31% in 2021.

## 2.1.5.1. Government health spending levels in perspective

GHE-S across LMICs is insufficient for serving the health needs of country populations (27, 41-44). The most recent data on domestic spending levels fall far short of the 2001 Abuja declaration pledge by African Union countries to spend 15% of government budgets on the health sector (45), both when looking only at GHE-S and when including external transfers distributed by governments (on-budget development assistance for the health sector) (27) (Figure 2.3).



Figure 2.3: Solid lines: Government Health Expenditure as a Source (GHE-S) as a fraction of General Government Expenditure (GGE); Dotted lines: GHE-S + external transfers distributed by government (on-budget development assistance for health), in Low- and Middle Income Countries (LMICs), 2000-2021 (27). Levels are compared to the Abuja target of 15% government health spending (both domestically and externally derived) / GGE (45).

Funding levels also fall short of the so-called "Chatham House goal" set by Di Mcintyre and Filip Meheus that GHE-S levels in LMICs should be at least 5% of Gross Domestic Product (GDP) to make progress towards Universal Health Coverage (UHC) (27, 41, 43) (Figure 2.2).

The level of GHE-S that a government spends depends on two key fiscal factors: A) public revenue from domestic sources, i.e. the total amount of money a government raises (excluding grants from external sources), and B) the proportion of this revenue that is spent on health. Concerning the first factor, based on WB data, average LMIC public revenue estimates (excluding grants) have fluctuated between 15-20% of GDP since 2005, compared to High-Income Country (HIC) figures between 24-27% (46). However, these levels vary substantially across countries, with Western economies such as Norway, Denmark and France raising between 40-50% of GDP as public revenue, and Sub-Saharan African (SSA) LICs such as Uganda, Malawi and Burkina Faso raising between 12-18% of GDP as public revenue (46). As for the second factor, the proportion spent on health, GHE-S as a proportion of General Government Expenditure (GGE) was lower in 2021 (6.4%) than in 2000 (6.9%) for LICs (Figure 2.3). For LMICs, improvement has been seen from 5.5% in 2000 to 7% in 2021, and in UMICs from 9% to 10.3%. In comparison, HICs have spent a median 12-15% of GGE on health, or 4-6% of GDP over the same time period (27). Again, these summary figures hide the inequality between global spearheads of government funded healthcare such as Sweden, Canada or the United Kingdom that spend 18-20% of GGE as GHE-S, or 7-10% of GDP, compared to the ten lowest-spending LICs that spend 2-4% of GGE as GHE-S, or less than 1% of GDP on health (27).

# 2.2. Development assistance definitions, terminology, levels and trends

## 2.2.1. Official Development Assistance

The main functions of international official donors and creditors are to give grants and loans. The main official international financial flows from HIC governments, multilateral organizations and private philanthropies to governments and Non-Government Organizations (NGOs) in partner LMICs are: Official Development Assistance (ODA), which is given either as grants or as concessional loans, and non-ODA loans given on non-concessional terms. Grants from private foundations are also reported in the Organisation for Economic Cooperation and Development (OECD) Creditor Reporting System (CRS) database as private financing, not as ODA (47).

Since 2018, ODA has been defined by the OECD Development Assistance Committee (DAC) as flows with a development purpose which are "*concessional in character*". This implies a grant element of:

- "45 per cent in the case of bilateral loans to the official sector of LDCs [Least Developed Countries] and other LICs (calculated at a rate of discount of 9 per cent).
- **15 per cent** in the case of bilateral loans to the official sector of LMICs (calculated at a rate of discount of 7 per cent).
- **10 per cent** in the case of bilateral loans to the official sector of UMICs (calculated at a rate of discount of 6 per cent).
- **10 per cent** in the case of loans to multilateral institutions ... (calculated at a rate of discount of 5 per cent for global institutions and multilateral development banks, and 6 per cent for other organisations, including sub-regional organisations)" (48).

Before 2018, the OECD definition of ODA was flows with a development purpose which were "concessional in character and convey[ed] a grant element of at least 25 per cent (calculated at a rate of discount of 10 per cent)" (48). Concessionality is defined by the OECD DAC as a measure of credit softness reflecting "the benefit to the borrower compared to a loan at market rate" (49). This relies on the grant element. The grant element is calculated as "the difference between the face value of a loan and the discounted present value of the service payments the borrower will make over the lifetime of the loan, expressed as a percentage of the face value." (49). This in turn depends on the interest rate (% per year), grace period (time from loan commitment until repayments have to start), maturity (time from commitment until last repayment has to be made) and the discount rate as per above (49). As an example, a loan provided via the World Bank (discount rate: 5% per year) with an interest rate of 4% per year, a grace period of 5 years, and maturity of 30 years, has a grant element of 10.9% and is therefore considered concessional (50).

The term Official Development Assistance<sup>+</sup> (ODA<sup>+</sup>) for health is used in this research, a term which has previously been used by Countdown to 2030 to capture ODA and Bill and Melinda Gates Foundation (BMGF) grants for health purposes due to their magnitude (51). The Institute for Health Metrics and Evaluation (IHME) use the term Development Assistance for Health (DAH), which includes private donations from individuals and corporations (52). As this research looks at official development partner activities, it generally uses the terms ODA for health purposes and ODA<sup>+</sup> for health with BMGF grants included, and generally relies on ODA data from the Organisation for Economic Cooperation and Development (OECD) Creditor Reporting System (CRS) database to examine official development partner financial

assistance (47). The terms "development assistance for the health sector" and "external health financing" are used to broadly refer to externally derived transfers for LMIC health sectors when the use of less specific terminology is more appropriate.

## 2.2.2. Development assistance levels and trends

According to OECD CRS data, in 2022, gross total ODA was about US\$ 277 billion per year (current US\$) (47). \$203 billion were disbursed as grants, and \$72 billion as loans. Out of total ODA, approximately \$39 billion were channelled to health purposes<sup>4</sup>. Both total ODA and ODA for health purposes have grown substantially over the recent decades. The latest increases seen since 2020 were to a large extent due to increased financing to help countries cope with the Covid-19 pandemic and funds allocated to receive refugees from Ukraine in donor-countries (47, 53, 54).





# 2.3. Debt terminology, levels and trends

# 2.3.1. Debt terminology

LMIC governments generally borrow from other government credit/development agencies bilaterally, International Financial Institutions (IFIs) – including the WB and the IMF, but also regional development banks such as the African Development Bank and Islamic Development Bank - other multilaterals and international private creditors, such as commercial banks or individual bondholders. Some of the bilateral and multilateral official lending may count as concessional and thereby as ODA as per above, while the rest is considered non-concessional.

<sup>&</sup>lt;sup>4</sup> Calculated by adding categories 120: "Health, total" and 130: "Population Policies/Programmes & Reproductive Health, Total".

#### 2.3.2. Debt levels and trends

The Public and Publicly Guaranteed (PPG) external debt stock held by LMICs has grown from about \$1.1 trillion in year 2000 to about \$3.4 trillion in 2022 (current US\$) (46) (Figure 2.5). Debt servicing levels have grown accordingly, with LMIC governments repaying about \$414 billion towards these loans in 2022, about US\$137 billion more than funds received as ODA (46, 47). In the 1990's and 2000's, the Heavily Indebted Poor Countries (HIPC) and Multilateral Debt Relief Initiative (MDRI) led to substantial debt relief for the most indebted LMICs, with an absolute decrease in the PPG external debt stock (46, 55). With corresponding strong GDP growth rates, the debt stock plummeted relative to GDP (46) (Figure 2.5). This trend however reversed during the 2010's as countries completed the HIPC and MDRI schemes, kept accumulating debt, and GDP growth rates slowed.



Figure 2.5: PPG external debt stock and service in LMICs from 2000-2022 in absolute terms and relative to GDP (46).

These aggreggate figures conceal a strong disparity in debt burdens between countries. 25 of the most indebted LMICs spend more than a fifth of their tax revenue on external debt servicing, and 55% of Least Developed Countries (LDCs) and other LICs are assessed as being at high risk of, or in, debt distress (2024 UN analysis based on 2023 data) (53).

# 2.4. IMF/WB mandates, programmes and loan conditionality terminology

The International Monetary Fund (IMF) has as its mandate to promote international monetary cooperation and financial stability, including for exchange rates and balance of payments, and encourage the expansion of trade, economic growth and prosperity of its member countries (56, 57). It does so by providing loans/credit and policy/technical advice and assistance (56).

The majority of IMF loans and credit are non-concessional, while funds disbursed through the IMF Concessional Trust Fund count as ODA (47).

Two of the World Bank Group's branches provide loans to governments: The International Bank for Reconstruction and Development (IBRD) and the International Development Association (IDA) (58). The mandate of the IBRD is to invest in capital for productive purposes, promote private foreign investment, economic growth and trade, improving living standards and labour conditions in its member countries, with an emphasis on economies disrupted by war and "less developed countries" (59). The IDA has as its mandate to promote economic development, increase productivity and raise living standards also in less developed countries, on terms more flexible and concessional than the IBRD (60). The WB provides non-concessional loans to Middle-Income Countries (MICs) via the IBRD (61), and concessional loans and grants to LICs and eligible LMCs via the IDA (62). The WB also functions as a global knowledge bank, offering policy advice, data, research and analysis, and technical assistance (63).

#### 2.4.1. Programme/mechanism terminology

The IMF and WB have a number of different lending facilities, financing mechanisms and country support programme types. These have changed over time. In the 1980's and 1990's their main programs were the Structural Adjustment Programs (64, 65). Following extensive criticism over the social implications of structural adjustment, the IMF replaced their structural adjustment facility with the Poverty Reduction and Growth Facility (PRGF) in 1999 (66). The WB and IMF now required country governments to develop comprehensive multi-sectoral Poverty Reduction Strategy Papers (PRSPs) for lending eligibility (67), which are still used today. In 2024, the IMF had 12 different lending facilities (68). Using different criteria and terms of lending, these facilities provide access to three different accounts. A) The General Resources Account (GRA), which has non-concessional terms with market-based interest rates (68). B) The Poverty Reduction and Growth Trust (PRGT) for low-income countries at low/zero interest rates (68). C) The Resilience and Sustainability Trust (RST) established in 2022 for addressing longer-term challenges such as climate change in low-income-, lower-middle income- and small states, with tiered and varying interest rates roughly between 0.6% to 5.2% (68-71).

#### 2.4.2. Loan conditionality terminology

While all loans have general terms of lending (interest rate, grace period, maturity period etc.), IMF and WB loans come with particular policy recommendations and loan conditionalities (72, 73), which forms a central argument for focusing on these two creditor institutions in this research. IMF conditionality may take the form of "prior actions", "structural benchmarks",

"quantitative performance criteria" and "indicative targets" (72). Prior actions and quantitative performance criteria are binding conditions, meaning they are mandatory for initiation or continuation of the IMF programme (72, 74). Prior actions are structural steps a borrower country agrees to take before the IMF will approve financing or complete a review, e.g. measures to increase revenue (72, 74). Quantitative performance criteria are quantitative conditions for IMF lending such as a ceiling on the public external debt (72, 74). Structural benchmarks are non-binding, non-quantitative progress indicators such as "strengthening tax administration" or "reforming state-owned enterprises" (72, 74). Indicative targets are nonbinding "flexible numerical trackers", used for monitoring a country's progress toward agreed upon economic adjustment objectives, e.g. placing a particular quantitative ceiling on the government wage bill (72, 74).

Through its Country Partnership Framework (CPF), the WB similarly sets out conditionalities upon which access to its lending and policy/technical advice depends (75, 76). The WB only counts legally binding conditions under the term loan conditionalities, which includes "prior actions" before a programme can begin, and "tranche release conditions" necessary for the release of funds, both of which are structural conditions as opposed to quantitative conditions (75-77).

Having established the key concepts and terminology for this research and levels and trends of examined variables, Chapter 3 presents the results of a review of the existing literature on the research questions and explains the economic hypotheses and conceptual framework for the research as informed by the literature. Chapter 4 describes the research methods used in this thesis and its associated epistemological position.

# References

1. Kutzin J. Health financing for universal coverage and health system performance: concepts and implications for policy. Bull World Health Organ. 2013;91(8):602-11.

New York: The International Bank for Reconstruction and Development / The World Bank **Oxford University Press** 

6. Merriam-Webster. 'Equity' and 'Equality': How they differ and overlap : Merriam-Webster; 2024 [cited 2024 January 10]. Available from: <u>https://www.merriam-</u> webster.com/grammar/equality-vs-equity-difference.

Schieber G, Baeza C, Kress D, Maier M. Financing Health Systems in the 21st Century. In: Jamison DT, 2. Breman JG, Measham AR, Alleyne G, Claeson M, Evans DB, et al., editors. Disease Control Priorities in Developing Countries. Washington (DC)

Copyright © 2006, The International Bank for Reconstruction and Development/The World Bank Group.; 2006. World Health Organization. Health financing 2024 [cited 2024 June 22]. Available from: 3. https://www.who.int/health-topics/health-financing#tab=tab\_1. 4. Goudge J, Akazili J, Ataguba J, Kuwawenaruwa A, Borghi J, Harris B, et al. Social solidarity and

willingness to tolerate risk- and income-related cross-subsidies within health insurance: experiences from Ghana, Tanzania and South Africa. Health Policy Plan. 2012;27 Suppl 1:i55-63.

McIntyre D. Beyond fragmentation and towards universal coverage: insights from Ghana, South Africa 5. and the United Republic of Tanzania. Bulletin of the World Health Organization. 2008;86(11):871-6.
7. Marx K. Critique of the Gotha Programme 1875 [cited 2024 January 10]. Available from: https://www.marxists.org/archive/marx/works/1875/gotha/ch01.htm.

Raphael DD. Equality and Equity. Philosophy. 1946;21(79):118-32.

Raphael DD. Equally and Equity. Fillosophy. 1940,21(79).110 Culyer A, Wagstaff A. Equity and equality in health and health

care. Journal of Health Economics. 1993.

10. Culyer A. Efficiency, equity and equality in health and health care. CHE Research Paper 120: University of York; 2015.

11. Aristotle. Nicomachean Ethics. Translated by W. D. Ross: Batoche Books, Kitchener; 1999 (original text 350 B.C.E).

12. World Bank. Analyzing Health Equity Using Household Survey Data 2024 [cited 2024 September 4]. Available from:

https://documents1.worldbank.org/curated/en/633931468139502235/pdf/424800ISBN978011OFFICIAL0USE00 NLY10.pdf.

13. Ataguba JE, Asante AD, Limwattananon S, Wiseman V. How to do (or not to do) ... a health financing incidence analysis. Health Policy Plan. 2018;33(3):436-44.

14. Kwesiga B, Ataguba JÉ, Abewe C, Kizza P, Zikusooka CM. Who pays for and who benefits from health care services in Uganda? BMC Health Serv Res. 2015;15:44.

15. Asante A, Price J, Hayen A, Jan S, Wiseman V. Equity in Health Care Financing in Low- and Middle-Income Countries: A Systematic Review of Evidence from Studies Using Benefit and Financing Incidence Analyses. PLoS One. 2016;11(4):e0152866.

16. Kakwani NC. Measurement of Tax Progressivity: An International Comparison. The Economic Journal. 1977;87(345).

17. Martinez-Alvarez M, Federspiel F, Singh NS, Schaferhoff M, Lewis Sabin M, Onoka C, et al. Equity of resource flows for reproductive, maternal, newborn, and child health: are those most in need being left behind? BMJ. 2020;368:m305.

18. Gini C. On the Measure of Concentration with Special Reference to Income and Statistics. Colorado College Publication, General Series. 1936;No. 208:73–9.

19. Cheng Q, Asante A, Susilo D, Satrya A, Man N, Fattah RA, et al. Equity of health financing in Indonesia: A 5-year financing incidence analysis (2015-2019). Lancet Reg Health West Pac. 2022;21:100400.

20. Yu CP, Whynes DK, Sach TH. Equity in health care financing: The case of Malaysia. Int J Equity Health. 2008;7:15.

21. Binyaruka P, Martinez-Alvarez M, Pitt C, Borghi J. Assessing equity and efficiency of health financing towards universal health coverage between regions in Tanzania. Social science & medicine (1982). 2024;340:116457.

22. World Health Organization. The world health report: health systems financing: the path to universal coverage. Geneva: World Health Organization; 2010.

23. Marmot M, Commission on Social Determinants of Health. Achieving health equity: from root causes to fair outcomes. Lancet. 2007;370(9593):1153-63.

24. McIntyre D, Garshong B, Mtei G, Meheus F, Thiede M, Akazili J, et al. Beyond fragmentation and towards universal coverage: insights from Ghana, South Africa and the United Republic of Tanzania. Bull World Health Organ. 2008;86(11):871-6.

25. Mathauer I, Saksena P, Kutzin J. Pooling arrangements in health financing systems: a proposed classification. Int J Equity Health. 2019;18(1):198.

26. OECD, Eurostat, WHO. A System of Health Accounts. OECD Publishing. 2011.

27. WHO. Global Health Expenditure Database 2024 [cited 2024 March 28]. Available from: <u>http://apps.who.int/nha/database</u>.

28. WHO. Health systems financing: the path to universal coverage 2010 [cited 2024 May 29]. Available from: <u>https://www.who.int/publications/i/item/9789241564021</u>.

29. Ortiz-Ospina E, Roser M. Taxation 2021 [cited 2021 December 10]. Available from: https://ourworldindata.org/taxation.

30. WHO. Voluntary health insurance: potentials and limits in moving towards UHC 2018 [cited 2023 December 14]. Available from: <u>https://www.who.int/publications/i/item/WHO-HIS-HGF-Health-Financing-Policy-Brief-18.5</u>.

31. Xu K, Evans DB, Carrin G, Aguilar-Rivera AM, Musgrove P, Evans T. Protecting households from catastrophic health spending. Health Aff (Millwood). 2007;26(4):972-83.

32. Makinen M, Waters H, Rauch M, Almagambetova N, Bitran R, Gilson L, et al. Inequalities in health care use and expenditures: Empirical data from eight developing countries and countries in transition. Bulletin of the World Health Organization. 2000;78(1):55-65.

33. Akazili J, McIntyre D, Kanmiki EW, Gyapong J, Oduro A, Sankoh O, et al. Assessing the catastrophic effects of out-of-pocket healthcare payments prior to the uptake of a nationwide health insurance scheme in Ghana. Glob Health Action. 2017;10(1):1289735.

34. Chuma J, Okungu V. Viewing the Kenyan health system through an equity lens: implications for universal coverage. Int J Equity Health. 2011;10:22.

35. Moreno-Serra R, Smith PC. Broader health coverage is good for the nation's health: evidence from country level panel data. J R Stat Soc Ser A Stat Soc. 2015;178(1):101-24.

36. Moreno-Serra R, Smith P. The Effects of Health Coverage on Population Outcomes: A Country-Level Panel Data Analysis. Results for Development Institute. 2011.

37. Qin VM, Hone T, Millett C, Moreno-Serra R, McPake B, Atun R, et al. The impact of user charges on health outcomes in low-income and middle-income countries: a systematic review. BMJ Glob Health. 2018;3(Suppl 3):e001087.

38. World Health Organization. What are the Overall Principles of HBP Design? : WHO; 2024 [cited 2024 June 21]. Available from: <a href="https://www.who.int/teams/health-financing-and-economics/economic-analysis/health-technology-assessment-and-benefit-package-design/resource-guide-for-the-use-of-hta-and-hbp-design-processes/what-are-the-overall-principles-of-hbp-design.</a>

39. World Health Organization. Making fair choices on the path to universal health coverage: Final report of the WHO Consultative Group on Equity and Universal Health Coverage 2014 [cited 2024 June 21]. Available from: <a href="https://iris.who.int/bitstream/handle/10665/112671/9789241507158">https://iris.who.int/bitstream/handle/10665/112671/9789241507158</a> eng.pdf?sequence=1.

40. Global Burden of Disease Health Financing Collaborator N. Trends in future health financing and coverage: future health spending and universal health coverage in 188 countries, 2016-40. Lancet. 2018;391(10132):1783-98.

41. McIntyre D, Meheus F, Rottingen JA. What level of domestic government health expenditure should we aspire to for universal health coverage? Health Econ Policy Law. 2017;12(2):125-37.

42. Dieleman JL, Sadat N, Chang AY, Fullman N, Abbafati C, Acharya P, et al. Trends in future health financing and coverage: future health spending and universal health coverage in 188 countries, 2016-40. The Lancet. 2018;391(10132):1783-98.

43. McIntyre D, Meheus F. Fiscal Space for Domestic Funding of Health and Other Social Services 2014 [cited 2019 September 17].

44. Stenberg K, Hanssen O, Edejer TT-T, Bertram M, Brindley C, Meshreky A, et al. Financing transformative health systems towards achievement of the health Sustainable Development Goals: a model for projected resource needs in 67 low-income and middle-income countries. The Lancet Global Health. 2017;5(9):e875-e87.

45. Organisation of African Unity. Abuja Declaration on HIV/AIDS, Tuberculosis and Other Related Infectious Diseases. Abuja, Nigeria; 2001.

46. World Bank. World Bank Open Data 2024 [cited 2024 March 28]. Available from: https://data.worldbank.org/.

47. OECD. Creditor Reporting System (CRS) 2024 [cited 2023 February 16]. Available from: https://stats.oecd.org/Index.aspx?DataSetCode=CRS1.

48. OECD DAC. Official development assistance – definition and coverage 2023 [cited 2023 December 18]. Available from: <u>https://www.oecd.org/development/financing-sustainable-development/development-finance-standards/officialdevelopmentassistancedefinitionandcoverage.htm</u>.

49. OECD. DAC Glossary of Key Terms and Concepts 2023 [cited 2023 December 18]. Available from: https://www.oecd.org/dac/dac-glossary.htm.

50. World Bank. Grant Element Calculator 2023 [cited 2023 December 18]. Available from: https://ida.worldbank.org/en/financing/debt/grant-element-calculator.

51. Arregoces L, Daly F, Pitt C, Hsu J, Martinez-Alvarez M, Greco G, et al. Countdown to 2015: changes in official development assistance to reproductive, maternal, newborn, and child health, and assessment of progress between 2003 and 2012. The Lancet Global Health. 2015;3(7):e410-e21.

52. Institute for Health Metrics and Evaluation. Financing Global Health 2012: The End of the Golden Age? : Seattle, WA: IHME; 2012 [cited 2023 December 18]. Available from: <u>http://www.healthdata.org/policy-report/financing-global-health-2012-end-golden-age</u>.

53. United Nations DESA FSDO. Financing for Sustainable Development Report 2024 2024 [cited 2024 April 17]. Available from: <a href="https://desapublications.un.org/publications/financing-sustainable-development-report-2024">https://desapublications.un.org/publications/financing-sustainable-development-report-2024</a>.

54. Alkhalil M, Alaref M, Ekzayez A, Mkhallalati H, El Achi N, Alzoubi Z, et al. Health aid displacement during a decade of conflict (2011-19) in Syria: an exploratory analysis. BMC Public Health. 2023;23(1):1562.

55. World Bank, IMF. Heavily Indebted Poor Countries (HIPC) Initiative and Multilateral Debt Relief Initiative (MDRI): Statistical Update 2019 [cited 2022 February 9]. Available from:

https://documents1.worldbank.org/curated/en/419661565316083523/pdf/Heavily-Indebted-Poor-Countries-HIPC-Initiative-and-Multilateral-Debt-Relief-Initiative-MDRI-Statistical-Update.pdf

56. IMF. What is the IMF? 2024 [cited 2024 May 20]. Available from:

https://www.imf.org/en/About/Factsheets/IMF-at-a-Glance.

57. (IMF) IMF. Articles of agreement 2020 [cited 2024 June 22]. Available from:

https://www.imf.org/external/pubs/ft/aa/pdf/aa.pdf.

58. World Bank. Who we are 2024 [cited 2024 June 22]. Available from: <u>https://www.worldbank.org/en/who-we-are</u>.

59. International Bank for Reconstruction and Development (IBRD). Articles of Agreement 2012 [cited 2024 June 22]. Available from: <u>https://thedocs.worldbank.org/en/doc/722361541184234501-0330022018/original/IBRDArticlesOfAgreementEnglish.pdf</u>.

60. International Development Association. Articles of Agreement 1960 [cited 2024 June 22]. Available from: <a href="https://thedocs.worldbank.org/en/doc/2a209939e876fdcd0d957036daebff6e-0410011960/original/IDA-Articles-of-Agreement-English.pdf">https://thedocs.worldbank.org/en/doc/2a209939e876fdcd0d957036daebff6e-0410011960/original/IDA-Articles-of-Agreement-English.pdf</a>.

61. World Bank. International Bank for Reconstruction and Development 2024 [cited 2024 January 9]. Available from: <u>https://www.worldbank.org/en/who-we-are/ibrd</u>.

62. World Bank. What is IDA? 2024 [cited 2024 January 9]. Available from:

https://ida.worldbank.org/en/what-is-ida.

63. The World Bank. What We Do 2024 [cited 2024 June 22]. Available from:

https://www.worldbank.org/en/about/what-we-do.print.

64. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

65. Thomson M, Kentikelenis A, Stubbs T. Structural adjustment programmes adversely affect vulnerable populations: a systematic-narrative review of their effect on child and maternal health. Public Health Rev. 2017;38:13.

66. IMF. IMF Lending to Poor Countries—How does the PRGF differ from the ESAF? 2001 [cited 2024 January 9]. Available from: <a href="https://www.imf.org/external/np/exr/ib/2001/043001.htm">https://www.imf.org/external/np/exr/ib/2001/043001.htm</a>.

67. The Poverty Reduction Strategy Initiative: Findings from 10 Country Case Studies of World Bank and IMF Support World Bank Operations Evaluation Department,

IMF Independent Evaluation Office, [cited 2024 January 9]. Available from:

https://ieg.worldbankgroup.org/sites/default/files/Data/reports/prsp\_country\_case\_studies.pdf.

68. IMF. IMF lending 2024 [cited 2024 January 8]. Available from:

https://www.imf.org/en/About/Factsheets/IMF-Lending.

69. IMF. Rates Query 2024 [cited 2024 January 9]. Available from:

https://www.imf.org/external/np/fin/data/queryoutput.aspx.

70. Center for Global Development. The IMF's New Resilience and Sustainability Trust (RST): A Technical Summary: Center for Global Development; 2022 [cited 2024 January 9]. Available from:

https://www.cgdev.org/sites/default/files/2022-05/IMF-RST-key-points.pdf. 71. IMF. The Resilience and Sustainability Facility (RSF) 2024 [cited 2024 January 9]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/Resilience-Sustainability-Facility-RSF.

72. IMF. IMF Conditionality 2023 [cited 2023 December 18]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/IMF-Conditionality.

73. World Bank. Development Policy Retrospective 2021 2023 [cited 2023 December 18]. Available from: https://documents1.worldbank.org/curated/en/099623509132210285/pdf/IDU0249804670b2fc0466f083850d1aad 1818915.pdf.

74. Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A, Stubbs T, editors: Oxford University Press; 2023. 0 p.

75. Brunswijck G. Flawed conditions: The impact of the World Bank's conditionality on developing countries: Eurodad; 2019 [cited 2024 January 9].

76. Stichelmans T. How international financial institutions and donors influence economic policies in developing countries 2016 [cited 2024 January 9]. Available from:

https://www.eurodad.org/how international financial institutions and donors influence economic policies in d eveloping countries.

77. Bull B, Jerve AM, Sigvaldsen E. The World Bank's and the IMF's use of Conditionality to Encourage Privatization and Liberalization: Current Issues and Practices 2006 [cited 2024 June 24]. Available from: https://www.duo.uio.no/bitstream/handle/10852/32692/Bull.pdf.

# 3. Chapter 3: Literature review, hypotheses and conceptual framework

This chapter provides an overview of the literature underlying the research in this thesis. This is organised according to the different dimensions of the research questions and the economic relationships investigated. It first summarises available theoretical policy influence analysis frameworks, and the literature on External Development Partner (EDP) influence on Low- and Middle-Income Country (LMIC) health financing. It then summarises the literature on the relationships between: Official Development Assistance (ODA), debt variables, International Monetary Fund (IMF) programmes and loan conditionalities, and government health spending, voluntary health insurance schemes and out-of-pocket financing. It then identifies some of the literature gaps sought to be addressed by this thesis, explains the economic hypotheses to be explored and the conceptual framework guiding this exploration based on the reviewed literature.

The methods used for reviewing the literature are described in Appendix 1.

### 3.1. Theoretical policy influence analysis frameworks

There are a number of frameworks available for thinking about and analysing how policy influence occurs more broadly. Among the better known of these is Kingdon's streams model (1), with streams of problems, politics and policy that come together in a policy window, through which policy happens, which also identifies how policy can be influenced via these streams and windows. Walt and Gilson's (1994) policy triangle model was specifically developed in the context of analysing the political economy of health policy reform in LMICs, highlighting the roles of context, policy content, policy process, and actors (2). The linear model of policy-making by Meier (1991) emphasised the influence from state-centred and society-centred forces on policy makers in the policy process, hence focusing on different groups of actors (3). A commonly used framework in the health financing policy process literature is the Stages model by Grindle and Thomas (1991), which depicts the policy making process as an algorithmic, linear process from an issue being on the agenda, a decision for reform being made, to successful implementation of a policy (4, 5). While this framework does not directly incorporate external actors, and real-world policymaking is likely more dynamic, it does identify junctures where EDPs can exert influence. The "3-i's" framework conceptualises how ideas, interests and institutions jointly form the basis for policy change, looking more at the "why" of policy change than the "how" (6-8). The "context, evidence, links" framework identifies key factors for policy influence as the political context, the evidence-base underlying a policy, and links/network factors, as well as external socio-economic/cultural factors (9-11).

Some authors have emphasised the agents of policy change, i.e. the actors, coalitions and policy communities that can influence policy (12). Others emphasise the mechanisms through which policy influence happens, which relates more directly to the research questions posed in Chapter 7, and includes the mechanisms of advice, lobbying, advocacy and activism (10, 13, 14).

Relating more directly to the influence of EDPs on LMIC governments is the Spheres of Influence Framework, that articulates how "bilateral activism" occurs via international science and finance, grounded in underlying political ideas, in an interaction with recipient country political resources and policy culture (15). On policy advice as a mechanism of EDP influence in development cooperation, Haldenwang and Anker (2009) distinguish between policy advice that is more scientific and politically neutral, and types of policy advice that are more political (16). Specifically for analysing the political economy of health financing reform, Sparkes et al. (2019) have developed a framework that includes the role of external politics, however without specific formalisation of external influence mechanisms on domestic health financing sources (17). Fox and Reich (2015) instead used a combination of Hall's 3-i's framework (6) and Kingdon's stream model (1) as a comprehensive theoretical framework for analysing how politics affects UHC reform in LMICs at different stages of the policy cycle (18, 19), though again without a specific formalisation of EDP influence on different health financing sources. On the World Bank (WB), Bazbauers (2017) described how the transfer of international norms, ideas and practices occurred via WB technical assistance, shaped by the political and social relations that surround them (20).

## 3.2. EDP influence on LMIC health financing

Several studies have investigated EDP influence on LMIC health financing. Most of the identified literature examines EDP influence on health financing reform such as the introduction of national health insurance schemes, introduction or removal of user fees, or piloting new health financing schemes in aid-recipient countries, while others have examined influence on domestic health financing allocations. This literature is predominantly qualitative, using methods including literature/document reviews, policy and stakeholder analysis, key stakeholder interviews and participant observation.

### 3.2.1. EDP influence on health financing reform

The existing literature has identified influence from EDPs+IMF on user fees, from earlier promotion especially by IMF/WB in the 1980's and 1990's to later promotion of user fee exemption schemes. In the 2000's EDPs have promoted national health insurance schemes and Community Based Health Insurance (CBHI) in different contexts. In terms of their used

mechanisms of influence, EDPs are generally found to provide technical and financial support to health financing reform.

In Ghana, the "cash-and-carry" system of paying user fees for health services was described as "constructed by the IMF and the World Bank in the 1980s"<sup>5</sup> - a notion supported by other authors (21). The IMF and WB later changed their position along with other partners in the 2000s in favour of a nationwide health insurance scheme (22). Donors were then found to use their financial leverage and technical expertise to gain access to health policy agenda-setting, highlighting the role of UNICEF in promoting fee exemption for maternity care (23). In Liberia, the positions of international humanitarian NGOs against user fees was also viewed as influential in the government's decision to suspend user fees, hoping to retain their partners (24). In Uganda, EDPs were described as wielding "formidable political power to influence the health policy process" in 2004 (25), originally in support of user fees, informed by the 1987 WB report: "Financing Health Services in Developing Countries", up until the abolition of user fees in 2001 (25-27). A 2017 review of the literature on government ownership of Universal Health Coverage (UHC) policy decisions in Sub-Saharan Africa (SSA) found that governments had varying levels of ownership, which was found to more often be due to private sector influence than donor influence. Donors and governments were then often seen to work handin-hand, e.g. in designing user fee exemption policies (5).

EDPs have been found to give technical and financial support to the development of the national health insurance scheme in Nigeria (28), and later advocated and provided technical and financial support for the Nigerian basic health care provision fund (29). EDPs have also been acknowledged as influential in health financing reforms in Zimbabwe (30, 31) and Uganda (25, 27, 32). In Pakistan, EDPs were described by the World Health Organisation (WHO) and Ministry of Health staff as having provided technical assistance in the development of national health financing reforms for UHC (33). Investigating the formation of a migrant health insurance scheme in Thailand, Herberholtz (2020) found that while finance and technical expertise were found to be mechanisms of influence at different stages of the policy process, intersectoral leverage instrumentalised as international reports and rating/carding systems applied to export sectors with potential harmful economic and reputational effects, were found to be much more influential (34).

<sup>&</sup>lt;sup>5</sup> As explained in Chapter 1, the IMF technically cannot be classified as an EDP, but the quote mentions both IMF and WB, and the study most appropriately fitted under an EDP influence subheading.

The value of EDPs in providing financial and technical assistance to the piloting and scale-up of CBHI has been described in Ethiopia (35), a value noted by other researchers (36). Po et al. (2010) praised EDP-supported pilot projects of new health financing schemes, leading to the nationwide scale-up of health equity funds in Cambodia (37). Evidence from Tajikistan however showed how EDP-supported pilots can also be incoherent and ineffective in supporting health financing reform in the absence of strong donor coordination and harmonization (38).

Other authors have critiqued how CBHI has been pushed by EDPs in Senegal, allegedly stifling progress toward UHC by more than a decade, which was attributed to poor coordination and inherent equity issues with CBHI compared to other modes of pooled health financing (39, 40).

#### 3.2.2. EDP influence on domestic health financing allocations

While not the focus of this thesis, EDPs have also been found to influence domestic health financing allocations in terms of the disease areas receiving government health budget priority.

In the 2000's and early 2010's, a study by Nagemi et al. (2021) found that vertical donor financing priorities took precedence over the Ugandan government's priorities, crowding out government health financing allocations to donor-prioritised disease areas (41). In the MDG era, high EDP prioritisation and financing for HIV/AIDS was found to converge with a disproportionately high government budget allocation to this disease relative to its disease burden in Brazil and India, but not Russia (42). A case study from Uganda has highlighted the importance of fiscal space and fiscal policy frameworks allowing for additional health funding being allocated to donor priorities: In 2002, Uganda was awarded a grant from the Global Fund to Fight AIDS, TB and Malaria (GFATM) on condition of matched funding from the domestic health budget, however the Ugandan public budget was fixed for three years in a Medium-Term Expenditure Framework (MTEF) agreed upon with the IMF (43). This study illustrates an intersection between our different actors and their activities, in this case an EDP seeking to mobilise additional domestic resources for health, allocated toward their priorities, and IMF fiscal policy frameworks preventing this. Another study from Uganda has also shown the role of EDPs in co-determining health financing allocations, with EDPs prioritising diseasespecific targets rather than broader health system strengthening as envisioned in the national development plan for the health sector (44).

Co-financing requirements from EDPs, e.g. Gavi, can also help mobilise additional government funds for different health programs (45, 46). Jha et al. (2024) have highlighted the

necessity of matched funds being net additional to avoid displacement from other social sector priorities (45).

The following sections summarizes the literature examining the relationship between ODA and domestic health financing, which is predominantly quantitative studies.

## 3.3. Development assistance for the health sector and sources of domestic health financing

## 3.3.1. Development assistance for the health sector and government health expenditure

The association between development assistance for the health sector and Government Health Expenditure (GHE) is described mainly in a body of literature centered around fungibility. Fungibility in this context refers to when development assistance for the health sector does not translate into a 1:1 net addition of funds due to a reduction in domestic government health expenditure resulting from having received these external funds for health (this is also referred to in the literature as "displacement" or "crowding out") (47-49). The general concept of fungibility is a well-known and acknowledged phenomenon in the general development assistance literature (50-56).

15 original panel data studies of associations between GHE and development assistance for the health sector were identified, published between 2009-2021 (48, 49, 55, 57-68) (see Appendix 1 for details). Panel sizes varied between 9 and 143 countries and time-periods between 1980-2015. Development assistance data sources used were the Organisation for Economic Cooperation and Development (OECD) Creditor Reporting System (CRS), WHO National Health Accounts (NHA), Institute for Health Metrics and Evaluation (IHME) and the WB. Health financing data came from the IMF and WHO NHA databases (the latter is available as the WHO Global Health Expenditure Database (GHED)). Measures of development assistance for the health sector varied considerably, with some studies disaggregating this into development assistance channeled via the government and others not. For GHE, some studies measured Government Health Expenditure as Agent (GHE-A, which is the same as GHE) while others narrowed to GHE-S (i.e. from domestic revenue) as the dependent variable. Estimation methods used included Ordinary Least Squares (OLS), Generalized Least Squares (GLS), Fixed Effects (FE) models and Generalized Method of Moments (GMM). Results varied greatly. One study found an additional 50 cents per capita being "crowded in" from domestic resources for every \$1 of development assistance for the health sector per capita received (57). Another found near complete fungibility: -\$0.86 GHE-S spent for every \$1 Development Assistance for Health channeled via Government (DAH-G) received (65). In total, eight studies found evidence of a decrease in government health spending (GHE-S/A (per Gross Domestic Product (GDP))) of more than 10% of the increase in development assistance for the health sector (total or channeled via government (/GDP)) (48, 49, 58-60, 64, 65, 67). Five studies found evidence of an effect size less than this or reported "little if any fungibility" or "little or no evidence of displacement" (55, 62, 63, 66, 68) (3 negative and two positive direction of effect), and two studies found evidence of an increase in government health spending of more than 10% of the increase in development assistance for the health sector (57, 61). While estimates varied greatly between the reviewed studies, 11 out of 15 studies indicated a negative direction of the association between receiving development assistance to the health sector and GHE, so in total, the literature supports the hypothesis of the presence of fungibility more than the hypothesis of the contrary. The observed variability between studies may to a large extent be due to the different methods described above.

## 3.3.2. Development assistance for the health sector, Out Of Pocket payments and Voluntary Health care Payment Schemes

The quantitative literature on associations between development assistance to the health sector and Out Of Pocket payments (OOP) and Voluntary Health care Payment Schemes (VHPS) is scarce and does not provide any solid indication of a clear, general relationship. The six following empirical studies were identified:

Xu et al. (2011) found that a 1% increase in external health financing was associated with a 0.03-0.06% increase in OOP in lower-middle income countries, with no significant association in LICs and UMICs (60).

Younsi et al. (2016) found OOP to increase by between 0.01-0.04% (depending on model specifications) for every 1% increase in development assistance for the health sector in LMICs (67).

Ali et al. (2020) found that neither overall Official Development Assistance (ODA) or ODA for the health sector per capita affected OOP measured as a proportion of private health spending in SSA between 1995-2015 (69).

Frimpong et al. (2022) examined the effects of external health spending on OOP using System GMM on data from 43 SSA countries from 2000-2017 and found a significant negative relationship (elasticity -0.06 to -0.09) (70).

Based on household survey data from 65 LMICs receiving above-average DAH per capita from 2000-2016, Gabani et al. (2024) found no overall association between DAH per capita

and OOP being larger than 10% of total household spending, OOP per total household spending, or impoverishment due to health expenditure. When channelled via the government however, significant negative associations were seen with most outcome variables, i.e. improved financial risk protection (71).

Patenaude (2021) found a significant positive relationship between DAH-G and prepaid private health financing (0.6\$ increase in prepaid private for \$1 increase in DAH-G) in some model specifications but not others, and not with OOP, and he found no association between Development Assistance for Health via Non-Governmental channel (DAH-NG) and these outcome variables<sup>6</sup> (68).

In a panel study of 154 countries from 2000-2015, including HICs, Leiter and Theurl (2021) found no significant association between DAH per capita and voluntary prepayment as a percent of total prepaid health financing<sup>7</sup> (74).

## 3.4. Non-health development assistance and government health spending

Three studies also looked at the effects of development assistance to other sectors than health on government health spending. By disaggregating concessionary loans ear-marked for different sectors using World Bank and IMF data from the 1970's and 80's across 38 LMICs, Feyzioglu et al. (1998) found that loans to the transport and communication sector stimulated public spending on health, thereby providing evidence for intersectoral fungibility benefiting the health sector (75). However their findings for loans given to the health sector were insignificant.

Using Fixed-effects IV estimation for a panel of 45 SSA countries from 1995-2015, Ali et al. (2020) found no effect of overall ODA received per capita on government health expenditure as a share of GDP (69).

A cross-sectional study of 2001 data from 44 African countries using OLS and robust least absolute error estimation found that "foreign aid" per capita correlated positively with "real per-capita health expenditure<sup>"8</sup> (76).

<sup>&</sup>lt;sup>6</sup> It deserves mention that the majority of his Hansen-tests have a p-value of the implausibly high number 1.000 ((72)), i.e. the models are in all likelihood overspecified and suffer from instrument proliferation (73), and the remainder have a P-value of 0.000, i.e. reject the null-hypothesis that the generated instruments are valid (73). No reference to the number of instruments generated or efforts to limit them is reported. Some of these results also suffer from serial correlation in the levels equation, and trusting the validity of these findings is therefore difficult. <sup>7</sup> In dynamic models.

<sup>&</sup>lt;sup>8</sup> Foreign aid interpreted as total ODA but unclear what was included in this variable. Also unclear which types of health expenditure were included in the dependent variable. A caveat applies for using correlational evidence from OLS for causal inference, in particular for relatively small panel and one year of data.

A single-country case study of Ghana from 1980-2014 using 3SLS found that increases in ODA were associated with increases in GHE/GDP<sup>9</sup> (77).

## 3.5. Public External Debt and Domestic Health Financing

### 3.5.1. Public External Debt and Government Health Expenditure

A few studies have examined the associations between debt and government health spending, with more studies finding evidence for a negative relationship between indebtedness and government health spending than the contrary.

Liang et al. (2014) investigated the associations between general Government Gross Debt (GGD: All government debt i.e. including both internal and external government debt) and GHE and GHE-S in LMICs between 1995-2010. They found a 1% increase in per capita GGD to be associated with a 0.27% increase in GHE, and a 0.10% increase for GHE-S (66).

Fosu (2007) looked at public expenditure shares in a panel of 35 African countries from 1975-1994 (78). He found that a one-standard deviation in public external debt servicing was associated with nearly one-third reduction in the share allocation to the health sector. A onestandard deviation increase in ODA/GDP was associated with a much smaller increase in the share allocation to the health sector, and this small responsiveness was found to be *"indicative of considerable fungibility of aid"* (78). The author found that debt relief would be more effective at increasing social sector funding allocation than aid (78). He also found a strong negative effect of predicted debt servicing on government health spending in a followup study (79).

Lora and Olivera (2006) studied a panel of about 50 LMICs between 1985-2003 and found higher public debt to reduce social spending, an effect that was attributed mainly to the stock of debt as opposed to debt service payments, which was interpreted as due to governments not desiring to take on more debt than they already had outstanding (80).

In a selected panel of seven South and Southeast Asian countries from 1980-2010, Shabbir and Yasin (2015) found a 1 percent increase in debt servicing to be associated with an 0.25% decrease in social sector spending (81).

Behera et al. (2019) found both significant positive, negative and insignificant associations between overall government debt servicing on government health expenditure in a panel of 85 LMICs from 2000-2013, based on which time period and countries were included (82). The relationship was significantly positive before the 2008 financial crisis, significantly negative after, insignificant in LICs, and significantly negative in MICs<sup>10</sup>.

<sup>&</sup>lt;sup>9</sup> Unclear if external contributions via government were excluded, which would explain positive correlation due to double counting.

<sup>&</sup>lt;sup>10</sup> Likely issues with overspecification and instrument proliferation as all Sargan test p-values 1.000, and Hansentest should be reported for two-step system GMM (72, 73). No mention of instrument numbers or efforts to curtail

Patenaude (2021) included general government debt per GDP as a covariate when analysing the impacts of DAH on the different main sources of domestic health financing (GHE-S, prepaid private health expenditure and OOP), and found a significant negative relationship between general government debt per GDP and GHE-S, and no relationship with prepaid private health spending and OOP<sup>11</sup> (68). Similarly, Lahiani et al. (2022) used system GMM to find that total government debt per GDP correlated negatively with total government health spending per GDP, and a separate negative effect of episodes of fiscal consolidation, in a selected panel of 23 "emerging" and middle-income countries from 2009-2018 (83).

In a case study of Senegal, Ouattara (2006) modeled the relationships between aid, debt and overall public spending between 1970-2000 (84). He found that 41 percent of aid disbursements and 14 percent of Senegalese government revenue had been used to repay debt, a negative elasticity of -0.13 between debt service increases and domestic expenditure as a share of GDP, and no significant elasticity between aid and domestic expenditure. He concluded that his findings indicated that reducing public debt in Senegal would be a more effective tool for increasing domestic expenditure than additional aid, making analogous inferences for health and other social sectors though without specifically disaggregating their spending (84).

A discussion paper by Kimalu (2002) examined national trends in Kenya, noting increasing debt stocks and -servicing throughout the 1980's and 90's, lowered government health expenditure per capita over the same period, and worsening public health indicators during the 1990's (mainly driven by the HIV/AIDS epidemic). The author argued that "*Kenya handles its external debt and its regular servicing at the expense of such vital life programmes as health care, education, and other social services*"<sup>12</sup> (85). Another discussion paper from Kenya has also used descriptive statistics from 2019-21 in Kenya, Zambia and Nigeria to argue that debt constrains government health spending in these countries (86). Also from Kenya, an Oxfam case study has used interviews, focus groups and national-, county- and household level statistics to explore debt impacts on the health sector including government health spending, concluding a constraining effect on the latter (87).

A WB discussion paper from 2022 projected that debt interest repayments are expected to limit government health spending per capita across LMICs until 2027 (88). 2023 analyses by the Center for Global Development (CGD) highlighted sharp increases in LMIC government interest repayments and that these were likely to constrain overall public spending, including for health sectors (89, 90). Similar conclusions were made by Birungi et al. (2022) for

them. Possible serial correlation in the levels equation for some regressions. Therefore caveats for trusting the validity of these findings.

<sup>11</sup> Findings subject to the caveats described in Section 3.3.2.

<sup>12</sup> Caveat: It is not possible to claim causality, effects, impacts or any other such terms based on a description of trends.

G20 Debt Service Suspension Initiative-participant countries (91).

Debt crises have also been found to negatively affect both government social and health spending across a panel of 108 LMICs from 1991-2019 (92).

## 3.5.2. Public External Debt, Out Of Pocket Payments and Voluntary Health care Payment Schemes

Only two studies were identified investigating these relationships. They included OOP as a dependent variable, not VHPS.

Said and Sani (2020) found a significant negative relationship between the public debt burden and OOP in 43 SSA countries from 2000-2014 in a robustness check, however the exact variables or regression outputs were not reported for this association (93).

Patenaude (2021) found no significant relationship between general government debt per GDP and OOP (68).

# 3.6. IMF programmes, loan conditionalities and domestic health financing

The literature investigating the links between IMF loan conditionalities and domestic health financing is roughly divided into two camps: one camp consisting of academics and civil society organisations finding that IMF conditionalities constrain domestic government health spending, and another of IMF/WB staff that find no constraining effect, or that government health spending increases under IMF programmes. There are methodological issues with the publications from both groups (described in Appendix 1), and some of the literature on both sides is not externally peer-reviewed, i.e. both IMF and civil society reports. The literature applies both qualitative and quantitative methods, some are policy analyses and others editorials/opinion pieces. Econometric studies of IMF effects on government health spending are characterized by significant heterogeneity in data used, specification choices, estimation methods, with corresponding divergence of findings, particularly between the described camps. The econometric literature is summarised in the below paragraphs, while further details are provided in Appendix 1.

Among the first group of academics and civil society, policy analyses and opinion pieces have proposed that IMF fiscal policy conditionalities under Structural Adjustment Programs (SAPs), Medium Term Expenditure Frameworks (MTEFs) and Poverty Reduction Strategy Papers (PRSPs) - the new name after the period of structural adjustment for the fiscal policy plans necessary to qualify for loans or debt relief from the IMF/WB - directly constrain government health sector spending through various mechanisms. These mainly revolve around austerity measures, including setting sectoral spending targets (94) (opinion piece), and public wage

expenditure ceilings (95, 96) (opinion piece/editorial; mixed-methods empirical study) motivated by the need to meet certain macro-economic targets such as specific levels of inflation (95), but also indirectly through allegedly undermining domestic industrialization and thus the tax base for health expenditures (97) (opinion piece). Critical analyses of IMF-driven austerity effects constraining social/health spending are numerous (43, 94-96, 98-115) (this includes a range of empirical work using both quantitative (including panel data studies) and qualitative methods identifying constraining effects, policy analyses and opinion pieces). Some nuances have been found in quantitative empirical work such as differential health spending effects in democracies vs. non-democracies (107), between different regions (116), between IMF programme participation and loan conditionalities (117), or no effect of IMF conditionalities on government health spending (77) (see Appendix 1 for details). Decentralization and privatization reforms of the Mexican health sector in the 1980's is another example of health sector reform resulting from IFI programmes and their conditionalities (118)

(policy analysis study).

While much of the above criticism and findings of constraining effects on social/health spending relate to austerity policies of the 1980's, 90's and early 2000's, policy analyses and quantitative empirical studies have continued up until today and found that while the IMF itself may claim to have learned from the adverse social impacts of imposing austerity measures on poor countries via SAPs, little has changed over time (99, 109-112). Change has however been identified in qualitative and quantitative text analyses of the evolution of WB policy over time, signifying a transition from more dogmatically neoliberal/Washington Consensus policies before the turn of the millenium towards increasing focus on pro-poor social policy (119, 120).

On the side of the International Financial Institutions (IFIs), the former WB vice president for Africa, Callisto Modavo, commented in an interview study that "*The resources that could be used for education and health are being used for debt servicing*" (121). A 2007 Center for Global Development working group report, authored by three WB staff members and one retired IMF staff member, however arrived at a number of ways in which the IMF may have constrained health spending in poor countries (98). These admissions are in stark opposition to the working papers and commentaries by IFI staff, who have rejected the validity of studies finding negative associations between IMF policy and health spending (122) (commentary), and defended the protection of the health sector from austerity measures or even increased health spending in IMF-programme countries as compared to non-IMF programme countries (122, 123) (commentary and panel data study). A number of panel data regression analyses and descriptive statistical analyses conducted by IMF and WB staff have had similar findings

providing empirical support to this position, including from the IMF's Independent Evaluation Office (IEO) (124-132).

No studies were identified investigating associations between IMF programmes or their conditionalities and OOP or VHPS.

## 3.7. Literature gaps

As illustrated, there is a considerable body of literature on the influence of EDPs on health financing policy in LMICs. However, a number of gaps, scarcities and limitations in the literature are identified in the following areas: exploring different mechanisms of influence used; IMF/WB policy influence on domestic health financing at the country level; influence of development assistance to the health sector on OOP and VHI; non-health ODA effects on domestic health financing; debt effects on domestic health financing; effect of IMF programs and conditionalities on OOP; comparing effect sizes of both aid, debt and IMF conditionalities on government health spending in the same model. There are also methodological limitations and quality issues in many of the existing econometric studies, and most of this literature is somewhat dated, relating to the MDG era and before.

Most of the identified literature sought to establish the role of EDPs in specific health financing reforms, at times allowing for establishment of *what* influence EDPs had and *why*. While some studies mentioned e.g. financial and technical support, no identified studies had as their primary focus an exploration of the *how*, i.e. the modes or mechanisms of influence, and a systematic unpacking of the mechanisms of EDP influence on the composition of domestic health financing sources is thus lacking in the literature. Addressing this question allows for a comprehensive and comparative analysis of the different pathways and directions of effect across different health financing sources, how they compare and relate, and whether there are areas of inconsistency or policy incoherence. The recent critical commentary from Senegal by Ridde et al. (2024) describing an external push for Community Based Health Insurance (CBHI) by EDPs (39) also motivates a broader empirical enquiry into the role of EDPs in determining the mix of health financing sources present in Senegal and considering equity implications hereof.

While the IMF and WB have faced substantial criticism for the effects of their policies on health financing in the 1980s-2000's, there are fewer recent analyses of their influence on health financing, and in particular a dearth of systematic, qualitative empirical enquiry at the country level. Much of the literature is quantitative, but with sectoral or nationwide policy recommendations and conditionalities, a qualitative approach exploring mechanisms of

influence and detailed policy content is also warranted. Aside from one study examining IMF policy documents for influence on GHE across 16 West African countries between 1995-2014, with brief mention of Senegal (96), a dedicated interrogation of IMF/WB influence on the different domestic health financing sources in Senegal is lacking in the literature. Given the generally highly influential position of these two organisations on fiscal policy in LMICs, and sectoral policy in the case of the WB, a such interrogation could help bring to light areas and mechanisms of influence that might co-determine health financing in Senegal. Such information would be relevant to national stakeholders in health financing, nonetheleast the Senegalese population as they would ultimately be subject to any influence identified. Lessons learnt could potentially also prove relevant for health financing stakeholders in other countries in the region and more broadly, and the IFIs themselves.

While there is a predominance of literature studying fungibility of development assistance to the health sector, the literature examining the relationship between development assistance to the health sector and OOP and voluntary health insurance is considerably scarcer.

There are very few studies that look at how non-health ODA affects health financing, although these intersectoral dynamics should be key questions for a comprehensive understanding of the domestic fiscal implications for the health sector of externally derived development financing.

Studies of how debt affects health spending are also rather limited and very heterogenous in study design, with some divergence of findings. Many of these studies look at general rather than external debt, and broad social spending. Only two studies have taken the step further to examine how government indebtedness affects the reliance on OOP health spending, however they have done so not as the primary research focus, but as covariates and in robustness checks, and not specifically investigating the role of public *external* debt. Apart from a 2002 discussion paper and a 2021 Oxfam case study both from Kenya (85, 87), no other single-country case studies of how debt affects health financing in their country were identified<sup>13</sup>. With the recent expansion of LMIC debt burdens, such questions at the country level are becoming increasingly pressing as more countries are seeing their fiscal space for providing social services threatened by growing debt obligations (88, 89, 133-135).

<sup>&</sup>lt;sup>13</sup> Ouattara (2006) studied overall government spending, not health spending specifically (84).

As for the effects of IMF programmes and conditionalities on health spending, this quantitative literature has focused on effects on government health spending and not whether IMF programmes and conditionalities in turn translate into increased reliance on OOP.

The literature also lacks a comprehensive comparative analysis of the strengths and sizes of the different associations to be investigated, as no studies were identified that compared the effects of aid, debt and IMF conditionalities on health spending in the same model. Such a comparison allows for essential questions of "what matters more" - does aid shift domestic health financing in a particular direction more than debt repayments or vice versa<sup>14</sup>; are IMF conditionalities more important than aid and debt in determining reliance on a given health financing source, etc. Addressing such questions could help focus policy work where it matters most for expanding fiscal space for health and reducing dependence on user fees.

Most of the econometric literature is dated to the Millenium Development Goal era or before, and a new exploration of these relationships into the Sustainable Development Goal (SDG) era is needed.

The System Generalised Method of Moments (GMM) method is one of the newer tools in the econometric toolbox, and many of the cited researchers had to rely on older methods that suffer from more statistical issues, as described in this chapter and in the Methods chapter (Chapter 4). Using System GMM should help overcome some of these issues. A number of the cited studies also suffer from quality issues as described in the footnotes of this chapter, issues that are hoped to be addressed to the extent possible with the estimation approach presented in Chapters 4 and 6.

This thesis aims to address these knowledge gaps by using both global-level econometric methods and country-level mixed-methods to unpack mechanisms, directions and sizes of EDP+IMF influence on domestic health financing sources, enabling a joint interpretation and discussion of findings across methods and across the global/national level, providing original, evidence-based policy and research recommendations.

<sup>&</sup>lt;sup>14</sup> Some authors have included ODA variables and debt variables in the same model, however without also including IMF variables (e.g. (68, 78)).

## 3.8. Hypotheses and conceptual framework

Informed by the reviewed literature, this thesis investigates the following main hypothesized relationships. These are conceptualised as illustrated in the overall conceptual framework for this thesis, presented in Figure 3.1.



Figure 3.1: Main hypothesized relationships in thesis. The objective numbers show which objectives cover each relationship. Objective 1 is explored in Chapter 6, objective 2 in Chapter 7, and objective 3 in Chapter 8.

ODA<sup>+</sup> for health from EDPs can either be fully additional and not displace the government's own health spending from domestic revenue, or it can lead the recipient government to lower its health spending, which is referred to as fungibility. Alternatively, ODA<sup>+</sup> for health could lead to synergy or co-financing, i.e. that the recipient government provides additional funds to the health sector as a result of receiving ODA<sup>+</sup> for health. EDPs can disburse their funds to the central government as budget support or earmark them to the health sector. Both of these can affect domestic government expenditure allocations to the health sector. These relationships are explored mainly in Chapter 6 at the global level and Chapter 7 at the country level. Following the literature presented in the present chapter, this thesis focuses on exploring mainly a fungibility hypothesis whereby domestic government funds for the health sector are displaced by external health funds.

In relative terms, one could hypothesize the following principal relationships between ODA+ for health, OOP and Voluntary Health Insurance (VHI) as shares of Current Health Expenditure (CHE) (referred to as *"reliance on"* OOP and VHI). CHE is a term from the WHO SHA that refers to total consumption of health goods and services by the residents of a country (136, 137). It includes both internal and external financing sources, and does not include capital spending.

- A: *ODA+ for health could decrease reliance on OOP and VHI*. This could occur through supporting the health system via governmental and non-governmental channels leading to a lower reliance on OOP and VHI (covering costs / external subsidy) (71).
- B: ODA+ for health could increase reliance on OOP and VHI. This could occur indirectly via ODA+ for health displacing GHE-S; ODA<sup>+</sup> for health directly leading to increased supply of NGO/private-run facilities with some level of co-payment and thereby increased reliance on OOP; and/or funding pilots and scale-up of CBHI, increasing VHI/CHE (71).

The main hypothesis used is that the displacement effect of ODA<sup>+</sup> for health on GHE-S is the main (indirect) mechanism of effect, increasing reliance on OOP and VHI. Secondarily, the hypothesis is applied that ODA+ for health directly promotes CBHI and thereby VHI in contexts where CBHI receives EDP funding (e.g. (35, 36, 39, 40)), and that ODA+ for health mostly works to cover/subsidise user fees, more than the contrary. The latter hypothesis is used on the background of a diverging literature (60, 67-71). The fungibility mechanism would then lead to a synergistic influence promoting VHI and a mutually counteracting influence on reliance on OOP, as illustrated in Figure 3.1.

These relationships are explored at the global level in Chapter 6 and at the country level in Chapter 7.

Borrowing external funds is a commonly used government strategy to increase fiscal space in the short term (138). This space could be used to increase social spending in the short term and funds could be invested to increase the revenue base for future health spending in the longer term. This could lead to a positive association with government health spending. However, the resulting outward resource flow from borrower countries to creditors as loans and its interests are repaid could constrain fiscal space for health and thereby GHE-S in the longer term and lead to an increased reliance on OOP over time (e.g. (e.g. (78, 80, 81). The Public and Publicly Guaranteed (PPG) external debt stock might also constrain health

spending in the present due to the prospect of future loan repayments. These relationships are explored in Chapters 6 and 8.

Alongside financial transactions are the non-financial influence mechanisms of EDPs+IMF. For EDPs, technical assistance for health financing policy reform working toward UHC appears to be a relatively common phenomenon, while other influence mechanisms are less explored. It is hypothesized that the different non-financial influence mechanisms, that this thesis aims to unpack, follow the general mandates of EDPs to promote equity, UHC and the SDGs, which is conceptualised as relative decreases in OOP and relative increases in financing sources that are pooled including VHI and GHE-S.

Specifically pertaining to loans, these also come with loan conditionalities, and in the case of the IMF, which countries turn to when experiencing balance of payments problems and other kinds of macroeconomic instability (139, 140), these are provided under programs with a macroeconomic framework and accompanying policy recommendations. These policy recommendations and conditionalities generally pertain to the macroeconomic and fiscal level, with the aim of achieving a sustainable balance of payments, macroeconomic stability, economic growth and ensuring that countries are able to repay loans from the IMF and other international creditors (139-141). This has historically involved measures of fiscal austerity, including wage ceilings, trade and private sector liberalization, privatization of state bodies and public health assets (95, 96, 99, 100, 110, 140). In the short term, this might constrain the fiscal space for health leading to lowered GHE-S (95, 96, 99, 100). Loan conditionalities may also require stimulation of the private health service market (95, 99, 100), and combined with privatization measures, this could shift the payment pattern in the health system more toward OOP and VHI. Longer term, positive growth and revenue effects from IMF policy influence could kick in and work to expand the fiscal space for health again. Following the bulk of the independent academic literature, the general hypothesis was however applied that the different non-financial influence mechanisms of the IMF have the effect of decreasing GHE-S, including per CHE, and increasing VHI and OOP / CHE. The WB is a special case in its position as an EDP and at the same time being the sister institution of the IMF. The hypothesis was applied that by having a development mandate, their direction of influence would align more with the UHC-promoting positions of EDPs, however the critical literature warranted a more agnostic approach to exploring their influence in Senegal that could potentially also prove to push the health system toward being characterized more by a predominant private health sector with a higher degree of reliance on OOP and VHI and less GHE-S. These relationships are explored in Chapter 6 at the global level (IMF conditionalities) and Chapter 8 at the country level.

The different presented financial relationships are further motivated and substantiated in the methods chapter (Chapter 4).

As for the presumed relationship between GHE-S on one side and VHI and OOP on the other, these form the bulk of domestically sourced components of CHE in LMICs (137) and were thus presumed to be inversely correlated if measured in relative terms as proportions of CHE, *ceteris paribus* (142).

## **3.9.** Further delineation of research topic

The intricate relationships between EDP+IMF activities and public revenue generation or Domestic Resource Mobilisation (DRM) was not empirically explored, as this constitutes its own separate and complicated area of research. In the same line, no empirical investigations were performed of the relationships between public revenue generation or DRM and GHE-S or social spending, as this again is another extensive, separate body of research, and instead the research rests on the general assumption that more DRM will generally tend to increase the fiscal space for health, provided the health sector is prioritised by the government.

While both are essential, this research focuses on who contributes and not who benefits from health financing, meaning it does not investigate the third health financing function of purchasing health services, or health financing allocations (143). This choice was made for two main reasons: A: To focus the scope of the research; B: Due to the presence of complete, standardised health financing source data for LMICs in the WHO GHED, including for Senegal (137). Correspondingly, matters pertaining to Results-Based Financing and Performance-Based Financing are not included in this thesis, which are provider payment mechanisms and relate to service purchasing. Some considerations of pooling of funds are made in Chapter 7, as this emerged from the findings from Senegal in the case of CBHI, warranting discussion.

The case study of Senegal (Chapter 8) looks at both IMF and WB conditionalities. However the econometric study (Chapter 6) only looks at IMF conditionalities, because these have been formally collected and standardised in a quantitative dataset by IMF Monitor (140), which is not the case for WB conditionalities. The literature review in Section 3.6 on the quantitative relationships between IMF conditionalities and health financing thus reflect these data limitations, and the methods used in Chapter 6 are delineated accordingly.

This thesis focuses on the two largest and most powerful IFIs in terms of their central position in determining domestic policy at the fiscal level (IMF and WB) and the health sector level (WB) (100, 110, 144). This focus comes at the expense of an in-depth interrogation of the content of policy conditionalities from other IFIs that as a general consideration have historically enjoyed less privileged positions, such as the African Development Bank, Islamic Development Bank and others, but also some bilateral EDPs that may have proven influential e.g. in Senegal. This choice was made to delineate the topic where the strongest influence was expected, but it may come at the cost of a more in-depth understanding of the influence from other IFIs and some bilateral creditors. This is further discussed in the thesis limitations section (Chapter 9).

Having reviewed the main literature underlying this thesis and introduced the main hypotheses to be explored as informed by the literature, Chapter 4 describes the methods used for doing so.

## References

1. Kingdon JW. Agendas, alternatives, and public policies: Boston : Little, Brown, [1984] ©1984; 1984. ISBN:

2. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. Health Policy Plan. 1994;9(4):353-70.

3. Meier GM. Politics and policy making in developing countries : perspectives on the new political economy. San Francisco, Calif. :: ICS Press; 1991.

4. Grindle M, Thomas JW. Public choices and policy change : the political economy of reform in developing countries. Baltimore: Johns Hopkins University Press; 1991. ISBN: 0801841550

5. Gautier L, Ridde V. Health financing policies in Sub-Saharan Africa: government ownership or donors' influence? A scoping review of policymaking processes. Glob Health Res Policy. 2017;2:23.

6. Hall PA. The Role of Interests, Institutions, and Ideas in the Comparative Political Economy of the Industrialized Nations Cambridge University Press 1997. ISBN: 0521586682

7. National Collaborating Centre for Healthy Public Policy. Understanding Policy Developments and Choices Through the "3-i" Framework: Interests, Ideas and Institutions. 2014 [cited 2024 July 29]. Available from: https://ccnpps-ncchpp.ca/docs/2014\_ProcPP\_3iFramework\_EN.pdf.

8. Mulvale G, McRae SA, Milicic S. Teasing apart "the tangled web" of influence of policy dialogues: lessons from a case study of dialogues about healthcare reform options for Canada. Implement Sci. 2017;12(1):96.

9. Overseas Development Institute (ODI). ROMA: A guide to policy engagement and influence. 2014 [cited April 9 2024]. Available from: <u>https://odi.cdn.ngo/media/documents/odi\_roma\_guide.pdf</u>.

10. Start D, Hovland I. Tools for Policy Impact: A Handbook for Researchers. 2004 [cited 2024 July 29]. Available from: <a href="https://cdn.odi.org/media/documents/194.pdf">https://cdn.odi.org/media/documents/194.pdf</a>.

11. Crewe E, Young J. Bridging Research and Policy: Context, Evidence and Links. 2002 [cited 2022 July 6]. Available from: <a href="https://cdn.odi.org/media/documents/184.pdf">https://cdn.odi.org/media/documents/184.pdf</a>.

12. Lindquist EA. Discerning Policy Influence: Framework for a Strategic Evaluation of IDRC-Supported Research. 2001 [cited April 9 2024]. Available from:

https://www.betterevaluation.org/sites/default/files/discerning\_policy.pdf.

13. Cathexis Consulting. A Guide to Policy-Influence Evaluation: Selected Resources and Case Studies. 2015 [cited April 9 2024]. Available from:

https://www.nccmt.ca/uploads/media/0001/01/c3374b8be4b35e1340385f1b593d3bb9f50f6a38.pdf.

14. De Raeve P, Bolzonella F, Davidson PM. A Theoretically Derived Approach to Impact: Implementing Policy Influence Strategies. Policy Polit Nurs Pract. 2022:15271544221100164.

15. Steinberg PF. Understanding Policy Change in Developing Countries: The Spheres of Influence Framework. Global Environmental Politics. 2003;3(1):11-32.

16. Haldenwang Cv, Alker M. Policy advice in development cooperation: a distinct type of policy advice. 2009. Contract No.: 15/2009.

17. Sparkes SP, Bump JB, Ozcelik EA, Kutzin J, Reich MR. Political Economy Analysis for Health Financing Reform. Health Syst Reform. 2019;5(3):183-94.

18. Fox AM, Reich MR. The Politics of Universal Health Coverage in Low- and Middle-Income Countries: A Framework for Evaluation and Action. Journal of Health Politics, Policy and Law. 2015;40(5):1023-60.

19. Rizvi SS, Douglas R, Williams OD, Hill PS. The Political Economy of Universal Health Coverage: A Systematic Narrative Review. Health Policy and Planning. 2020;35(3):364-72.

20. Bazbauers AR. World Bank Technical Assistance: The Relational Dynamics of Policy Movement. Public Administration and Development. 2017;37(4):246-59.

21. Akortsu MA, Abor PA. Financing public healthcare institutions in Ghana. J Health Organ Manag. 2011;25(2):128-41.

 Kusi-Ampofo O, Church J, Conteh C, Heinmiller BT. Resistance and change: a multiple streams approach to understanding health policy making in Ghana. J Health Polit Policy Law. 2015;40(1):195-219.
 Koduah A, van Dijk H, Agyepong IA. The role of policy actors and contextual factors in policy agenda setting and formulation: maternal fee exemption policies in Ghana over four and a half decades. Health Res Policy Syst. 2015;13:27.

24. Meessen B, Hercot D, Noirhomme M, Ridde V, Tibouti A, Tashobya CK, et al. Removing user fees in the health sector: a review of policy processes in six sub-Saharan African countries. Health Policy Plan. 2011;26 Suppl 2:ii16-29.

25. Kajula PW, Kintu F, Barugahare J, Neema S. Political analysis of rapid change in Uganda's health financing policy and consequences on service delivery for malaria control. Int J Health Plann Manage. 2004;19 Suppl 1:S133-53.

26. World Bank. Financing Health Services in Developing Countries. 1987 [cited 2023 September 5]. Available from: <u>https://elibrary.worldbank.org/doi/abs/10.1596/978-0-8213-0900-1</u>.

27. Okuonzi SA, Macrae J. Whose policy is it anyway? International and national influences on health policy development in Uganda. Health Policy Plan. 1995;10(2):122-32.

28. Onoka CA, Hanson K, Hanefeld J. Towards universal coverage: a policy analysis of the development of the National Health Insurance Scheme in Nigeria. Health Policy Plan. 2015;30(9):1105-17.

29. Alawode G, Adewoyin AB, Abdulsalam AO, Ilika F, Chukwu C, Mohammed Z, et al. The Political Economy of the Design of the Basic Health Care Provision Fund (BHCPF) in Nigeria: A Retrospective Analysis for Prospective Action. Health Syst Reform. 2022;8(1):2124601.

30. Mhazo AT, Maponga CC. The political economy of health financing reforms in Zimbabwe: a scoping review. Int J Equity Health. 2022;21(1):42.

 Witter S, Chirwa Y, Chandiwana P, Munyati S, Pepukai M, Bertone MP. The political economy of results-based financing: the experience of the health system in Zimbabwe. Glob Health Res Policy. 2019;4:20.
 Nannini M, Biggeri M, Putoto G. Health Coverage and Financial Protection in Uganda: A Political Economy Perspective. Int J Health Policy Manag. 2022;11(9):1894-904.

33. Khalid F, Afzal S, Shirazi A, Quick I, Mataria A. Coordinating external assistance for UHC: Pakistan's early experience of the SDG3 GAP. Health Policy and Planning. 2024;39(2):247-51.

34. Herberholz C. The role of external actors in shaping migrant health insurance in Thailand. PLoS One. 2020;15(7):e0234642.

35. Mulat AK, Mao W, Bharali I, Balkew RB, Yamey G. Scaling up community-based health insurance in Ethiopia: a qualitative study of the benefits and challenges. BMC Health Serv Res. 2022;22(1):473.

36. Kiendrébéogo JA, Meessen B. Ownership of health financing policies in low-income countries: a journey with more than one pathway. BMJ Glob Health. 2019;4(5):e001762.

37. Ir P, Bigdeli M, Meessen B, Van Damme W. Translating knowledge into policy and action to promote health equity: The Health Equity Fund policy process in Cambodia 2000-2008. Health Policy. 2010;96(3):200-9.
38. Jacobs E. The politics of the basic benefit package health reforms in Tajikistan. Glob Health Res Policy. 2019;4:14.

39. Ridde V, Caffin JH, Hane F. External influences over Senegalese health financing policy: delaying universal health coverage? Health Policy Plan. 2024;39(1):80-3.

40. Ridde V, Ba MF, Guyot M, Kane B, Mbow NB, Senghor I, et al. Factors that foster and challenge the sustainability of departmental health insurance units in Senegal. International Social Security Review. 2022;75(2):97-117.

41. Nagemi C, Mwesigwa C. Donor Power and Prioritization in Development Assistance for Health Policies: The Case of Uganda. Journal of Developing Economies. 2021;2(1):54-74.

42. Sridhar D, Gomez EJ. Health financing in Brazil, Russia and India: what role does the international community play? Health Policy Plan. 2011;26(1):12-24.

43. ActionAid International USA, Global AIDS Alliance, Student Global AIDS Campaign, RESULTS Educational Fund. Blocking Progress: How the Fight against HIV/AIDS is Being Undermined by the World Bank and International Monetary Fund. 2004. Available from: <u>http://www.hivpolicy.org/Library/HPP000506.pdf</u>.

44. Colenbrander S, Birungi C, Mbonye AK. Consensus and contention in the priority setting process: examining the health sector in Uganda. Health Policy Plan. 2015;30(5):555-65.

45. Jha A, Kolesar RJ, Comas S, Gribble J, Ugaz J, Gonzalez-Pier E. Getting ready for reduced donor dependency: the co-financing of family planning commodities. Health Policy Plan. 2024;39(1):87-93.

46. Mao W, McDade KK, Huffstetler HE, Dodoo J, Abankwah DNY, Coleman N, et al. Transitioning from donor aid for health: perspectives of national stakeholders in Ghana. BMJ Glob Health. 2021;6(1).

47. Ooms G, Decoster K, Miti K, Rens S, Van Leemput L, Vermeiren P, et al. Crowding out: are relations between international health aid and government health funding too complex to be captured in averages only? The Lancet. 2010;375(9723):1403-5.

48. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJL. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

49. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

50. Sanjeev KS, Shyam N. Is Donors' Concern About the Fungibility of Foreign Aid Justified?: A Panel Data Analysis. The Journal of Developing Areas. 2010;43(2):299-311.

51. Feyzioglu T, Swaroop V, Zhu M. A Panel Data Analysis of the Fungibility of Foreign Aid. The World Bank Economic Review. 1998;12(1):29-58.

52. Devarajan S, Swaroop V. The Implications of Foreign Aid Fungibility for Development Assistance (English). Policy, Research working paper. 1998;no. WPS 2022.

53. Canagarajah S, Williamson T. Uganda's Virtual Poverty Fund Pro-Poor Spending Reform. World Bank. 2008;288(43849).

54. Chatterjee S, Giuliano P, Kaya I. Where Has All the Money Gone? Foreign Aid and the Quest for Growth. IZA Discussion Paper June 2007;No. 2858.

55. Barkat K, Mrabet Z, Alsamara M. Does Official Development Assistance for Health from Developed Countries Displace Government Health Expenditure in Sub-Saharan Countries? Economics Bulletin. 2016;36(3):1616-35.

56. Morrissey O. Aid and Government Fiscal Behavior: Assessing Recent Evidence. World Development. 2015;69:98-105.

57. Mishra P, Newhouse D. Does health aid matter? J Health Econ. 2009;28(4):855-72.

58. Farag M, Nandakumar AK, Wallack SS, Gaumer G, Hodgkin D. Does funding from donors displace government spending for health in developing countries? Health Aff (Millwood). 2009;28(4):1045-55.

59. Stuckler D, Basu S, McKee M. International Monetary Fund and aid displacement. Int J Health Serv. 2011;41(1):67-76.

60. Xu K, Saksena P, Holly A. The determinants of health expenditure: A country-level panel data analysis. Results for Development Institute WHO; 2011.

61. Gebrehanna S, Upadhyay MP. Fungibility of Foreign Aid in Selected African Countries. Indian Development Review. 2012;10(1):63-80.

62. Van de Sijpe N. The fungibility of health aid reconsidered. 2013.

63. Van de Sijpe N. Is Foreign Aid Fungible? Evidence from the Education and Health Sectors. World Bank Economic Review. 2013;27(2):320-56.

64. Fernandes Antunes Á, Xu K, James CD, Saksena P, Van de Maele N, Carrin G, et al. General budget support: has it benefited the health sector? Health economics. 2013;22(12):1440-51.

65. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

66. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

67. Younsi M, Chakroun M, Nafla A. Robust analysis of the determinants of healthcare expenditure growth: evidence from panel data for low-, middle- and high-income countries. International Journal of Health Planning and Management. 2016;31(4):580-601.

68. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

69. Ali MEM, Ebaidalla M, Rizk R. Foreign aid and out-of-pocket health expenditure in Sub-Saharan Africa: does institutional development matter. International Journal of Sustainable Economy. 2020;12(3).

70. Frimpong AO, Amporfu E, Arthur E. Effects of public and external health spending on out-of-pocket payments for healthcare in sub-Saharan Africa. Health Policy Plan. 2022;37(9):1129-37.

71. Gabani J, Suhrcke M, Neelsen S, Eozenou PH, Smitz MF. Does health aid matter to financial risk protection? A regression analysis across 159 household surveys, 2000-2016. Social science & medicine (1982). 2024;356:117148.

72. Roodman D. How to do xtabond2: An introduction to difference and system GMM in Stata. Stata Journal. 2009;9(1):86-136.

73. Roodman D. A Note on the Theme of Too Many Instruments. Oxford Bulletin of Economics and Statistics. 2009;71(1):135-58.

74. Leiter AM, Theurl E. Determinants of prepaid systems of healthcare financing: a worldwide country-level perspective. Int J Health Econ Manag. 2021;21(3):317-44.

Finkelstein A, et al. The Oregon Health Insurance Experiment: Evidence from the First Year. 2011.
Murthy VN, Okunade AA. The core determinants of health expenditure in the African context: some econometric evidence for policy. Health Policy. 2009;91(1):57-62.

77. Boachie MK, Agyemang J, Immurana M. Health sector funding in Ghana: The effect of IMF conditionalities. Dialogues Health. 2022;1:100045.

78. Fosu AK. The External Debt-Servicing Constraint and Public Expenditure Composition. Helsinki, Finland: UNU-WIDER; 2007.

79. Fosu AK. The External Debt-Servicing Constraint and Public-Expenditure Composition in Sub-Saharan Africa. African Development Review. 2010;22(3):378-93.

80. Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 2007;8(4):299-310.

81. Shabbir S, Yasin HM. Implications of Public External Debt for Social Spending: A Case Study of Selected Asian Developing Countries. Lahore Journal of Economics. 2015;20(1):71-103.

82. Behera DK, Dash U. Impact of macro-fiscal determinants on health financing: empirical evidence from low-and middle-income countries. Glob Health Res Policy. 2019;4:21.

 Lahiani A, Mtibaa A, Gabsi F. Fiscal Consolidation, Social Sector Expenditures and Twin Deficit Hypothesis: Evidence from Emerging and Middle-Income Countries. Comp Econ Stud. 2022;64(4):710-47.
 Ouattara B. Aid, Debt and Fiscal Policies in Senegal. Journal of International Development. 2006;18(8):1105-22.

85. Kimalu PK. Debt relief and health care in Kenya. World Institute for Development Economics (UNU-WIDER). 2002;WIDER Discussion Papers(No. 2002/65, ISBN 9291902551, UNU-WIDER, Helsinki).
86. Yiega V. The Impact of Foreign Debt on Economic and Social Rights: A Case Study on the Right to

86. Yiega V. The Impact of Foreign Debt on Economic and Social Rights: A Case Study on the Right to Health in Kenya, Nigeria and Zambia. ADHR (CFS). 2022; (Research Paper 06/06).

87. OXFAM. Policy report: Impact of debt servicing on social spending and wellbeing of low- income households in Kenya: a case study of Nairobi city and Mombasa counties. 2021 [cited 2024 March 26]. Available from: <a href="https://kenya.oxfam.org/latest/publications/impact-debt-servicing-social-spending-and-wellbeing-low-income-households-kenya">https://kenya.oxfam.org/latest/publications/impact-debt-servicing-social-spending-and-wellbeing-low-income-households-kenya</a>.

88. Kurowski C, Evans DB, Tandon A, Eozenou PH-V, Schmidt M, Irwin A, et al. From Double Shock to Double Recovery – Implications and Options for Health Financing in the Time of COVID-19. 2022 [cited 2023 September 15]. Available from: <u>https://openknowledge.worldbank.org/server/api/core/bitstreams/76d5786b-9501-5235-922a-caa71f99f0fc/content</u>.

89. Fan V, Gupta S. What's Rising Debt Got to Do With Health Spending? 2023 [cited 2023 September 27]. Available from: <u>https://www.cgdev.org/blog/whats-rising-debt-got-do-health-spending</u>.

90. Clements B, Gupta S, Khamidova S. A Fiscal Straitjacket is Stifling Health Spending in Developing Countries. 2023 [cited 2023 September 28]. Available from: <u>https://www.cgdev.org/blog/fiscal-straitjacket-stifling-health-spending-developing-countries</u>.

91. Birungi C, Azcona JA, Munevar D. A pandemic triad: HIV, COVID-19 and debt in low- and middleincome countries. Afr J AIDS Res. 2022;21(2):110-22.

92. Nguyen TC, Castro V, Wood J. The impact of financial crises on social spending: Delving into the effects in developed and developing countries. International Journal of Finance & Economics. 2023.

93. Said R, Morai AS. Relationship between Public Debt Burden and Health Expenditure in Sub-Saharan African Countries: The Role of Institutional Quality. Journal of Business and Social Review in Emerging Economies. 2020;6(2):493-502.

94. Ooms G, Schrecker T. Expenditure ceilings, multilateral financial institutions, and the health of poor populations. The Lancet. 2005;365(9473):1821-3.

95. Stuckler D, Basu S. The International Monetary Fund's effects on global health: before and after the 2008 financial crisis. Int J Health Serv. 2009;39(4):771-81.

96. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

97. Rowden R. Why Health Advocates Must Get Involved in Development Economics: The Case of the International Monetary Fund. International Journal of Health Services. 2010;40(1):183-7.

Center for Global Development. Does the IMF Constrain Health Spending in Poor Countries? . 2007
 [cited 2024 March 19]. Available from: <u>https://www.cgdev.org/sites/default/files/14103\_file\_IMF\_report.pdf</u>.
 Brunswijck G. Unhealthy Conditions. Eurodad; 2018.

100. Stuckler D, Basu S. The Body Economic: Why Austerity Kills: Basic Books; 2013. ISBN: 978-0465063987

101. Abocejo F. The Impact of International Monetary Fund (IMF) Structural Adjustment Policies (SAP) on the Philippines. Recoletos Multidisciplinary Research Journal. 2014;2(1):19-24.

102. Isiani MC, Anthonia Obi-Ani N, Obi-Ani P, Chidume CG, Okoye-Ugwu S. Interrogating the International Monetary Fund (IMF) Policies in Nigeria, 1986–2018. Cogent Arts & Humanities. 2021;8(1).

103. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

104. Foley EE. Your pocket is what cures you: The Politics of health in Senegal: Rutgers University Press; 2010. ISBN: 978-0-8135-4667-4

105. Meurs M, Seidelmann L, Koutsoumpa M. How healthy is a 'healthy economy'? Incompatibility between current pathways towards SDG3 and SDG8. Global Health. 2019;15(1):83.

106. Birn AE, Nervi L, Siqueira E. Neoliberalism Redux: The Global Health Policy Agenda and the Politics of Cooptation in Latin America and Beyond. Development and Change. 2016;47(4):734-59.

107. Nooruddin I, Simmons JW. The Politics of Hard Choices: IMF Programs and Government Spending. International Organization. 2006;60(04).

108. Ruckert A, Labonte R. The financial crisis and global health: the International Monetary Fund's (IMF) policy response. Health Promot Int. 2013;28(3):357-66.

109. Kentikelenis A, Stubbs T. IMF Social Spending Floors: A fig leaf for austerity? 2023 [cited 2023 September 29]. Available from: <u>https://oxfamilibrary.openrepository.com/bitstream/handle/10546/621495/bp-imf-social-spending-floors-130423-en.pdf?sequence=4</u>.

110. Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A, Stubbs T, editors: Oxford University Press; 2023. 0 p. ISBN: 9780190637736

111. Stubbs T, Kentikelenis A, Gabor D, Ghosh J, McKee M. The return of austerity imperils global health. BMJ Glob Health. 2023;8(2).

112. Stubbs T, Kentikelenis A, Ray R, Gallagher KP. Poverty, Inequality, and the International Monetary Fund: How Austerity Hurts the Poor and Widens Inequality. Journal of Globalization and Development. 2022;13(1):61-89.

113. Cornia GA, Jolly R, Stewart F. Adjustment with a human face, Vol. 1. Oxford [Oxfordshire]; Oxford [Oxfordshire]; New York: Clarendon Press ; Oxford University Press; 1987. ISBN: 0198286104

114. Odaga J, Lochoro P. Budget ceilings and health in Uganda. 2006. Available from:

https://www.researchgate.net/publication/265273554 Budget ceilings and health in Uganda. 115. Stubbs T, Kentikelenis A. Targeted social safeguards in the age of universal social protection: the IMF and health systems of low-income countries. Critical Public Health. 2017;28(2):132-9.

116. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

117. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

118. Homedes N, Ugalde A. Twenty-five years of convoluted health reforms in Mexico. PLoS Med. 2009;6(8):e1000124.

119. Cormier B, Manger MS. The Evolution of World Bank Conditionality: A Quantitative Text Analysis. 2020.
120. Vetterlein A. Economic Growth, Poverty Reduction, and the Role of Social Policies: The Evolution of the

World Bank's Social Development Approach. Global Governance. 2007;13(4):513-33.

121. Abbasi K. The World Bank and world health: Changing sides. Bmj. 1999;318(7187):865-9.

122. Gupta S. Can a causal link be drawn? A commentary on "the impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations". Social science & medicine (1982). 2017;181:199-201.

123. Clements B, Gupta S, Nozaki M. What happens to social spending in IMF-supported programmes? Applied Economics. 2013;45(28):4022-33.

124. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs. International Monetary Fund; 2004.

125. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

126. IMF Independent Evaluation Office. Fiscal Adjustment in IMF-Supported Programs. 2003 [cited 2024 March 19]. Available from: <u>https://www.imf.org/external/np/ieo/2003/fis/pdf/main.pdf</u>.

127. Clements B, Gupta S, Nozaki M. What Happens to Social Spending in IMF-Supported Programs? . 2011. Available from: <u>http://www.imf.org/external/pubs/ft/sdn/2011/sdn1115.pdf</u>.

128. IMF. Social safeguards and program design in PRGT and PSI-supported programs. 2017 [cited 2024 March 20]. Available from: <a href="https://www.imf.org/en/Publications/Policy-Papers/Issues/2017/06/01/pp042117social-safeguards-and-program-design-in-prgt-and-psi.">https://www.imf.org/en/Publications/Policy-Papers/Issues/2017/06/01/pp042117social-safeguards-and-program-design-in-prgt-and-psi.</a>

129. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

130. A Strategy for IMF Engagement on Social Spending. Policy Papers. 2019;19(017).

131. IMF. A Strategy for IMF Engagement on Social Spending. 2019 [cited 2023 August 22]. Available from: <u>https://www.imf.org/en/Publications/Policy-Papers/Issues/2019/06/10/A-Strategy-for-IMF-Engagement-on-Social-</u> Spending-46975.

132. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs 2004. Available from: http://www.ieo-imf.org/ieo/files/completedevaluations/BP041.pdf.

133. United Nations DESA FSDO. Financing for Sustainable Development Report 2024. 2024 [cited 2024 April 17]. Available from: <u>https://desapublications.un.org/publications/financing-sustainable-development-report-2024</u>.

134. United Nations Global Crisis Response Group. A world of debt: A growing burden to global prosperity. 2023 [cited 2024 March 25]. Available from: <u>https://www.un.org/regionalcommissionsnyoffice/news/launch-un-report-%E2%80%9C-world-debt%E2%80%9D.</u>

135. Bank W. International Debt Report 2023. 2023 [cited 2024 February 22]. Available from: https://www.worldbank.org/en/programs/debt-statistics/idr/products.

136. OECD, Eurostat, WHO. A System of Health Accounts. OECD Publishing. 2011.

137. WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from:

http://apps.who.int/nha/database.

138. IMF. Back to Basics -- Fiscal Space: What It Is and How to Get It. 2005 [cited 2024 June 19]. Available from: <u>https://www.imf.org/external/pubs/ft/fandd/2005/06/basics.htm</u>.

139. IMF. IMF conditionality. 2023 [cited 2023 August 22]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/IMF-Conditionality.

140. IMF Monitor. IMF Conditionality Dataset, 1980-2019. 2023 [cited 2023 November 2]. Available from: https://imfmonitor.org/conditionality/.

141. IMF. IMF Conditionality. 2023 [cited 2023 December 18]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/IMF-Conditionality.

142. McIntyre D, Meheus F, Rottingen J-A. What Level of Domestic Government Health Expenditure Should We Aspire to for Universal Health Coverage? Health Economics, Policy and Law. 2017;12(2):125-37.

143. Schieber G, Baeza C, Kress D, Maier M. Financing Health Systems in the 21st Century. In: Jamison DT, Breman JG, Measham AR, Alleyne G, Claeson M, Evans DB, et al., editors. Disease Control Priorities in Developing Countries. Washington (DC) New York: The International Bank for Reconstruction and Development / The World Bank

Oxford University Press Copyright © 2006, The International Bank for Reconstruction and Development/The World Bank Group.; 2006.

144. Millet D, Toussaint E. Who Owes Who? 1 ed: Zed Books; 2004. 185 p. ISBN: 1842774271

## 4. Chapter 4: Methods

This chapter describes the methods used in this thesis. It begins with a description of the overall study approach, explains the study design for the multi-country panel data analysis and the country case study in Senegal, and ends with discussions of the ethical considerations and epistemological position of the research. Methodological limitations are discussed in Chapter 9 – Discussion.

## 4.1. Study approach

This research employs a mixed-methods approach using panel data regression analysis, descriptive statistical analysis, interviews and document review. Mixed methods can be used to gain both a factual, descriptive understanding of a research question, including by crosschecking information, while also reaching a deeper and more comprehensive understanding of the research question. This can occur through exploring concepts that are not easily guantifiable and eliciting potential explanations for observations made, which improves the interpretability of findings and expands the scope for generation of hypotheses and theories (7, 8). A mixed methods approach was deemed necessary to be able to unpack and understand the complex interplay between External Development Partners (EDPs) + International Monetary Fund (IMF), their funding flows and associated non-financial influence pathways, global and local macroeconomic circumstances, and national health financing sources and associated government policy. The results chapters progress from the global to the country level and begin by exploring the "what" of the research questions, seeking to establish directions and sizes of effect at the global level. The following chapters then unpack the "how", by performing an in-depth exploration of mechanisms of effect as identified in Senegal.

## 4.2. Global panel data study

The global-level panel data study attempted to answer the research question of whether and how official development assistance, public external debt repayments and IMF loan conditionalities affect the degree of reliance on domestic government health financing vs. OOP in recipient/borrower countries. For this study, secondary observational data from a number of databases were compiled as outlined below.

### 4.2.1. Variables

#### 4.2.1.1. Dependent variables

Dependent variables were Government Health Expenditure as a Source (GHE-S)/Current Health Expenditure (CHE), Out Of Pocket payments (OOP)/CHE, GHE-S/Gross Domestic

Product (GDP), and OOP/GDP. GHE-S and OOP were chosen as these two sources jointly make up nearly 80% of health financing in Low- and Middle-Income Countries (LMICs) (10), with contrasting levels of (/absent) progressivity, financial risk protection, prepayment, pooling, and cross-subsidy, with associated consequences for health service access for the poor and in turn health outcomes, as explained in Chapter 2 (11-18). Interpretations are thereby clearer than for Voluntary Health Insurance (VHI), which makes up only 5% of LMIC health financing and may cover a range of different financing schemes for different population groups in different countries, challenging interpretation at the cross-country level (see Chapter 9: Discussion for further details) (10).

The degree of progressivity of health financing can be measured in financing incidence analysis using the Kakwani index (11, 20) (e.g. (21, 22)), however cross-country time-series data of health financing progressivity are not available. Instead, GHE-S/CHE and OOP/CHE were used, based on the general rationale that when household OOP becomes a more predominant source of health financing and GHE-S less so, the burden of payment is shifted from taxpayer, who contributes in accordance with their ability to pay, onto the patient at the point of care, irrespective of their ability to pay. At a crude macro-level, the degree of health financing progressivity should resultingly worsen. These two variables measure each health financing source relative to total current health financing. They respond to absolute increases and decreases in OOP and GHES, as well as changes in the other sources constituting the denominator of CHE. They also have a mutual accounting relationship, whereby GHE-S + OOP + Voluntary Health care Payment Schemes (VHPS) + Social Health Insurance (SHI) + External Health Financing (EXT)  $\approx$  CHE, which means that changes in the proportion of one health financing source relative to CHE will by definition lead to changes in the proportions of the others relative to CHE, all things equal. The dependent variables GHE-S/GDP and OOP/GDP were therefore added. By dividing by GDP, the introduction of spurious correlations from dividing dependent and independent financial variables with the same deflator numbers was avoided, which would have been necessary to adjust for inflation, if financial variables were measured as absolute amounts or per capita (25). It also meant that a population covariate could be omitted (necessary in the case of measuring variables directly), increasing parsimoniousness and degrees of freedom and reducing the risks of multicollinearity, instrument proliferation and overidentification. Measuring directly would also make some economic interpretations less meaningful (e.g. "entering into an IMF programme resulted in X\$ higher/lower OOP expenditure", or X% if log-transformed).

Changes in GHE-S/GDP have been commonly analysed in the literature (e.g. (29-32)), however no other literature was identified that examined dynamics of OOP/GDP as a

65

dependent variable, but OOP was divided by GDP as for GHE-S for consistency. Doing so accounts for the fact that larger economies have overall higher levels of health spending, both public and private, and changes in these variables are thus reflective of a relative expansion or contraction of the given health financing source compared to overall growth rates in the economy. Growth and inflation rates are thus accounted for, and increases in GHE-S/GDP should, if spent effectively, translate into a real expansion of health services, beyond what would be expected if the government was raising a fixed proportion of GDP as revenue and allocating a fixed proportion of General Government Expenditure (GGE) to health. This implies increased fiscal space for health and/or increased political priority of the health sector. Increases in GHE-S/CHE on the other hand can occur merely as a result of decreases in any other health source. Technically, this should improve overall financing progressivity, however it is necessary to consider and interpret changes in GHE-S/CHE along with changes in the other 3 dependent variables. Increases in OOP/GDP reflects a relative expansion of direct payments to the health system beyond what would be expected as a result of economic growth and inflation. This implies a relative expansion of private health sector services and/or public health services funded by user fees. Increases in OOP/CHE mean that user fees are on aggregate expanding more than other sources, or as before, that other sources are decreasing relative to OOP. A joint interpretation of all four variables is thus necessary and should together provide an impression of the overall effects of independent variables on the balance between the two main health financing sources in LMICs, public vs. private health financing, and ensuing degree of overall health financing progressivity.

The data source for GHE-S and OOP was the World Health Organisation (WHO) Global Health Expenditure Database (GHED) (10), while GDP data were obtained from the World Bank (WB) databank (33).

#### 4.2.1.2. Independent variables

#### 4.2.1.2.1. Official development assistance variables

Four variables were used to measure the influence of Official Development Assistance<sup>+</sup> (ODA<sup>+</sup>) on the outcomes of interest: Disbursements on ODA<sup>+</sup> for health / GDP and ODA<sup>+</sup> for non-health purposes / GDP, both disaggregated by channel of assistance via government or civil/private sector. Debt relief, administrative expenses and other in-donor country expenses were removed to count only real transfers to the recipient country. Based on the literature (29-31, 34-45), as per Chapter 2, the main hypothesis was established that increases in  $\frac{ODA^+ for health}{GDP}$  would be associated with decreases in GHE-S/GDP and decreases in GHE-S/CHE as well, partly because EXT ( $\approx$  ODA<sup>+</sup> for health) is part of CHE. Following several authors (29-31, 39, 40, 42), it was further hypothesized that this would depend on funds being channeled "on-budget", i.e. via the government. The mirror-hypothesis for non-health ODA<sup>+</sup>

was that fungibility between sectors could occur when disbursements were channeled via the government, leading to an increase in GHE-S/GDP, and potentially in GHE-S/CHE. The hypothesis for GHE-S/GDP relies on positive intersectoral fungibility effects benefiting the health sector being larger than positive economic growth effects.

OOP/CHE could correspondingly be positively associated with  $\frac{ODA^+ for health}{GDP}$ , and negatively associated with  $\frac{ODA^+ for non-health}{GDP}$ , both mediated via effects on GHE-S/CHE. Two possible effects act counter to the main hypothesized positive relationship between  $\frac{ODA^+ for health}{GDP}$  and OOP/CHE: A direct replacement effect from EDPs covering health service fees for users would decrease OOP/CHE, and the accounting relationship between EXT and OOP both being parts of CHE would also drive a negative association.

Data on ODA<sup>+</sup> disbursements were obtained from the Organisation for Economic Cooperation and Development (OECD) Creditor Reporting System (CRS) database (46).

#### 4.2.1.2.2. Public external debt variables

Public and Publicly Guaranteed (PPG) external debt service / GDP, and PPG external debt stock / GDP were used as the main debt variables. The inclusion of these two variables follows the hypothesis of a constraining effect of the public external debt burden on fiscal space for health and hence on GHE-S, as indicated by most of the available studies on the topic (31, 43, 47-51). Both the servicing flows and the stock of debt were included following Lora and Olivera (2006) (47), and because the stock of debt / GDP could serve an adjustment purpose: Adjusted for the size of the debt relative to GDP, are increases in debt servicing / GDP associated with decreases in GHE-S/GDP and GHE-S/CHE? This was done as paying off large chunks of a small debt could reflect healthy fiscal conditions and ample external currency reserves, allowing a government to get rid of some of its relatively small external debt. Conversely, small repayments on a large external debt could indicate a struggling government where debt is amassing but repayments do not follow.

Studies on debt and OOP were scarce and inconsistent (31, 52). A positive relationship between  $\frac{PPG \text{ external debt service}}{GDP}$  and OOP/CHE was hypothesized, mediated via the hypothesized decreases in GHE-S/GDP and GHE-S/CHE from debt servicing.

Data for the debt variables were obtained from the WB International Debt Statistics database (53).

#### 4.2.1.2.3. IMF variables

IMF programmes and their conditionalities are key factors determining fiscal conditions in IMF programme participant countries, which could have implications for all sectors including the

health sector. A variable counting the number of IMF conditions in a country-year was therefore included, along with an IMF programme participation dummy, taking the value of 1 when there were conditions in place in a country-year and 0 when not. The main hypothesis was that having more IMF conditionalities could lead to fiscal austerity, spending cuts, privatisation, wage ceilings etc., leading to lower government health spending and a higher reliance on OOP. Additionally, being in an IMF programme could by itself have some of these effects via policy advice, technical advice, transfers of knowledge, beliefs, ideology or other exchanges that exist around conditionalities themselves (2, 54-65).

The research from IMF/WB staff runs counter to this general hypothesis. Their findings would support a hypothesis stating that IMF programmes and their conditionalities support countries struggling economically to reduce their fiscal deficit, generate revenue, improve budget management and execution, and social sectors could be insulated from any spending cuts via e.g. spending floors, which could lead to more government spending on health and less reliance on OOP (66-70).

Data for IMF conditionalities came from the IMF Monitor conditionality dataset<sup>15</sup> (72).

#### 4.2.1.3. Covariates

A number of covariates were included to control for other factors that could influence the dependent variables.

GDP per capita was included as this could co-determine the outcome variables (e.g. government health spending (2, 4, 29, 73, 74)), independent variables of interest (e.g. aid (75)), and other covariates (e.g. Infant Mortality Rate (IMR) (76)) (33). Adjusting for GDP ( / capita) is standard practice across the literature examined exploring the same/similar outcome variables (examples: (2, 30, 37, 43, 50, 77)). These data were obtained from the WB Databank (33).

IMR was included as a proxy for unmet population health need (78-80). Increases in IMR could be codetermined by decreases in GHE-S, and conversely, increases in IMR could drive both GHE-S and OOP. IMR has been found to be a stronger determinant of ODA for health than the Under-5 Mortality Rate (U5MR) and HIV prevalence (79). IMR was preferred to alternative health need variables for this reason, and because e.g. HIV prevalence was deemed to be a

<sup>&</sup>lt;sup>15</sup> The IMF's own MONA database (71) has been heavily criticized for an ad hoc manner of data collection, the presence of duplicates, a break in reporting in 2002, certain key pieces of information on conditions missing, and underreporting and misclassification of conditions being ubiquitous (2), so it was therefore decided to use the IMF monitor database instead.

stronger driver of ODA for health in the (early/mid) Millenium Development Goal era, and it is much more epidemiologically concentrated than IMR, potentially serving as a de-facto proxy for Southern Africa, rather than a more general health need predictor. The dependency ratio has the issue that in Upper-Middle Income Countries (UMICs) it may predominantly proxy the health needs of the elderly, whereas in Low Income Countries (LICs) it may almost exclusively proxy the health needs of children, with resulting challenges for modelling and interpretation. IMR data were obtained from the United Nations Inter-Agency Group for Child Mortality Estimation (81).

Two variables respectively measuring government effectiveness and control of corruption were included from the Worldwide Governance Indicators database (82). These are surveybased and capture perspectives from respondents on different dimensions of respectively government effectiveness and corruption, which are aggregated into single indicators, and measured as standard deviations on a normal distribution (82). More corrupt governments have been found to spend less on health (83, 84), and the same has been found for less effective governments (10, 82). The variables are frequently used in the development assistance literature as they are found to codetermine both flows and effectiveness of development assistance (e.g. (85-93)). Some have found evidence of development assistance being completely offset by decreases in domestic revenue among the most corrupt countries (94). Counter to intuition, more corrupt governments have however been found to implement aid to the health sector more effectively, leading to better health service coverage (95). This finding was explained as follows: "They seek to please donors with progress in the health sector, to justify additional flows and continue their rent-seeking behavior in more lucrative sectors" (95). Additionally, country governance may jointly determine and be determined by IMF programme participation, and which recommendations they provide to a country (96).

An armed conflict covariate was included, as armed conflict has been linked to lower government health spending (34). Conflict might also lead to more OOP/CHE due to destabilisation of government functions in terms of revenue generation, planning, fiscal management and budget execution for health, along with competing military expenditures, and health insurance schemes might also destabilise. Effects on OOP/GDP are more difficult to predict as both might drop but at different speeds, or OOP might increase depending on particular circumstances in the given conflict, although user fees require functional health services to pay for. Conflict has by some been found to determine development assistance and lending flows (97, 98), and by others not (99). The Uppsala Conflict Data Program / Peace Research Institute Oslo Battle-Related Deaths Dataset was used for this variable (100). It was

measured as a dummy variable taking the value of 0 if the number of battle-related deaths in a country-year was less than 1 per 1 million population and 1 if equal to or higher.

Lastly three dummy variables were included to measure colonial legacy from either UK, France or Spain using the Issue Correlates Of War dataset (101). Country health systems, financing policies and mechanisms, as well as other systems and institutions, are partly shaped by their colonial legacy and the continued policy influence from the formerly colonizing bilateral donors (102-105). This could co-determine the dependent variables by shaping the payment patterns seen in the health sector, as different former colonizers could have different ideas and preferences for how health care should be financed in their former colonies, perhaps to some extent reflecting the systems and schemes present in their own countries. Bilateral donors are also known to give more development assistance to their former colonies (e.g. (95, 106-108)).

#### 4.2.2. Variable transformations and data sample

#### 4.2.2.1. Variable transformations

The following variables were natural log-transformed due to right skew: GHE-S/GDP, OOP/GDP,  $\frac{ODA^+ for health}{GDP}$  and  $\frac{ODA^+ for non-health}{GDP}$  (separated by government and civil/private sector channel),  $\frac{PPG \text{ external debt service}}{GDP}$ ,  $\frac{PPG \text{ external debt stock}}{GDP}$ , and GDP/capita. This was also done to facilitate the interpretation of relationships as elasticities, and it is commonly applied in the econometric health financing literature (e.g. (31, 37, 45, 48)).

As motivated in Section 4.2.1, all dependent and independent financial variables were measured per GDP similar to other authors (e.g. (29, 30, 47, 78)) (except of course GDP/capita).

Values of all current US\$-financial variables in the models including alternative models in sensitivity tests were deflated into constant 2020 US\$ using annual average Consumer Price Index (CPI) data (33).

#### 4.2.2.2. Data sample

Data availability allowed us to examine the time period 2005-2019 for a final panel of 105 LMICs (see Appendix 4 for full list). Countries with more than five years of missing data for the main dependent and independent variables of interest or GDP were excluded. This threshold corresponds to 33% of the data for a given time series, which means that a linear regression for a such country would be based only on ten time units, possibly disjointed by significant gaps, with resulting reservations regarding the quality and validity of a such regression.

Countries excluded in this process were mainly relatively small countries with likely limited ability to gather and report national statistics (examples: Kiribati, Saint Lucia, Dominica), and countries severely affected by conflict with large gaps in data reporting (examples: Syria, Libya, Afghanistan). Gaps in one key data series might also reflect more general issues with national statistical capacity, and the process of exclusion may therefore have reduced the degree of measurement error in the final dataset (see Chapter 9: limitations).

#### 4.2.3. Choice of estimator and diagnostic testing

The final dataset was an unbalanced panel of n = 105 and t = 15. For this panel shape, the Generalized Method of Moments (GMM) estimator is most appropriate and comes with several advantages over alternative estimators (19, 26, 109-112). After its invention, GMM was broadly applied across the econometric literature (109). It has the advantage of enabling statistical inference when the probability distribution of the data is not known (109), which is typically the case in socioeconomic panel data sets. It does not rely on prior distributional assumptions, and under certain regularity conditions it provides consistent, efficient and asymptotically normal estimates (23). Specifically, the system GMM estimator was used, a subtype of GMM estimators. System GMM has been specifically designed for panel analysis with a small number of time periods and a large number of units of observation (19). It builds a system, or a matrix, of equations that include both the direct levels of variables included, and first-differenced equations, which greatly improves efficiency (19). It also allows for a large set of lagged instruments, or moment conditions, to be generated, after which they can be "collapsed" (19). This removes redundant 0-values and leaves a set of instruments that help describe the relationship between explanatory and outcome variables in a manner that is more independent from the influence of the error term (i.e. it helps address the endogeneity problem) (113). Endogeneity was expected in most of the variables through measurement error (see Chapter 9: limitations) and simultaneity. For example ODA might affect government health spending but countries with lower government health spending might receive more ODA, increases in IMR might drive increases in public and private health spending to address increasing health needs, or IMR might increase as a result of lower health spending as previously mentioned, etc. System GMM also allows one to keep factors in the levels equation that may apply across multiple countries and have fixed effects over time such as the colonial legacy variables (and alternative variables in sensitivity testing), as opposed to difference GMM where these are cancelled out (19, 114). Specifically, two-step system GMM was used, as it is a more efficient estimator than regular GMM (19, 112).

In comparison, using other common panel data estimators can be problematic: The Ordinary Least Squares (OLS) estimator is the Best Linear Unbiased Estimator (BLUE) only when all

Gauss-Markov assumptions are met. These can rarely be met for panel data that may suffer from endogeneity, heteroskedasticity and autocorrelation, and the probability distribution of the data may not be known, as mentioned previously. The latter also rules out the use of other Maximum Likelihood Estimation (MLE) methods. The Two-Staged Least Squares (2SLS) estimator, an Instrumental Variable (IV)-estimator, would be feasible, but it mainly performs well for balanced datasets<sup>16</sup> under homoskedasticity (19). GMM models the error structure comparatively more accurately in the presence of heteroskedasticity and serial correlation and does not have the same issues with unbalanced data (19).

One problem with the GMM method is that it produces downward-biased standard errors (19, 115). This problem can, however, be corrected with the Windmeijer-correction, which was applied (115). Following the literature, the Hausman test (116) was used to examine whether a random- or fixed-effects model was more appropriate for describing the data (examples: (37, 43, 50, 77, 117)), which fell in strong favour of a fixed-effects approach such as GMM (p=0.0000 for all models). The Arellano-Bond test was used to test for first-order serial correlation in levels (19, 110, 118, 119). The Hansen test (26) was used to examine the validity of instruments while avoiding instrument proliferation and overidentification (118, 120, 121) (Box 1).

<sup>&</sup>lt;sup>16</sup> 2SLS drops those observations where lagged observations are missing, thus magnifying gaps in the dataset, which system GMM does not (19).
#### Box 1: The issue of instrument proliferation in GMM

The main advantage of the system GMM estimator is also its weakness, namely the introduction of time-lagged variables into models to address endogeneity (6). As the number of instruments increases, the explanatory power of each independent variable is diluted while the  $R^2$  of the model goes toward 1.0, the precision of estimates declines and the model becomes unusable for prediction (6). This is referred to as "overfitting" the model (6, 19). Another problem with instrument proliferation is excessive "overidentification", i.e. if one's model contains so many moment conditions that it makes it impossible to estimate whether one's instruments are exogenous (i.e. they do not correlate with the error term) and hence valid (19, 23). The original Sargan- (24) and Hansen tests (26) for overidentification have been shown to be invalid in the face of instrument proliferation (6, 27, 28) – which naturally tend to coincide. An immediate sign of instrument proliferation is a Sargan/Hansen test p-value of the implausibly high number of 1.000 (6, 19). The issue of instrument proliferation necessitates limitation of the number of instruments used.

The null hypothesis of the Hansen test is that the generated instruments (or overidentifying restrictions) are exogenous, i.e. they do not correlate with the error term, and are thereby jointy valid (6). There is no hard and fast rule about the exact values of the Hansen-test that can be considered "safe", but a value between 0.05 and 0.1 is generally not considered enough to convincingly accept the null (118, 120, 121). The different recommended limits in the theoretical literature were examined as well as those used in the empirical literature, and the value of 0.15 was adopted as a lower safety threshold, but this decision has an unavoidable element of arbitrarity. When instrument proliferation becomes an issue is also not clear, and different recommendations and empirical applications were similarly examined, arriving at p=0.6 as an upper limit, as well as the rule of thumb that instruments must not outnumber units of observation (19, 120).

Recommendations by Roodman (2009) were further followed, and the difference-in-Hansen test was applied for the subset of GMM instruments generated in the levels equation, which also has the null-hypothesis that these are exogenous, i.e. valid (6, 19). For both this and the Arellano-Bond test, a lower cut-off of p=0.15 was again applied as a safety threshold for consistency.

In order to be able to achieve acceptable diagnostic test statistics, the instrument matrix was collapsed, and the number of instruments generated was curtailed by applying lag-limits (6, 42, 118, 119). The robustness of the findings to lag-limits was then scrutinised by adjusting these in sensitivity analysis.

Before deciding on a final method of estimation, the Wooldridge test was applied to check for serial correlation in the dependent variables, which was present as expected (122, 123). A Breusch-Pagan test for heteroskedasticity was performed, which was present in two models (124). These test results supported GMM as the best available estimator. The models were also checked for the presence of multicollinearity, and reduced-form models were generated to address this, which is further explained in Chapter 6.

Heckman selection models were generated to check for selection bias into IMF programmes as described in Box 2 (1, 9, 66). This was relevant for full models where IMF variables were still included. A number of unit root tests for dependent variables were performed as detailed in Chapter 6 and its Appendix 4. These consistently rejected the null-hypotheses that panels contained unit roots.

Following the motivations and hypotheses for each variable, all independent variables except for colonial legacy and year-dummies were treated as endogenous, while colonial legacy and year-dummies were treated as exogenous (19, 118, 119).

Microsoft Excel was used to compile the data (125), Stata was used to analyse the data (126), the command "Xtabond2" was used to perform GMM (19) and the command "Asdoc" to export regression results (127).

#### Box 2: Heckman selection models and IMF programme participation

Entering into an IMF programme happens for a reason - countries generally approach the IMF under conditions of economic turmoil/instability (1-5). These underlying conditions might in turn affect the dependent variables, leading to selection bias, or in other words: unobserved variables that make IMF programme participation more likely could be associated with the dependent variables. To check for this, one has to construct a Heckman selection model (9). This requires first building a Probit regression model of factors that might help explain why a given binary state occurs, in this case IMF programme participation. This was built following the available literature on factors that may help determine IMF programme participation (1-5), which included the current account balance, reserves, exchange rate, growth rates and a proxy for democracy. The Heckman selection model then captures the calculated hazard for participation in an IMF programme as the Inverse Mills Ratio, which can then be added to one's main model to adjust for selection bias, if present (1-5, 9). In the case of this study however, the Inverse Mills Ratio was consistently insignificant in all models and led to substantial loss of degrees of freedom, and was therefore not included, but it was explored whether including it would change the results in sensitivity analysis.

#### 4.2.4. Lag structure and model selection

The effects studied might occur with a time delay, warranting the investigation of different model lag structures. To decide on a lag structure and select a final model, advice on GMM model selection was adapted mainly from Kiviet (118), Kripfganz (119) and Roodman (19) to the specific data. In summary, 0-2-year lags were first jointly examined for all independent variables of interest, and lag-years were then sequentially reduced until arriving at a parsimonious specification that maximised t-statistics for each variable while achieving acceptable diagnostic test-statistics. This turned out to generally and considerably favour 1year lags for nearly all financial independent variables of interest in all models, which were therefore adopted universally rather than "cherry-picking" alternative lag structures for a few variables, as an attempt at lowering the degree of confirmation bias in the model selection process. 1-year lags are commonly used in the reviewed literature, as some of the fiscal and macroeconomic dynamics investigated often may have a delayed effect that manifests in the following (budget) year (e.g. (2, 47)). An important exception from this were IMF-variables: Some authors treat IMF-variables as lagged (2, 5, 73) while others do not (3), arguing for both contemporaneous and delayed effects. The results in Chapter 6 tended to show higher, at times significant or borderline significant t-statistics when unlagged, so these results were followed and IMF-variables were kept unlagged. This was deemed economically reasonable since when a program starts and conditions apply and these were thereby counted in the dataset, they apply to the current budget year and may affect government spending in the present, as well as having further delayed implications, which however were insignificant in all models. Alternative lag structures (0-2 years) were tested for all variables, including IMFvariables, in sensitivity analysis.

When deciding on the number of GMM-style lag limits, the model with the lowest number of instruments, the lowest Hansen test p-value and otherwise acceptable diagnostic tests that showed significant results for independent variables of interest, was generally selected (6, 19, 119). This was done to avoid type II error in the first instance while minimising the risk of overfitting and overidentifying models. The robustness of significant results was then interrogated in sensitivity analysis, where lag limits were manipulated.

The full models included a number of variables and tested multiple hypotheses at once. Reduced form models were generated to increase parsimoniousness, reduce the risk of type II error due to multicollinearity, and conversely lower the risk of significant findings being due to multiple comparisons. Inspired by Kiviet (2020) (118), independent variables of interest with a p-value above 0.2 were sequentially removed, starting with the variable with the highest pvalue (or lowest t-value) first. All covariates were still kept in these models and then changed/removed in the respective sensitivity analyses. This yielded parsimonious models that isolated significant relationships while still adjusting for covariates.

#### 4.2.5. Model specification

The full model specification was:

$$\begin{split} Y_{it}' &= \beta_0 + \beta_1' Y_{i,t-1}' + \beta_2' ln \left( \frac{ODA^+ for health}{GDP} \right)_{i,t-1}' + \beta_3' ln \left( \frac{ODA^+ for non-health}{GDP} \right)_{i,t-1}' \\ &+ \beta_4 ln \left( \frac{debt \ service}{GDP} \right)_{i,t-1} + \beta_5 ln \left( \frac{debt \ stock}{GDP} \right)_{i,t-1} \\ &+ \beta_6 IMF \ conditionalities_{it} + \beta_7 IMF \ participation_{it} + \beta_8 ln \left( \frac{GHES}{GDP} \right)_{it} \\ &+ \beta_9 ln \left( \frac{GDP}{cap} \right)_{it} + \beta_{10} IMR_{it} + \beta_{11} gov. \ effectiveness_{it} \\ &+ \beta_{12} corruption \ control_{it} + \beta_{13} conflict_{it} + \beta_{14}' independence \ from'_{it} + \mu_i \\ &+ \nu_t + \varepsilon_{it} \end{split}$$

Details of each individual variable measurement and transformation is available in Chapter 6. The terms  $\mu_i$  is country-specific, time-invariant effects (country fixed effects),  $v_t$  is time-invariant cross-country effects (year fixed effects, such as shocks), and  $\varepsilon_{it}$  is the error term. Country fixed effects are automatically included in the levels equation in System GMM and eliminated from the difference equation (19, 114), while year dummies capturing year fixed effects were manually added in all models, as is recommended in the theoretical literature (19, 118, 119) and common practice (e.g. (40, 128)).

#### 4.2.6. Sensitivity analysis

The sensitivity analysis is described in Chapter 6 and Appendix 4. Econometric data compilation and model building involves a great degree of choice, as discussed in the limitations section of Chapter 9. Alternative choices might have produced different results. Showing the implications of these different choices is therefore essential to ensure a model is robust and begin to be able to make claims about, and for readers to be able to critically assess, the validity of one's findings. The following sensitivity analyses were therefore performed as robustness checks:

- Models with no lags and with 2-year lags for independent variables of interest.
- Alternative lag-limits.
- Per capita financial variables instead of per GDP.
- U5MR and MMR instead of IMR (81, 129).
- Alternative conditionality variable (72).

- Alternative deflator.
- Alternatives to colonial legacy dummies:
  - No colonial legacy dummies
  - Regional dummy for Sub-Saharan Africa (SSA) instead, to adjust for the range of different dynamics specific to this region that may not apply elsewhere (also used as interaction term).
  - Lower-Middle income Country (LMC) and LIC World Bank income group dummies instead (meaning that UMIC was then the baseline state), to investigate the different relationships differentiated by country income group without splitting the dataset into problematically small groups for GMM estimation, with resulting narrowing of the scope for causal inference (also used as interaction terms).
  - No log-transformation of relevant variables, as this is essentially a data manipulation, although it was well justified due to substantial right skew of many variables and to facilitate economic interpretation of relationships as elasticities.
  - Aggregate ODA-variables as just health and non-health purposes, which lowered the ability to identify fungibility but increased parsimoniousness.
  - Full models with Inverse Mills Ratios, to adjust for any degree of selection bias in IMF programme participation, even decidedly insignificant, which came at the expense of 13 countries and 214 observations.
- Interaction terms: These were multiplied onto the main independent variables of interest to look for signs of effect modification from the following set of variables:
  - Government effectiveness, as debt, aid and IMF programmes might affect health financing differently depending on the level of government effectiveness (see Section 4.2.1.2).
  - Corruption, for the same reasons (see Section 4.2.1.2).
  - GHE-S/GDP multiplied on  $\frac{PPG \text{ external debt service}}{GDP}$ , to test the hypothesis that debt servicing could have differential impact on the health system payment pattern at different levels of GHE-S.

## 4.3. In-depth case study in Senegal

## 4.3.1. Study setting

#### 4.3.1.1. Motivation for Senegal as a case country

The case country was selected based on the following criteria:

- Being relatively peaceful and politically stable within the most recent decades, to describe the relationships between EDPs+IMF and the government in a time of stability rather than extraordinary circumstance. This criteria was set to optimise the opportunity for obtaining findings that might be generalizable to other contexts.
- Being a SSA country, as many countries in this region rely heavily on OOP (10), the region has the highest unmet health needs globally, the highest levels of poverty, the most resource constrained public sectors, it receives relatively high amounts of development assistance and International Financial Institution (IFI) loans, and many countries in the region were and still are experiencing a strongly increasing debt burden (33).
- It should preferably be less corrupt than the SSA average, anticipating that interview responses from ministries, agencies and other organisations would be more credible and less likely to be motivated by personal gain or to hide a truth. This criteria was also set to improve the expected reliability of numbers and official reports and policy documents, to not be misled in the research and increase the chances of observations made being due to legitimate EDP/IMF-government interactions and financial flows and not due to corruption.
- The country should have an IMF and WB office and existing programs in place from these institutions.
- It should have country offices from some of the world's major bilateral and multilateral donors in the health sector, as measured by global ODA disbursements for health. This was to optimise the scope for learning lessons that might be informative to major donors and all countries they work in.
- It should have received a reasonable amount of ODA per capita and ODA<sup>+</sup> for health per capita, roughly in the range of the SSA average or above, to be able to identify an influence of financing flows, associated non-financial develop cooperation, and fungibility dynamics.
- It should be relatively externally indebted, as compared to the regional average for SSA (Table 4.1), to optimise chances of identifying an impact of the external debt on health financing.
- It should have a relatively high reliance on OOP, to investigate the hypothesis of external determinants of this reliance as per the research questions.

Table 4.1 displays selected statistics that aided the choice of Senegal as a case country. The statistics displayed in Table 4.1 of course differ somewhat from the numbers used to choose the case country in 2018, however the most recent numbers are provided here to give an up-to-date overview. Notably, Senegal changed status from LIC to LMC in 2019 (130).

As seen, Senegal fit the selection criteria well. Until recently, it had been relatively politically stable and less corrupt than the SSA average (further discussed in Chapters 7 and 9). It is more externally indebted than the averages for SSA, LICs and LMCs (33). It receives ODA and ODA<sup>+</sup> for health at a comparable level to the SSA average. It has a stronger reliance on OOP than the SSA average. It has IMF and WB country offices, as well as country offices from major international development partners including the United States, France, Canada, UNICEF, WHO and more.

	Senegal	SSA	LIC	LMC
GDP (billion US\$, 2022) (33)	28	44	11	151
GDP per capita (US\$, 2022) (33)	1,599	1,700	750	2,562
PPG external debt stock, (billion US\$, 2022) (33)	15	481	121	1,194
PPG external debt service (billion US\$, 2022) (33)	1	45	6	109
PPG external debt stock per GDP (%, 2022) (33)	52	23	23	15
PPG external debt service per GDP (%, 2022) (33)	4	2	1	1
ODA per capita (US\$, 2022) (33)	46	49	66	27
ODA <sup>+</sup> for health per capita (US\$, 2022) (46)	18	14	13	4
GHE-S per CHE (%, 2021) (10)	26	33	24	42
OOP per CHE (%, 2021) (10)	47	33	38	35
VHI per CHE (%, 2021) (10)	6	6	7	4
EXT per CHE (%, 2021) (10)	18	25	32	18
GHE-S per GGE (%, 2021) (10)	4	7	6	9
GHE-S per GDP (%, 2021) (10)	1	2	1	3
GHE-S per capita (US\$, 2021) (10)	18	52	10	75
Transparency International corruption perceptions index (2023, higher score ~ less corrupt) (131, 132)	43 / 100	33 / 100	-	-

Table 4.1: Selected statistics for Senegal as compared to averages in Sub-Saharan Africa (SSA, excluding high-income countries), Low-Income Countries (LICs), and Lower-Middle income Countries (LMCs). All US\$ are 2022 current US\$. GDP: Gross Domestic Product. PPG: Public and Publicly Guaranteed. ODA: Official Development Assistance. GHE-S: Government Health Expenditure as Source. CHE: Current Health Expenditure. GGE: General Government Expenditure. GDP: Gross Domestic Product.

#### 4.3.1.2. Case country overview

Senegal is a West African country with approximately 18 million inhabitants (2023) (133). Compared to the rest of the African continent, Senegal has enjoyed relative political stability as a constitutional multiparty democracy with generally peaceful transitions of power, albeit at certain times partly achieved via the arrest of demonstrators and imprisonment of political opponents, and with significant recent political instability around the 2024 presidential election (134-138). Senegal became independent from France in 1960. It had a GDP per capita of around \$1,599 (33). Over the last 20 years, ODA has fluctuated around approximately \$100 per capita per year (constant 2021 US\$), which is about 1.5-2 times the SSA average (46). ODA for health purposes in Senegal has risen from about \$7 in 2007 to about \$18 in 2022 (OECD CRS data, constant 2021 US\$), which is at a comparable level to the rest of SSA (46). The largest bilateral development partners in the health sector in Senegal in 2022 as measured by ODA disbursements (grants + loans) were the United States and Japan, who transferred respectively US\$49 million and \$42 million, followed by France (US\$ 24 million), Korea (\$20 million) and Canada (\$19 million) (46). The largest multilateral partners in the health sector were the WB International Development Association (IDA) branch (\$73 million), the Global Fund to Fight Aids, Tuberculosis and Malaria (GFATM) (\$30 million), the Bill and Melinda Gates Foundation (BMGF) (\$17 million) and Gavi (\$9 million) (46).

Following the Heavily Indebted Poor Countries (HIPC) initiative and the Multilateral Debt Relief Initiative (MDRI), where Senegal received \$2.836 billion (2015 US\$) in debt relief (139), Senegal spent less of its government revenue on external debt servicing (Figure 4.1) (33, 140). This trend however reversed in 2007 at 4% to grow to 32% in 2020 (Figure 4.1).



Figure 4.1: Public and Publicly Guaranteed (PPG) external debt servicing per government revenue in Senegal from 2000-2022 [4, 16].

Chapter 8 provides an in-depth description of the composition and evolution of PPG external debt in Senegal. Appendix 2 provides an overview of key health indicators in Senegal.

The case study interviews were conducted in the wealthier, predominantly urban region of Dakar and the poorer, predominantly rural region of Tambacounda, to obtain a diversity in perspective from different interviewees (see Section 4.3.2.1). Figure 4.2 shows a regional map of Senegal.



Figure 4.2: Regional map of Senegal. (From: Wikimedia Commons, URL: <u>https://commons.wikimedia.org/wiki/File:Senegal, administrative divisions - en - monochrome.svg</u>. License: <u>https://creativecommons.org/licenses/by-sa/3.0/deed.en</u>. Image unedited.)

Table 4.2 shows selected socioeconomic and health system indicators illustrating great inequality between the regions of Dakar and Tambacounda, with strong health system infrastructure and human resource concentration in the Dakar region (141-143).

Indicator	Dakar	Tambacounda
Population (million, 2023)	3.9	1.0
Poverty rate (%, 2018/19)	9	62
Hospitals (2020)	14	1
Health centres (2020)	25	7
Nurses (2020)	1168	75
Midwives (2020)	550	121
Doctors (2020)	651	12

 Table 4.2: Selected socioeconomic and health system indicators for the Senegalese regions of Dakar and Tambacounda (141-143).

#### 4.3.1.3. Health financing sources in Senegal

The largest source of health sector financing in Senegal is OOP, followed by GHE-S, EXT, voluntary prepayments and SHI (Figure 4.3) (10). OOP has increased over time while GHE-S has been relatively stagnant. Adjusted for inflation, GHE-S in absolute terms was similar in

2019 (272 million constant 2021 US\$) compared to 2006 (\$258 million) (10). Apart from a significant increase in GHE-S in response to the Covid-19 pandemic in 2020, GHE-S as a percent of GGE has decreased steadily from 9% in 2000 to 4% in 2021, and from 1.8% of GDP in 2006 to 1.1% in 2021 (10). Compared to the rest of SSA, Senegal's per capita GHE-S is relatively low at \$18 per capita vs. a \$62 per capita average across SSA, while OOP is at a similar level (\$34 per capita in Senegal vs. \$33 per capita across SSA) (2021 data) (10). OOP health expenditures per capita have remained substantially higher than GHE-S since about 2009, reaching \$34 in 2021 vs \$18 from the government (10). External health financing has grown to \$13 per capita, increasingly comparable to the level of health financing from the government, while SHI contributions from employers and employees, and voluntary prepayments have remained relatively small by comparison since about 2013 (10).



Figure 4.3: Main sources of health financing in Senegal, 2000-2021 (10). GHE-S: Government Health Expenditure as a Source. OOP: Out-Of-Pocket payments. EXT: External health financing. Social ins: Social health insurance. Vol. pre: Voluntary prepayments.

Using national poverty surveys conducted in 2005 and 2011, Sow et al. (2013) investigated the changes in levels of household spending on health (OOP) and capacity to pay in Senegal (144). The authors found that the proportion of OOP as a share of households' capacity to pay had increased substantially, especially for the poorest quintile where this share increased from approx. 5.0% to 11.9% compared to 3.3% to 4.2% in the wealthiest quintile. An estimated 2.59% of households in the survey incurred catastrophic health expenditures in 2011, and 1.78% were pushed into poverty as a result of these expenditures.

#### 4.3.2. Qualitative Methods

#### 4.3.2.1. Interview participants

Interviewees were sampled based on an initial stakeholder mapping across bilateral and multilateral development partner institutions, ministries and agencies in the central government, institutions at the regional and district level in Tambacounda, health system management staff, civil society organisations and academia. Development partner and government representatives were selected due to their central placement in the nexus of influence relevant to the research questions. The inclusion of civil society organisations, academics and health system managerial staff served as a critical counterbalance to the accounts obtained from these larger, official, political institutions interviewed. Participants in Tambacounda were included to obtain accounts far from the political and economic centre of Dakar, based on the notion that the reality and lived experiences in a poorer, in-land region at the intermediary and primary health system level might differ from those obtained at the central level.

38 candidate interviewees were identified, of which 32 responded and participated in the study (Table 4.3). Candidate participants from two multilateral organisations, two hospitals, and two academics were unavailable for interview or did not respond.

Interviews were conducted primarily in-person in Dakar and in the Tambacounda region of Senegal, while two were conducted remotely over Skype. Table 4.3 shows the distribution of interview participants across stakeholder groups and geographical regions.

Stakeholder group	Number of participants		
	Total	Dakar	Tambacounda
Bilateral development partners	4	4	0
Multilateral development partners	4	4	0
Central government*	9	9	0
Regional/District government	3	0	3
Hospital management	4	1	3
Civil Society Organisations	5	4	1
Academics	3	3	0
Total interviews	32	25	7

Table 4.3: Interview participants. \*Central government participants came from four different ministries, however the specific ministries are intentionally not mentioned to protect the anonymity of participants.

#### 4.3.2.2. Interview data collection

Semi-structured interviews were conducted with study participants. These have a loose structure with open-ended questions that allow for further exploration of the topic discussed (145). This interview method was chosen to enable an in-depth exploration of the research topic. The general questionnaire structure was maintained between stakeholder groups, but individual questions were tailored to focus on areas of respondent knowledge and expertise. Participant information sheets and consent forms were provided in French or English depending on the preferred language of the participant. Interviews were conducted in the preferred language of the participant interpreter was used where necessary (18 out of 32 interviews). Interview recordings were transcribed by a transcriptionist. Handwritten notes were taken from all interviews.

Interview questions centered around domestic health sector financing and the different schemes and policies; stakeholder priorities and opinions regarding the different contribution sources in Senegal and the balance between public and private financing; EDP+IMF influence on the public health budget; development assistance fungibility including in the health sector; development lending including to the health sector; external debt effects on the health budget; IMF/WB programs and conditionalities and their influence on the health sector and its financing; and perceived impacts of the above on health service access for the poor. Questions regarding external debt effects on the health budget were omitted from interviews in Tambacounda to match the expected knowledge and expertise of interviewees in this region.

#### 4.3.2.3. Interview data analysis

Interviews were analysed using the Framework method (146). The Framework method enables a systematic establishment of general themes and patterns inductively from interview data that may illustrate principal phenomena, whilst also allowing for the deductive application of themes established a priori based on the research questions and existing literature and frameworks (147-149). Thematic analysis also moves beyond e.g. mere quantification of certain words or phrases, which can be performed using the alternative content analysis (149), and instead attempts to qualitatively describe ideas and patterns in interviews. The interview data analysis process was both inductive and deductive: Before analysis, a set of themes were established informed by the theory underlying the research and the different key questions asked. An initial round of coding of 1/3 of interviews (n=11) was then performed, adding themes to the thematic framework as they emerged from the data. When no more themes emerged, the resulting set of themes was revised, themes with a high degree of overlap were merged while less clear concepts were clarified and better delineated. The final

coding framework was validated by an independent researcher who applied it to one sample interview. This led to the addition of one sub-theme. The final coding framework is available in Appendix 5.

NVivo 12 was used to code interviews (150). The final coding framework was applied to all transcripts, and summaries of participant statements on each theme were put into a framework matrix in Microsoft Excel (125). Central and illustrative quotations were also added to the framework matrix. After having coded all transcripts, the coding framework was reassessed for any revisions/merging of categories needed (similar to Ward et al. 2013 (151)). No changes were deemed necessary. An example interview data framework matrix is shown in Table 4.4, illustrating its layout. The actual framework matrix can be made available in anonymised form upon request for bona fide research or academic review purposes only, to protect participant confidentiality.

	Participant #1	Participant #2	
Theme 1	Summary	Summary	
		Quote	
Theme 2	Summary	Summary	

Table 4.4: Example interview data framework matrix.

During and after the completion of the matrix, patterns emerging mainly horizontally (between interviews and stakeholder groups) were then described in the results sections of Chapter 7 and 8, as relevant to each separate study question, supported by direct quotes illustrating observations made.

#### 4.3.2.4. Document review data collection

A purposive document review was also conducted. Documents were identified from EDP+IMF and government websites relating to health financing, health financing policy, as well as policy/strategy papers and lending agreements from the IMF and WB in Senegal. This included the Senegalese National Statistics and Demography Agency (ANSD) (152), the Universal Health Coverage Agency (*"Agence de la Couverture Maladie Universelle (CMU)"* (153), The Ministry of Health and Social Action (154), The Ministry of the Economy, Planning and Cooperation (155), the Ministry of Finance and Budgeting (156, 157), the major bilateral and multilateral donors present in Senegal at the time (USA, France, Canada, Japan, Belgium, Luxembourg, Spain, the GFATM, BMGF and Gavi (46)), the IMF (158, 159) and WB (134). Google Scholar, PubMed, Embase, Cochrane, CINAHL, Scopus, Web of Science and EconLit were also searched for relevant academic literature, using keywords such as Senegal, health financing, equity, debt, loan conditionalities, etc. Full details of the search strategy are

provided in Chapters 7 and 8. No time period constraint was applied to the search. A total of 329 full-text documents were retrieved and reviewed.

#### 4.3.2.5. Document data extraction and analysis

For Chapter 7, documentation of EDP activities deemed to have direct implications for the mix of domestic health financing sources in Senegal was included and integrated into the results section, including facts that confirmed or rebutted interview statements (triangulation). For Chapter 8, information regarding Senegal's external borrowing and main economic policy discussions, recommendations and conditionalities from the IMF and the WB was summarised in written and table form. This included all available key policy documents from the IMF and WB and all WB project loan agreements for the health sector in Senegal since 1966. Policy recommendations and conditionalities directly relating to the levels of and balance between domestic health financing sources and health financing scheme coverage were included in the results section. These findings were also triangulated against interview statements and quantitative data.

#### 4.3.3. Quantitative methods

To investigate patterns relevant to the research questions and triangulate and contextualise qualitative findings, descriptive quantitative analysis of the health financing composition in Senegal since 2000 was performed using the WHO GHED as the main source of national health accounts data (10). These data are supplied to the database by the Senegalese government according to WHO standards. GHE-S, EXT, OOP and Voluntary Health Insurance (VHI) (*Mutuelles*)) were examined. For Chapter 8, levels and trends of PPG loan disbursements separated by creditor type, and PPG external debt stock and debt service were examined using WB data (33). Total official development loans received by sector from the OECD CRS database were also included to examine official creditor priority of the health sector in Senegal (46). We focused on the period after 2000, as this was the period when WHO GHED data became available.

Through the analysis of the results for EDP+IMF influence on domestic health financing sources, analytical frameworks for understanding this influence emerged, which are presented in Chapter 7 and 8 (the analytical framework in Chapter 7 was developed as a synthesis and modification of existing frameworks combined with the empirical findings, while the framework in Chapter 8 was also informed by the available literature and economic theory).

#### 4.3.4. Ethical considerations

The key ethical considerations for this research revolved around participant confidentiality and ensuring interview data integrity for the case study in Senegal. The remaining data were publicly available secondary data, and ethical considerations for these parts of the research were therefore limited to general research ethics considerations as stipulated in relevant LSHTM academic regulations and research practice policy documents (160-162).

#### 4.3.4.1. Interview participant confidentiality

Interview participants were able to choose their preferred level of anonymity, as shown in the ethical consent form in Appendix 5. This ranged from no anonymity over some degree of anonymity e.g. name and organisation, to complete anonymity. However, upon writing up Chapter 7 and 8, I realised that the most appropriate solution to ensure that I protected the confidentiality of interviewees was to simply anonymise all of them completely. This was mainly due to the fact, that the group of government officials, EDP officials, academics and Civil Society Organisation (CSO) representatives working specifically within health financing in Dakar is not very large, and people could still potentially be identified from statements made if the name of their organisation or government agency, or even agency policy, projects or financial transactions made, were mentioned. For this reason, all interviews were treated as fully anonymous when the related chapters were written up, only referring to which general participant group the interviewee belonged to (development partner, government, CSO etc.). This change was submitted as an amendment to the LSHTM Research Governance and Integrity Office and approved as a non-substantial change.

#### 4.3.4.2. Data management

Interviews were recorded using a digital voice recorder, and stored on an encrypted, password protected USB-drive. Handwritten notes were taken from all interviews, written up on a computer, and stored on the protected USB-drive. Interview recordings were shared with a professional transcriptionist, who was bound by a London School of Hygiene & Tropical Medicine (LSHTM) Standard Mutual Confidentiality Agreement. Recordings and their transcripts were shared between me and the transcriptionist as encrypted, password-protected files. After completing the transcripts, the transcriptionist deleted all files from her computer as per our agreement. All used secondary datasets and documents were used in accordance with their license agreements and cited appropriately. The interview codebook, participant information sheet, and example interview topic guide were uploaded to the LSHTM data repository (163) and are available in Appendix 5.

#### 4.3.4.3. Ethical approvals

Ethical approvals were obtained from the London School of Hygiene Observational / Interventions Research Ethics Committee (*ref: 16420*) and the National Health Research Ethics Committee in Senegal (*protocol SEN19/56, ref: 00172*).

## 4.4. Epistemology and ontology

By going from a global-level quantitative study to an in-depth mixed methods case study, the research conducted in this thesis can be seen as subscribing to the *pragmatist* worldview (164, 165). In this worldview, emphasis is placed first and foremost on the research problem, and different methods are then used to approach this problem, acknowledging the strengths and limitations of each (164, 165).

Within this approach, the quantitative methods applied in this thesis generally originate from a *postpositivist* tradition (164). Positivism was characterized by a firm belief in an absolute truth waiting to be discovered by the use of the scientific method, and the postpositivists rejected this belief when studying the behaviour and actions of humans, acknowledging biases in the research process (164-166). Both rely heavily on quantitative methods, and seek to reduce human and social complexity to numbers, from which relationships or even knowledge of cause and effect can be derived (164-166). The global-level econometric study in Chapter 6 can be seen as mostly leaning toward this worldview in its quantitative, critical hypothesis testing. It however acknowledges its limitations for drawing causal inferences from observational data covering complex social, political and economic relations.

The qualitative methods applied originate from a *constructivist* tradition that seeks to understand the world through the study of the subjective meanings of different individuals (164, 165). In this tradition, these meanings are thought to be constructed partly via interaction with others (social constructivism), and the constructivist researcher concerns themself with these meanings, interactions and the historical, social, political and cultural context that shape them (164, 166). However, as this thesis concerns itself with what is believed to be real economic phenomena, its qualitative components rather belong to an ontologically *realist* worldview, believing that there is an external reality that exists independently of people's beliefs or understandings about it (165). It then uses interviews and document review, among quantitative tools, to help understand this reality, without applying a belief that reality was co-constructed in interviews with participants.

By applying a *pragmatist* approach, this thesis leverages the research tools developed in these different epistemological traditions to explore its research questions at different levels and from multiple angles, reaching discussion points and interpretations that could not have been reached using only one of these methods.

Having described the methods used and epistemological position of this research, the following chapter presents an analytical commentary building the case for the importance of the research on debt and health financing presented in Chapter 6 and 8.

## References

1. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

2. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

3. Clements B, Gupta S, Nozaki M. What Happens to Social Spending in IMF-Supported Programs? . 2011. Available from: <u>http://www.imf.org/external/pubs/ft/sdn/2011/sdn1115.pdf</u>.

4. Nooruddin I, Simmons JW. The Politics of Hard Choices: IMF Programs and Government Spending. International Organization. 2006;60(04).

5. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

6. Roodman D. A Note on the Theme of Too Many Instruments. Oxford Bulletin of Economics and Statistics. 2009;71(1):135-58.

7. O'Cathain Á, Murphy E, Nicholl J. Why, and how, mixed methods research is undertaken in health services research in England: a mixed methods study. BMC Health Serv Res. 2007;7:85.

8. Greene JC, Caracelli VJ, Graham WF. Toward a Conceptual Framework for Mixed-Method Evaluation Designs. Educational Evaluation and Policy Analysis. 2016;11(3):255-74.

Heckman JJ. Sample Selection Bias as a Specification Error. Econometrica. 1979;47(1).
 WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from:

http://apps.who.int/nha/database.

11. Ataguba JE, Asante AD, Limwattananon S, Wiseman V. How to do (or not to do) ... a health financing incidence analysis. Health Policy Plan. 2018;33(3):436-44.

12. WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available from: <u>https://www.who.int/publications/i/item/9789241564021</u>.

13. Xu K, Evans DB, Carrin G, Aguilar-Rivera AM, Musgrove P, Evans T. Protecting households from catastrophic health spending. Health Aff (Millwood). 2007;26(4):972-83.

14. Moreno-Serra R, Smith PC. Broader health coverage is good for the nation's health: evidence from country level panel data. J R Stat Soc Ser A Stat Soc. 2015;178(1):101-24.

15. Moreno-Serra R, Smith P. The Effects of Health Coverage on Population Outcomes: A Country-Level Panel Data Analysis. Results for Development Institute. 2011.

16. Qin VM, Hone T, Millett C, Moreno-Serra R, McPake B, Atun R, et al. The impact of user charges on health outcomes in low-income and middle-income countries: a systematic review. BMJ Glob Health. 2018;3(Suppl 3):e001087.

17. Makinen M, Waters H, Rauch M, Almagambetova N, Bitran R, Gilson L, et al. Inequalities in health care use and expenditures: Empirical data from eight developing countries and countries in transition. Bulletin of the World Health Organization. 2000;78(1):55-65.

18. Akazili J, McIntyre D, Kanmiki EW, Gyapong J, Oduro A, Sankoh O, et al. Assessing the catastrophic effects of out-of-pocket healthcare payments prior to the uptake of a nationwide health insurance scheme in Ghana. Glob Health Action. 2017;10(1):1289735.

19. Roodman D. How to do xtabond2: An introduction to difference and system GMM in Stata. Stata Journal. 2009;9(1):86-136.

20. Kakwani NC. Measurement of Tax Progressivity: An International Comparison. The Economic Journal. 1977;87(345).

21. Yu CP, Whynes DK, Sach TH. Equity in health care financing: The case of Malaysia. Int J Equity Health. 2008;7:15.

22. Binyaruka P, Martinez-Alvarez M, Pitt C, Borghi J. Assessing equity and efficiency of health financing towards universal health coverage between regions in Tanzania. Social science & medicine (1982). 2024;340:116457.

23. Verbeek M. A Guide to Modern Econometrics. 4th ed: Wiley; 2012. ISBN: 9781119951674
24. Sargan JD. The Estimation of Economic Relationships using Instrumental Variables. Econometrica. 1958;26(3).

25. Kronmal RA. Spurious Correlation and the Fallacy of the Ratio Standard Revisited. Journal of the Royal Statistical Society Series A (Statistics in Society). 1993;156(3).

26. Hansen LP. Large Sample Properties of Generalized Method of Moments Estimators. Econometrica. 1982;50(4).

27. Andersen TG, Sorensen BE. GMM Estimation of a Stochastic Volatility Model: A Monte Carlo Study. Journal of Business and Economic Statistics. 1996;14(3):328-52.

28. Bowsher CG. On testing overidentifying restrictions in dynamic panel data models. Economics Letters. 2002;77(2):211-20.

29. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJ. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

30. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

31. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

32. Ali MEM, Ebaidalla M, Rizk R. Foreign aid and out-of-pocket health expenditure in Sub-Saharan Africa: does institutional development matter. International Journal of Sustainable Economy. 2020;12(3).

33. World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from:

https://data.worldbank.org/.

34. Mishra P, Newhouse D. Does health aid matter? J Health Econ. 2009;28(4):855-72.

35. Farag M, Nandakumar AK, Wallack SS, Gaumer G, Hodgkin D. Does funding from donors displace government spending for health in developing countries? Health Aff (Millwood). 2009;28(4):1045-55.

36. Stuckler D, Basu S, McKee M. International Monetary Fund and aid displacement. Int J Health Serv. 2011;41(1):67-76.

37. Xu K, Saksena P, Holly A. The determinants of health expenditure: A country-level panel data analysis. Results for Development Institute WHO; 2011.

38. Gebrehanna S, Upadhyay MP. Fungibility of Foreign Aid in Selected African Countries. Indian Development Review. 2012;10(1):63-80.

39. Van de Sijpe N. The fungibility of health aid reconsidered. 2013.

40. Van de Sijpe N. Is Foreign Aid Fungible? Evidence from the Education and Health Sectors. World Bank Economic Review. 2013;27(2):320-56.

41. Fernandes Antunes Á, Xu K, James CD, Saksena P, Van de Maele N, Carrin G, et al. General budget support: has it benefited the health sector? Health economics. 2013;22(12):1440-51.

42. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

43. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

44. Barkat K, Mrabet Z, Alsamara M. Does Official Development Assistance for Health from Developed Countries Displace Government Health Expenditure in Sub-Saharan Countries? Economics Bulletin. 2016;36(3):1616-35.

45. Younsi M, Chakroun M, Nafla A. Robust analysis of the determinants of healthcare expenditure growth: evidence from panel data for low-, middle- and high-income countries. International Journal of Health Planning and Management. 2016;31(4):580-601.

46. OECD. Creditor Reporting System (CRS). 2024 [cited 2023 February 16]. Available from: https://stats.oecd.org/Index.aspx?DataSetCode=CRS1.

47. Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 2007;8(4):299-310.

48. Behera DK, Dash U. Impact of macro-fiscal determinants on health financing: empirical evidence from low-and middle-income countries. Glob Health Res Policy. 2019;4:21.

49. Shabbir S, Yasin HM. Implications of Public External Debt for Social Spending: A Case Study of Selected Asian Developing Countries. Lahore Journal of Economics. 2015;20(1):71-103.

50. Fosu AK. The External Debt-Servicing Constraint and Public Expenditure Composition. Helsinki, Finland: UNU-WIDER; 2007.

51. Fosu AK. The External Debt-Servicing Constraint and Public-Expenditure Composition in Sub-Saharan Africa. African Development Review. 2010;22(3):378-93.

52. Said R, Morai AS. Relationship between Public Debt Burden and Health Expenditure in Sub-Saharan African Countries: The Role of Institutional Quality. Journal of Business and Social Review in Emerging Economies. 2020;6(2):493-502.

53. World Bank. International Debt Statistics. 2024 [cited 2024 April 24]. Available from:

https://databank.worldbank.org/source/international-debt-statistics#

54. Ooms G, Schrecker T. Expenditure ceilings, multilateral financial institutions, and the health of poor populations. The Lancet. 2005;365(9473):1821-3.

55. Center for Global Development. Does the IMF Constrain Health Spending in Poor Countries? . 2007
[cited 2024 March 19]. Available from: <u>https://www.cgdev.org/sites/default/files/14103\_file\_IMF\_report.pdf</u>.
56. Brunswijck G. Unhealthy Conditions. Eurodad; 2018.

57. Stuckler D, Basu S. The International Monetary Fund's effects on global health: before and after the 2008 financial crisis. Int J Health Serv. 2009;39(4):771-81.

58. Stuckler D, Basu S. The Body Economic: Why Austerity Kills: Basic Books; 2013. ISBN: 978-0465063987

59. Abocejo F. The Impact of International Monetary Fund (IMF) Structural Adjustment Policies (SAP) on the Philippines. Recoletos Multidisciplinary Research Journal. 2014;2(1):19-24.

60. Isiani MC, Anthonia Obi-Ani N, Obi-Ani P, Chidume CG, Okoye-Ugwu S. Interrogating the International Monetary Fund (IMF) Policies in Nigeria, 1986–2018. Cogent Arts & Humanities. 2021;8(1).

61. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

62. Foley EE. Your pocket is what cures you: The Politics of health in Senegal: Rutgers University Press; 2010. ISBN: 978-0-8135-4667-4

63. Meurs M, Seidelmann L, Koutsoumpa M. How healthy is a 'healthy economy'? Incompatibility between current pathways towards SDG3 and SDG8. Global Health. 2019;15(1):83.

64. Birn AE, Nervi L, Siqueira E. Neoliberalism Redux: The Global Health Policy Agenda and the Politics of Cooptation in Latin America and Beyond. Development and Change. 2016;47(4):734-59.

65. Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A, Stubbs T, editors: Oxford University Press; 2023. 0 p. ISBN: 9780190637736

66. Clements B, Gupta S, Nozaki M. What happens to social spending in IMF-supported programmes? Applied Economics. 2013;45(28):4022-33.

67. Gupta S. Can a causal link be drawn? A commentary on "the impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations". Social science & medicine (1982). 2017;181:199-201.

68. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs. International Monetary Fund; 2004.

69. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

70. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

71. IMF. Monitoring of Fund Arrangements (MONA) database. 2019 [cited 2019 May 15]. Available from: https://www.imf.org/external/np/pdr/mona/Arrangements.aspx.

72. IMF Monitor. IMF Conditionality Dataset, 1980-2019. 2023 [cited 2023 November 2]. Available from: https://imfmonitor.org/conditionality/.

73. Stubbs T, Kentikelenis A. Targeted social safeguards in the age of universal social protection: the IMF and health systems of low-income countries. Critical Public Health. 2017;28(2):132-9.

74. Boachie MK, Agyemang J, Immurana M. Health sector funding in Ghana: The effect of IMF conditionalities. Dialogues Health. 2022;1:100045.

75. Stubbs TH, Kentikelenis AE, King LP. Catalyzing Aid? The IMF and Donor Behavior in Aid Allocation. World Development. 2016;78:511-28.

76. O'Hare B, Makuta I, Chiwaula L, Bar-Zeev N. Income and child mortality in developing countries: a systematic review and meta-analysis. J R Soc Med. 2013;106(10):408-14.

77. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJL. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

78. Feyzioglu T, Swaroop V, Zhu M. A Panel Data Analysis of the Fungibility of Foreign Aid. The World Bank Economic Review. 1998;12(1):29-58.

79. Lee SA, Lim J-Y. Does International Health Aid Follow Recipients' Needs? Extensive and Intensive Margins of Health Aid Allocation. World Development. 2014;64:104-20.

80. Reidpath DD, Allotey P. Infant mortality rate as an indicator of population health. J Epidemiol Community Health. 2003;57(5):344-6.

81. UN Inter-agency Group for Child Mortality Estimation. Child Mortality, Stillbirth, and Causes of Death Estimates. 2024 [cited February 9 2024]. Available from: <u>https://childmortality.org/</u>.

82. Kaufmann D, Kraay A. Worldwide Governance Indicators. 2023 [cited 2024 March 21]. Available from: https://www.govindicators.org/.

83. Jajkowicz O, Drobiszová A. The Effect of Corruption on Government Expenditure Allocation in OECD Countries. Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis. 2015;63(4):1251-9.

84. Hashem EA. The Effects of Corruption on Government Expenditures: Arab

Countries Experience. Journal of Economics and Sustainable Development. 2014;5(4).

85. Oryema JB, Gyimah-Brempong K, Picone G. The impact of debt relief on under five mortality rate in Sub-Saharan Africa. Glob Public Health. 2017;12(6):795-807.

86. In'airat M. Aid allocation, selectivity, and the quality of governance. Journal of Economics Finance and Administrative Science. 2014;19(36):63-8.

87. Sterck O, Roser M, Ncube M, Thewissen S. Allocation of development assistance for health: is the predominance of national income justified? Health Policy Plan. 2018;33(suppl\_1):i14-i23.

88. Rahim M. Quazi. Effects of Foreign Aid on GDP Growh and Fiscal Behavior: An Econometric Case Study of Bangladesh. The Journal of Developing Areas. 2005;38:95-117.

89. Craig Burnside, David Dollar. Aid, Policies, and Growth: Revisiting the Evidence. World Bank Policy Research Working Paper 3251. 2004.

90. Jung Y, Kim J, Kim K. Whom is economic aid meant for? The push vs. pull determinant factors of official development assistance. International Review of Economics & Finance. 2024;89:173-95.

91. Nanda VP. The "Good Governance" Concept Revisited. The ANNALS of the American Academy of Political and Social Science. 2016;603(1):269-83.

92. Fielding D. Health aid and governance in developing countries. Health Econ. 2011;20(7):757-69.

93. Alesina A, Weder B. Do Corrupt Governments Receive Less Foreign Aid? American Economic Review. 2002;92(4):1126-37.

94. Clements B, Gupta S, Pivovarsky A, Tiongson ER. Foreign Aid: Grants versus Loans. IMF: Finance & Development. 2004(September):46-9.

95. Dietrich S. The Politics of Public Health Aid: Why Corrupt Governments Have Incentives to Implement Aid Effectively. World Development. 2011;39(1):55-63.

96. IMF. IMF and Good Governance. 2024 [cited 2024 February 9]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/The-IMF-and-Good-Governance.

97. World Bank. World Bank Group Assistance to Low-Income Fragile and Conflict-Affected States: An Indepedent Evaluation. 2014:166.

98. Yogo UT, Mallaye D. Health Aid and Health Improvement in Sub Saharan Africa. Munich Personal RePEc Archive. 2012(44938).

99. Collier P, Hoeffler Á. Aid, Policy, and Growth in Post-Conflict Societies. World Bank Policy Research Working Paper 2902. October 2002.

100. Davies S, Pettersson T, Öberg M. Organized violence 1989–2022, and the return of conflict between states. Journal of Peace Research. 2023;60(4):691-708.

101. Hensel PR. ICOW Colonial History Data Set, version 1.1. 2023 [cited 2023 November 17]. Available from: http://www.paulhensel.org/icowcol.html.

102. Jones P. History matters: New evidence on the long run impact of colonial rule on institutions. Journal of Comparative Economics. 2013;41(1):181-200.

103. Robin M. Grier. Colonial Legacies and Economic Growth. Public Choice. 1999;98(3/4):317-35.

104. Acemoglu D, Robinson JA, Johnson S. The Colonial Origins of Comparative Development: An Empirical Investigation. American Economic Review 2001;91(5):1369-401.

105. Azevedo MJ. The State of Health System(s) in Africa: Challenges and Opportunities. In: Azevedo MJ, editor. Historical Perspectives on the State of Health and Health Systems in Africa, Volume II: The Modern Era. Cham: Springer International Publishing; 2017. p. 1-73.

106. Alesina A, Dollar D. Who Gives Foreign Aid to Whom and Why? Journal of Economic Growth. 2000;5(1):33-63.

107. Matteis AD. Whose poverty really matters when deciding aid volumes? International Journal of Public Policy. 2016;12(1/2).

108. Neumayer E. The Determinants of Aid Allocation by Regional Multilateral Development Banks and United Nations Agencies. International Studies Quarterly. 2003;47(1):101-22.

Hall AR. Generalized Method of Moments: Oxford University Press; 2005. 408 p. ISBN: 0198775202
Arellano M, Bond S. Some Tests of Specification for Panel Data: Monte Carlo Evidence and an

Application to Employment Equations. The Review of Economic Studies. 1991;58(2):277-97.
111. Arellano M, Bover O. Another look at the instrumental variable estimation of error-components models. Journal of Econometrics. 1995;68(1):29-51.

112. Blundell R, Bond S. Initial conditions and moment restrictions in dynamic panel data models. Journal of Econometrics. 1998;87(1):115-43.

113. Ullah S, Akhtar P, Zaefarian G. Dealing with endogeneity bias: The generalized method of moments (GMM) for panel data. Industrial Marketing Management. 2018;71:69-78.

114. Statalist. System GMM: does it includes fixed effects? 2019 [cited 2023 March 23]. Available from: <u>https://www.statalist.org/forums/forum/general-stata-discussion/general/1520513-system-gmm-does-it-includes-fixed-effects</u>.

115. Windmeijer F. A finite sample correction for the variance of linear efficient two-step GMM estimators. Journal of Econometrics. 2005;126(1):25-51.

116. Hausman JA. Specification Tests in Econometrics. Econometrica. 1978;46(6).

117. Sanjeev KS, Shyam N. Is Donors' Concern About the Fungibility of Foreign Aid Justified?: A Panel Data Analysis. The Journal of Developing Areas. 2010;43(2):299-311.

118. Kiviet JF. Microeconometric dynamic panel data methods: Model specification and selection issues. Econometrics and Statistics. 2020;13:16-45.

119. Kripfganz S. Generalized method of moments estimation of linear dynamic panel-data models. London Stata Conference2019.

120. Adeleye BN. (Stata 13): How to Estimate Two-Step System GMM. 2019 [cited 2023 November 9]. Available from: https://www.youtube.com/watch?v=JkBFrjYXEyM.

121. Roodman D. Doubts about the evidence that foreign aid for health is displaced into non-health uses. The Lancet. 2012;380(9846):972-3.

122. Drukker DM. Testing for Serial Correlation in Linear Panel-data Models. The Stata Journal: Promoting communications on statistics and Stata. 2003;3(2):168-77.

123. Wooldridge JM. Econometric Analysis of Cross Section and Panel Data. Cambridge, Massachusetts: The MIT Press; 2002. ISBN:

124. Breusch TS, Pagan AR. A Simple Test for Heteroscedasticity and Random Coefficient Variation. Econometrica. 1979;47(5).

125. Microsoft® 2024. Microsoft® Excel for Mac, version 16.81.

126. StataCorp 2015. Stata Statistical Software: Release 14.2. College Station, TX: StataCorp LP.

127. Shah A. ASDOC: Stata module to create high-quality tables in MS Word from Stata output. Statistical Software Components S458466, Boston College Department of Economics. 2018.

128. Liang L-L, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

129. WHO, UNICEF, UNFPA, World Bank Group, UNDESA/Population Division. Trends in maternal mortality 2000 to 2020: estimates by WHO, UNICEF, UNFPA, World Bank Group and UNDESA/Population Division. 2023 [cited 2024 February 29]. Available from: <u>https://www.who.int/publications/i/item/9789240068759</u>.

World Bank. New country classifications by income level: 2019-2020. 2019 [cited 2024 May 27].
 Available from: <u>https://blogs.worldbank.org/en/opendata/new-country-classifications-income-level-2019-2020</u>.
 Transparency International. Corruption Perception Index. 2024 [cited 2024 April 24]. Available from: <a href="https://www.transparency.org/en/cpi/2023/index/sen">https://blogs.worldbank.org/en/opendata/new-country-classifications-income-level-2019-2020</a>.
 Transparency International. Corruption Perception Index. 2024 [cited 2024 April 24]. Available from: <a href="https://www.transparency.org/en/cpi/2023/index/sen">https://www.transparency.org/en/cpi/2023/index/sen</a>.

132. Transparency International. Corruption Perceptions Index 2023 Report 2023 [cited 2024 May 27]. Available from: <a href="https://images.transparencycdn.org/images/CPI-2023-Report.pdf">https://images.transparencycdn.org/images/CPI-2023-Report.pdf</a>.

133. Agence nationale de la statistique et de la demographie (ANSD). Le Sénégal En Bref. 2023 [cited 2023 August 11]. Available from: <u>https://www.ansd.sn/</u>.

134. World Bank. Open Knowledge Repository. 2023 [cited 2023 October 10]. Available from: <u>https://openknowledge.worldbank.org/</u>.

135. Encyclopædia Brittanica. Senegal: Government And Society. 2018 [cited 2018 August 16]. Available from: <u>https://www.britannica.com/place/Senegal/Government-and-society</u>.

136. Amy Niang. 2024 Senegal election crisis points to deeper issues with Macky Sall and his preferred successor. 2024 [cited 2024 April 24]. Available from: <u>https://theconversation.com/2024-senegal-election-crisis-points-to-deeper-issues-with-macky-sall-and-his-preferred-successor-223035</u>.

Elodie Toto. How Macky Sall Provoked a Constitutional Crisis in Senegal. 2024 [cited 2024 May 5].
 Available from: <u>https://foreignpolicy.com/2024/03/01/senegal-election-macky-sall-sonko-faye-constitutional-coup/</u>.
 Haque N. From prisoner to president in 20 days, Senegal's Diomaye Faye takes office. 2024 [cited 2024 May 5].
 Available from: <u>https://www.aljazeera.com/features/2024/4/2/from-prisoner-to-president-in-20-days-senegals-diomaye-faye-takes-office</u>.

139. World Bank, IMF. Heavily Indebted Poor Countries (HIPC) Initiative and Multilateral Debt Relief Initiative (MDRI) - Statistical Update. 2017 [cited 2018 August 20]. Available from:

http://pubdocs.worldbank.org/en/175131505738008789/WB-HIPC-stat-update-2017.pdf.

140. IMF. Public Finances in Modern History. 2024 [cited 2024 June 10]. Available from: https://www.imf.org/external/datamapper/datasets/FPP.

141. Agence nationale de la statistique et de la demographie (ANSD). Regions. 2024 [cited 2024 May 28]. Available from: <u>https://www.ansd.sn/taxonomy/term/7</u>.

142. Agence nationale de la statistique et de la demographie (ANSD). Situation economique et sociale de la region de Tambacounda. 2021 [cited 2024 May 28]. Available from: <u>https://www.ansd.sn/sites/default/files/2024-01/SES-Tambacounda\_2020-2021.pdf</u>.

143. Agence nationale de la statistique et de la demographie (ANSD). Situation economique et sociale de la region de Dakar. 2021 [cited 2024 May 28]. Available from: <u>https://www.ansd.sn/sites/default/files/2024-01/SES-Dakar\_2020-2021.pdf</u>.

144. Sow M, Diop M, Mbengue A, AW A, Sall FL, Cissé A, et al. Rapport d'analyse sur les dépenses catastrophiques de santé et leur impact sur l'appauvrissement et l'utilisation des services au Sénégal, 2005 et 2011. The African Health Monitor. 2013(17):46-50.

Britten N. Qualitative interviews in medical research. BMJ (Clinical research ed). 1995;311(6999):251-3.
Ritchie J, Spencer L. Qualitative Data Analysis for Applied Policy Research. In: Bryman A, Burgess R, editors. Analyzing Qualitative Data. 1st ed: Routledge; 1994.

147. Braun V, Clarke V. Using thematic analysis in psychology. Qualitative Research in Psychology. 2006;3(2):77-101.

 Guest G, MacQueen KM, Namey EE. Applied Thematic Analysis. 2012 2018/09/10. Thousand Oaks, California: SAGE Publications, Inc. Available from: <u>http://methods.sagepub.com/book/applied-thematic-analysis</u>.
 Vaismoradi M, Turunen H, Bondas T. Content analysis and thematic analysis: Implications for

conducting a qualitative descriptive study. Nurs Health Sci. 2013;15(3):398-405.

150. QSR International Pty Ltd. NVivo qualitative data analysis software. Version 12, 2020.

151. Ward DJ, Furber C, Tierney S, Swallow V. Using Framework Analysis in nursing research: a worked example. J Adv Nurs. 2013;69(11):2423-31.

152. Agence nationale de la statistique et de la demographie (ANSD). ansd.sn. 2024 [cited 2024 July 25]. Available from: <u>https://www.ansd.sn/index.php</u>.

153. Agence de la Couverture Maladie Universelle. agencecmu.sn. 2024 [cited 2024 July 25]. Available from: https://agencecmu.sn/.

154. Ministère de la Santé et de l'Action Sociale. <u>https://sante.gouv.sn/</u>. 2024 [cited 2024 September 19]. Available from: <u>https://sante.gouv.sn/</u>.

155. Ministère de l'Economie du Plan et de la Coopération. economie.gouv.sn. 2023 [cited 2023 October 10]. Available from: <u>https://www.economie.gouv.sn/</u>.

156. Ministère des Finances et du Budget. Direction de la Dette Publique (DDP). 2023 [cited 2023 October 10]. Available from: <u>https://www.sentresor.org/structures-du-tresor/directions-administratives-da/direction-de-la-dette-publique-ddp/presentation-de-la-ddp/</u>.

157. Ministère des Finances et du Budget. senfinances.org. 2023 [cited 2023 October 10]. Available from: https://www.senfinances.org/. 158. IMF. Senegal. 2023 [cited 2023 September 25]. Available from: https://www.imf.org/en/Countries/SEN.

159. IMF. IMF eLIBRARY: Senegal. 2023 [cited 2023 October 10]. Available from: https://www.elibrary.imf.org/subject/SN.

160. London School of Hygiene and Tropical Medicine. Good research practice policy. 2021 [cited 2024 February 20]. Available from: <u>https://www.lshtm.ac.uk/sites/default/files/good-research-practice-policy.pdf</u>.

161. London School of Hygiene and Tropical Medicine. LSHTM Academic Manual 2023-24. 2023 [cited 2024 February 20]. Available from: <u>https://www.lshtm.ac.uk/aboutus/organisation/quality-and-academic-standards/academic-regulations</u>.

162. London School of Hygiene and Tropical Medicine. Research Degrees Handbook For Doctoral (PhD and DrPH) and MPhil Students 2023-2024. 2023 [cited 2024 February 20]. Available from:

https://www.lshtm.ac.uk/sites/default/files/research-degrees-handbook.pdf

163. London School of Hygiene and Tropical Medicine. LSHTM Data Compass. 2023 [cited 2023 November 14]. Available from: <u>https://datacompass.lshtm.ac.uk/</u>.

164. Creswell JW. Research design: qualitative, quantitative, and mixed methods approaches. 4th ed: SAGE Publications; 2018. ISBN: 978-1-4522-2609-5

165. Ritchie J, Lewis J. Qualitative Research Practice: A Guide for Social Science Students and Researchers: SAGE Publications; 2003. ISBN: 0 7619 7109 2

166. Green J, Thorogood N. Qualitative Methods for Health Research: SAGE Publications; 2004. ISBN: 0 7619 4770 1



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

## **RESEARCH PAPER COVER SHEET**

Please note that a cover sheet must be completed for each research paper included within a thesis.

#### SECTION A - Student Details

Student ID Number	lsh1601722	Title	Dr.
First Name(s)	Frederik		
Surname/Family Name	Federspiel		
Thesis Title	External Development Partner and International Monetary Fund Influence on Domestic Health Financing Sources in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal		
Primary Supervisor	Josephine Borghi		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

#### SECTION B - Paper already published

Where was the work published?	Global Health Action		
When was the work published?	22 June 2022		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

\*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

#### SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	

Improving health worldwide

www.lshtm.ac.uk

Stage of publication	Choose an item.

## SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I conceived of this short communication, conducted the literature review, data collection and analysis, and wrote the first draft of the manuscript. My supervisors and I then critically revised the manuscript, with me being responsible for performing the revisions necessary.
---	---

#### SECTION E

Student Signature		
Date	20/9-24	

Supervisor Signature		
Date	23/09/24	

## 5. Chapter 5: Growing Debt Burden in Low- and Middle-Income Countries During Covid-19 May Constrain Health Financing

Authors: Frederik Federspiel, Josephine Borghi, Melisa Martinez-Alvarez Published in: Global Health Action, 22 June 2022<sup>17</sup>

#### Chapter overview:

The analytical commentary presented in this chapter further builds the case for the research presented in Chapters 6 and 8, by illustrating how the public external debt repayments of LMICs have outgrown government health expenditures in many borrower countries. A brief research agenda is proposed for empirically exploring how this growing debt burden could impact countries' ability to expand domestic government health spending before making some preliminary policy recommendations.

Of note, this chapter is not a main empirical piece in itself, but rather a short commentary and analysis performed during the course of the research, to bring attention to the possible implications of the growing debt burdens of LMICs, communicate a research agenda for better understanding these and provide early policy recommendations arising from the analysis. Minor changes have been made from the published version for consistency across chapters.

<sup>&</sup>lt;sup>17</sup> Evidence of retention of copyright is provided in Appendix 3

## 5.1. Abstract

Debt burdens are growing steadily in Low- and Middle-Income Countries (LMICs), compounded by the Covid-19 economic recession, threatening to crowd out essential health spending. In 2019, 54 LMICs spent more on servicing their debt to foreign creditors than on financing their health services. While development loans may have positive effects on population health, the ensuing debt servicing requirements may have detrimental effects on health through constrained fiscal space for government health spending. However, the existing evidence is inadequate for a general understanding of whether, how and under what circumstances, debt may constrain government health spending. We call for more research on the impacts of debt on health financing and call on creditors and borrowers to carefully consider the potential impacts of lending on borrower countries' ability to finance their health services.

### 5.2. Introduction

Low- and Middle-Income Country (LMIC) governments have become increasingly indebted over the past decade (Figure 5.1) (1). Concurrently, debt repayment levels have grown steadily, channelling funds out of LMIC government budgets to public and private creditors in High-Income Countries (HICs) (Figure 5.2) (1). The number of LMICs that spend more on debt servicing than health has increased substantially from 33 in 2010 to 54 in 2019 (Figure 5.3 and Figure 5.4) (1, 2). In 2019, four countries, Angola, Benin, Cameroon and the Republic of the Congo, spent at least five times more on external debt servicing than they spent on health. Reductions in the public budget due to debt servicing will likely have implications for sectoral budget allocations, including health. Government health spending is essential for the functioning of all publicly funded health services from prevention to cure. The deeper the levels of poverty in a population, the less people can afford or have access to private health services, and government health financing is crucial to achieve Universal Health Coverage (UHC), serving all of the population, reducing health inequalities, and protecting people from financial risk when using health services (3, 4). The growing debt burden in LMICs seen in Figures 1-3 therefore gives cause for concern of constrained fiscal space for health and the implications hereof. Indeed, based on observational data similar to that presented in Figures 2-4, the Jubilee Debt Campaign have brought attention to the countries spending more on debt than health in the beginning of the pandemic and called for debt service cancellation for the year 2020 (5).



Figure 5.1: Public external debt stock as a proportion of Gross Domestic Product (GDP) in Low- and Middle-Income Countries (LMICs) from 2000-2020.



Figure 5.2: Public external debt service as a proportion of Gross Domestic Product (GDP) in Low- and Middle-Income Countries (LMICs) from 2000-2020.



*Figure 5.3: Number of Low- and Middle-Income Countries (LMICs) that spent more on public external debt servicing than health, 2000-2019.* 



Figure 5.4: Public external debt service and government health expenditure from domestic revenue (GHE-S) as fractions of Gross Domestic Product (GDP) (%) among Low-and Middle-Income Countries (LMICs) in 2019. The diagonal line indicates equal spending on debt servicing and health. Countries above the line spend more on external debt than on health and vice versa.

# 5.3. Covid-19 has led to more development lending which may increase debt levels

Covid-19 has triggered the largest economic recession since World War II (6). This has eroded the tax base in many LMICs, contracting government revenue and leaving a rift between countries expected to recover economically and others experiencing lasting damage to government revenue (6). Early WHO estimates from countries with available national health accounts data for 2020 - mainly HICs - indicate public health spending increases, which may reflect increased funds to help manage the pandemic (7). This may however not be the case for LMICs operating under severe public resource constraints, and where government revenue does not show signs of recovery after the initial shock (6). In such settings, Covid-19 may take up already scarce health resources for its prevention and management, and studies are ongoing to assess the impacts of emergency expenditure reallocation to manage Covid-19 on the rest of the health budget in Pakistan and South Africa (8). Covid-19 also adds to the health burden through worsened mental (9) and maternal health (10), amongst others, increasing the need for health financing.

In addition to these impacts, Covid-19 also appears to have affected the levels of development assistance committed by donors and the balance between loans and grants. Total aid

commitments made in 2020 increased by 19% in real terms compared to 2019 (11). This increase was made up almost entirely by development loan commitments, presumably to help countries cope with the impacts of the pandemic, while grant commitments stayed at approximately the same level (11). While increased donor commitments is positive, it will be important to monitor first of all disbursement rates of these commitments over the following years, but also if a new normal will be established, where development loans are a more prominent mode of providing development assistance, which will in turn further increase debt levels.

# 5.4. What do we know about the relationship between debt and health expenditure?

Our empirical understanding of the full impacts of development lending, the ensuing debt stock and its servicing on health financing in borrower countries is, however, limited and conflicting. Three studies of countries in Africa (using data from 1975-1994) (12), Latin America (1985-2003) (13) and South and Southeast Asia (1980-2010) (14) have found a higher debt burden to be associated with less government health spending or less overall social spending, while a larger study of 120 LMICs found the opposite (1995-2010) (15). Other research has identified constraining effects on government health spending of IMF loan conditionalities (e.g. (16, 17)). The identified empirical studies (12-17) are all quantitative multi-country studies. We therefore have a limited understanding of which particular social, economic and political country characteristics and which lending arrangements constrain health spending or support it. Neither do we fully understand whether or how debt servicing affects equity by impacting the distribution of health financing in a given country and resultingly health service access for different population groups. Such in-depth analysis may be necessary to be able to provide relevant policy recommendations to creditors and borrowers.

Another important consideration is that the debt of LMICs originates from various sources, with much coming from concessional Official Development Assistance (ODA) loans. These may have gone to the health sector in the first place, such as for constructing health centres, and may thus have a beneficial effect on population health. The proportion of ODA loan disbursements going to health is however rather small, at less than 4% of all ODA loans to LMICs in 2019 (11). Development lending to other sectors such as the productive, economic and infrastructure sectors may also have effects on health, which may be beneficial, for example through improved food security or improved financial or geographical access to health services, or harmful, for example through pollution or increased availability of health-harmful commodities. Additionally, development lending to these sectors may lead to

economic growth which, if taxed and partly allocated to the health sector, may expand government health spending. These considerations and the data presented in Figures 1-4 give rise to several questions: Does more public external debt servicing lead to less government health spending, and vice versa? Do the positive effects of development lending on health financing outweigh the negative effects, if any, of debt repayments? What are the effects of development loans and the associated debt burden on equity in health financing in borrower countries?

In the 2000s, a steady decline in debt repayments occurred after the Heavily Indebted Poor Countries (HIPC) initiative and the Multilateral Debt Relief Initiative (MDRI) were launched in 1996 and 2005 respectively, offering debt relief to 39 heavily indebted LMICs (18, 19). While some researchers have discussed the potential of these initiatives to free up funds for health financing (20, 21), to the authors' knowledge, no statistical testing of the actual health financing impacts of this natural policy experiment have been performed, and neither have dedicated case-country policy analyses. Nor are we aware of any analyses of the real health financing impacts of the recent G20 Debt Service Suspension Initiative, temporarily delaying debt servicing requirements for 2020 and 2021 for participating countries (22). As such it is not known whether and under what circumstances the assumption of decreased debt servicing freeing up fiscal space actually translates into more health spending.

## 5.5. Recommendations

Figure 5.1 and Figure 5.2 show concerning trends in the debt obligations of LMICs, while Figure 5.3 and Figure 5.4 illustrate the level of debt servicing relative to health expenditure in LMICs. While Covid-19 has shown the world the necessity and importance of health spending, it has also eroded government revenue in many countries and has elicited an increase in development lending. These events prompt us to call for more empirical, peer-reviewed research analysing the following questions:

First, does increased external debt servicing result in lower government health spending? If so, under what circumstances? Conversely, does increased fiscal space generated from decreased debt servicing lead to increased health spending? If so, under what circumstances? Second, do development loans to health and other sectors have a net positive effect on health financing in the long run, when debt repayments are taken into account? What are some individual country experiences in regard to this question?

Third, what are the implications of development loans and ensuing external debt for equity in health financing in borrower countries?

We believe a range of different methodological approaches from the social and public health sciences, both quantitative and qualitative, will be necessary to address these questions. The answers may vary greatly between countries, and in-depth case country studies may be necessary to address these questions in sufficient detail and understand individual country realities.

We ask for creditors to be keenly aware of the illustrated growth in external debt obligations of borrower countries, and to carefully consider the potential short- and long-term impacts of lending on the ability of these countries to finance their health services. The growing debt burdens in LMICs may mean that debt relief will at some point again have to be considered to ensure continued sustainable development in borrower countries. Debt-to-health swaps, where funds freed up from debt relief are earmarked for health purposes, are one way to ensure that debt relief translates into increased health spending (alongside debt-to-education swaps, debt-to-environment swaps, etc.) (23).

For borrower countries, we ask policy makers to prioritize social sectors and safeguard these from budget cuts when debt repayments have to be made. We ask them to carefully consider the short- and long-term social and environmental implications of taking up a development loan, and to exercise strong vigilance about the type of creditor and lending arrangements engaged in, to avoid a disproportional burden of repayment being passed on to the next generation, with ensuing possible implications for the fiscal space for social spending.

## References

1. World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: <u>https://data.worldbank.org/</u>.

2. WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from: http://apps.who.int/nha/database.

6. World Bank. From Double Shock to Double Recovery – Implications and Options for Health Financing in the Time of COVID-19. 2021 [cited 2022 March 22]. Available from:

<sup>3.</sup> McIntyre D, Mooney G. The Economics of Health Equity. 1st ed: Cambridge University Press; 2007. ISBN: 978-0521705066

<sup>4.</sup> WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available from: <a href="https://www.who.int/publications/i/item/9789241564021">https://www.who.int/publications/i/item/9789241564021</a>.

<sup>5.</sup> Jubilee Debt Campaign. Sixty-four countries spend more on debt payments than health. 2020 [cited 2022 March 14]. Available from: <u>https://jubileedebt.org.uk/press-release/sixty-four-countries-spend-more-on-debt-payments-than-health</u>.

https://openknowledge.worldbank.org/bitstream/handle/10986/35298/From\_Double\_Shock\_to\_Double\_Recovery %20Revision%2020210916.pdf?sequence=7&isAllowed=y.

<sup>7.</sup> WHO. Global expenditure on health: Public spending on the rise? 2021 [cited 2022 March 3]. Available from: <u>https://www.who.int/publications/i/item/9789240041219</u>.

<sup>8.</sup> Allan S, Connolly D, Tariq K. The opportunity cost of covid-19 emergency expenditure reallocation. 2021 [cited 2022 March 23]. Available from: <u>https://www.disasterprotection.org/blogs/the-opportunity-cost-of-covid-19-emergency-expenditure-reallocation</u>.

<sup>9.</sup> Santomauro DF, Mantilla Herrera AM, Shadid J, Zheng P, Ashbaugh C, Pigott DM, et al. Global prevalence and burden of depressive and anxiety disorders in 204 countries and territories in 2020 due to the COVID-19 pandemic. The Lancet. 2021;398(10312):1700-12.

10. Chmielewska B, Barratt I, Townsend R, Kalafat E, van der Meulen J, Gurol-Urganci I, et al. Effects of the COVID-19 pandemic on maternal and perinatal outcomes: a systematic review and meta-analysis. The Lancet Global Health. 2021;9(6):e759-e72.

11. OECD. Creditor Reporting System (CRS). 2024 [cited 2023 February 16]. Available from: <u>https://stats.oecd.org/Index.aspx?DataSetCode=CRS1</u>.

12. Fosu AK. The External Debt-Servicing Constraint and Public Expenditure Composition. Helsinki, Finland: UNU-WIDER; 2007.

13. Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 2007;8(4):299-310.

14. Shabbir S, Yasin HM. Implications of Public External Debt for Social Spending: A Case Study of Selected Asian Developing Countries. Lahore Journal of Economics. 2015;20(1):71-103.

15. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

16. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

17. Brunswijck G. Unhealthy Conditions. Eurodad; 2018.

18. IMF. Debt Relief Under the Heavily Indebted Poor Countries (HIPC) Initiative. 2021 [cited 2021 October 12]. Available from: <u>https://www.imf.org/en/About/Factsheets/Sheets/2016/08/01/16/11/Debt-Relief-Under-the-Heavily-Indebted-Poor-Countries-Initiative</u>.

19. IMF. Multilateral Debt Relief Initiative — Questions and Answers. 2017 [cited 2021 December 20]. Available from: <u>https://www.imf.org/external/np/exr/mdri/eng/mdrians.htm#q01</u>.

20. Kaddar M, Furrer E. Are current debt relief initiatives an option for scaling up health financing in beneficiary countries? Bull World Health Organ. 2008;86(11):877-83.

21. Gupta S, Clements B, Guin-Siu MT, Leruth L. Debt relief and public health spending in heavily indebted poor countries. Bulletin of the World Health Organization. 2002;80(2):151-7.

22. World Bank. Debt Service Suspension Initiative. 2022 [cited 2022 March 14]. Available from: https://www.worldbank.org/en/topic/debt/brief/covid-19-debt-service-suspension-initiative.

23. Wuennenberg L. Debt-To-Health Swaps: Financing health system resilience beyond the COVID-19 pandemic. 2022 [cited 2022 March 14]. Available from: <u>https://www.iisd.org/sustainable-recovery/debt-to-health-swaps-financing-health-system-resilience-beyond-the-covid-19-pandemic/</u>.



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

## **RESEARCH PAPER COVER SHEET**

Please note that a cover sheet must be completed for each research paper included within a thesis.

#### SECTION A - Student Details

Student ID Number	lsh1601722	Title	Dr.
First Name(s)	Frederik		
Surname/Family Name	Federspiel		
Thesis Title	External Development Partner and Influence on Domestic Health Finan Middle-Income Countries: Panel Da in Senegal	Internationa neing Sourc ata Analysi	al Monetary Fund ses in Low- and s and Case Study
Primary Supervisor	Josephine Borghi		

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

#### SECTION B – Paper already published

Where was the work published?			
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Choose an item.	Was the work subject to academic peer review?	Choose an item.

\*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

#### SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	Health Economics
Please list the paper's authors in the intended authorship order:	Frederik Federspiel, Josephine Borghi, Melisa Martinez Alvarez, Henning Tarp Jensen

Improving health worldwide

www.lshtm.ac.uk

Stage of publication	Not vet submitted
etage et pasiteation	not jot oublinted

#### SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I designed this study as Principal Investigator (PI) in collaboration with my supervisors. I produced the study protocol and research ethics applications with input from supervisors. I assembled the data, performed the analysis, interpreted the results, and wrote the first draft paper, with the input necessary along the way from my supervisors. My supervisors and I then critically revised the article, with me being responsible for performing the revisions necessary.

#### SECTION E

Student Signature		
Date	20/9-2024	

Supervisor Signature		
Date	23/09/24	L.

## 6. Chapter 6: Development Assistance, Public External Debt, IMF conditionalities and Domestic Health Financing in Low- and Middle-Income Countries

## 6.1. Chapter overview:

This chapter presents the first piece of empirical research conducted for this thesis. It uses regression analysis on multi-country panel data looking for quantitative evidence at the global level for some of the main relationships explored in this thesis. It explores whether aid, debt and IMF conditionalities have discernible effects on GHE-S and OOP on aggregate across countries, seeking to establish the "*what*" at the global level for this research, looking at directions and sizes of effect. By operating at the global level, it sacrifices local knowledge, detail and local applicability, but gains in its ability to test the principal economic hypotheses posed in this thesis. These relationships and accompanying non-financial relationships that are not easily quantifiable are then further explored in Chapters 7 and 8.

The present chapter begins by introducing the relevant literature and hypotheses. It then describes the data used and estimation approach, followed by the results for development assistance, debt and IMF programmes and conditionalities, including sensitivity testing. The chapter finishes with a discussion of its findings in relation to the existing literature, study limitations and conclusions.
## 6.2. Abstract

## Background

Across Low- and Middle-Income Countries (LMICs), Official Development Assistance (ODA) is growing over time, but so are debt burdens and the number of conditionalities that come with borrowing from the International Monetary Fund (IMF). The United Nations, The World Bank, and a number of Civil Society Organisations and academics have deemed debt levels unsustainable and expressed concern for the ability to finance health under this fiscal pressure. In our study, we jointly analyse the relationships between ODA<sup>+</sup> (including funding from the Bill and Melinda Gates Foundation), public external debt, IMF loan conditionalities and key recipient country health financing sources.

#### Methods

We performed a panel data study of 105 LMICs from 2005-2019, investigating associations between Government Health Expenditure as a Source (GHE-S) and Out-Of-Pocket Payments (OOP) as dependent variables, and a set of ODA<sup>+</sup>-variables, public external debt stock and servicing, IMF programme participation and conditionalities and covariates as independent variables. We used the Generalised Method of Moments estimator and performed a range of sensitivity tests to check the robustness of our findings.

## Results

We found some evidence that ODA<sup>+</sup> channelled via the recipient country public sector was associated with reductions in both OOP and GHE-S, measured per GDP. We also found some evidence that increases in public external debt servicing levels per GDP were associated with relative increases in OOP and relative decreases in GHE-S per Current Health Expenditure (CHE). Our main findings showed no relationship between IMF programme participation or conditionalities and GHE-S or OOP. Our findings were variably robust to sensitivity tests.

## Conclusions

We confirm previous findings of fungibility of on-budget ODA<sup>+</sup> for health and add that both onand off-budget ODA<sup>+</sup> for health also appear to subsidise OOP, all with modest effect sizes. Our findings for debt indicated a small shift in the burden of payment for health services from the government onto the user from increasing public external debt service obligations. This provides some additional empirical support to recent calls for debt resolution among more heavily indebted LMICs to avoid the negative health service access implications for the poor associated with health systems relying on OOP. We encourage more country case studies of the effects of growing debt burdens on countries' ability to finance health.

## 6.3. Introduction

Each year, international development agencies and International Financial Institutions (IFIs, i.e. the International Monetary Fund (IMF), the World Bank (WB) and others), send billions of dollars of development assistance and foreign credit to recipient countries, aiming to reduce poverty and achieve socioeconomic development, and in the case of the IMF, to achieve macroeconomic stability including stabilising the balance of payments, improve credit ratings and ensure a country's ability to repay its debts (1). Part of this external financial assistance is provided as loans that require repayment, and come with specific policy conditionalities, that often extend beyond the basic terms of repayment. Specifically, loans obtained from the IMF come with loan conditionalities, traditionally involving measures of austerity, fiscal consolidation and decentralization, with emphasis on private sector-led growth. These different external factors: development assistance, external debt and IMF conditionalities, have been on the rise in recent years.

Official Development Assistance (ODA), including for health, has expanded substantially in real terms, in particular after the Covid-19 pandemic (2). So have the public external debt burdens of Low- and Middle-Income Countries (LMICs) (Figure 6.1) (3, 4), leading the United Nations (5) and the World Bank (WB) (6), among a number of civil society organisation and academics (e.g. (7-12)), to express concern of unsustainable debt levels and how this may affect countries' ability to finance health under this growing fiscal pressure. In particular, Sub-Saharan African (SSA) governments have the highest external debt burdens relative to their budgets (Figure 6.1) (3, 4), and this is where health financing is already scarcest (13). Contrary to common beliefs, after a decline in the 2000's, the mean number of structural conditions in IMF programs have increased since 2008, with some countries having experienced more than 1000 total IMF conditions between 1980-2019 (14).



Figure 6.1: Public and Publicly Guaranteed (PPG) external debt service out of total government expenditure in 83 LMICs separated by region, 2000-2021 (%) (3, 4). Data from 83 countries with complete data.

The influence pathways and associated hypotheses explored in this chapter are outlined in detail in Chapter 3. In summary, these pertain to the relationships between ODA, public external debt and IMF loan conditionalities and the levels and balance of Government Health Expenditure as a Source (GHE-S) and Out Of Pocket Payments (OOP) in LMICs. The effects of development assistance for health sectors on domestic government health spending have been rather extensively studied (15-29) (variably disaggregated as government health spending from domestic revenue or from all revenue), generally reporting a negative effect, consistent with fungibility i.e. crowding out/displacement. Several of these authors have shown that displacement only occurs when development assistance for the health sector is channelled through the recipient country government (20, 21, 23, 24, 29) (also done by Patenaude (2021) (28)).

If on-budget development assistance to the health sector can displace GHE-S, then on-budget development assistance to non-health sectors might also displace government expenditure from those sectors to the benefit of the health sector. These relationships are much less studied. Fosu (2007 and 2010) found a positive effect of overall ODA per Gross Domestic Product (GDP) on SSA governments' share allocation to health sectors (30, 31), while

Antunes et al. (2012) found that General Budget Support (GBS) did not significantly affect GHE-S (22).

As for the relationship between development assistance and OOP, Frimpong et al. (2022) found a negative relationship between external health spending and OOP in SSA (32), and Gabani et al. (2024) found negative relationships between Development Assistance for Health (DAH) and different measures of OOP per household spending only when DAH was channelled via the government (33). Patenaude (2021) found no relationship between DAH and OOP at the global level (28), and Ali et al. (2020) found that neither overall development assistance or development assistance for the health sector per capita affected OOP measured as a proportion of private health spending in SSA between 1995-2015 (34). Contrarily, Younsi et al. (2016) found OOP to increase by between 0.01-0.04% (depending on model specifications) for every 1% increase in DAH in LMICs (27), while Xu et al. (2011) found a 1% increase in external funds for health per capita to correlate with a 0.03-0.06% increase in OOP per capita in lower-middle income countries, with no significant correlation in Low-Income Countries (LICs) and Upper Middle-Income Countries (UMICs) (18).

In terms of debt, there is more evidence for a negative relationship between indebtedness and government health spending than the contrary (variably disaggregated as government health spending from domestic revenue or from all revenue), however the datasets used and model specifications vary greatly. A study of 134 LMICs found a negative effect of a higher debt-to-GDP ratio on overall domestic government health financing from 2000-2015 (28). Others have similarly found that the general debt-to-GDP ratio but not interest payments was negatively correlated with social spending per GDP (50 LMICs, 1985-2003) (35). Variable effects were found (positive, negative and insignificant) in another study of overall government debt servicing on government health expenditure across 85 LMICs from 2000-2013 depending on time subperiod and country subset specifications (36). In a selected panel of seven South and Southeast Asian countries from 1980-2010, a study found a 1% increase in debt servicing to be associated with an 0.25% decrease in social sector spending (37). A study of LMICs between 1995-2010 found a 1% increase in per capita general government debt to be associated with a 0.27% increase in Government Health Expenditure (GHE), and a 0.10% increase for GHE-S (25). A study of SSA countries from 1975-94 found that a one-standard deviation in public external debt servicing was associated with nearly one-third reduction in the government share allocation to the health sector in (30), and strong negative effects of predicted debt servicing were also found in a follow-up study (31).

As for relationships between debt and OOP, one study found no significant relationship between general government debt per GDP and OOP (28), while others found a negative relationship in SSA (exact variables not reported for this association) (38).

The effects of IMF programs and loan conditionalities on health financing have been debated for decades. On one side are academics and civil society, generally criticising austerity measures from the IMF for constraining government health financing based on quantitative econometric analysis, qualitative / mixed-methods policy analysis and opinion writing (39-53). There are however some exceptions to this (54, 55), and IMF and WB staff members have rejected the above findings and criticisms and themselves shown that health spending is protected/increases under IMF programs (56-61) (though also with some exceptions (62)). To the best of our knowledge, associations between IMF conditionalities and OOP are unexplored.

None of the identified studies investigate all of our determinant variables of interest jointly, thus not allowing for a joint comparison of effect sizes and directions. Such a comparison allows one to examine which of the examined macro-fiscal indicators most strongly determines dependence on GHE-S vs. OOP, and thereby which might be the most effective policy levers for pursuing Universal Health Coverage (UHC). In general, there are much fewer studies of the effects of the examined macroeconomic indicators on OOP than on GHE-S. Studies rarely specify dependent variables as proportions of Current Health Expenditure (CHE) as we do in this study, and few studies discuss any equity implications of their findings. Any shift in the balance between the main sources of health financing in LMICs: GHE-S and OOP (13), our dependent variables of interest, affects the degree of overall progressivity in the mix of health financing sources, i.e. the degree of alignment with ability to pay, which tends to be higher for tax-based contributions as compared to user fees (63-67). This in turn has implications for financial inequality, health service access and health outcomes, particularly among poorer population groups (67-75). By examining effects on the degree of reliance on the main domestic health financing sources, GHE-S and OOP, out of CHE, our study uses novel model specifications to explore relationships with direct relevance to UHC, financing progressivity and health service access for poor population groups in LMICs.

In this paper, we aim to investigate the relationships between development assistance, public external debt and IMF loan conditionalities and the levels and balance of GHE-S and OOP. We do so by performing a cross-country panel data study of 105 LMICs between 2005-2019.

## 6.4. Methods

#### 6.4.1. Data sources, time period and variables used

Most of our data were retrieved between 16. October and 1. November<sup>18</sup>, 2023. The data availability for our included variables allowed us to examine the time period 2005-2019. The upper bound of our time period was determined by the IMF conditionality dataset (76) and the lower bound by OECD CRS data separated by channel (2). Our dataset included all countries that were counted as LMICs by the WB in October 2023 (n=134) (77), excluding countries with more than five years of missing data in our time period for health financing, development assistance, debt variables, IMF conditionality variables or GDP (n=29), resulting in 105 countries (see Appendix 4 for full list). Data were compiled in a Microsoft Excel (78) spreadsheet and analysed in Stata (79). Table 6.1 shows summary descriptive statistics for the variables used in our models. A detailed variable description, motivation and explanation of tested hypotheses is available in Appendix 4.

Variable	Obs	Mean	Std. Dev.	Min	Max
OOP / CHE	1566	40.21	19.4	2.17	84.79
OOP / GDP	1566	2.17	1.4	.13	14.95
GHE-S / CHE	1566	40.23	19.05	4.16	80.5
GHE-S / GDP	1566	2.2	1.41	.14	6.82
ODA <sup>+</sup> for health (public) / GDP	1573	.36	.59	0	6.34
ODA <sup>+</sup> for health (civ./priv.) / GDP	1547	.16	.3	0	1.97
ODA <sup>+</sup> for non-health (public) / GDP	1574	2.38	3.54	0	66.13
ODA <sup>+</sup> for non-health (civ./priv.) / GDP	1574	.47	.78	0	7.19
PPG external debt service / GDP	1574	1.89	2.51	0	46.71
PPG external debt stock / GDP	1574	25.76	20.08	.12	232.42
IMF conditionalities	1575	12.48	20.08	0	122
IMF programme participation	1575	.35	.48	0	1
GDP per capita	1574	3537.77	3053.67	200.36	15432.25
Infant mortality rate	1575	35.95	23.25	2.4	124.6
Government effectiveness	1575	57	.55	-2.32	1.16
Corruption control	1575	6	.52	-1.7	1.16
Conflict	1575	.17	.38	0	1
Colonial independence from UK	1575	.26	.44	0	1
Colonial independence from France	1575	.22	.41	0	1
Colonial independence from Spain	1575	.06	.23	0	1

Table 6.1: Descriptive statistics for variables used in models.

<sup>&</sup>lt;sup>18</sup> Except for: ODA<sup>+</sup> for health separated as grants and loans, added on November 10, 2023; US consumer price index data, retrieved on December 5, 2023; country mortality rate data, retrieved on December 20, 2023, and Development Assistance for Health (DAH) from the Institute for Health Metrics and Evaluation (IHME) (<u>https://doi.org/10.6069/HX1C-J716</u>), retrieved on February 28, 2024; ODA<sup>+</sup> for health channelled via NGO's & civil society and private sector institutions, retrieved February 29, 2024. Additional variables for Heckman selection model retrieved from the WB Databank on March 21, 2024.

#### 6.4.1.1. Variables and data sources

#### 6.4.1.1.1. Dependent variables

We used the following dependent variables  $Y'_{it}$ :

OOP per CHE in a given country *i* in year t (in %) (model 1). A higher proportion means that in relative terms, more of a country's health financing comes directly from users at the point of care, relative to other sources of health financing.

OOP per GDP (in %; natural logged) (model 2). This variable measures how much households in a country spend on health services out of pocket relative to the overall size of the economy. GHE-S per CHE (in %) (model 3). This variable measures the level of government health spending out of its own revenue, excluding external transfers, relative to total CHE.

GHE-S per GDP (in %; natural logged) (model 4). This measures the level of government health spending out of its own revenue, excluding external transfers, relative to the overall size of the economy.

Changes in the per CHE-variables reflect relative changes in the burden of payment for health services. Increases can reflect relative decreases in other health financing sources and vice versa, including faster and slower growth rates, and they do not provide any information about the absolute amount of financing.

Increases in the per GDP-variables can also reflect relative decreases in GDP and vice versa, and faster and slower growth rates. However, an increase on average in a health financing source relative to GDP will in our dataset, where year-on-year GDP growth rates were positive in most years, reflect a real expansion of health sector activity financed by OOP or GHE-S respectively.

We used health financing data from the World Health Organisation (WHO) Global Health Expenditure Database (GHED) (13) and GDP data from the WB (13, 77).

## 6.4.1.1.2. Independent variables

Our independent variables included the following: Official Development Assistance (ODA) plus disbursements from the Bill and Melinda Gates Foundation (BMGF), together ODA<sup>+</sup> (80), which counts ODA disbursements (minus debt relief, administrative expenses, in-donor country expenses and promotion of development awareness) plus BMGF grants for health and non-health purposes respectively, measured per GDP (in %; natural logged; lagged). Following the literature, we disaggregated both our health and non-health ODA-variables and associated questions by on-budget and off-budget channel for consistency, with our pathways of interest being the effects of on-budget aid on GHE-S (and in turn on OOP). Data were obtained from the Organisation for Economic Cooperation and Development (OECD) Creditor Reporting System (CRS) database (2). We included Public and Publicly Guaranteed (PPG) external debt service and stock respectively per GDP (in %; natural logged; lagged), obtained

from the WB International Debt Statistics database (4). We also included the number of binding and non-binding conditions in place in a given country-year, obtained from the IMF Monitor conditionality dataset (76). We added IMF participation as a dummy variable for when a country has an IMF programme in place, as measured by the presence or absence of conditionalities as recorded in the dataset.

### 6.4.1.1.3. Covariates

The following covariates were included: GHE-S per GDP, which was not included when GHE-S was part of the dependent variable due to covariance. GDP per capita in constant 2020 US\$ (natural logged) obtained from the WB (77). Infant Mortality Rate (IMR) defined as the number of children dying in their first year of life per 1,000 live births from the United Nations Inter-Agency Group for Child Mortality Estimation (81). Government effectiveness, which is a measure of the perceived quality of government, public and civil services, policy formulation and implementation, measured in standard deviations from the Worldwide Governance Indicators (WGI) database (82). Corruption control, defined as the perceived degree of exercise of public power for private gain and state capture by elite and private interests, measured in standard deviations, also from the WGI database (82). A conflict dummy variable attaining the value of 0 if the number of battle-related deaths in a country-year is less than one per one million population and 1 if equal to or higher, using the Uppsala Conflict Data Program / Peace Research Institute Oslo Battle-Related Deaths Dataset (83). Three binary variables capturing whether a country obtained its colonial independence from the United Kingdom, France or Spain, using Issue Correlates Of War data (84).

## 6.4.1.2. Variable transformations

As per above, all variables showing a skewed distribution were natural-log transformed. This A) normalised their distribution, and B) eased economic interpretation as elasticities. All financial variables except per CHE-variables entered models as proportions of GDP, similar to other authors (24, 29, 35, 85) (in %; natural logged), to examine the effects of financial flows relative to the size of the economy. This was done primarily to avoid introducing spurious correlations by dividing all financial variables with the same deflator numbers, which would be necessary to adjust for inflation if examining absolute or per-capita amounts. It also allowed for the omission of a population covariate necessary in the case of measuring health financing variables directly, in turn increasing our degrees of freedom. In terms of economic interpretation, our resulting health financing measures reflect real expansions/contractions of health financing sources beyond what would be expected simply as a result of background changes in the size of the economy and following inflation levels. The effects of absolute aid and debt flows may also greatly depend on at what level relative to the size of the economy they occur.

#### 6.4.2. Econometric methods

#### 6.4.2.1. Full model specification

Our full model specification was as follows:

$$\begin{split} Y_{it}' &= \beta_{0} + \beta_{1}' Y_{i,t-1}' + \beta_{2}' ln \left( \frac{ODA^{+} for health}{GDP} \right)_{i,t-1}' + \beta_{3}' ln \left( \frac{ODA^{+} for non-health}{GDP} \right)_{i,t-1}' \\ &+ \beta_{4} ln \left( \frac{debt \ service}{GDP} \right)_{i,t-1} + \beta_{5} ln \left( \frac{debt \ stock}{GDP} \right)_{i,t-1} \\ &+ \beta_{6} IMF \ conditionalities_{it} + \beta_{7} IMF \ participation_{it} + \beta_{8} ln \left( \frac{GHES}{GDP} \right)_{it} \\ &+ \beta_{9} ln \left( \frac{GDP}{cap} \right)_{it} + \beta_{10} IMR_{it} + \beta_{11} gov. \ effectiveness_{it} \\ &+ \beta_{12} corruption \ control_{it} + \beta_{13} conflict_{it} + \beta_{14}' independence \ from'_{it} + \mu_{i} \\ &+ \nu_{t} + \varepsilon_{it} \end{split}$$

 $\mu_i$  is country-specific, time-invariant effects (country fixed effects),  $\nu_t$  is time-invariant crosscountry effects (year fixed effects, e.g. shocks, included as year dummies), and  $\varepsilon_{it}$  is the error term.

#### 6.4.2.2. Variable reduction

As our full model specification tested multiple hypotheses at once, some of which turned out not to be supported by the results, we ran backward stepwise regressions to generate reduced models. We sequentially removed independent variables with a p-value > 0.2, starting with the variable with the highest p-value. This was done to A) increase parsimoniousness, B) reduce the risk of type II error due to multicollinearity, and C) lower the risk of significant findings being due to multiple comparisons. The resulting reduced form models are presented in Table 6.2. The full model regression results are presented in Appendix 4.

Table 6.2: Two-step	system GMM	results for reduc	ed form mode	l specifications
---------------------	------------	-------------------	--------------	------------------

Dependent variable	1a) OOP/CHE	2a) Ln(OOP	3a) GHE-S/CHE <sup>†</sup>	4a) Ln(GHE-S
		/GDP)		/GDP)
L. dependent variable	0.662*** [0.059]	0.928*** [0.061]	0.710*** [0.080]	0.694*** [0.082]
L2. Dependent variable	-	-	0.229*** [0.075]	
L. $\ln(ODA^+ \text{ for health (public) } /$	-0.412 [0.268]	-0.024** [0.010]	-	-0.026** [0.013]
GDP)				
L. ln(ODA+ for health (civ./priv.)	-0.339 [0.215]	-0.012** [0.006]	-	-
/ GDP)				
L. ln(ODA+ for non-health	-	-0.038** [0.018]	0.789* [0.467]	-
(civ./priv.) / GDP)				
L. ln(debt service/GDP)	0.725** [0.350]	-	-0.664** [0.287]	-
ln(GHE-S / GDP)	-8.228*** [1.628]	0.072** [0.036]	-	-
ln(GDP/cap)	-1.340 [1.801]	-0.251*** [0.074]	1.395 [2.191]	0.110 [0.218]
IMR	-0.144*** [0.053]	-0.003* [0.002]	-0.019 [0.058]	-0.001 [0.003]
Gov. effectiveness	1.144 [1.420]	-0.002 [0.045]	-1.144 [1.237]	-0.030 [0.063]
Corruption control	-2.241 [1.617]	0.063 [0.059]	3.180** [1.310]	0.054 [0.078]
Conflict	0.387 [0.748]	0.039* [0.021]	-0.064 [0.552]	0.037 [0.031]
Ind. from UK	-1.966 [1.486]	-0.019 [0.045]	0.180 [0.941]	0.041 [0.077]
Ind. from France	0.274 [1.589]	0.043 [0.043]	0.461 [1.164]	-0.083 [0.063]
Ind. from Spain	-0.303 [1.626]	0.018 [0.056]	1.842 [1.150]	-0.004 [0.089]
Constant	-31.033* [16.747]	1.273** [0.535]	-5.071 [17.666]	-0.697 [0.763]
Observations	1429	1429	1356	1460
Countries	105	105	105	105
Instruments	77	97	88	73
Lag limits (years)	5	7	8	7
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
AB-test AR(2) (p-level)	0.944	0.785	0.659	0.194
Hansen test (p-level)	0.402	0.469	0.242	0.346
Diffin-Hansen test (p-level)	0.150	0.543	0.908	0.725
F-statistic	495.52***	128.67***	1794.7***	117.58***

AB-test AR(2) = Arellano-Bond test for first-order autocorrelation in the levels equation. Diff.-in-Hansen test: Difference-in-Hansen test for exogeneity of GMM instruments in levels equation. Govt.= Government. IMR= Infant Mortality Rate. Ind. = Independence (colonial). L. = 1-year lag. L2. = 2-year lag. Ln= natural logarithm.

<sup>†</sup> 2<sup>nd</sup>-order lag of dependent variable included as independent variable in model 3a to avoid serial correlation as determined by the AB-test.

Windmeijer-corrected robust standard errors in brackets; \*p < 0.1, \*\*p < 0.05, \*\*\*p < 0.01. We used a significance level of 0.05 for all regressions and associated discussions, while results significant at the 0.10-level were only discussed for sensitivity tests where main model specification results were significant at the 0.05-level, except for the case of contemporaneous IMF programme participation (appendix), which warranted separate discussion.

#### 6.4.2.3. Estimation strategy

#### 6.4.2.3.1. Generalized Method of Moments

To overcome serial autocorrelation and heteroskedasticity (see Appendix 4), we adopted twostep system Generalized Method of Moments (GMM) estimation (86-91), using fixed effects following the Hausman test (92), employed with the Stata command "Xtabond2" (88). The twostep GMM is a more efficient estimator and has been designed for panel data analysis with a relatively small number of time periods and a larger number of units of observation (countries) (88, 91), as in our case.

All explanatory variables except colonial independence variables and year-dummies were treated as endogenous (GMM-style instrumentation), while colonial legacy and year-dummies were treated as exogenous (Instrumental Variable (IV)-style instrumentation) (88, 93, 94). We

used the Windmeijer-correction to correct for the tendency of two-step system GMM to produce downward-biased standard errors (88, 95). To lower the instrument count, we collapsed the instrument matrix (88, 93, 94, 96). We also applied upper bounds to the number of year-lags used for GMM-style instrumentation (the so-called "lag limit" (88)), and ensured that each model was valid by adjusting the lag limit, guided by the instrument count and the following tests:

We used the Arellano-Bond test to check for first-order autocorrelation in the levels equation (88, 89, 93, 94). We used the Hansen test (86) to explore the validity of our generated moment conditions and avoiding overidentification, ensuring p-values between 0.15 and 0.6 (93, 97, 98). We used the difference-in-Hansen test to ensure exogeneity of the generated GMM-style instruments in the levels equation (88, 96).

As expected, moderate to strong serial correlation was present in our dependent variables. We have explained our approach to this issue in Appendix 4.

### 6.4.2.3.2. Lag structure

For choosing the number of lags on our independent variables of interest in our main model specifications, we first specified 2-year lags for all of these and then sequentially reduced the lag-level (93, 94). This process showed that a parsimonious lag-specification of 1-year lags for all financial independent variables of interest, without contemporaneous or two-year lagged variable versions, generally, with a few exceptions mentioned in the results section, identified the highest level of significance and optimised the above test statistics. Contemporaneous variable versions, with the noted exceptions, were generally found to be insignificant, or in the case of joint insignificance at both t and t-1, had lower t-values than at t-1. When including both unlagged and 1-year lagged independent variables of interest, strong issues of multicollinearity were encountered, degrees of freedom lost, and issues of overspecification as well as instrument proliferation from inclusion of insignificant explanatory variables substantially reduced the room for manoeuvre in the search for an appropriate model within acceptable diagnostic test parameters, in turn increasing the risk of committing type-II errors. Contemporaneous variable version for financial independent variables of interest were therefore not included. This specification type follows the work of other authors in our field (15, 35, 39, 99). Its consistency and parsimoniousness also limited confirmation bias or "cherrypicking" of significant results, to the extent possible with the GMM method (see limitations). Economically, a 1-year lag of the financial variables of interest also made sense as the fiscal implications of these macroeconomic factors may not be realised until the following year.

IMF conditionalities and IMF programme participation, however, apply to the current financial year, they did not show the same significance patterns for their lags, and were therefore kept unlagged.

## 6.4.2.4. Diagnostic tests

We performed twelve unit root tests for each dependent variable (48 in total, available in Appendix 4): 4 variations of the Fisher-type test (augmented Dickey-Fuller and Phillips-Perron), Im-Pesaran Shin (IPS) test and Levin-Lin-Chu test – with trend, demeaned with trend, and with or without Akaike Information Criteria-determined lags (Appendix 4). As a significant linear time trend was present in all our dependent variables, we did not perform these tests without accounting for a time trend. All 48 tests rejected the null-hypotheses that panels contained unit roots.

We tested for multicollinearity by calculating Variance Inflation Factors (VIFs) for all models (Appendix 4). This showed the expected collinearity of 1<sup>st</sup> and 2<sup>nd</sup> order lags of the dependent variable in model 3a (VIFs 17-18), moderate multicollinearity of ODA-variables in full models (VIFs 5.7-6.4), and moderate multicollinearity of  $ln\left(\frac{GDP}{cap}\right)$  in full models (VIF about 5). This among other factors led us to generate the reduced form models as seen in Table 6.2, where VIFs were less than 5 for all variables, except for the dependent variable lags in model 3a, which were necessary to convincingly pass (p>0.15) the Arellano-Bond test for serial correlation in the levels equation. This approach should have adequately addressed the issue of multicollinearity and the associated risk of Type II error, and we did find a number of significant correlations. Correlation matrices are available in Appendix 4.

We tested for selection bias into an IMF programme using Heckman selection models (55, 100, 101) (described in Appendix 4). The results of these tests were insignificant, meaning that unobserved variables that make IMF programme participation more likely were not significantly associated with our dependent health financing variables, and an Inverse Mills Ratio was therefore not included in our main regressions.

## 6.4.2.5. Sensitivity analysis

We ran a series of sensitivity tests and robustness checks to critically interrogate the internal validity of our findings.

We ran our models with no lags and with 2-year lags on the independent variables of interest. We examined the robustness of our results to changing the number of generated GMM-style lags by altering the lag limit. We ran our models with all financial variables, including dependent variables, being per capita instead of per GDP, as is often also done in the literature (e.g. (15, 102)). We used the same US\$ Consumer Price Index (CPI) data for deflation of all per-capita financial variables (77). We tried swapping IMR with Under-5 Mortality Rate (U5MR) (81) and Maternal Mortality Ratio (MMR) as alternative health need indicators (103).

We explored the effect of using alternative measures of conditionality, i.e. including only binding conditions or not, and attributing different weights to binding vs. non-binding conditions, as binding conditions could be more influential than non-binding conditions (76).

To adjust for any regional and income group effects on our dependent variables, we ran our models with a regional dummy for SSA instead of colonial independence dummies; with Lower Middle-income Country (LMC) and Low-Income Country (LIC) WB income group dummies instead of colonial independence dummies; and without any of these dummies altogether, as they were generally found to be insignificant.

We explored a battery of interaction terms between our main independent variables of interest and the region being SSA, being a LIC, having a higher degree of government effectiveness, having less corruption, and having a higher level of GHE-S relative to GDP. These tested the respective sets of hypotheses that ODA<sup>+</sup> for health, PPG external debt service and IMF conditionality could have different effects on GHE-S and OOP under these five conditions. For example: ODA<sup>+</sup> for health could displace GHE-S more in SSA compared to other regions; debt service could shift the burden of payment more from the government onto the user in LICs; more effective governments could be subjected to milder conditionality with less impact on health financing – or could implement conditionality more effectively resulting in a larger impact; ODA<sup>+</sup> for health could be more effective at reducing OOP in less corrupt countries; and debt servicing could have differential impact on the health system payment pattern at different levels of GHE-S.

We ran our models without log-transformation.

We tested the effect of using US GDP deflator data instead of US consumer price index data for deflation (77).

We tried aggregating our ODA-variables into just health and non-health purposes, as was done in some of the earlier fungibility literature (e.g. (15)).

Finally, we tried adding Inverse Mills Ratios to our full models to adjust for any selection bias for IMF programme participation present even at a non-significant level (i.e. p>0.05).

Regression outputs were exported using the Stata command "Asdoc" (104) and are available in Appendix 4.

## 6.5. Results

Table 6.2 shows the results of our four final reduced form model specifications. The results of our full model specifications are available in the Appendix 4, and these are described in the sensitivity analysis section (6.5.4). Findings are summarised for each set of independent variables of interest along with sensitivity testing. Significant findings for our covariates are described in Appendix 4.

## 6.5.1. Official development assistance

In our analysis of ODA-variables, we found some evidence that ODA<sup>+</sup> for health channelled via the recipient country public sector was significantly associated with a reduction in OOP expenditures as well as a decrease in GHE-S measured as a share of GDP (evidence of displacement of both). Specifically, we found that a 1% increase in the lag of ODA<sup>+</sup> for health purposes disbursed via the recipient country public sector per GDP was associated with a - 0.024% reduction of OOP/GDP (p=0.02), and a -0.026% decrease in GHE-S/GDP (p=0.045). When channelled via the civil/private sector, the negative association with OOP/GDP remained but not with GHE-S/GDP, i.e. there was no evidence of off-budget ODA<sup>+</sup> for health leading governments to lower GHE-S. We also did not find any evidence for our hypothesized positive effect of on-budget ODA<sup>+</sup> for other purposes than health on GHE-S. We found no significant association between ODA<sup>+</sup> for health via the public sector per GDP and OOP/CHE or GHE-S/CHE. Overall, effect sizes were small.

## 6.5.2. Debt

For debt, we found that a 1% increase in lagged PPG external debt servicing per GDP was positively associated with a 0.007 %-point increase in OOP/CHE (p=0.041). It was also associated with a -0.007 %-point decrease in GHE-S/CHE (p=0.023).

Lagged PPG external debt servicing did not correlate with the remaining dependent variables. The level of PPG external debt stock per GDP was not associated with any dependent variables. Again, effect sizes were small.

## 6.5.3. IMF programmes and conditionalities

IMF programme participation and the number of IMF conditionalities were removed during our model reduction process, as they were not found to significantly influence any dependent variables in our full models or in sequential model reduction steps.

## 6.5.4. Sensitivity testing

## 6.5.4.1. Full models, alternative lag specifications and covariate swaps

Our significant results for ODA-variables were robust to some specification changes but not to others (Appendix 4 and below). The significant negative association between on-budget

ODA<sup>+</sup> for health purposes per GDP with GHE-S/GDP remained in our full model, but not for OOP/GDP (p=0.086).

Our finding that on-budget ODA<sup>+</sup> for health purposes per GDP was significantly associated with a reduction of OOP/GDP and decrease in GHE-S/GDP was also present at the 2<sup>nd</sup> lag, but not in a contemporaneous specification. Its correlation with OOP/GDP was robust to increasing the GMM-style lag limit, but not to decreasing it, and it was robust to most but not all covariate swaps (Appendix 4). Its negative correlation with GHE-S/GDP was only robust to increasing the lag limit by 1 but not to decreasing it, however it was robust to all covariate swaps (Appendix 4).

The significant positive correlations between lagged PPG external debt servicing and OOP/CHE and negative with GHE-S/CHE were sensitive to some model modifications with the latter being more robust (Appendix 4). The significantly positive association between lagged PPG external debt servicing per GDP and OOP/CHE was not robust to changes in lag limits, and only robust to excluding/swapping colonial history variables but not mortality indicators. The negative association with GHE-S/CHE was robust to increasing, but not decreasing lag limits, and it was robust to all covariate swaps. None of our debt variables were significant at the 2-year lag or unlagged. The variables were insignificant in full models (the positive OOP/CHE correlation for debt servicing was near-significant at p=0.057).

We found a borderline significant negative association between IMF programme participation and GHE-S/CHE only in our full model (p=0.08), which was significant in a contemporaneous full model version (-1.13 %-point decrease from IMF participation (p=0.029) (Appendix 4). In this model version, the finding was still present at GMM-style lag limits 3, 4 and 5, but not 2 and 6 (model overspecified at lag limit 6). It was robust to most covariate swaps. We checked for reverse causation, which was not present. IMF variables were not significant at the 1<sup>st</sup> or 2<sup>nd</sup> lag and did not become significant by modifying lag limits.

#### 6.5.4.2. Interaction terms

None of the tested interaction terms had any robustly significant effects on any of our relationships of interest within acceptable model test parameters (Appendix 4).

#### 6.5.4.3. Alternative variable specifications

When measuring our financial variables per capita instead of per GDP, including dependent variables, we found some evidence of crowding out/displacement from other sectors benefiting the health sector: lagged ODA<sup>+</sup> for non-health purposes channelled via the public sector per capita had a positive association with GHE-S per capita (semi-elasticity: -0.0004,

p=0.019) (full model). Similar to our main findings, lagged debt service per capita had a negative association with GHE-S/CHE (p=0.02). Our findings for ODA<sup>+</sup> for health via the public sector were not present when specifying per capita.

## 6.5.4.4. No log-transformation

When measuring our variables directly and not log-transformed, including dependent variables, this meant that we deviated from much of the literature in our field and introduced substantial right-skew of many variables into the models. The findings with this specification deviated substantially from our main model findings. We confirmed that no unit root was present for direct dependent variables. We were unable to arrive at a specification for model 1 that convincingly passed diagnostic tests without overspecification. We again found a significant negative correlation between ODA<sup>+</sup> for health channelled via the public sector per GDP and OOP/GDP in both reduced and full models. We also found a new positive association between IMF programme participation two years prior and OOP/GDP (0.17 %-points, p=0.001). We found a new significant negative association between lagged PPG external debt stock per GDP and GHE-S/CHE (-0.06 %-points, p=0.015), and a new positive association between ODA<sup>+</sup> for health per GDP channelled via the civil/private sector and GHE-S/GDP (0.17 %-points, p=0.037).

## 6.5.4.5. Alternative deflation data

Using US GDP deflator data instead of US consumer price index data for deflation of percapita variables had no significant effect on any of our results (77).

## 6.5.4.6. Aggregating ODA-variables

When aggregating our logged ODA-variables into just health and non-health purposes irrespective of channel of assistance, the "fungibility effect" of a decrease in GHE-S/GDP from an increase in lagged ODA<sup>+</sup> for health per GDP was not found again, while the negative association with OOP/GDP remained (p=0.01) (Appendix 4). Again, the lag of debt service per GDP was positively associated with OOP/CHE (p=0.04), but no longer with a decrease in GHE-S/CHE.

## 6.5.4.7. Adding Inverse Mills Ratios

Finally, adding Inverse Mills Ratios to our full models did not cause any substantive changes to our overall results or conclusions.

Apart from when changing the variable being measured (per capita, not log-transformed, aggregated etc.), effect sizes did not change substantially in sensitivity tests<sup>19</sup>.

<sup>&</sup>lt;sup>19</sup> Some changes in coefficients and p-values were seen when including Inverse Mills Ratios, which was most likely due to the loss of degrees of freedom resulting from the inclusion of new explanatory variables with incomplete data for our Heckman selection model and the following need to adjust some model lag limits to avoid overspecification (see Appendix 4).

## 6.6. Discussion

The objective of this study was to explore the relationships between external development financing, public external debt, loan conditionalities from the IMF, and key recipient country health financing sources. We discuss our main findings in relation to the literature in the below sections. Policy implications and research recommendations are discussed in the Conclusions section.

## 6.6.1. ODA<sup>+</sup> for health and health financing sources

Our first main finding was that both lagged on-budget and off-budget ODA<sup>+</sup> for health were associated with reductions in OOP/GDP, but not OOP/CHE. This offers some evidence that these types of assistance from External Development Partners (EDPs) modestly displace or subsidise OOP, but do not do so in a manner in which the degree of reliance on OOP out of total CHE is reduced. The latter could at least in part be explained by the finding that on-budget ODA<sup>+</sup> for health also displaces GHE-S (measured per GDP) with a negative elasticity of a comparable magnitude, although effects on other health financing sources were not investigated.

These findings broadly align with results from two previous studies. One panel study showed a negative relationship between external health spending and OOP in SSA (32), and a fixed-effects/pseudo-panel study based on household survey data among 65 LMICs showed a negative relationship between on-budget DAH (DAH-G) per capita and health OOP per total household spending (33). Compared to the latter study, our findings however extend to also show an OOP-displacing effect for off-budget development assistance to the health sector. Our findings however differ from other authors, but methodological differences readily explain

this, and direct comparisons are difficult to make. Studies have found no association between DAH-G or DAH-NG and OOP on either a log-log scale or level scale<sup>20</sup> (28), and significant positive relationships between total DAH and OOP measured directly (i.e. without denominator) (27) and between external funds for health / cap (essentially DAH/cap) and OOP/cap only in Lower-MICs<sup>21</sup> (18).

<sup>&</sup>lt;sup>20</sup> Most of these results suffer from serial correlation in the levels equation.

<sup>&</sup>lt;sup>21</sup> Xu et al. split their panel into subgroups with only 27 countries and 329 observations in some groups. Splitting our panel would necessitate a change of estimation method to an alternative IV regression method e.g. two-stage-least-squares (2SLS), which behaves inappropriately in unbalanced panels and requires homoskedasticity for efficiency ((88)). This would result in losing the superior performance of GMM in the face of unbalanced data, heteroskedasticity, serial correlation and endogeneity (88), which would further limit the scope for causal inference than is already the case for GMM estimation on observational data.

All of the cited studies differ internally and from ours in model specification, variable transformation, country inclusion and grouping choices, time period, and extent of diagnostic and sensitivity testing performed. Common to the latter three articles (18, 27, 28) is also that all/nearly all of their GMM models have Sargan/Hansen p-values of 1.00, indicating overspecification, and without a description of an approach to limit instrument proliferation (88, 96). Other authors have measured OOP as a proportion of private health spending, which again renders direct comparisons challenging (34). Our finding was not fully robust to sensitivity tests, and we resultingly advise some caution when interpreting this finding and its potential policy implications.

Our second main finding was that lagged ODA<sup>+</sup> for health channelled via the recipient country public sector was associated with reductions in GHE-S/GDP. This effect was not found when channelled outside of government or when measured jointly, i.e. irrespective of channel. This supports the interpretation of a modest displacing effect specifically of on-budget ODA<sup>+</sup> for health on GHE-S (fungibility). This finding aligns with the majority of the extensive, and at times conflicting, body of literature on the fungibility of development assistance to the health sector (15-29). In particular, the fact that we were unable to identify fungibility when not distinguishing between on- and off-budget development assistance for health aligns with similar/analogous findings by others (20, 21, 23, 24, 28, 29). A government can only respond to what it can see, and if ODA<sup>+</sup> for health is distributed directly from an EDP to the civil or private sector, this may not be in clear view for the government and may thus not trigger any fiscal redistributions.

The caveat needs to be made that some of the above literature has also faced methodological criticism (29, 98), and the same concerns mentioned in the previous section again apply to the same studies<sup>22</sup>. Again, our finding was not fully robust to all sensitivity tests, and the effect size was small. We have however subjected the fungibility hypothesis for on-budget ODA<sup>+</sup> for health to extensive scrutiny, going beyond existing studies, and our main findings are confirmatory.

Of note, in an alternative specification measuring our variables per capita, we did not find fungibility of ODA<sup>+</sup> for health, but instead that ODA<sup>+</sup> for non-health purposes provided via the public sector had a significant positive association with GHE-S, in this specification supporting

<sup>&</sup>lt;sup>22</sup> Further quality issues deserve mention such as no/limited sensitivity testing of results ((16, 105)), or using two-way, fixed effects Generalised-Least Squares (GLS) estimation of observational data to make causal claims and policy recommendations ((16)), which has been called into question ((106)).

the interpretation that external funding injections to other public sectors allow governments to shift some additional funds to the health sector. Our main specification did not confirm this relationship. This highlights the sensitivity of these types of models to variable choices. The fungibility hypothesis for on-budget development assistance to the health sector is now well supported in the literature, and future research could further scrutinize its inverse: if development assistance to non-health sectors free up domestic funds to the benefit of the health sector.

## 6.6.2. Debt and health financing sources

We found evidence that increases in lagged public external debt servicing levels per GDP were associated with relative increases in OOP and relative decreases in GHE-S measured as proportions of CHE. The association with GHE-S/CHE was more robust than that for OOP/CHE, and neither was fully robust to all sensitivity tests. The fact that OOP/CHE was found to increase by a similar amount as GHE-S/CHE decreased points to the possibility of changes in OOP/CHE being mediated by changes in GHE-S/CHE. Overall these findings provide some novel cross-sectional evidence in support of the hypothesis that increasing public external debt servicing reduces the remaining envelope for other government expenses, including for public health, resulting in a shift in the burden of payment for health services from the government toward the user. This can be seen as worsening the overall progressivity of health financing. On average, OOP and GHE-S each constituted 40% of CHE across our sample of country-years, meaning our dependent variables reflect the bulk of health financing across the studied LMICs and years.

The effects seen were however not identifiable when measuring GHE-S and OOP per GDP or per capita, which challenges the above interpretations, unless the effects from debt are indeed more compositional and relative in terms of government budget allocations and overall health system payment patterns, but neutral in proportion to increases over time in economic output and population.

This part of our findings broadly align to some extent with the majority of the relevant identified econometric literature. This includes a regional study that found negative associations between general debt servicing and social spending in 7 South- and Southeast Asian countries<sup>23</sup> (37), and three studies that found negative associations between debt burdens in SSA and countries' share allocation to the health sector out of the total government budget (30, 31, 38)<sup>24</sup>. One of these reported a negative relationship between the public debt burden

<sup>&</sup>lt;sup>23</sup> Caveat for limited reporting of GMM diagnostic tests.

<sup>&</sup>lt;sup>24</sup> About <sup>3</sup>/<sub>4</sub> of observations missing for panel (172 observations across 43 countries and 15 years).

and OOP in SSA as a robustness check, though without showing those regression results (38). At the global level, another study found a negative association between the general government debt-to-GDP ratio and domestic government health financing across 134 LMICs, and no significant relationship with OOP (28). Others have had mixed findings for public health spending (36) (same methodological caveats apply as above)<sup>25</sup>, and one study has found positive associations with GHE-S using FE-2SLS (above caveats apply) (25).

Differences between previous findings and ours may likely be due to significant differences in data used, model specification and estimation technique, and we have addressed some methodological issues that may have been present in previous studies, including instrument proliferation. Viewed jointly, while our study adds to the side of the econometric literature finding negative (and absent) links between debt and government health financing, we deem the identified literature, our study included, inadequate for forming a final, general verdict on this relationship. Rather, the cross-sectional evidence available, ours included, should be seen as one type of evidence improving our understanding of these relationships, that can then be supplemented by other types of evidence, e.g. country-case studies.

# 6.6.3. IMF programme participation, conditionalities and health financing sources

We found no association between IMF programme participation or number of IMF conditionalities and our dependent variables in our main models. Only in a contemporaneous full model version did we find a partially robust, significant negative association between IMF participation and GHE-S/CHE, and not with GHE-S/GDP (Appendix 4). Viewed jointly, our findings do not convincingly support the hypothesis of a constraining effect of IMF programmes and conditionalities on GHE-S, nor of a promotion of OOP.

As with the above literature, there are numerous differences in data used, specification choices and estimation methods amongst previous studies of IMF effects on health financing, with equal divergence in the empirical findings from econometric studies. Our mostly negative findings align most with IMF/WB's own work, that has used a variety of methods generally finding that government health spending is not impacted under IMF programs, or even that it increases under IMF programs (the latter of which our findings do not support), including from its Independent Evaluation Office (57-61) (although with some exceptions (62)). While academics have at times found similar results (54, 55, 108), criticisms and findings of IMF

<sup>&</sup>lt;sup>25</sup> Reported Sargan statistics were all 1.000, no approach to instrument curtailing was described, and for two-step system GMM (same as our method) under conditions of heteroskedasticity and serial correlation in the error term, the Hansen test should be relied upon (107)

programmes and conditionalities constraining government health spending prevail on the part of academics and civil society (39-53).

While this debate is yet unresolved in the econometric literature, qualitative investigations into the health financing impacts of IMF conditionality are scarce (e.g. (39)). Such studies might help elucidate why findings differ, and more importantly generate learnings at the country level for which IMF policies and conditionalities help promote pooled health spending and which do not.

## 6.6.4. Limitations

Using observational data, no matter the level of sophistication of the statistical techniques applied, limits the scope for causal inference compared to using experimental data. We have sought to use the optimal econometric methods currently available for exploring our research questions with the cross-sectional data available. The system GMM estimator eliminates time-invariant confounders (fixed effects) from the first-differenced but not the levels equation, while time-variant confounders remain (88, 109). Our choices of covariates and sensitivity testing with variable swaps should have addressed some confounding, but residual confounding will remain in ours and any other model. As an example, there are limitless different health needs that may drive both health spending but also development assistance for health, all of which cannot be captured by any single variable. No sensitivity tests swapping single variables or adding a few can ever fully address this, however the GMM estimator partially mitigates this issue by instrumenting, addressing the issue of endogeneity and estimating a purer and more accurate relationship between independent and dependent variables (110).

By prioritising valid application of the best available estimator for our data, our study was limited in its ability to inform sub-group questions, such as differences in effect across all regions and income groups. This may have caused us to overlook especially regional effect modification present in other regions than SSA. We thus advise caution in making inferences at the regional level - and in particular at the country level, as for other global-level studies.

We have addressed and discussed the potential issue of selection bias into IMF programmes, and our choice of covariates should have partially addressed the issue of selection bias in terms of who obtains development assistance including loans. Residual selection bias will however necessarily remain. Measurement error is also an omnipresent and underrecognized issue in the general development economic and global health economic literature (111). We have used standardised data from the best available data sources for our research questions, removed countries with significant missing data, performed a range of sensitivity tests, and used GMM that helps overcome this source of endogeneity (112), however some degree of bias from measurement error will remain by default (111). Also, what is counted as ODA constitutes a heated topic of debate, and measurement methods change over time (113). However, we had to rely directly on OECD data to be able to disaggregate on- and off-budget ODA.

We have explored whether non-linear relationships were present by examining relevant interaction terms, however further non-linear relationships are conceivable.

Finally, the variables in our main reduced form models did not turn out significant in ways that allowed for a deeper comparison of effect sizes between independent variables as envisaged. Our study was thus limited in its ability to answer comparative questions such as: "Does aid displace government health spending more than debt servicing constrains it?", similar to e.g. (30). Alternative study designs, perhaps with longer panels if omitting IMF variables, might help address this.

In summary, we believe our analysis has provided the foundation for a careful causal interpretation for some of the associations identified, with some caveats, as described in the conclusions.

## 6.7. Conclusions

## 6.7.1. Policy implications

Our study provided confirmatory evidence for the existing fungibility hypothesis of on-budget ODA<sup>+</sup> for health with GHE-S, and added some evidence that both on- and off-budget ODA<sup>+</sup> for health also appear to displace or subsidise OOP, when measured per GDP. The latter should encourage EDPs in the sense that while it appears they do to some extent act as subsidising agents for recipient governments, it also seems they help subsidise health service costs paid with the most inequitable source of financing in LMICs.

Providing GBS overcomes the issue of fungibility of sectoral funding disbursements at the expense of EDP control of funds. Agreements for minimal levels of government health spending could accompany GBS to ensure benefit to the health sector. Co-financing requirements is an available alternative policy tool for addressing fungibility. This should however ideally only be considered in adherence with the Paris and Accra principles for development cooperation, particularly the principle of country ownership (114), and keeping in mind the broader fiscal repercussions of multiple EDPs requiring co-financing for their preferred policy area. Importantly, while displacement of GHE-S was only identified for on-

budget ODA<sup>+</sup> for health, this should not be interpreted as an argument for providing off-budget ODA<sup>+</sup> for health, as this may be associated with inefficiencies from poor/absent coordination, duplication of efforts, and may undermine domestically owned development plans for the health sector.

We also provided novel evidence that public external debt servicing led to relative shifts in the mix of domestic health financing sources from GHE-S toward OOP, measured per CHE, however not when measured per GDP or per capita. Our results thus point to a compositional effect in health system payment patterns from government onto user, found to be neutral in proportion to economic output and population, but implying decreasing progressivity in the mix of domestic health financing sources with increasing public external debt servicing obligations. This new identified relationship provides modest added support to the argument for avoiding large public external burdens from a health financing perspective and should be considered by both external creditors and governments in LMICs. The argument has been made by the United Nations (5), The WB (6), and a number of civil society organisations and academics (e.g. (7-12)), that the critical debt levels in many countries may put government health financing at risk, and resolution is needed to ensure that all public sectors, health included, can deliver their essential services. Higher debt servicing obligations can be avoided by a number of ways, including more concessional terms of lending, debt restructuring, debt cancellation, debt-to-health swaps, and more (115, 116). The latter works to ensure that health sectors do indeed benefit from funds freed up from debt relief.

Our main findings showed no relationship between IMF programme participation or conditionalities and GHE-S or OOP. The latter absent relationships with OOP have not been explored before this study. Only in an alternative model specification did we find evidence of IMF programme participation, not the number of IMF conditionalities, affecting GHE-S per CHE negatively, and not GHE-S per GDP. Overall, these findings mostly support the interpretation that GHE-S is not impacted by IMF programmes, and that OOP is not promoted either. For the IMF, the obvious policy implication is that they should focus on developing policy at the central fiscal level that supports GHE-S increases across a Global South where government health spending levels are wholly inadequate for meeting population needs, particularly in LICs (13, 117-119). The predominant IMF policy of using social spending floors seemingly does not achieve increases in GHE-S on aggregate, and other policy measures such as dedicated health spending floors, or objectives for GHE-S increases in adjustment programs for low-spending countries should be considered, as allowed within the IMF mandate.

## 6.7.2. Research recommendations

While our study adds knowledge on public external debt impacts on domestic health financing at the global level, future studies should focus on exploring how domestic health financing is impacted by debt obligations in individual countries. Negative effects have already been identified in some countries in SSA (120-122). As debt burdens grow and warnings of health financing impacts are released, a key role of researchers would be to elicit single country experiences, both showing any negative implications, but also identifying policies that have been successful at insulating health sectors from debt servicing obligations.

Similarly, while our study has explored impacts of IMF programmes and conditionalities on domestic health financing sources at the aggregate, global level, future studies could unpack the complexity of IMF policy at the country level from a health financing perspective, using qualitative or mixed methods. Such studies might help inform which particular policies support increasing progressivity of health financing sources, and which do not, which would be helpful for informing future IMF policy making that aligns with health sectoral objectives for UHC.

## References

1. IMF. What is the IMF? 2024 [cited 2024 May 20]. Available from:

https://www.imf.org/en/About/Factsheets/IMF-at-a-Glance. 2. OECD. Creditor Reporting System (CRS). 2024 [cited 2023 February 16]. Available from: https://stats.oecd.org/Index.aspx?DataSetCode=CRS1.

3. IMF. Public Finances in Modern History. 2024 [cited 2024 June 10]. Available from: https://www.imf.org/external/datamapper/datasets/FPP.

4. World Bank. International Debt Statistics. 2024 [cited 2024 April 24]. Available from: https://databank.worldbank.org/source/international-debt-statistics#.

United Nations Global Crisis Response Group. A world of debt: A growing burden to global prosperity.
 2023 [cited 2024 March 25]. Available from: <u>https://www.un.org/regionalcommissionsnyoffice/news/launch-un-report-%E2%80%9C-world-debt%E2%80%9D.</u>

6. World Bank. Health Financing in a Time of Global Shocks: Strong Advance, Early Retreat. 2023 [cited 2025 March 25]. Available from: <u>https://openknowledge.worldbank.org/handle/10986/39864</u>.

7. The L. Economic storms threaten to cast health adrift. Lancet. 2024;403(10424):325.

8. Fan V, Gupta S. What's Rising Debt Got to Do With Health Spending? 2023 [cited 2023 September 27]. Available from: <u>https://www.cgdev.org/blog/whats-rising-debt-got-do-health-spending</u>.

9. OXFAM. No country should be forced to choose between paying back debts or providing health care. 2024 [cited 2024 March 26]. Available from: <u>https://www.oxfam.org/en/no-country-should-be-forced-choose-between-paying-back-debts-or-providing-health-care</u>.

10. Jubilee Debt Campaign. Sixty-four countries spend more on debt payments than health. 2020 [cited 2022 March 14]. Available from: <u>https://jubileedebt.org.uk/press-release/sixty-four-countries-spend-more-on-debt-payments-than-health</u>.

11. Amnesty International. Debt and the Right to Health: Closing Observations from Our Series. 2024 [cited 2024 March 26]. Available from: <u>https://www.amnesty.org/en/latest/campaigns/2023/10/debt-and-the-right-to-health-closing-observations-from-our-series/</u>.

12. Development Finance International. The worst ever global debt crisis. 2023 [cited 2024 March 26]. Available from:

https://assets.nationbuilder.com/eurodad/pages/3195/attachments/original/1696947958/Debt Service Watch Bri efing\_Final\_Word\_EN\_0910.pdf?1696947958.

 WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from: <u>http://apps.who.int/nha/database</u>.
 Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A,

14. Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A, Stubbs T, editors: Oxford University Press; 2023. 0 p. ISBN: 9780190637736

15. Mishra P, Newhouse D. Does health aid matter? J Health Econ. 2009;28(4):855-72.

16. Farag M, Nandakumar AK, Wallack SS, Gaumer G, Hodgkin D. Does funding from donors displace government spending for health in developing countries? Health Aff (Millwood). 2009;28(4):1045-55.

17. Stuckler D, Basu S, McKee M. International Monetary Fund and aid displacement. Int J Health Serv. 2011;41(1):67-76.

18. Xu K, Saksena P, Holly A. The determinants of health expenditure: A country-level panel data analysis. Results for Development Institute WHO; 2011.

19. Gebrehanna S, Upadhyay MP. Fungibility of Foreign Aid in Selected African Countries. Indian Development Review. 2012;10(1):63-80.

20. Van de Sijpe N. The fungibility of health aid reconsidered. 2013.

21. Van de Sijpe N. Is Foreign Aid Fungible? Evidence from the Education and Health Sectors. World Bank Economic Review. 2013;27(2):320-56.

22. Fernandes Antunes A, Xu K, James CD, Saksena P, Van de Maele N, Carrin G, et al. General budget support: has it benefited the health sector? Health economics. 2013;22(12):1440-51.

23. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

24. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

25. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

26. Barkat K, Mrabet Z, Alsamara M. Does Official Development Assistance for Health from Developed Countries Displace Government Health Expenditure in Sub-Saharan Countries? Economics Bulletin. 2016;36(3):1616-35.

27. Younsi M, Chakroun M, Nafla A. Robust analysis of the determinants of healthcare expenditure growth: evidence from panel data for low-, middle- and high-income countries. International Journal of Health Planning and Management. 2016;31(4):580-601.

28. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

29. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJ. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

30. Fosu AK. The External Debt-Servicing Constraint and Public Expenditure Composition. Helsinki, Finland: UNU-WIDER; 2007.

31. Fosu AK. The External Debt-Servicing Constraint and Public-Expenditure Composition in Sub-Saharan Africa. African Development Review. 2010;22(3):378-93.

32. Frimpong AO, Amporfu E, Arthur E. Effects of public and external health spending on out-of-pocket payments for healthcare in sub-Saharan Africa. Health Policy Plan. 2022;37(9):1129-37.

33. Gabani J, Suhrcke M, Neelsen S, Eozenou PH, Smitz MF. Does health aid matter to financial risk protection? A regression analysis across 159 household surveys, 2000-2016. Social science & medicine (1982). 2024;356:117148.

34. Ali MEM, Ebaidalla M, Rizk R. Foreign aid and out-of-pocket health expenditure in Sub-Saharan Africa: does institutional development matter. International Journal of Sustainable Economy. 2020;12(3).

35. Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 2007;8(4):299-310.

36. Behera DK, Dash U. Impact of macro-fiscal determinants on health financing: empirical evidence from low-and middle-income countries. Glob Health Res Policy. 2019;4:21.

37. Shabbir S, Yasin HM. Implications of Public External Debt for Social Spending: A Case Study of Selected Asian Developing Countries. Lahore Journal of Economics. 2015;20(1):71-103.

38. Said R, Morai AS. Relationship between Public Debt Burden and Health Expenditure in Sub-Saharan African Countries: The Role of Institutional Quality. Journal of Business and Social Review in Emerging Economies. 2020;6(2):493-502.

39. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

40. Ooms G, Schrecker T. Expenditure ceilings, multilateral financial institutions, and the health of poor populations. The Lancet. 2005;365(9473):1821-3.

41. Center for Global Development. Does the IMF Constrain Health Spending in Poor Countries? . 2007 [cited 2024 March 19]. Available from: <u>https://www.cgdev.org/sites/default/files/14103\_file\_IMF\_report.pdf</u>.

42. Brunswijck G. Unhealthy Conditions. Eurodad; 2018.

43. Stuckler D, Basu S. The International Monetary Fund's effects on global health: before and after the 2008 financial crisis. Int J Health Serv. 2009;39(4):771-81.

44. Stuckler D, Basu S. The Body Economic: Why Austerity Kills: Basic Books; 2013. ISBN: 978-0465063987

45. Abocejo F. The Impact of International Monetary Fund (IMF) Structural Adjustment Policies (SAP) on the Philippines. Recoletos Multidisciplinary Research Journal. 2014;2(1):19-24.

46. Isiani MC, Anthonia Obi-Ani N, Obi-Ani P, Chidume CG, Okoye-Ugwu S. Interrogating the International Monetary Fund (IMF) Policies in Nigeria, 1986–2018. Cogent Arts & Humanities. 2021;8(1).

47. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

48. Foley EE. Your pocket is what cures you: The Politics of health in Senegal: Rutgers University Press; 2010. ISBN: 978-0-8135-4667-4

49. Meurs M, Seidelmann L, Koutsoumpa M. How healthy is a 'healthy economy'? Incompatibility between current pathways towards SDG3 and SDG8. Global Health. 2019;15(1):83.

50. Birn AE, Nervi L, Siqueira E. Neoliberalism Redux: The Global Health Policy Agenda and the Politics of Cooptation in Latin America and Beyond. Development and Change. 2016;47(4):734-59.

51. Nooruddin I, Simmons JW. The Politics of Hard Choices: IMF Programs and Government Spending. International Organization. 2006;60(04).

52. Stubbs T, Kentikelenis A. Targeted social safeguards in the age of universal social protection: the IMF and health systems of low-income countries. Critical Public Health. 2017;28(2):132-9.

53. Kentikelenis A, Stubbs T. Social protection and the International Monetary Fund: promise versus performance. Global Health. 2024;20(1):41.

54. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

55. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

56. Gupta S. Can a causal link be drawn? A commentary on "the impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations". Social science & medicine (1982). 2017;181:199-201.

57. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs. International Monetary Fund; 2004.

58. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

59. IMF Independent Evaluation Office. Fiscal Adjustment in IMF-Supported Programs. 2003 [cited 2024 March 19]. Available from: https://www.imf.org/external/np/ieo/2003/fis/pdf/main.pdf.

60. Clements B, Gupta S, Nozaki M. What Happens to Social Spending in IMF-Supported Programs? . 2011. Available from: <u>http://www.imf.org/external/pubs/ft/sdn/2011/sdn1115.pdf</u>.

61. IMF. Social safeguards and program design in PRGT and PSI-supported programs. 2017 [cited 2024 March 20]. Available from: <u>https://www.imf.org/en/Publications/Policy-Papers/Issues/2017/06/01/pp042117social-safeguards-and-program-design-in-prgt-and-psi</u>.

62. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

63. McIntyre D, Mooney G. The Economics of Health Equity. 1st ed: Cambridge University Press; 2007. ISBN: 978-0521705066

64. Wagstaff A, Van Doorslaer E, Paci P. Equity in the Finance and Delivery of Health Care: Some Tentative Cross-Country Comparisons. Oxford Review of Economic Policy. 1989;5(1):89-112.

65. Mtei G, Makawia S, Ally M, Kuwawenaruwa A, Meheus F, Borghi J. Who pays and who benefits from health care? An assessment of equity in health care financing and benefit distribution in Tanzania. Health Policy Plan. 2012;27 Suppl 1:i23-34.

66. Martinez-Álvarez M, Federspiel F, Singh NS, Schaferhoff M, Lewis Sabin M, Onoka C, et al. Equity of resource flows for reproductive, maternal, newborn, and child health: are those most in need being left behind? BMJ. 2020;368:m305.

67. World Health Organization. The world health report: health systems financing: the path to universal coverage. Geneva: World Health Organization; 2010.

68. Xu K, Evans DB, Carrin G, Aguilar-Rivera AM, Musgrove P, Evans T. Protecting households from catastrophic health spending. Health Aff (Millwood). 2007;26(4):972-83.

69. Makinen M, Waters H, Rauch M, Almagambetova N, Bitran R, Gilson L, et al. Inequalities in health care use and expenditures: Empirical data from eight developing countries and countries in transition. Bulletin of the World Health Organization. 2000;78(1):55-65.

70. Akazili J, McIntyre D, Kanmiki EW, Gyapong J, Oduro A, Sankoh O, et al. Assessing the catastrophic effects of out-of-pocket healthcare payments prior to the uptake of a nationwide health insurance scheme in Ghana. Glob Health Action. 2017;10(1):1289735.

71. WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available from: <u>https://www.who.int/publications/i/item/9789241564021</u>.

72. Chuma J, Okungu V. Viewing the Kenyan health system through an equity lens: implications for universal coverage. Int J Equity Health. 2011;10:22.

73. Moreno-Serra R, Smith PC. Broader health coverage is good for the nation's health: evidence from country level panel data. J R Stat Soc Ser A Stat Soc. 2015;178(1):101-24.

74. Moreno-Serra R, Smith P. The Effects of Health Coverage on Population Outcomes: A Country-Level Panel Data Analysis. Results for Development Institute. 2011.

75. Qin VM, Hone T, Millett C, Moreno-Serra R, McPake B, Atun R, et al. The impact of user charges on health outcomes in low-income and middle-income countries: a systematic review. BMJ Glob Health. 2018;3(Suppl 3):e001087.

76. IMF Monitor. IMF Conditionality Dataset, 1980-2019. 2023 [cited 2023 November 2]. Available from: https://imfmonitor.org/conditionality/.

77. World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: https://data.worldbank.org/.

78. Microsoft® 2024. Microsoft® Excel for Mac, version 16.81.

79. StataCorp 2015. Stata Statistical Software: Release 14.2. College Station, TX: StataCorp LP.

80. Arregoces L, Daly F, Pitt C, Hsu J, Martinez-Alvarez M, Greco G, et al. Countdown to 2015: changes in official development assistance to reproductive, maternal, newborn, and child health, and assessment of progress between 2003 and 2012. The Lancet Global Health. 2015;3(7):e410-e21.

81. UN Inter-agency Group for Child Mortality Estimation. Child Mortality, Stillbirth, and Causes of Death Estimates. 2024 [cited February 9 2024]. Available from: <u>https://childmortality.org/</u>.

82. Kaufmann D, Kraay A. Worldwide Governance Indicators. 2023 [cited 2024 March 21]. Available from: https://www.govindicators.org/.

83. Davies S, Pettersson T, Öberg M. Organized violence 1989–2022, and the return of conflict between states. Journal of Peace Research. 2023;60(4):691-708.

84. Hensel PR. ICOW Colonial History Data Set, version 1.1. 2023 [cited 2023 November 17]. Available from: <u>http://www.paulhensel.org/icowcol.html</u>.

85. Feyzioglu T, Swaroop V, Zhu M. A Panel Data Analysis of the Fungibility of Foreign Aid. The World Bank Economic Review. 1998;12(1):29-58.

86. Hansen LP. Large Sample Properties of Generalized Method of Moments Estimators. Econometrica. 1982;50(4).

87. Hall AR. Generalized Method of Moments: Oxford University Press; 2005. 408 p. ISBN: 0198775202
88. Roodman D. How to do xtabond2: An introduction to difference and system GMM in Stata. Stata
Journal. 2009;9(1):86-136.

89. Arellano M, Bond S. Some Tests of Specification for Panel Data: Monte Carlo Evidence and an Application to Employment Equations. The Review of Economic Studies. 1991;58(2):277-97.

90. Arellano M, Bover O. Another look at the instrumental variable estimation of error-components models. Journal of Econometrics. 1995;68(1):29-51.

91. Blundell R, Bond S. Initial conditions and moment restrictions in dynamic panel data models. Journal of Econometrics. 1998;87(1):115-43.

92. Hausman JA. Specification Tests in Econometrics. Econometrica. 1978;46(6).

93. Kiviet JF. Microeconometric dynamic panel data methods: Model specification and selection issues. Econometrics and Statistics. 2020;13:16-45.

94. Kripfganz S. Generalized method of moments estimation of linear dynamic panel-data models. London Stata Conference2019.

95. Windmeijer F. A finite sample correction for the variance of linear efficient two-step GMM estimators. Journal of Econometrics. 2005;126(1):25-51.

96. Roodman D. A Note on the Theme of Too Many Instruments. Oxford Bulletin of Economics and Statistics. 2009;71(1):135-58.

97. Adeleye BN. (Stata 13): How to Estimate Two-Step System GMM. 2019 [cited 2023 November 9]. Available from: <u>https://www.youtube.com/watch?v=JkBFrjYXEyM</u>.

98. Roodman D. Doubts about the evidence that foreign aid for health is displaced into non-health uses. The Lancet. 2012;380(9846):972-3.

99. Forster T, Kentikelenis AE, Stubbs TH, King LP. Globalization and health equity: The impact of structural adjustment programs on developing countries. Social science & medicine (1982). 2020;267:112496.
100. Clements B, Gupta S, Nozaki M. What happens to social spending in IMF-supported programmes? Applied Economics. 2013;45(28):4022-33.

101. Heckman JJ. Sample Selection Bias as a Specification Error. Econometrica. 1979;47(1).

102. Liang L-L, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

103. WHO, UNICEF, UNFPA, World Bank Group, UNDESA/Population Division. Trends in maternal mortality 2000 to 2020: estimates by WHO, UNICEF, UNFPA, World Bank Group and UNDESA/Population Division. 2023 [cited 2024 February 29]. Available from: https://www.who.int/publications/i/item/9789240068759.

104. Shah A. ASDOC: Stata module to create high-quality tables in MS Word from Stata output. Statistical Software Components S458466, Boston College Department of Economics. 2018.

105. Fernandes Antunes A, et al. General Budget Support: Has It Benefited the Health Sector? Health Economics. 2013;22(12):1440-51.

106. Imai K, Kim IS. On the Use of Two-Way Fixed Effects Regression Models for Causal Inference with Panel Data. Political Analysis. 2020;29(3):405-15.

107. Statalist. xtabond2: System GMM robust estimation - Do I use Hansen or Sargan test results?? 2024 [cited 2024 March 18]. Available from: <u>https://www.statalist.org/forums/forum/general-stata-</u>

discussion/general/1432620-xtabond2-system-gmm-robust-estimation-do-i-use-hansen-or-sargan-test-results. 108. Boachie MK, Agyemang J, Immurana M. Health sector funding in Ghana: The effect of IMF conditionalities. Dialogues Health. 2022;1:100045. 109. Statalist. System GMM: does it includes fixed effects? 2019 [cited 2023 March 23]. Available from: <u>https://www.statalist.org/forums/forum/general-stata-discussion/general/1520513-system-gmm-does-it-includes-fixed-effects</u>.

110. Ullah S, Akhtar P, Zaefarian G. Dealing with endogeneity bias: The generalized method of moments (GMM) for panel data. Industrial Marketing Management. 2018;71:69-78.

111. Jerven M. Poor Numbers: How we are misled by African development statistics and what to do about it. 1 ed: Cornell University Press; 2013. 187 p. ISBN: 978-0-8014-7860-4

112. Biørn E. The measurement error problem in dynamic panel data analysis: Modeling and GMM estimation. University of Oslo, Department of Economics. 2012;Memorandum 02/12.

113. Ritchie E. The DAC debates: why aid measurement matters for development. 2023 [cited 2024 March 25]. Available from: <u>https://devinit.org/blog/dac-debate-why-aid-measurement-matters/</u>.

114. OECD. The Paris Declaration on Aid Effectiveness and the Accra Agenda for Action. 2005, 2008 [cited 2024 July 29]. Available from: <u>https://web-archive.oecd.org/temp/2021-08-02/73869-</u>

parisdeclarationandaccraagendaforaction.htm.

115. United Nations. Our Common Agenda - Policy Brief 6: Reforms to the International Financial Architecture. 2023 [cited 2024 May 6]. Available from: <u>https://www.un.org/sites/un2.un.org/files/our-common-agenda-policy-brief-international-finance-architecture-en.pdf</u>.

116. Wuennenberg L. Debt-To-Health Swaps: Financing health system resilience beyond the COVID-19 pandemic. 2022 [cited 2022 March 14]. Available from: <u>https://www.iisd.org/sustainable-recovery/debt-to-health-swaps-financing-health-system-resilience-beyond-the-covid-19-pandemic/</u>.

117. McIntyre D, Meheus F, Rottingen J-A. What Level of Domestic Government Health Expenditure Should We Aspire to for Universal Health Coverage? Health Economics, Policy and Law. 2017;12(2):125-37.

118. Dieleman JL, Sadat N, Chang AY, Fullman N, Abbafati C, Acharya P, et al. Trends in future health financing and coverage: future health spending and universal health coverage in 188 countries, 2016-40. The Lancet. 2018;391(10132):1783-98.

119. Dieleman JL, Campbell M, Chapin A, Eldrenkamp E, Fan VY, Haakenstad A, et al. Future and potential spending on health 2015-40: Development assistance for health, and government, prepaid private, and out-of-pocket health spending in 184 countries. The Lancet. 2017;389(10083):2005-30.

120. Kimalu PK. Debt relief and health care in Kenya. World Institute for Development Economics (UNU-WIDER). 2002;WIDER Discussion Papers(No. 2002/65, ISBN 9291902551, UNU-WIDER, Helsinki).

121. Yiega V. The Impact of Foreign Debt on Economic and Social Rights: A Case Study on the Right to Health in Kenya, Nigeria and Zambia. ADHR (CFS). 2022; (Research Paper 06/06).

122. OXFAM. Policy report: Impact of debt servicing on social spending and wellbeing of low- income households in Kenya: a case study of Nairobi city and Mombasa counties. 2021 [cited 2024 March 26]. Available from: <a href="https://kenya.oxfam.org/latest/publications/impact-debt-servicing-social-spending-and-wellbeing-low-income-households-kenya">https://kenya.oxfam.org/latest/publications/impact-debt-servicing-social-spending-and-wellbeing-low-income-households-kenya</a>.



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

## **RESEARCH PAPER COVER SHEET**

Please note that a cover sheet must be completed for each research paper included within a thesis.

Student ID Number	lsh1601722	Title	Dr.
First Name(s)	Frederik		
Surname/Family Name	Federspiel		
Thesis Title	External Development Partner and International Monetary Fund Influence on Domestic Health Financing Sources in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal		
Primary Supervisor	Josephine Borghi		

## SECTION A - Student Details

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

#### SECTION B – Paper already published

Where was the work published?			
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Choose an item.	Was the work subject to academic peer review?	Choose an item.

\*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

#### SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	Health Policy and Planning
Please list the paper's authors in the intended authorship order:	Frederik Federspiel, Josephine Borghi, Elhadji Mamadou Mbaye, Henning Tarp Jensen, Melisa Martinez Alvarez

#### Improving health worldwide

www.lshtm.ac.uk

Stage of publication	Undergoing revision

## SECTION D - Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I designed this study as Principal Investigator (PI) in collaboration with my supervisors and co-PI in Senegal. I produced the study protocol and research ethics applications with input from supervisors and co-PI. I collected, analysed and interpreted the data, and wrote the first draft paper, with the input necessary along the way from my supervisors. My supervisors, local co-PI and I then critically revised the article, with me being responsible for performing the revisions necessary.

## SECTION E

Student Signature		
Date	20/9-2024	

Supervisor Signature		
Date	27/09/2024	

## 7. Chapter 7: External Development Partner Influence on Domestic Health Financing Sources in Senegal: A Mixed Methods Case Study

## 7.1. Chapter overview

Chapter 6 found some evidence at the global level for Official Development Assistance<sup>+</sup> (ODA<sup>+</sup>) for health displacing both Government Health Expenditure as a Source (GHE-S) and Out Of Pocket payments (OOP), as well as a slight shift in the burden of payment from the government toward the user resulting from debt servicing obligations. The study was however unable to directly investigate underlying mechanisms of effect and non-financial mechanisms that are not easily quantifiable and not measured and available in cross-country panel datasets. The following chapters seek to build on the understanding developed in Chapter 6 using a mixed-methods approach in a case study in Senegal.

Chapter 7 presents the first part of this case study. It uses interviews, document review and descriptive quantitative analysis to address the second objective of this thesis: To explore the mechanisms of External Development Partner (EDP) influence on the mix of domestic health financing sources in Senegal. It first introduces the relevant background literature and concepts, its analytical framework, study setting and methods, before summarising its findings structured by the four identified influence mechanisms: setting aims and standards, lobbying/negotiation, providing policy/technical advice, and providing external financing. It then discusses its findings in relation to the literature, to then make recommendations encouraging the use of the developed framework to help identify dynamics and areas of policy incoherence in other contexts that do not optimally support progress toward Universal Health Coverage (UHC).

Chapter 8 further unpacks the influence mechanisms identified in this chapter by specifically looking at the health financing implications from the external financing modality of lending and exploring the specific influence of the International Monetary Fund (IMF) and the World Bank (WB) on domestic health financing sources in Senegal.

Chapter 7 has been submitted to Health Policy and Planning and re-submitted after implementing reviewer comments. Minor language and formatting changes from the re-submitted version have been made to ensure consistency and clarity across chapters in the thesis.

## 7.2. Abstract

## Background

Sustainable and equitably contributed domestic health financing is essential for improving health and making progress towards Universal Health Coverage in Low- and Middle-Income Countries. In this study, we explore the pathways through which development partners influence the mix of domestic health financing sources in Senegal.

## Methods

We performed a qualitative case study comprised of 32 key stakeholder interviews and a purposive document review, supplemented by descriptive statistical analysis of WHO and OECD data on health financing sources in Senegal. We developed a novel framework to analyse the different mechanisms and directions of development partner influence on domestic health financing contributions.

## Results

We identified development partner influence via four mechanisms: setting aims and standards, lobbying/negotiation, providing policy/technical advice, and providing external financing. Overall, development partners worked to increase tax-based government contributions and expand community-based health insurance (CBHI), which is seemingly equity enhancing. Fungibility and intrinsic equity issues related to CBHI may however limit equity gains.

## Conclusions

We encourage stakeholders in the health financing sphere to use our framework and analysis to unpack how development partners affect domestic health financing in other settings. This could help identify dynamics that do not optimally enhance equity and support progress towards UHC, to help achieve more coherent policy-making across all domains of development partner activities in support of UHC. Future research should investigate the role of international creditors, lending and loan conditionalities on domestic health financing in recipient countries, including equity implications.

## 7.3. Introduction

Equity in health financing contributions has long been recognized as essential to improving health indicators and making progress towards Universal Health Coverage (UHC) in Low- and Middle-Income Countries (LMICs), protecting patients and their families from financial risk (1-4). We define equity of health financing contributions as funds being contributed in proportion to ability to pay; being prepaid so funds can be made available to those who need it irrespective of their ability to pay at the time of seeking a health service; and being pooled across many individuals to allow for financial risk sharing (3, 5-12).

Many recipient countries of development assistance are highly donor-dependent for their health sector financing (13), and development partners exert great influence over national health policy (e.g. (14-16)) including health financing policy (e.g. (15, 17-25)) in various contexts including Senegal. In a recent commentary by (21), the authors explain how Community-Based Health Insurance (CBHI) in Senegal was pushed by external development partners, stifling progress toward UHC by more than a decade due to poor coordination and intrinsic equity issues with CBHI.

In the quantitative literature, some econometric studies have examined the relationship between external and domestic financing in the form of fungibility, i.e. whether the health budget increases by less than the amount injected as development assistance for the health sector due to an associated decrease in Government Health Expenditure as a Source (GHE-S), i.e. from domestic revenue (e.g. (26-29)), with most authors finding a fungibility effect. Other studies have examined the relationship between external health financing and Out-Of-Pocket payments (OOP), finding no effect or a crowding-in effect (30-32). However, these studies do not go beyond relationships between financing flows to explore the different potential pathways of development partner influence.

Understanding through which mechanisms development partners influence domestic health financing in aid recipient countries is important: This can help to inform development partner efforts, ensuring they do indeed work toward sustainable, equitably contributed health financing in the countries they support.

Building on the above work and using the case of Senegal, our study aims to explore through which pathways development partners may influence the composition of domestic funding sources for health, and whether this influence has been equity enhancing or not in the case of Senegal. We use qualitative methods (interviews and document review), supplemented by descriptive statistics to examine the presence, pathways and nature of external development partner influence on domestic health financing contributions, since 2000. We first describe the main financing sources and mechanisms in Senegal. We then introduce and apply a novel analytical framework to identify and examine the different ways development partners may exert influence on domestic health financing contributions in Senegal.

## 7.4. Methods

## 7.4.1. Equity definition

As explained above, we define equity of health financing contributions as funds contributed in proportion to ability to pay, that are prepaid and pooled. This implies cross-subsidies of health funds from rich to poor and from the healthy to the sick (33, 34). With this definition, OOP payments are considered least equitable, as they do not take into account a person's ability to pay, they are not prepaid and there is no pooling of funds (3). Health insurance is more equitable than OOP, but to a varying degree depending on the level of contributions made relative to ability to pay of insurance pool members, the level of cross-subsidy from rich to poor, and the size of pools.

Government health financing is considered most equitable, as taxes are overall progressive, although VAT and some excise taxes can be regressive. The individual benefiting from a fully tax-funded health service experiences no personal financial cost at the point of care, and funding pools can be as large as covering a whole nation's population, resulting in maximal risk sharing and cross subsidy across the income- and wealth spectrums of a nation. We thus consider donor influence on the composition of funding sources toward more government financing and less OOP, and support for pooled financing mechanisms over no pooling as equity enhancing.

## 7.4.2. Analytical framework

The framework used in this study for analysing development partner influence on domestic health financing contributions is illustrated in Figure 7.1.



Figure 7.1: Framework for analysing external development partner influence on domestic health financing contributions. Development partners can influence domestic health financing contributions via setting aims and standards, advising, lobbying and negotiating, and providing financing. Development partner financing may have a synergistic effect or a displacing effect on government health financing. These different mechanisms of influence may result in changes in the balance between different health financing sources – government health financing from taxes, health insurance premiums, and out-or-pocket payments – which in turn has implications for the degree of equity of domestic health financing contributions. This figure is an original visualisation based on previous work by (35-37), and on the interview, document and quantitative findings of our study in Senegal. The arrows reflect our findings in Senegal

We developed the framework iteratively, both inductively and deductively. We first conducted a literature review to identify existing policy influence analysis frameworks from the broader policy analysis literature, including health policy analysis frameworks. Existing frameworks have focused on e.g. the nature of problems, politics and policy (38); context, policy content, policy process, and actors (39); policymaking processes (20, 40); agents of policy change (41); or on the ideas, institutions and interests underlying policy influence and reform (42-44). Others have emphasized political context, existing evidence and links/network factors, (35, 45, 46); mechanisms of policy influence more broadly (35, 36, 47); and the role of international science and finance in determining LMIC policy (37). A framework developed by Sparkes et al. for analysing the political economy of health financing reform has focused on the different dimensions of politics e.g. bureaucratic politics, leadership politics or external politics (15). Fox and Reich (2015) have combined Hall's 3-i's framework (42) and Kingdon's stream model (38)

to analyse how politics affects UHC reform in LMICs at different stages of the policy cycle (48, 49).

Three of the identified frameworks were found to emphasize policy influence mechanisms (35-37). These were adapted and adjusted into one that reflected the dimensions of EDP influence present in our data. We then applied the framework to our data, making any final adjustments needed based on the findings in the data.

We found that development partners can exert influence on domestic health financing contributions via setting aims and standards, providing policy/technical advice, lobbying and negotiating, and by providing finance. Development partner financing may elicit or require co-financing by the recipient government or displace government funds (fungibility/subsidy). These different modes of influence may result in changes in the balance between different health financing sources which in turn affect the degree of equity of domestic health financing contributions (3, 5, 10, 11).

The influence mechanisms via giving policy advice or technical advice, and by lobbying and negotiating, are derived from (35) and (36). Policy and technical advice can be viewed as evidence-based knowledge production and dissemination, often enacted through publishing official reports and briefings, allowing for the "*diffusion of sector-specific know-how*" offering solutions to policy problems (35, 36, 50). Lobbying/negotiating can be thought of as the art of persuasion, often involving high-level networking through people-to-people interactions in both formal as well as informal settings (35, 51).

The mechanism of providing external health financing is derived from (37). Based on our study results, we have added the policy influence mechanism: "setting aims and standards", meaning influence through the establishment of aspirational concepts such as the Sustainable Development Goals or (evidence-based) best practice recommendations; and we have further separated external financing into that which has an increasing or a displacing effect on domestic government health financing.

## 7.4.3. Study setting

Senegal is a Francophone democratic republic in West Africa with a population of 18 million (2023) (52). Classified as a lower middle-income country, its GDP per capita was \$1,599 in 2022 (53).
#### 7.4.3.1. Overview of health financing sources in Senegal

Over the past two decades, domestic health financing in Senegal has been characterized by a strong reliance on user contributions and a smaller and decreasing reliance on government contributions, with the exception of the year 2020 when government health financing saw a transient increase due to the Covid-19 outbreak (Figures Figure 7.2 and Figure 7.3). Following a period of steady rise from 2000-2006, GHE-S was \$290 million in 2006 and \$280 million in 2019 (constant 2022 US\$) (Figure 7.2) (13, 53). This corresponds to a per capita decrease from \$26 in 2006 to \$18 in 2019 (constant 2022 US\$) (13, 53), and a decrease per GDP from 1.8% in 2006 to 1.1% in 2019 (13) (Figure 7.2). Before the Covid-19 pandemic, GHE-S also received decreasing budget priority, declining as a share of General Government Expenditure (GGE) from 8% in 2006 to 4% in 2019, getting further from the Abuja target of 15% (13) (Figure 7.2).



Figure 7.2: Government health expenditure from domestic revenue in absolute terms (panel A); as a proportion of general government expenditure (panel B); as a proportion of Gross Domestic Product (GDP) (panel C); and per capita (panel D) (13, 53). (Constant US\$ data deflated using WB US\$ GDP deflator data (53)).

Following a period of relative decline from 2004-2006, OOP contributions correspondingly made up a growing proportion of all health financing in Senegal, from their lowest point in 2006



at 37% to reach 49% in 2019 (Figure 7.3) (13). This corresponds to a per capita increase from \$21 in 2006 to \$36 in 2019 (constant 2022 US\$) (13, 53).

Figure 7.3: Sources of health financing in Senegal as percent of current health expenditure, 2000-2021 (13). Dom. Gov: Domestic government revenue; Ext: External financing; Soc: Social insurance contributions; Vol. Pre: Voluntary prepayment; OOP: Out-Of-Pocket Payments. Visualisation method as per (13). Note: There may be some additional government health financing captured within Soc. and Vol. Pre. as subsidies to these schemes.

Partly mitigating the previous lack of growth in domestic government health financing, realterm external health financing has expanded substantially from \$14 million in 2000 to \$236 million in 2021 (constant 2022 US\$) (13, 53). This corresponds to an increase from 4% to 18% of Current Health Expenditure (CHE) over the same time period (Figure 7.3), or from \$1 to \$14 per capita (constant 2022 US\$) (13, 53).

As seen in Figure 7.4, total Official Development Assistance (ODA) for health disbursed between 2002-2022 has been provided predominantly as grants (81%) and as project-type interventions (70%) (54).





*Figure 7.4: Total official development assistance for health in Senegal disbursed between 2002-2022, separated by flow types and types of aid (OECD, 2024).* 

## 7.4.3.2. Health financing schemes

Various health financing schemes exist in Senegal, as summarised in Table 7.1. The current composition of health financing schemes in Senegal is strongly influenced by the 2014 "*Plan Sénégal Émergent*" ("Emerging Senegal") that charted a course for all sectors including health sectoral reform (55). Currently, it mainly consists of a set of exemption schemes for vulnerable groups, priority services and drugs called the *Gratuités*; compulsory health insurance schemes for formal sector workers and their families; and CBHI schemes called the *Mutuelles* for all Senegalese though mainly targeting informal sector workers and the rural poor (56-63).

Scheme type	Scheme name	Target groups	Funding source	Pooling
Exemptions	Gratuités	People > 60 years, children < 5 years, caesarean sections, dialysis, antiretroviral and tuberculosis drugs	Government contributions, donor contributions	National level
Compulsory health insurance	Assurance Maladie Obligatoire (Imputation budgetaire, Institutions de Prévoyance Maladie, and more)	Civil servants + families, formal sector employees + families, retired state and private sector employees + families, university students, occupational injury and illness coverage and more	Member contributions, employer/organisation contributions, private donations	Scheme members
Community based health insurance	Mutuelles de santé	All Senegalese, though mainly informal sector workers, rural poor	Member contributions, state contributions*, private donations	Members, community level*
Private health insurance	Various	Anyone, though mainly wealthier groups	Member contributions	Scheme members

Table 7.1: Overview of health financing schemes in Senegal (56-63). \*For most of those enrolled in Mutuelles, the state pays 50% of the nationally fixed annual premium of 7000 CFA per person (62, 63) (approximately US\$12), while certain very poor or disabled groups can obtain 100% subsidy (57, 59, 62). Mutuelles are currently undergoing consolidation from the community to the departmental level (21, 22, 63).

## 7.4.4. Study design, sampling, data collection and management

This study was a qualitative case study comprised of key stakeholder interviews and a purposive document review, supplemented by descriptive quantitative analysis of health financing in Senegal. Interviews were conducted in Senegal between October 2019 and January 2020. Documents and quantitative data were collected before, during and after this period, and analysed after an initial analysis of interviews.

## 7.4.5. Purposive document review

To further investigate EDP influence on domestic health financing sources, we searched government and development partner websites for webpage articles and official reports on health financing and health financing policy in Senegal. This was done by screening websites from 11 EDPs and the Senegalese government for available links and references that could potentially discuss health financing (e.g. "Our work" => "global health") (Figure 7.5).



Figure 7.5: Flowchart for purposive document review. EDP: External Development Partner. GFATM: Global Fund to fight Aids, Tuberculosis and Malaria. BMGF: Bill and Melinda Gates Foundation. LSHTM: London School of Tropical Medicine & Hygiene. CINAHL: Cumulative Index to Nursing and Allied Health Literature.

Websites from the following organisations were screened: the Senegalese government (the National Statistics and Demography Agency, the Universal Health Coverage Agency, the Ministry of Health and Social Action, the Ministry of Finance and Budgeting, the Ministry of the Economy, Planning and Cooperation), the major bilateral and multilateral donors present in Senegal (USA, France, Canada, Japan, Belgium, Luxembourg, Spain, the Global Fund to Fight Aids, Tuberculosis and Malaria (GFATM), Bill and Melinda Gates Foundation (BMGF) and Gavi) (54) (Figure 7.6), as well as the World Bank.



*Figure 7.6: Main official development partners in the health sector in Senegal by disbursements made in 2022 (54). "Other" sums disbursements from 28 multilateral organisations, bilateral organisations and private foundations, each less than \$3 million.* 

We also searched Google Scholar, PubMed, Embase, Cochrane, CINAHL, Scopus, Web of Science and EconLit for relevant academic literature, using keywords including Senegal, health financing and equity. 157 full-text articles and reports were retrieved for full-text review (118 from organisation websites and 39 via academic databases). Articles/reports documenting development partner activities with direct implications for domestic health financing contributions or providing facts confirming/rebutting cited statements from interviews were included and integrated into the results section. 42 out of the 157 full-text documents reviewed were included in the results. No time period constraints were applied to the document review in order to also obtain a broader historical understanding of EDP activities and health financing policy in Senegal, however we focus our results on the period after 2000.

## 7.4.6. Semi-structured interviews

We conducted an initial mapping of key external and domestic stakeholders engaged in health financing in Senegal by searching websites and through discussion with contacts in academia and government. Representatives from the main international official donors, government ministries and agencies, Senegalese civil society organisation leaders and academics, and management and administrative staff at the regional, district and hospital level were included. Once the initial set of stakeholder institutions and persons were identified, snowball sampling was used to identify additional participants (Table 7.2). We also included participants from the region of Tambacounda, as this region is the largest geographical region in the country with

an estimated population of nearly 1 million people in 2023 (65) and has one of the highest poverty rates in the country at 62% in 2018/19 (66). Repetition of similar observations and positions became apparent towards the end of interview data collection, indicating that data saturation was reached (67).

We conducted 32 interviews, 25 at the national level in Dakar and seven at the district and regional level in Tambacounda (Table 7.2). Representatives from two multilaterals, two hospitals, and two academics were unavailable/did not respond.

Stakeholder group	Number of participants			
	Total	Dakar	Tambacounda	
Bilateral development partners	4	4	0	
Multilateral development partners	4	4	0	
Central government*	9	9	0	
Regional/District government	3	0	3	
Hospital management	4	1	3	
Civil Society Organisations	5	4	1	
Academics	3	3	0	
Total interviews	32	25	7	

Table 7.2: Interview participants. \*Central government participants came from four different ministries, however the specific ministries are intentionally not mentioned to protect the anonymity of participants.

We used an interview topic guide to elicit the participant's organisation's activities, views/positions and observations relevant to the composition and degree of equity of health financing in Senegal and development partner influence hereon. Informed consent was obtained from all participants. Interviews were recorded where consent for this was given (n = 27). In the five instances where interviews were not recorded, FF took written notes and the interviews were used to broadly further his understanding of the research topic in Senegal and to inform the interviewees' general views on this. Interviews lasted from about 30 minutes to just over an hour. 30 interviews were conducted face-to-face and two remotely. Interviews were conducted in the preferred language of the participant. In most instances this was French and in some English. Interview materials were provided in the corresponding language. A professional interpreter was used for the majority of French-language interviews until FF had reached adequate proficiency for conducting interviews in French independently. Interviews were transcribed by a professional transcriptionist. FF controlled the quality and validity of transcripts by comparing segments from all interviews were analysed and interpreted in their

original language. Written notes were taken from all interviews. All interviews were treated as anonymous. The participant information sheets, informed consent forms and an example interview topic guide can be found in Appendix 5.

## 7.4.7. Interview data analysis

Interviews were analysed using the Framework method (68). We used NVivo for interview coding (69). We developed our coding framework based on a combination of deduction of predetermined themes and induction of themes from the data. FF first coded a third of all interviews in an exploratory manner to establish themes (n=11), ensure conceptual clarity and avoid overlap or omission of themes present within the data (70). FF, JB and MMA then agreed on the final coding framework. An independent researcher then co-coded a transcript for validation, after which the final coding framework was applied to all transcripts. Summaries and central/illustrative quotations were entered into the framework matrix. The final dataset was then systematically reviewed for patterns and relevant opinions, factual statements and explanatory accounts.

## 7.4.8. Quantitative data

We used the WHO Global Health Expenditure Database (GHED) (13) to perform descriptive quantitative analysis of health financing sources (GHE-S, External Health Financing (EXT), OOP and Voluntary Health Insurance (VHI) (*Mutuelles*)). We looked at trends in the composition of health financing sources over time from 2000-2020 (all available data), to contextualise and triangulate information from our other sources.

## 7.5. Results

We identified partner influence on domestic health financing contributions via four mechanisms: setting aims and standards, lobbying/negotiation, providing policy/technical advice, and providing external financing (Figure 7.1). Our findings generally indicated an equity promoting role of development partners in regard to domestic contributions, however concerns were raised as to their actual effect as government health funding cuts had been observed.

## 7.5.1. Setting aims and standards

Commenting on the slow growth in government health spending seen in Senegal, seven out of eight development partner representatives interviewed stated that they wanted to see stronger increases in the government health budget, with several referring to the Abuja target of domestic government health expenditure making up at least 15% of GGE (71). This desire was echoed across all stakeholder groups, including Ministerial/government agency representatives.

"The country signs agreements, and in these agreements, it is asked to make a budget that is approved for health that must reach 15%. This is an external pressure, and the country is bound to make efforts to achieve this." (Ministry/government agency).

The Abuja declaration comes from African nations themselves (71), but donors used this as a normative standard towards which they wanted the Senegalese government to aspire. An internally derived aim thus became a partly externally promoted aim. However, over the period 2006-19, there was no discernible increase in real-term government health financing, while GHE-S/GGE decreased from about 8% to 4% (Figure 7.2). This indicates that this normative/aspirational influence pathway from both EDPs and African nations jointly has been ineffective in Senegal in this time period.

An academic also referred to the 1978 Alma Ata declaration (72) as another international standard used to promote UHC.

"We have now the universal health coverage, is it coming from Senegal? No Senegal has to implement it because we signed it. In 1978 when the world decided on primary health care, we signed it and we started implementing." (Academic)

Overall, there was mostly universal agreement between interviewee stakeholder groups, including donors, that Senegal should aim to reduce OOP. In terms of their overall policy stance, most donor representatives interviewed stated that they wanted the future development of Senegal's health sector to be characterized by higher government contributions and less reliance on OOP:

"[Donor] encourages the countries to work on means to reduce user fees at service delivery points. Those are barriers in accessing health services and we very much support implementation of measures that facilitate access to health services by all populations, especially the poor population." (Donor)

These positions follow the 2017 Senegalese national health financing strategy (73) and the 2019 National Plan for Health and Social Development (64), and are as such consistent with official government policy.

As elaborated below, an academic also highlighted the WB publication "Investing in Health" as influential on the Senegalese government in promoting primary health care financing, which indicates external aim/standard-, or norm-setting by the WB (74).

Development partner and government interviewees uniformly viewed Senegal as a nation with a high level of sovereignty and self-governance, setting its own targets and development partners getting behind those targets: "In an organized country (Senegal), where there is a benchmark that serves as a reference, a partner cannot come and invest just anywhere... We are the ones who send funding requests to partners... The funding that is requested is always within the framework of what we want in terms of priorities." (Ministry/government agency)

Some interviewees in academia and civil society however disagreed with this view, arguing that donor funding priorities dictated government health program priorities:

"Each partner comes with their priorities, and the state in order to have the financing accepts everyone's priorities... Usually the priority is dictated by the funder." (CSO)

For this study, we did not identify any partner documents externally setting binding standards or aims for domestic health financing in Senegal, consistent with views expressed by government and donor representatives (64, 73).

#### 7.5.2. Lobbying/negotiation and policy/technical advice

Statements of lobbying, negotiating or "pushing", as well as providing policy/technical advice for increased government health financing were given by some development partner representatives and academics. Policy/technical advice supporting UHC and CBHI was described in partner documents as well.

To help the ministry of health attain a higher budget and support its execution, a donor gave both technical and negotiation support as follows:

"What we are supporting is the planning process of the budget formulation. So, we are supporting the minister of health in the negotiation with the ministry of economy and finance ... for additional resources in the health sector... We are also trying to support the execution of the budget ... Training of some officials in the ministry of health about the procedures and the requirements of the budget execution." (Donor)

This suggests external support in internal negotiations to mobilize more government funds for health, thus having dimensions both of technical advice and negotiation. Referring to UHC, an academic described technical advice received from the WHO:

"Senegal cannot really isolate itself and say no I'm not listening to the world experts ... You decide on the basis of advice that the international donors are advising. The technical guidance should be all of us, should be behind WHO whose mandate is to orient, guide and support our countries." (Academic)

They also described the World Bank "pushing" the Senegalese government to view health spending as an investment:

"In 2004 (original publication 1993), the World Bank published a document, that inspired our government which is Investing in Health... For the first time, the World Bank found

that investing in health has a return... It helped... When they said investing in health, they started pushing the government to invest more money in primary healthcare which was good." (Academic)

This relates both to aim/standard-setting as described above, but this was then described as followed by a "push" (categorised as lobbying/negotiation) by the World Bank toward the Senegalese government once this new aim had been established.

Partner websites and reports listed several examples of working to expand domestic health financing contributions by strengthening CBHI as described below. We categorise these as policy/technical advice. USAID and UNICEF were helping the Senegalese government develop and implement their national health financing strategy to expand UHC and *Mutuelles* (75-77). World Bank support for the *Couverture Maladie Universelle* and *Mutuelles* included technical advice, e.g. "supporting new institutional arrangement to promote greater efficiency in internal processes of the UHI ["Couverture Maladie Universelle" or UHC] scheme" (78, 79). The GFF provided "technical support on developing a Theory of Change to further inform implementation of the Investment Case…", which includes consolidation of *Mutuelle* risk pools (80).

#### 7.5.3. Financing

Providing health financing was identified as a key way development partners sought to influence domestic health financing contributions. These findings generally illustrated development partners seeking to increase government health financing and expand and consolidate CBHI. This can be seen as equity enhancing by better aligning payments with ability to pay in the case of increased government health financing, and to some degree for *Mutuelles* given that 50-100% of premiums are paid for by the state. It also shows attempts to consolidate health insurance pooling at the departmental level, which increases financial risk sharing and cross-subsidisation, although with disagreement between partners along the way (21, 81).

USAID, AFD and the World Bank provided external support for the rollout of *Mutuelles* across the country during the past decade (79, 82-85). Using a mix of loans and grants, the World Bank together with multiple donors gave financial support to the Senegalese government for strengthening the *Couverture Maladie Universelle* programme including the *Mutuelles* (78, 79). This illustrates external financial support for strengthening domestic health financing schemes. Since 2014, there has however apparently been initial disagreement between partners about the need for consolidation of *Mutuelles*, with Enabel (Belgium) for and USAID and WB against (21, 81, 86). Informed by USAID-, WB- and Enabel-supported pilots of funding

pool consolidation, the Senegalese government has begun moving financial risk pooling from the community level to the departmental level (21, 22, 63). External financing has thereby indirectly led to a consolidation of funding pools, which is equity enhancing. In 2019, the Global Financing Facility (GFF) partnership, consisting of France (AFD), Gavi, GFATM, GFF, Japan (JICA), World Bank, USAID, UNICEF and other UN agencies, provided a \$140 million loan and \$10 million grant to support "...*the government's commitment to increase the share of its health budget from 4 percent* (of total government expenditure) to 10 percent by 2022" (80) (this number was 4% in 2019 and 6% in 2020 (13) (Figure 7.2)), extending Mutuelle insurance premium exemption for the poorest members and aggregating *Mutuelle* pools at the departmental level. This shows external financial support for expansion of government health financing and CBHI.

## 7.5.3.1. Co-financing vs. fungibility

In spite of the above investments, some interviewees did, however, call into question whether financial support from development partners stimulated an increase in government health spending (co-financing) or a decrease (subsidy / fungibility) (Figure 7.1).

#### 7.5.3.1.1. Co-financing

We found examples of donors leveraging government finance include GFATM, Gavi and UNFPA which have government co-financing requirements for their health programs (87, 88). A donor explained:

"[Donor] provides resources and the government has to provide also the cost share ... The conditionalities are that you have to put at minimum 25% of the total envelope [Donor] is providing you" (Donor).

Another example was the provision of \$154 million from the WB to help co-finance the Senegalese government's Covid-19 response in 2021 (89), during which a great increase in domestic government health financing was seen (Figure 7.2).

## 7.5.3.1.2. Fungibility

Several of our interviewees claimed that development partner financing however led to decreases in government health spending, i.e. fungibility. Some government officials denied the presence of fungibility, while others believed it took place.

"Most of the time, when donors intervene, we are asked to give counterparts [i.e. cofinancing], and we try to satisfy these counterparts. Without taking into account that we have to readjust... Especially when it comes to budget support, fungibility exists when it comes to budget support." (Ministry/government agency).

A donor representative described cuts to the health budget during the government fiscal year as an explanation for why government health spending has not increased much in Senegal, and the interviewee attributed these cuts to a high presence of donors in the health sector: "... What they [the government] did during the development of the budget at the beginning of the fiscal year, so they give the amount... At the middle of the year, they introduced what they call the amending finance law ... And they cut the budget ... In the health sector... Because there are more donors in the health sector. So that means that the donor resource funding influences the decision of the government in reducing the budget... If they cut the budget, the first target population who will be impacted is the poor and vulnerable population." (Donor)

Of the 15 years where amending finance laws were available since 2000, the government expenditure budget for the Ministry of Health (MoH) was cut eight times compared to six times for the Ministry of Education (MoE) (90)<sup>26</sup>. On average, the MoH lost 0.3% of its initial budgets through these amendments while the MoE gained 0.6% (90). Some expenditure for health and education however exists outside of these ministries. Furthermore, internal versus external revenue source for a given ministry's spending is not delineated in these documents. These numbers also do not elucidate the drivers behind budget cuts, and whether the presence of donors plays a role as claimed is thus not possible to verify using our other data sources.

Another government official emphasized a positive effect of within-sector fungibility of development partner financing for health by freeing up government resources for other social/health purposes:

"... Where partners put in a lot of resources, for example when we speak of certain priority diseases, we see that the state puts less resources... They indirectly influence domestic financing by permitting the state to put many more resources into neglected aspects" (mentions social protection and NCDs) (Ministry/government agency).

While these key-informant statements did not provide hard evidence for the presence of fungibility, which can be difficult to assert, they elicited fungibility as a potential mechanism constraining domestic government health financing. Figure 7.7 displays real-term absolute levels and year-on-year changes in government spending for sectors with available data and external health financing between 2000-2021 (13, 53). As illustrated, these time series do not allow for any judgment regarding the presence or absence of fungibility in the health sector, underlining the importance of key-informant observations.

<sup>&</sup>lt;sup>26</sup> This citation covers 23 budget documents available via the link provided in the reference.





Figure 7.7: A: Government spending for sectors with available data and external health financing in Senegal, 2000-2021 (13, 53) (US\$ GDP deflator used to transform current to constant US\$ for all 4 categories, which leads to slightly lower health financing values than deflated WHO GHED values). B: Real-term year-on-year changes in government spending for sectors with available data and external health financing in Senegal, 2001-2021 (based on constant 2021 US\$) (13, 53).

## 7.6. Discussion

The main objective of this study was to examine the pathways through which development partners influence the mix of domestic funding sources for health in Senegal. Our analysis

identified four potential pathways of influence: setting aims and standards, lobbying/negotiation, providing policy/technical advice, and financing (Figure 7.1). The influence identified generally appeared to be equity enhancing, mainly in terms of expanding government health financing, supporting existing insurance mechanisms and promoting an increase in the size of insurance scheme risk pools. Development assistance fungibility was however identified by some key informants as a dynamic potentially dampening the level of domestic government health financing. Some of the identified pathways of development partner influence were similar to those reported in studies exploring broader EDP influence on recipient governments, including on health and health financing policy. Within gender equality promotion and migration control in Senegal, (91) found consultants and people-to-people exchanges similar to our lobbying/negotiation and policy/technical advice which tend to occur through such interactions. (91) found infrequent presence of tied aid and no evidence of aid conditionality and attributed this to alignment between donor and government objectives. This resonates well with our finding that Senegal was generally seen to set its own development objectives and that partners aligned with these. Technical expertise and financing/financial incentives are frequently cited EDP modes of influence on health policy in other contexts, e.g. in Tanzania (92), Uganda (93, 94), Pakistan and Cambodia (95), and for health financing reform e.g. in Ghana (96), Nigeria (18, 97), Pakistan (98) and Thailand (99). Lobbying has been conceptualised as a general mode of influence in the politial economy of UHC reform in LMICs (48), which our findings support in the case of EDPs in Senegal.

Some authors have emphasized normative power and the diffusion/transfer of international norms (e.g. (100, 101)), by some described as rooted in neoliberal ideals in the 1990's and enacted by the international financial institutions, and how this promoted marketization of health systems and limited government health spending in partner countries (e.g. (102)). This dimension was reflected in our aims and standards category. In Senegal, other authors have found that the use of CBHI as the primary instrument in the path toward UHC, was influenced by a coalition of national and international actors, in part shaped by the ideas, institutions and interests of EDPs (86, 103). On the allocation side of health financing, EDPs have been found to act as "diffusion entrepreneurs", inducing diffusion of performance-based financing policy across SSA (104-106). As further confirmed by other authors (17-22, 24, 25), the role of EDPs in shaping health financing policy in SSA both across contributions and allocations appears well supported, with our study shedding further light on influence mechanisms for health financing contributions. Some authors have also used the case of the international response against HIV, tuberculosis and malaria in the 1990's and 2000's as examples of homogenous, vertical approaches in a heterogenous Africa, with associated marginalisation of African states (e.g. (24, 108)). Fred Eboko (108) saw hope for a return to agency, which we saw manifested

across our interviews in Senegal, and signs of successful government ownership have been found in health financing policy reform across SSA, including for user fees exemption policies (20). Future research could extend our findings and investigate the differential responsiveness to and integration of the different pathways identified on the part of recipient governments.

Our results did not provide examples of EDPs using public advocacy as a means of influence (35, 36). This is consistent with EDP influence occurring more in direct exchange with the Senegalese government, rather than via advocating publicly. This may reflect a functional and intricate collaboration between the Senegalese government and its external partners, where appealing to the government indirectly is unnecessary for EDPs.

For the health sector, our findings illustrate how it is important for development partners to consider to what extent all of their technical, political, and financial activities support partner governments in progressing towards improved equity of domestic health financing contributions and achieving UHC. There may be inconsistencies, where one branch of activities supports the government in mobilising more funds for health, while another helps expand user-fee contributions or regressive insurance premiums. The identified health financing policy analysis frameworks do not specifically emphasize mechanisms or pathways of EDP policy influence, while the identified broader policy influence analysis frameworks focusing on mechanisms/pathways stem from the broader development space without specific application to health financing policy reform (15, 20, 35-49). Our analytical framework fills this gap in the literature by focusing on the different pathways or mechanisms of EDP influence on different health financing sources. In doing so, the derived framework may help understand how the different main EDP activities pursue certain directions in the mix of domestic health financing sources. This may facilitate identification of areas of EDP policy incoherence on the path towards UHC. Once identified, this could form the basis for constructive discussion between government and EDP on how to address or minimise these. Due to a relatively high degree of sovereignty, political vision and quality of policy formulation, we generally saw a high degree of donor alignment and thus analysed EDP influence jointly. This may however vary greatly across contexts. If applying our framework individually across multiple EDPs in a country (e.g. first the World Bank, then the WHO, etc.), this would allow the analyst to map out health financing policy incoherence, separated by EDP and by mechanism. This could potentially add a degree of nuance that could further the utility of our framework as a diagnostic tool for EDP health financing policy incoherence, helping to identify which branches of activities in which organisations promote reliance on which health financing sources. Next steps for further developing our analytical framework could be to integrate co-determinants arising from the domestic political economy, which would require new dedicated empirical

enquiry. It is also conceivable that some pathways might be present in some countries but not in others, necessitating corresponding amendments. The financing pathway could also be further exploded into loans and grants and investigate the downstream effects of debt repayments arising from loans. Another avenue would be to integrate the "3-i's" (42) analogous to (48), exploring the underlying determinants for the EDP influence seen (see limitations).

In Senegal, development partners have provided their support of CBHI in the form of Mutuelles, however the Mutuelles have been critiqued. Issues have included relying on user co-payment, creating relatively small pools with variable financial sustainability, limiting cross subsidy from rich to poor and financial risk protection of poor members, and instituting voluntary enrolment with limited reach (21, 62, 63, 109-111). Efforts to consolidate Mutuelles at the departmental level are ongoing (21, 111), which however does not raise risk pooling to the national level (63, 81). Arguments for the decentralized CBHI model included management being rooted in communities with a higher degree of community ownership, and the historical presence of CBHI in Senegal, leading to higher social acceptability (86, 103). Arguments for a joint departmental model have included administrative professionalization, improved risk pooling, efficiency and financial viability (21, 86, 112). The proposed administrative centralization in the departmental model, enabling increased cross-subsidy among many more members, can be viewed as adhering more to a social welfarist ideology as opposed to the decentralized model, by some described as rooted in neoliberal ideology (81). This exemplifies how differences in priorities and ideologies between partners can cause conflict in the search for a preferred UHC strategy. The value of EDP-supported pilots of alternative health financing mechanisms before broader scale-up, including CBHI, has been noted in Cambodia (113), Ethiopia (114), and across LICs (115), but has also been critiqued as incoherent and ineffective in supporting health financing reform due to poor donor coordination and harmonization in Tajikistan (116). Viewed together, this highlights the importance of the principles of effective development cooperation from the Paris Declaration, Accra Agenda and Busan Partnership for successful EDP-supported health financing reform (115, 117, 118).

Several interviewees identified development assistance fungibility as a mechanism limiting government health spending. The fungibility dynamic in development assistance for health is a well described phenomenon (26, 27, 30, 31, 119-122), and while undesirable for donors, it has been viewed as rational redistribution of funds by others (123, 124). Most of the ODA for health since 2002 has been disbursed as project-type interventions (70%), which are more tightly ear-marked (125), and only 7% as sectoral budget support, which limits the scope for fungibility of the injected funds themselves. Crowding-out of government funds is however still

possible, if the government deems that externally funded projects cover certain population health needs and then decides, for whatever reason, to withdraw or not to supplement with funding for the same population health needs.

With the mentioned caveats of a possible fungibility effect and the small financial risk pools of decentralized CBHI, which is now being reformed, EDPs in Senegal generally appear to have been a force for improved equity of domestic health financing sources. They have used their identified influence pathways to promote a mix of domestic health financing sources characterized more by progressive, tax-based contributions from the government and less by OOP. In the complex political economy of UHC reform, the incrementalist approach building on existing CBHI structures may however have limited the overall scope for equity improvements compared to a more universalist approach (21, 48, 86). The enduring predominance of OOP in Senegal and limited real-term growth in GHE-S indicates that in spite of their efforts, EDPs have not been successful in achieving a more equitable domestic health financing mix. Differing interests from both different domestic stakeholders, including domestic policy makers and the Senegalese mutualist movement, as well as between different EDPs (USAID, Enabel and the WB), may have co-determined the limited progress seen over time (21, 81, 86).

## 7.6.1. Limitations

The dynamics we have investigated in this study result from policy processes that often occur behind closed doors and are subject to unspoken ideology, power dynamics and political considerations (100, 126-129). Interview participants may also have held incorrect or imprecise information, and causal pathways from external development partner actions to domestic health financing impacts may be complex. Also, there is no counterfactual, and it is not possible to truly know how domestic health financing would have differed in the absence of development partner influence. These circumstances inevitably limited the extent to which we could access the "truth" of our research question. We sought to mitigate this limitation by interviewing a broad range of stakeholders, offering them anonymity so they could speak freely, and using a range of other sources of information to triangulate statements. Future studies could search for natural policy experiments, possibly at the regional or district level, where comparable geographical entities are subjected to different EDP-supported health financing reforms, such as the Enabel-funded pilot of departmental aggregation of CBHI in Senegal. Such studies should however bear in mind the historical and present influence from other EDPs, and finding a true "untouched" control seems improbable.

We were also unable to measure actual equity of financing through financing incidence analysis, but rather sought to provide an indication of whether and how development partners influenced financing mechanisms that are typically more or less equitable.

Our results did not allow us to explore the underlying reasons for why development partners used a particular mechanism or tried to push health financing contributions in a certain direction. The "3-*i*'s" framework by Hall 1997 (ideas, interests and institutions) is one possible basis for approaching this question (42-44), as exemplified by (48, 86, 130, 131). On the allocation side, neoliberal ideology has been pointed out as a reason for development partners promoting performance-based financing in Senegal (81). Future research should further interrogate the role of IFI policy recommendations and loan conditionalities in determining domestic health financing contributions and allocations in Senegal.

Certain nuances may have been lost in translation during interviews. To mitigate this, a professional interpreter was used for French-language interviews until this was no longer necessary. Interviews were also transcribed in their original language, so all nuances in wordings were retained in the data and could be interpreted post hoc. Where interviews were conducted in English as per the stated preference of the participant, this may still not have been their primary working language, and some depth and nuance may have been lost as a result.

While the manuscript was written in English, the proficiency of our team in both French and English and the use of a professional interpreter and transcriptionist should have rendered any impacts on our results and interpretation from French-English translation altogether minimal.

Five participants did not allow for the interview to be recorded or used directly in the paper, which greatly limited the analytical utility of these interviews. As an attempt to mitigate this effect, FF took notes from these interviews to help understand the interviewee's general position on interview topics, which helped inform the research, however superficially compared to recorded interviews. To honour the wishes of these participants and follow the lower level of detail present in handwritten interview notes, we only referred to findings from these interviews as part of broad statements such as "*X* was echoed across all stakeholder groups".

Representatives from two multilateral organisations, two hospitals and two academics were unavailable for interview or did not respond. Had these representatives participated, perhaps our results might have been slightly less favourable for the government. However, as seen in Table 7.2, our final interview group was well balanced, with 9 interviewees in central government vs. 15 outside of the central government-donor nexus, and 12 of the latter were from hospitals, civil society and academia, which were generally more critical groups.

Finally, as our purposive document review was not a full systematic literature review, it is conceivable that we could have overlooked relevant studies.

## 7.7. Conclusions

We identified setting aims and standards, lobbying/negotiation, providing policy/technical advice and financing as avenues for development partner influence on domestic health financing contributions in Senegal, and with a seemingly equity enhancing influence. Fungibility and intrinsic equity issues related to CBHI may however have limited equity gains. We encourage stakeholders in the health financing sphere to use our framework and analysis to unpack how development partners affect domestic health financing, including equity, in other settings. This could serve as a basis for identifying dynamics that do not optimally support progress towards UHC, and facilitate working towards coherent policy-making across all domains of development partner activities, that all support UHC. Our framework and analysis should be expanded and amended in other contexts as appropriate. The role of international creditors, lending and loan conditionalities on domestic health financing in recipient countries should also be further explored, including equity implications.

## References

1. Makinen M, Waters H, Rauch M, Almagambetova N, Bitran R, Gilson L, et al. Inequalities in health care use and expenditures: Empirical data from eight developing countries and countries in transition. Bulletin of the World Health Organization. 2000;78(1):55-65.

2. Xu K, Evans DB, Carrin G, Águilar-Rivera AM, Musgrove P, Evans T. Protecting households from catastrophic health spending. Health Aff (Millwood). 2007;26(4):972-83.

3. WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available from: <a href="https://www.who.int/publications/i/item/9789241564021">https://www.who.int/publications/i/item/9789241564021</a>.

4. Marmot M, Commission on Social Determinants of Health. Achieving health equity: from root causes to fair outcomes. Lancet. 2007;370(9593):1153-63.

5. McIntyre D, Mooney G. The Économics of Health Equity. 1st ed: Cambridge University Press; 2007. ISBN: 978-0521705066

6. Wagstaff A, Van Doorslaer E, Paci P. Equity in the Finance and Delivery of Health Care: Some Tentative Cross-Country Comparisons. Oxford Review of Economic Policy. 1989;5(1):89-112.

7. Mtei G, Makawia S, Ally M, Kuwawenaruwa A, Meheus F, Borghi J. Who pays and who benefits from health care? An assessment of equity in health care financing and benefit distribution in Tanzania. Health Policy Plan. 2012;27 Suppl 1:i23-34.

8. Martinez-Alvarez M, Federspiel F, Singh NS, Schaferhoff M, Lewis Sabin M, Onoka C, et al. Equity of resource flows for reproductive, maternal, newborn, and child health: are those most in need being left behind? BMJ. 2020;368:m305.

9. Mathauer I, Saksena P, Kutzin J. Pooling arrangements in health financing systems: a proposed classification. Int J Equity Health. 2019;18(1):198.

10. Kutzin J. Towards Universal Health Care Coverage: A Goal-oriented Framework for Policy Analysis. 2000 [cited 2024 July 25]. Available from:

https://documents1.worldbank.org/curated/pt/260141468779178780/820140748\_200405160101945/additional/K utzin010Toward0Universal1whole.pdf.

11. Kutzin J, Witter S, Jowett M, Bayarsaikhan D. Developing a national health financing strategy: a reference guide. Geneva: World Health Organization; 2017 2017. ISBN: 9789241512107 9789246512102 (Arabic)

12. Mills A, Ataguba JE, Akazili J, Borghi J, Garshong B, Makawia S, et al. Equity in financing and use of health care in Ghana, South Africa, and Tanzania: implications for paths to universal coverage. The Lancet. 2012;380(9837):126-33.

13. WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from: http://apps.who.int/nha/database.

14. Parkhurst J, Leir S, Walls H, Vecchione E, Liverani M. Evidence and Policy in Aid-Dependent Settings. In: Parkhurst J, Ettelt S, Hawkins B, editors. Evidence Use in Health Policy Making: An International Public Policy Perspective. Cham: Springer International Publishing; 2018. p. 201-19.

15. Sparkes SP, Bump JB, Ozcelik EA, Kutzin J, Reich MR. Political Economy Analysis for Health Financing Reform. Health Syst Reform. 2019;5(3):183-94.

16. Ridde V, Faye A. Policy response to COVID-19 in Senegal: power, politics, and the choice of policy instruments. Policy Design and Practice. 2022;5(3):326-45.

17. Kajula PW, Kintu F, Barugahare J, Neema S. Political analysis of rapid change in Uganda's health financing policy and consequences on service delivery for malaria control. Int J Health Plann Manage. 2004;19 Suppl 1:S133-53.

18. Alawode G, Adewoyin AB, Abdulsalam AO, Ilika F, Chukwu C, Mohammed Z, et al. The Political Economy of the Design of the Basic Health Care Provision Fund (BHCPF) in Nigeria: A Retrospective Analysis for Prospective Action. Health Syst Reform. 2022;8(1):2124601.

 Witter S, Chirwa Y, Chandiwana P, Munyati S, Pepukai M, Bertone MP. The political economy of results-based financing: the experience of the health system in Zimbabwe. Glob Health Res Policy. 2019;4:20.
 Gautier L, Ridde V. Health financing policies in Sub-Saharan Africa: government ownership or donors' influence? A scoping review of policymaking processes. Glob Health Res Policy. 2017;2:23.

21. Ridde V, Caffin JH, Hane F. External influences over Senegalese health financing policy: delaying universal health coverage? Health Policy Plan. 2024;39(1):80-3.

22. Ridde V, Ba MF, Guyot M, Kane B, Mbow NB, Senghor I, et al. Factors that foster and challenge the sustainability of departmental health insurance units in Senegal. International Social Security Review. 2022;75(2):97-117.

23. Sridhar D, Gomez EJ. Health financing in Brazil, Russia and India: what role does the international community play? Health Policy Plan. 2011;26(1):12-24.

24. Nagemi C, Mwesigwa C. Donor Power and Prioritization in Development Assistance for Health Policies: The Case of Uganda. Journal of Developing Economies. 2021;2(1):54-74.

25. Colenbrander S, Birungi C, Mbonye AK. Consensus and contention in the priority setting process: examining the health sector in Uganda. Health policy and planning. 2015;30(5):555-65.

26. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJ. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

27. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

28. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

29. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

30. Xu K, Saksena P, Holly A. The determinants of health expenditure: A country-level panel data analysis. Results for Development Institute WHO; 2011.

31. Younsi M, Chakroun M, Nafla A. Robust analysis of the determinants of healthcare expenditure growth: evidence from panel data for low-, middle- and high-income countries. International Journal of Health Planning and Management. 2016;31(4):580-601.

32. Ali MEM, Ebaidalla M, Rizk R. Foreign aid and out-of-pocket health expenditure in Sub-Saharan Africa: does institutional development matter. International Journal of Sustainable Economy. 2020;12(3).

33. Goudge J, Akazili J, Ataguba J, Kuwawenaruwa A, Borghi J, Harris B, et al. Social solidarity and willingness to tolerate risk- and income-related cross-subsidies within health insurance: experiences from Ghana, Tanzania and South Africa. Health Policy Plan. 2012;27 Suppl 1:i55-63.

34. McIntyre D. Beyond fragmentation and towards universal coverage: insights from Ghana, South Africa and the United Republic of Tanzania. Bulletin of the World Health Organization. 2008;86(11):871-6.

35. Start D, Hovland I. Tools for Policy Impact: A Handbook for Researchers. 2004 [cited 2024 July 29]. Available from: <u>https://cdn.odi.org/media/documents/194.pdf</u>.

36. De Raeve P, Bolzonella F, Davidson PM. A Theoretically Derived Approach to Impact: Implementing Policy Influence Strategies. Policy Polit Nurs Pract. 2022:15271544221100164.

37. Steinberg PF. Understanding Policy Change in Developing Countries: The Spheres of Influence Framework. Global Environmental Politics. 2003;3(1):11-32.

38. Kingdon JW. Agendas, alternatives, and public policies: Boston : Little, Brown, [1984] ©1984; 1984. ISBN:

39. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. Health Policy Plan. 1994;9(4):353-70.

40. Grindle M, Thomas JW. Public choices and policy change : the political economy of reform in developing countries. Baltimore: Johns Hopkins University Press; 1991. ISBN: 0801841550
41. Lindquist EA. Discerning Policy Influence: Framework for a Strategic Evaluation of IDRC-Supported Research. 2001 [cited April 9 2024]. Available from:

https://www.betterevaluation.org/sites/default/files/discerning\_policy.pdf.

42. Hall PA. The Role of Interests, Institutions, and Ideas in the Comparative Political Economy of the Industrialized Nations Cambridge University Press 1997. ISBN: 0521586682

43. National Collaborating Centre for Healthy Public Policy. Understanding Policy Developments and Choices Through the "3-i" Framework: Interests, Ideas and Institutions. 2014 [cited 2024 July 29]. Available from: https://ccnpps-ncchpp.ca/docs/2014 ProcPP 3iFramework EN.pdf.

44. Mulvale G, McRae SA, Milicic S. Teasing apart "the tangled web" of influence of policy dialogues: lessons from a case study of dialogues about healthcare reform options for Canada. Implement Sci. 2017;12(1):96.

45. Overseas Development Institute (ODI). ROMA: A guide to policy engagement and influence. 2014 [cited April 9 2024]. Available from: <u>https://odi.cdn.ngo/media/documents/odi roma guide.pdf</u>.

46. Crewe E, Young J. Bridging Research and Policy: Context, Evidence and Links. 2002 [cited 2022 July 6]. Available from: <u>https://cdn.odi.org/media/documents/184.pdf</u>.

47. Cathexis Consulting. A Guide to Policy-Influence Evaluation: Selected Resources and Case Studies. 2015 [cited April 9 2024]. Available from:

https://www.nccmt.ca/uploads/media/0001/01/c3374b8be4b35e1340385f1b593d3bb9f50f6a38.pdf.

48. Fox AM, Reich MR. The Politics of Universal Health Coverage in Low- and Middle-Income Countries: A Framework for Evaluation and Action. Journal of Health Politics, Policy and Law. 2015;40(5):1023-60.

49. Rizvi SS, Douglas R, Williams OD, Hill PS. The Political Economy of Universal Health Coverage: A Systematic Narrative Review. Health Policy and Planning. 2020;35(3):364-72.

50. Galanti MT. Policy Advice. In: Harris P, Bitonti Å, Fleisher ČS, Skorkjær Binderkrantz A, editors. The Palgrave Encyclopedia of Interest Groups, Lobbying and Public Affairs. Cham: Springer International Publishing; 2020. p. 1-5.

51. UK Parliament. Lobbying Parliament. 2024 [cited 2024 August 1]. Available from:

https://www.parliament.uk/get-involved/contact-an-mp-or-lord/lobbying-parliament/

52. Agence nationale de la statistique et de la demographie (ANSD). Le Sénégal En Bref. 2023 [cited 2023 August 11]. Available from: <u>https://www.ansd.sn/</u>.

53. World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: https://data.worldbank.org/.

54. OECD. Creditor Reporting System (CRS). 2024 [cited 2023 February 16]. Available from: https://stats.oecd.org/Index.aspx?DataSetCode=CRS1.

55. Government of Senegal. Plan Sénégal Émergent 2014 [cited 2024 March 27]. Available from: https://www.au-senegal.com/IMG/pdf/plan-senegal-emergent-2014.pdf.

56. Agence de la Couverture Maladie Universelle. Official Website. 2020 [cited 2020 November 4]. Available from: <u>http://agencecmu.sn/</u>.

57. Paul E, Ndiaye Y, Sall FL, Fecher F, Porignon D. An assessment of the core capacities of the Senegalese health system to deliver Universal Health Coverage. Health Policy Open. 2020;1:100012.
58. Division des Institutions de Prevoyance Maladie MdTedOp, Senegal, Les Institutions de Prevoyance Maladie (I.P.M.). [cited 2021 Sept. 30]. Available from:

https://fonctionpublique.sec.gouv.sn/sites/default/files/ipm.pdf.

59. Ministère de la Santé et de l'Action Sociale (MSAS). Strategie nationale de financement de la santé pour tendre vers la couverture sanitaire universelle. 2017 [cited 2024 July 29]. Available from:

https://www.sante.gouv.sn/sites/default/files/2%20MSAS%20Strat%C3%A9gie%20Nationale%20de%20Finance ment%20de%20Ia%20Sant%C3%A9.pdf.

60. ICAMO. Sources de financement des IPM. 2023 [cited 2023 August 3]. Available from: http://icamo.sn/sourcedefinancementipm.php.

61. Ministère du développement communautaire dleset. Agence de la Couverture Maladie Universelle (ANACMU). 2021 [cited 2021 May 19]. Available from: <u>https://equite.sec.gouv.sn/programme/agence-de-la-couverture-maladie-universelle-anacmu</u>.

62. Wood A. Patronage, partnership, voluntarism: Community-based health insurance and the improvisation of universal health coverage in Senegal. Social science & medicine (1982). 2023;319:115491.

63. Daff BM, Diouf S, Diop ESM, Mano Y, Nakamura R, Sy MM, et al. Reforms for financial protection schemes towards universal health coverage, Senegal. Bull World Health Organ. 2020;98(2):100-8.

64. Ministère de la Santé et de l'Action Sociale (MSAS). Plan National de Développement Sanitaire et Social (PNDSS). 2019 [cited 2024 July 29]. Available from:

https://www.sante.gouv.sn/sites/default/files/1%20MSAS%20PNDSS%202019%202028%20Version%20Finale.p

65. Agence nationale de la statistique et de la demographie (ANSD). ansd.sn. 2024 [cited 2024 July 25]. Available from: <u>https://www.ansd.sn/index.php</u>.

66. Agence Nationale de la Statistique et de la Démographie. Enquête harmonisée sur les Conditions de Vie des Ménages (EHCVM) au Sénégal. 2021 [cited 2022 May 12]. Available from:

https://www.ansd.sn/sites/default/files/2022-11/Rapport-final-EHCVM-vf-Senegal.pdf. 67. Saunders B, Sim J, Kingstone T, Baker S, Waterfield J, Bartlam B, et al. Saturation in qualitative research: exploring its conceptualization and operationalization. Qual Quant. 2018;52(4):1893-907.

68. Ritchie J, Spencer L. Qualitative Data Analysis for Applied Policy Research. In: Bryman A, Burgess R, editors. Analyzing Qualitative Data. 1st ed: Routledge; 1994.

69. QSR International Pty Ltd. NVivo qualitative data analysis software. Version 12, 2020.

70. Ritchie J, Lewis J. Qualitative Research Practice: A Guide for Social Science Students and Researchers: SAGE Publications; 2003. ISBN:

71. Organisation of African Unity. Abuja Declaration on HIV/AIDS, Tuberculosis and Other Related Infectious Diseases. Abuja, Nigeria; 2001.

72. WHO. Declaration of Alma-Ata. 1978 [cited 2024 July 29]. Available from:

https://cdn.who.int/media/docs/default-source/documents/almaata-declaration-en.pdf?sfvrsn=7b3c2167 2. 73. Ministère de la Santé et de l'Action Sociale (MSAS). Strategie nationale de financement de la santé pour tendre vers la couverture sanitaire universelle. 2017 [cited 2021 October 19]. Available from: https://www.sante.gouv.sn/sites/default/files/2%20MSAS%20Strat%C3%A9gie%20Nationale%20de%20Finance

ment%20de%20la%20Sant%C3%A9.pdf.

74. World Bank. Investing in Health. 1993 [cited 2024 July 29]. Available from:

https://openknowledge.worldbank.org/handle/10986/5976.

75. USAID. Senegal: Global Health. 2023 [cited 2023 February 26]. Available from:

https://www.usaid.gov/senegal/global-health.

76. USAID. Improving Health. [cited 2023 February 26]. Available from:

https://www.usaid.gov/sites/default/files/2022-05/Senegal\_USAID\_FactSheet\_Health.pdf

77. UNICEF. Equity, governance and social policy. 2020 [cited 2023 February 26]. Available from: https://www.unicef.org/senegal/en/equity-governance-and-social-policy.

78. World Bank. Official Documents- Financing Agreement for Credit 6495-SN (English). 2020 [cited 2023 February 28]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/742331586276881044/official-documents-financing-agreement-for-credit-6495-sn. 79. World Bank. Implementation completion and results report. 2019 [cited 2023 August 2]. Available from:

https://documents1.worldbank.org/curated/en/169851584393359384/pdf/Senegal-Health-Nutrition-Financing-Project.pdf

80. Global Financing Facility. Senegal: Focusing on Results to Accelerate Health Outcomes. 2022 [cited 2023 February 26]. Available from: <u>https://www.globalfinancingfacility.org/senegal</u>.

81. Jean Hugues Caffin. L'aide au développement et le financement basé sur la performance : quelle performativité ? : analyse du processus de conceptualisation et de diffusion du financement basé sur la performance dans la gestion des systèmes de santé africains par la Banque Mondiale et l'USAID : étude du cas du Programme national de financement basé sur les résultats du Ministère de la Santé du Sénégal: Ecole Doctorale de Management Panthéon-Sorbonne; 2018.

82. Ministère de la santé et de la prevention. Stratégie nationale d'extension de la couverture du risque maladie des Sénégalais. 2010 [cited 2024 July 29]. Available from: <u>https://www.social-</u>

protection.org/gimi/RessourcePDF.action;jsessionid=kKAs2zrwy5YcY\_iv6i2rMs2ffVZynczjqVsFmCS6XI5DtzHLo w\_D!1588761716?id=18177

83. Alenda-Demoutiez J. Les mutuelles de santé au Sénégal face aux difficultés de coordination de leurs acteurs. Revue internationale de l'économie sociale: Recma. 2017(345).

84. Bénédicte Fonteneau, Vaes S, Ongevalle JV. Toward Redistributive Social Protection? Insights from Senegal and Morocco. 2017 [cited 2024 July 25]. Available from: <u>https://www.befind.be/Documents/WPs/WP21</u>.
85. Ministère de la Santé et de la Prevention. Plan National de Développement Sanitaire. 2009 [cited 2024 July 29]. Available from:

https://extranet.who.int/countryplanningcycles/sites/default/files/planning\_cycle\_repository/senegal/pnds2009\_20\_18\_version\_finale\_finale.pdf.

86. Deville C. Same objective, different paths: An analysis of Universal Health Coverage policies in Benin and Senegal. 2018.

87. The Global Fund. Guidance Note: Sustainability, Transition and Co-financing. 2022 [cited 2024 July 29].
Available from: <u>https://www.theglobalfund.org/media/5648/core\_sustainabilityandtransition\_guidancenote\_en.pdf</u>.
88. Jha A, Kolesar R, Comas S, Gribble J. Principles for co-financing of family planning commodities. 2022
[cited 2024 July 25]. Available from: <u>http://www.healthpolicyplus.com/ns/pubs/18658-</u>
19135 cofinancingFPcommodities.pdf.

89. World Bank. World Bank financing helps to support Senegal in the fight against COVID-19. 2021 [cited 2023 August 4]. Available from: <u>https://www.worldbank.org/en/results/2021/08/31/world-bank-financing-helps-to-support-senegal-in-the-fight-against-covid-19</u>.

90. Direction Générale du Budget. Lois de finances rectificatives. 2024 [cited 2024 August 8]. Available from: https://budget.sec.gouv.sn/recherche-par-types-documents/loi-de-finances-rectificatives.

91. Olivie I. How is aid used to exert power? Gender equality promotion and migration control in Senegal. J Int Dev. 2022;34(8):1509-26.

92. Fischer SE, Strandberg-Larsen M. Power and Agenda-Setting in Tanzanian Health Policy: An Analysis of Stakeholder Perspectives. Int J Health Policy Manag. 2016;5(6):355-63.

93. Razavi SD, Kapiriri L, Abelson J, Wilson M. Who is in and who is out? A qualitative analysis of stakeholder participation in priority setting for health in three districts in Uganda. Health Policy Plan. 2019;34(5):358-69.

94. Nabyonga Orem J, Marchal B, Mafigiri D, Ssengooba F, Macq J, Da Silveira VC, et al. Perspectives on the role of stakeholders in knowledge translation in health policy development in Uganda. BMC Health Serv Res. 2013;13:324.

95. Khan MS, Meghani A, Liverani M, Roychowdhury I, Parkhurst J. How Do External Donors Influence National Health Policy Processes? Experiences of Domestic Policy Actors in Cambodia and Pakistan. Health Policy and Planning. 2018;33(2):215-23.

96. Koduah A, van Dijk H, Agyepong IA. The role of policy actors and contextual factors in policy agenda setting and formulation: maternal fee exemption policies in Ghana over four and a half decades. Health Res Policy Syst. 2015;13:27.

97. Onoka CA, Hanson K, Hanefeld J. Towards universal coverage: a policy analysis of the development of the National Health Insurance Scheme in Nigeria. Health Policy Plan. 2015;30(9):1105-17.

98. Khalid F, Afzal S, Shirazi A, Quick I, Mataria A. Coordinating external assistance for UHC: Pakistan's early experience of the SDG3 GAP. Health Policy and Planning. 2024;39(2):247-51.

99. Herberholz C. The role of external actors in shaping migrant health insurance in Thailand. PLoS One. 2020;15(7):e0234642.

100. Shiffman J. Knowledge, moral claims and the exercise of power in global health. Int J Health Policy Manag. 2014;3(6):297-9.

101. Bazbauers AR. World Bank Technical Assistance: The Relational Dynamics of Policy Movement. Public Administration and Development. 2017;37(4):246-59.

 Palier B, Mandin L. Le rôle des organisations internationales dans les réformes des systèmes nationaux de protection sociale. In: Schemeil Y, Wolf-Dieter E, editors. Normer le monde: L'Harmattan; 2009. p. 177-210.
 Deville C, Fecher-Bourgeois F, Poncelet M. Les mutuelles de santé subventionnées comme instruments de la Couverture Maladie Universelle au Sénégal. 2018.

104. Gautier L, Tosun J, De Allegri M, Ridde V. How do diffusion entrepreneurs spread policies? Insights from performance-based financing in Sub-Saharan Africa. World Development. 2018;110:160-75.

105. Gautier L. From ideas to policymaking : the political economy of the diffusion of performance-based financing at the global, continental, and national levels: Université de Montréal et Université Paris-Diderot; 2019.
106. Gautier L, Coulibaly A, De Allegri M, Ridde V. From Amsterdam to Bamako: a qualitative case study on diffusion entrepreneurs' contribution to performance-based financing propagation in Mali. Health Policy Plan. 2019;34(9):656-66.

107. Discussions at the Eighteenth Annual Conference of the NATIONAL ASSOCIATION FOR THE PREVENTION OF TUBERCULOSIS, July 21st to 23rd 1932. Discussions at the Eighteenth Annual Conference of the NATIONAL ASSOCIATION FOR THE PREVENTION OF TUBERCULOSIS, July 21st to 23rd 1932. 108. Eboko F. Vers une matrice de l'action publique en Afrique ? Approche trans-sectorielle de l'action

publique en Afrique contemporaine. 2015 [cited 2024 March 27 ]. Available from: https://journals.openedition.org/lectures/17120.

109. Mladovsky P, Ndiaye P. Solidarity in community-based health insurance in Senegal: rhetoric or reality? African Health Monitor. 2015.

110. Ly MS, Faye A, Ba MF. Impact of community-based health insurance on healthcare utilisation and outof-pocket expenditures for the poor in Senegal. BMJ Open. 2022;12(12):e063035.

111. Ridde V, Gaye I, Ventelou B, Paul E, Faye A. Mandatory membership of community-based mutual health insurance in Senegal: A national survey. PLOS Glob Public Health. 2023;3(9):e0001859.

112. Ridde V, Kane B, Mbow NB, Senghor I, Faye A. The sustainability of two departmental health insurance units in Senegal: A qualitative study. SSM - Health Systems. 2024;2.

113. Ir P, Bigdeli M, Meessen B, Van Damme W. Translating knowledge into policy and action to promote health equity: The Health Equity Fund policy process in Cambodia 2000-2008. Health Policy. 2010;96(3):200-9. 114. Mulat AK, Mao W, Bharali I, Balkew RB, Yamey G. Scaling up community-based health insurance in Ethiopia: a qualitative study of the benefits and challenges. BMC Health Serv Res. 2022;22(1):473.

115. Kiendrébéogo JA, Meessen B. Ownership of health financing policies in low-income countries: a journey with more than one pathway. BMJ Glob Health. 2019;4(5):e001762.

116. Jacobs E. The politics of the basic benefit package health reforms in Tajikistan. Glob Health Res Policy. 2019;4:14.

117. OECD. The Paris Declaration on Aid Effectiveness and the Accra Agenda for Action. 2005, 2008 [cited 2024 July 29]. Available from: <u>https://web-archive.oecd.org/temp/2021-08-02/73869-</u>parisdeclarationandaccraagendaforaction.htm.

118. OECD. Busan Partnership for Effective Development Co-operation. 2011 [cited 2024 July 29]. Available from: https://web-archive.oecd.org/temp/2023-10-09/57958-busanpartnership.htm.

119. Farag M, Nandakumar AK, Wallack SS, Gaumer G, Hodgkin D. Does funding from donors displace government spending for health in developing countries? Health Aff (Millwood). 2009;28(4):1045-55.

120. Stuckler D, Basu S, McKee M. International Monetary Fund and aid displacement. Int J Health Serv. 2011;41(1):67-76.

121. Fernandes Antunes A, Xu K, James CD, Saksena P, Van de Maele N, Carrin G, et al. General budget support: has it benefited the health sector? Health economics. 2013;22(12):1440-51.

122. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

123. Rana Z, Koch D-J. Is it time to 'decolonise' the fungibility debate? Third World Quarterly. 2019;41(1):42-57.

Martinez Alvarez M, Borghi J, Acharya A, Vassall A. Is Development Assistance for Health fungible?
Findings from a mixed methods case study in Tanzania. Social science & medicine (1982). 2016;159:161-9.
OECD. Methodological notes on the Development Co-operation Profiles 2023. 2023 [cited 2024 April 4].

Available from: <u>https://www.oecd-ilibrary.org/side/5d46d48-</u>

en/index.html?itemId=/content/component/5d646dd8-en

126. Erasmus E, Gilson L. How to start thinking about investigating power in the organizational settings of policy implementation. Health Policy Plan. 2008;23(5):361-8.

127. Anderson E-L. African health diplomacy: obscuring power and leveraging dependency through shadow diplomacy. International Relations. 2018;32(2):194-217.

128. Bachrach P, Baratz MS. Two Faces of Power. The American Political Science Review. 1962;56(4):947-52.

129. Lukes S. Power: A Radical Viiew. 2nd ed: Palgrave Macmillan; 2005. ISBN: 978-0-333-42092-8 978-0-230-80257-5

130. Mhazo AT, Maponga CC. The political economy of health financing reforms in Zimbabwe: a scoping review. Int J Equity Health. 2022;21(1):42.

131. Parkhurst J, Ghilardi L, Webster J, Snow RW, Lynch CA. Competing interests, clashing ideas and institutionalizing influence: insights into the political economy of malaria control from seven African countries. Health Policy Plan. 2021;36(1):35-44.



London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT

T: +44 (0)20 7299 4646 F: +44 (0)20 7299 4656 www.lshtm.ac.uk

## RESEARCH PAPER COVER SHEET

Please note that a cover sheet must be completed for each research paper included within a thesis.

Student ID Number	lsh1601722	Title	Dr.
First Name(s)	Frederik		
Surname/Family Name	Federspiel		
Thesis Title	External Development Partner and International Monetary Fund Influence on Domestic Health Financing Sources in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal		
Primary Supervisor	Josephine Borghi		

#### **SECTION A – Student Details**

If the Research Paper has previously been published please complete Section B, if not please move to Section C.

#### SECTION B - Paper already published

Where was the work published?			
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Choose an item.	Was the work subject to academic peer review?	Choose an item.

\*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

#### SECTION C - Prepared for publication, but not yet published

Where is the work intended to be published?	Globalization and Health
Please list the paper's authors in the intended authorship order:	Frederik Federspiel, Josephine Borghi, Henning Tarp Jensen, Elhadji Mamadou Mbaye, Melisa Martinez Alvarez

Improving health worldwide

www.lshtm.ac.uk

Stage of publication	Not yet submitted

## SECTION D – Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I designed this study as Principal Investigator (PI) in collaboration with my supervisors and co-PI in Senegal. I produced the study protocol and research ethics applications with input from supervisors and co-PI. I collected, analysed and interpreted the data, and wrote the first draft paper, with the input necessary along the way from my supervisors. My supervisors, local co-PI and I then critically revised the article, with me being responsible for performing the revisions necessary.
---	---

#### SECTION E

Student Signature		
Date	20/9-2024	

Supervisor Signature				
Date	23/	09/24		

# 8. Chapter 8: Influence of Public External Borrowing on Domestic Health Financing Sources: A Case Study of Senegal

## 8.1. Chapter overview

Chapter 6 identified negative relationships at the global level between Official Development Assistance (ODA)<sup>+</sup> for health and Government Health Expenditure as a Source (GHE-S) and Out Of Pocket payments (OOP), and found Public and Publicly Guaranteed (PPG) external debt servicing to be associated with shifts from GHE-S toward OOP in the mix of domestic health financing sources. Chapter 7 further explored the ways in which External Development Partners (EDPs) could influence the mix of health financing sources using a case study of Senegal, also identifying a fungibility dynamic while establishing some further non-financial influence mechanisms. The present chapter expands on these findings by exploring external influence on domestic health financing sources in Senegal via the financial mechanism of providing loans resulting in debt obligations. As outlined in Chapters 1, 2 and 3, loans specifically provided by the IMF and WB are provided under programs with extensive macroeconomic and fiscal policy recommendations and conditionalities that may determine government health expenditure and the mix of domestic health financing sources in borrower countries. This chapter uses mixed-methods, including interviews, document review and descriptive quantitative analysis to address objective three of this thesis: To explore how external official lending, public external debt obligations and IMF/WB policy recommendations and loan conditionalities affect the mix of domestic health financing sources in Senegal. It begins with a summary of the relevant literature, introduces the analytical framework and the study setting and methods, before summarizing its findings. It then discusses these findings in relation to the literature before making some recommendations for policy and future research based on its findings.

## 8.2. Abstract

#### Background

Public external debt burdens of Low- and Middle-Income Countries (LMICs) have grown substantially over the past decade linked to increased borrowing. Our empirical understanding of the ways in which external official loans, their ensuing debt burden and associated policy recommendations and loan conditionalities may impact the mix of domestic health financing sources, is limited, particularly at the country level. We conducted a mixed-methods case study of Senegal focusing on the role of the International Monetary Fund (IMF) and the World Bank (WB) to better understand these dynamics.

#### Methods

We conducted semi-structured interviews between October 2019 and January 2020 with 25 key informants from government ministries, development partners, academia, civil society and hospital management in Senegal. We also performed a purposive document review of reports from IMF, WB, government websites and academic literature, as well as a descriptive quantitative analyses of loan, debt and health financing statistics.

#### Results

External official lending has had a mixed influence on the different domestic health financing sources in Senegal. Our findings indicated that this has occurred via the following pathways: Loans have helped expand the government budget and have supported the health sector directly. Resulting debt obligations appear to have had a constraining effect on the government budget including for health. Specifically for IMF/WB loans over the recent two decades, these have come with policy recommendations and conditionalities promoting increased domestic government health spending, expansion of community-based health insurance and private sector health service delivery, along with fiscal austerity.

#### Conclusions

To mitigate the negative effect of debt repayment, official external creditors should consider more flexible debt repayment timelines, exploring options for debt relief including debt-tohealth swaps, and a specific government health expenditure floor at 10-15% of government expenditure. A legally binding agreement building on the Abuja pledge could be considered between the Senegalese government and its external partners to support sustained increases in government budget prioritisation of the health sector, supported by debt relief.

We encourage further country level research documenting debt impacts on health financing.

## 8.3. Introduction

In the late 1990's and early 2000's, the public external debt burdens of Low- and Middle-Income Countries (LMICs) were high. This was viewed as unsustainable, constricting fiscal space in turn limiting investments in social sectors. Debt relief was granted out of a motivation to help countries free up resources to develop and fund their hard-pressed education and health sectors (1, 2). This resulted in falling debt obligations during the 2000s (3-5) (Figure 8.1). The debt burdens of LMICs to foreign creditors have since been growing over the following decade, as debt relief decreased and public external borrowing increased (6). In 2021, 49 out of 116 LMICs with data spent more on public external debt servicing than on health compared to 34 in 2010 (7, 8), a trend concentrated among Sub-Saharan African (SSA) countries (7-10).





All loans provided by external official creditors have different terms of lending. Unlike other creditors, loans from the International Monetary Fund (IMF) and the World Bank (WB) are offered under programs with a range of macroeconomic and fiscal policy recommendations and conditionalities in addition to requirements for repayment. These have historically involved measures of fiscal consolidation, decentralization, privatization and trade liberalization, which can also affect domestic health financing.

Existing literature has examined the association between public external debt and government health spending through quantitative cross-country panel data analyses in African (11, 12),

Latin American (13), South- and Southeast Asian (14), and select "emerging" and middleincome countries (15) as well as across LMICs overall (16-18). A discussion report has examined trends in debt and government health spending in Kenya (19).

IMF/WB loan conditionalities were associated with the introduction of user fees for health services in the 1980's and 1990's across several Sub-Saharan African countries (20-24). Policy recommendations and conditionalities of austerity such as public spending cuts, containing the wage bill, and privatization of public entities from the IMF have been criticized for leading to constrained government health spending (12, 20, 25-35), including in Senegal (33). However, the IMF and WB have conversely claimed that government health spending is protected from austerity measures, and even increased in IMF Structural Adjustment Programme (SAP) countries compared to non-adjusting countries (36-40). Other researchers have reported findings that align with those of the IMF/WB across LMICs (41), SSA Low-Income Countries (LICs) though not for non-SSA LICs (42), and in Ghana (43). Only one of these studies has systematically investigated IMF policy documents in Senegal from a health sector perspective, briefly describing IMF documents for Senegal from 2004 and 2005 among data from 15 other SSA countries (25). We identified no studies systematically interrogating WB policy documents in Senegal from a health sector perspective.

The existing studies of both debt and loan conditionality impacts on government health spending consist primarily of quantitative, cross-sectional, multi-country analyses and desk-based document reviews. As such, they are limited in their ability to identify potential effect pathways and explain observed associations in the study countries.

While we explore effects of loans and ensuing debt from all external official creditors, we specifically focus on the role of IMF and WB policy recommendations and conditionalities in determining the mix of domestic health financing sources in Senegal. We focus on these two creditors because of their privileged and extensive role as the main global actors in macrofiscal policymaking historically across the Global South (44-46) and the reported influence on health financing (12, 20-22, 25-32, 34, 35) including in Senegal (33). These organisations have distinct mandates: IMF ensuring macroeconomic stability, and the WB promoting social and economic development (47-50). We therefore discuss their influence individually and comparatively as appropriate.

Our study aims to understand the influence of external official loans from External Development Partners (EDPs, including the WB) and the IMF, the ensuing debt obligations, and specifically the policy recommendations and loan conditionalities that come with IMF/WB

loans on the mix of domestic health financing sources, by performing an in-depth mixedmethods case study of Senegal.

## 8.4. Methods

## 8.4.1. Framework for mapping effect pathways

We developed a framework shown in Figure 8.2 of the potential pathways for how loans from external official creditors, the ensuing debt repayments and IMF/WB policy recommendations and loan conditionalities can affect the mix of domestic health financing sources. The framework was developed iteratively based on the available literature and economic theory, and our study findings. We used the derived general economic understanding as described below as the foundation for analysing and interpreting our data. For a glossary of the debt-related abbreviations used in this paper, see Table 10.3 (Appendix 6).



Figure 8.2: Framework illustrating how lending activities from external official creditors may affect government health spending, and how accompanying policy recommendations, targets, objectives and conditionalities may affect both central fiscal policy and health sectoral policy for the different sources of health financing.

External official creditors give loans to the recipient government as general budget support or earmarked funds to specific sectors, including the health sector. Over time, debt plus its interest accumulates and needs repayment from the general government budget. However, the government health expenditure budget is also determined by levels of Domestic Resource Mobilisation (DRM), development assistance, and political prioritisation. DRM relies in part on the amount of taxable economic activity, which is co-determined by external official lending that may help grow the economy. This results in a question of relative balance: Which sectors benefit more from development lending, and which sectors are "picking up the tab" for repaying

investments in other sectors? Importantly, the debt burden effects will depend on the terms of lending, and concessional loans will result in a lower debt servicing burden than non-concessional loans<sup>27</sup>, all things equal.

External official loans also come with associated policy recommendations, targets and objectives, as well as binding and non-binding conditionalities. IMF policy advice and loan conditionalities generally pertain to the central and structural levels such as balance of payments, fiscal deficit, tax revenue, public financial management, etc. (52). These factors co-determine a government's ability to finance its different sectors, the balance between the public and private sector, and how the government will prioritise its resources to ensure it meets IMF requirements. WB policy advice and loan conditionalities may target the central and the sectoral level, including project-level advice (53), thereby affecting government policy at all system levels.

For the health sector, these factors may in combination help determine how much of the government budget is allocated to the health sector. A larger government health budget may be able to reach more people and cover more services, resulting in a lower degree of reliance on user fees and other private health financing mechanisms in the health system, and vice versa. If IMF/WB policy recommendations and conditionalities conversely expand the role of the private health sector, this might lead to a similar expansion of private health insurance and user fees to pay for private health services. Together, these factors may affect the balance between the main domestic health financing sources: government health spending from domestic revenue (Government Health Expenditure as a Source (GHE-S)), health insurance payments and Out-Of-Pocket payments (OOP). This balance may determine the overall degree of equity in the mix of domestic health financing sources by influencing to what extent funds are prepaid, pooled and contributed progressively in accordance with ability to pay, which in turn affects health service access for the poor and associated health outcomes (54-62).

## 8.4.2. Study setting

Situated in Francophone West Africa, Senegal has been one of the most stable democracies in the region, and in Africa more broadly (63), except for political turmoil that erupted over former President Macky Sall's attempt to postpone the 2024 presidential elections (64). The country became independent from French colonial rule in 1960 (63). In 2023, the country had

<sup>&</sup>lt;sup>27</sup> Concessionality depends on the grant element of a loan and is defined by the OECD DAC as a measure of credit softness reflecting "*the benefit to the borrower compared to a loan at market rate*" (51).

a population of approximately 18 million (65) and a GDP per capita of \$1,599 in 2022 (3), resulting in a WB classification as a Lower Middle-income Country (LMC). We chose Senegal as a case country for four main reasons. A: Its political stability at the time of fieldwork, so our results were reflective of creditor-government dynamics under stable conditions. B: Its relatively low perceived corruption levels (2023 and earlier) to minimise the risk of changes in health financing occurring due to corruption and not the factors of interest to our study (66). C: A growing public external debt burden (3) (Figure 8.1 and Figure 8.4)<sup>28</sup>. D: A health financing composition increasingly characterized by user fees and less so by government contributions at the time our interviews were performed in 2019/20 (8) (Figure 8.3), inviting questions as to the determinants of this trend.





#### 8.4.3. Study design, data collection, management and analysis

This study was a mixed-methods case study comprised of a purposive document review, key stakeholder interviews together with descriptive statistical analysis of health financing contributions in Senegal. The document analysis and analysis of quantitative data was performed after the initial analysis of interview data, going back and forth between all sources of data while writing up the results.

<sup>&</sup>lt;sup>28</sup> In 2017, the Senegalese government changed its definition of public debt to also include parastatals, state-owned enterprises and central government issued guarantees, resulting in an increase in public debt per GDP of about 11% (67).

We considered the following as evidence of external official lending, public external debt servicing, and the IMF/WB influencing domestic health financing sources. A: statements from interviewees identifying a such influence or effect. B: written documentation of IMF/WB policy recommendations, targets, objectives or conditionalities directly pertaining to domestic health financing sources. C: Discernible patterns between levels and changes in external official lending, public external debt servicing and domestic health financing sources.

#### 8.4.3.1. Document review

We reviewed IMF (68, 69), WB (70) and Senegalese government websites (71-76) to identify key documents that contained information on external official borrowing, economic policy discussions, recommendations and loan conditionalities from the IMF and the WB (215 full text documents reviewed)<sup>29</sup>. We also used the London School of Hygiene & Tropical Medicine Discover platform and Google Scholar to search all academic publishing databases and collections, including PubMed, Embase, CINAHL, Cochrane, EconLit, Scopus and Web of Science, for academic literature mentioning external official lending, public external debt or loan-associated policy recommendations/conditionalities in Senegal<sup>30</sup>, both health-specific and not. 8 academic articles were identified for full-text review. From the identified loan agreements, reports and articles, we included and summarised A: the main IMF and WB policy and lending programmes in place, and B: all WB project-level loan agreements to the health sector, in Table 10.4 (Appendix 6). We did not apply any time period constraint to the document review. This was done to allow for a better understanding of how historical and more recent IMF/WB influence have helped shape health payment patterns in Senegal, and to understand any recent and current ramifications of IMF/WB lending such as debt obligations and conditionality, viewed in the light of the historical benefits to the health sector from their loans.

From included documents, we extracted the year an IMF/WB instrument or arrangement was instituted, amount borrowed under instrument/arrangement, loan duration, the main purpose and policy content, including policy recommendations, targets, objectives and binding and non-binding loan conditionalities directly relevant to domestic health financing sources (GHE-S, health insurance payments and OOP). Direct relevance to health financing was defined as a recommendation/ conditionality *directly* pertaining to the levels and balance of the different

<sup>&</sup>lt;sup>29</sup> Main policy programme documents and accompanying loan agreements reviewed. Individual WB development project loan agreement documents for non-health sectoral loans were not reviewed, while health sector specific loan agreements were reviewed.

<sup>&</sup>lt;sup>30</sup> Search string: (international OR external OR official OR development) AND (loan\$ OR credit\$ OR lend\$ OR borrow\$ OR debt OR conditionalit\$) AND Senegal. The search was carried out in March 2023 and repeated in August 2023.

domestic health financing sources and coverage of health financing schemes, and not working only indirectly through various upstream mechanisms. Documents were collected before, during and after conducting interviews in-country in Senegal, up until July 2024.

#### 8.4.3.2. Interviews

We first mapped key stakeholders including relevant donor- and creditor representatives, government representatives, academics, civil society leaders and hospital management staff. We then used snowball sampling to identify additional key informants. A total of 32 stakeholders were identified for interview of which 25 agreed to participate. Interviews for this study were conducted in Dakar (Table 8.1). Interviews were conducted between September 2019 - January 2020.

Stakeholder group	Number of participants
Bilateral development partners	4
Multilateral development partners	4
Central government*	9
Hospital management	1
Civil Society Organisations	4
Academics	3
Total interviews	25

 Table 8.1: Interview participants: \*Central government participants came from four different ministries, however the specific ministries are intentionally not mentioned to protect the anonymity of participants.

The interview method used was Semi-Structured Interviews (SSIs). We maintained the same overall interview topics<sup>31</sup> and key questions for all interviewees to enable valid comparison across stakeholders, while also tailoring some questions to better fit the interviewee's expertise. Interviews were conducted in French or English, using an interpreter for interviews in French. Interviews were recorded and transcribed by a professional transcriptionist, where consent was given for this.

Informed consent for interview was obtained from all interview participants. The participant information sheet used, consent form and an example interview topic guide can be found in the online LSHTM data repository (77) and Appendix 4. All interviewee statements were anonymised when writing up the paper. The interview data were analysed using the Framework method (78). Themes were established deductively based on the research questions as well as inductively from the interviews. The thematic framework was validated

<sup>&</sup>lt;sup>31</sup> Overarching topics: Stakeholder views and priorities for health sector financing in Senegal; EDP influence on public health budget, health sector implications of loans, debt repayments and IMF/WB programs and loan conditionalities. Please see online supplementary materials for further details (77).
by an independent researcher, agreed upon by the co-authors, and used to code all interview data. Interview data were coded using NVivo 12 (79). Data summaries and key quotations for each theme were written under each interviewee in a spreadsheet, or Framework "matrix" (78, 80), in Microsoft Excel (81). The final Framework matrix was then examined for emerging patterns which are described in corresponding results sections.

### 8.4.3.3. Quantitative data analysis

We triangulated and contextualised document and interview data for Senegal using simple, descriptive statistics of secondary data. We extracted data on Public and Publicly Guaranteed (PPG<sup>32</sup>) loan disbursements in Senegal separated by creditor type and PPG external debt stock and service from the WB (3, 6); government revenue and expenditure data from the IMF and WB (3, 82); total official development loans received by sector from the Organisation for Economic Cooperation and Development (OECD) (83); and GHE-S, External health financing (EXT), OOP and Voluntary Health Insurance (VHI) from the WHO (8).

Data was used as historically available in databases. WB data on public borrowing and debt were available from 1970-2022; IMF/WB data on government revenue and expenditure from 1994-2022; OECD CRS ODA data from 2002-2022; and WHO data on health financing from 2000-2021.

We analysed and presented the quantitative data as trends over time, comparing changes in public borrowing, debt burden and the levels and balance of different health financing sources over time, as well as the overall sectoral distribution of development lending in Senegal. This was done to A: examine the extent to which the health sector has benefited from development loans; B: explore associations between debt servicing levels and government health spending<sup>33</sup>, and C: add context to interpret and triangulate findings from other sources. For this analysis, the general hypothesis was applied of an inverse relationship between debt servicing and government health spending due to a general budget constraint. For comparison, we also explored levels and changes in government education spending to look for evidence of debt servicing impacting across social sectors or differentially between them, using data from the WB (3).

<sup>&</sup>lt;sup>32</sup> Public and publicly guaranteed borrowing includes debt owed by the government and by nongovernmental public and private debtors such as parastatal enterprises or private companies, that have their debt guaranteed for repayment by a public entity (3).

<sup>&</sup>lt;sup>33</sup> This was not done to establish causality, but rather to generate hypotheses and look for evidence supporting or challenging findings from other data sources.

## 8.5. Results

### 8.5.1. Lending and debt effects on government health spending

Results from our document review and interviews indicated that the debt in Senegal has constrained government health spending through constraining the overall government expenditure budget. However, quantitative data showed consistent real-term growth in General Government Expenditure (GGE) over the past decades, in part supported by development loans, which have also benefited the health sector. Our interviewees identified how the government prioritised economically profitable investments and servicing the public debt, resulting in limited real-term growth in GHE-S over time.

### 8.5.2. Trends in borrowing and debt

Real-term public borrowing in Senegal has increased substantially since 2000, along with overall government revenue, expenditure and GDP. GHE-S has not followed the economic expansion seen, instead showing limited growth over time. Development lending has benefited the health sector in Senegal, but debt obligations from external official lending appeared to also have had a constraining effect on government health spending.

Between 1970-2022, Senegal has borrowed approximately US\$46.7 billion<sup>34</sup> from its creditors (constant 2022 US\$) (6). Of this, approximately US\$12.6 billion has been borrowed from bilateral creditors, US\$9.2 billion from the WB IDA branch, US\$8.4 billion from other multilateral creditors, US\$6.4 billion via bonds<sup>35</sup>, US\$5.2 billion from commercial banks and other private creditors, US\$4.4 billion from the IMF<sup>34</sup>, and \$0.5 billion from the WB International Bank for Reconstruction and Development (IBRD) branch (3).

As shown in Figure 8.4, Senegalese public borrowing increased during the 1970's, leading to debt buildup, which was partly relieved in the Heavily Indebted Poor Countries (HIPC) initiative, providing 850 million US\$ in debt relief from 2000-2004 (84). Since 2000, Senegalese annual borrowing levels from foreign creditors have risen substantially, up from approximately \$0.3 billion in 2000 to \$1.7 billion in 2022 (constant 2022 US\$) (3). Increases were most pronounced from 2016 and onward, in part due to the issuance of US\$5.4 billion worth of bonds (Eurobonds) in separate tranches, beginning in 2009 (3, 85-87) (Figure 8.4). By 2022, Senegal owed a total US\$16.1 billion<sup>34</sup> to its creditors (3). Of this, \$4.1 billion was

<sup>&</sup>lt;sup>34</sup> This number includes drawings made on the IMF general resource account, not the reserve tranche, for which we were unable to find a data source. IMF disbursements and resulting outstanding credit are not counted in PPG external debt figures and were added to attain the aggregate numbers presented.

<sup>&</sup>lt;sup>35</sup> Some of the amount borrowed via bonds may be double counted in the other categories, as these creditor types may buy government bonds, however we were unable to identify more detailed data on the holders of Senegalese bonds to be able to avoid this.

owed to bondholders, US\$3.7 billion to the WB IDA, \$3.0 billion to other multilateral creditors, \$3.0 billion to bilateral creditors, \$1.6 billion to the IMF<sup>34</sup>, and \$0.8 billion to commercial banks and other private creditors (3, 6) (Figure 8.4).



Figure 8.4: Panel A: Public and Publicly Guaranteed (PPG) borrowing in Senegal separated by creditor type, 1970-2022 (billion constant 2022 US\$) (3, 6). Data are for disbursements, i.e. transfers made in each year (flows, as opposed to stock in Panel B). IMF loan disbursement data is only available for drawings made on the IMF general resource account, not the reserve tranche. Panel B: PPG debt stock in Senegal from 1970-2022 (6). IMF credit includes all IMF loans except amounts drawn against the country's reserve tranche position. It includes Special Drawing Rights (SDR) allocations (6). Note: There may be some overlap between "Bonds" and the other creditor categories as these may buy Senegalese bonds, however we were unable to identify more detailed data on the owners of Senegalese bonds to be able to avoid double counting. Panel C: Gross Domestic Product (GDP) (1970-2022), PPG external debt stock (1970-2022), and General Government Expenditure (GGE) in Senegal (1994-2022) (3, 82). Panel D: PPG external debt service (1970-2022) and

Government Health Expenditure as a Source (GHE-S) (2000-2021) in Senegal (3, 8). (Data deflated using WB US\$ GDP deflator (3). Note: PPG external debt does not include the use of IMF credit (6)).

The PPG external debt of Senegal corresponded to 52% of its GDP in 2022, well above the average for other LMCs (15%) and SSA countries (23%) (Figure 8.1) (3). In 2021, real-term PPG external debt servicing peaked at US\$1.5 billion (constant 2022 US\$), corresponding to 20% of GGE, outgrowing GHE-S by a factor of about 4.7 (Figure 8.4) (3, 8, 82).

Several interviewees across stakeholder groups were of the understanding that external official loans predominantly went to economically profitable sectors (natural resources, production, manufacturing etc.). The government and donors were described as being jointly interested in ensuring a return on investment:

"I think lending goes more into the economic sectors, and the health and social sectors are less important... The donors or partners who finance ... expect a return on investment. They have not invested much in the social, the social is not economically profitable, so many partners prefer to invest in the economic than in the social." (CSO)

A government interviewee concurred that they encouraged external official loans where these could be profitable:

"... Now more and more, we still want to guide partners to finance loans especially in areas that can be profitable". (Government)

Figure 8.5 shows the distribution of development loans received in Senegal between 2002-21 by sector (83). Health and population and reproductive health have received approximately 6.5% of all development loans, compared to 33% to economic sectors, 12% to production sectors, and 6% to the education sector (83).



Figure 8.5: Total official development loans received by sector, Senegal, 2002-22. This chart is based on OECD CRS data in constant 2022 US\$ (83). No data available before 2002.

As seen in Figure 8.6, a growing share of ODA for health is disbursed as loans in Senegal, up from 3% in 2015 to 67% in 2022 (83). This follows a general, though less marked trend for ODA across all sectors in Senegal.



*Figure 8.6: Share of official development assistance provided as loans in Senegal from 2002-2022 (83). "Health" includes both the categories "Health" and "Population policies/Programmes & reproductive health".* 

## 8.5.3. Lending directly benefiting the health sector

Although not the primary focus of ODA lending, the general observation can be made that ODA lending has benefited the health sector directly. This included the examined case of WB project lending to the health sector from the 1980's up until today (Table 10.4, Appendix 6). Identified WB project lending for the health sector has gone to health systems strengthening; family planning; infectious disease control; results-based financing for health and nutrition; establishing and expanding exemption schemes and CBHI; improving utilization of reproductive, maternal, neonatal, child and adolescent health and nutrition, and more (88-95)<sup>36</sup>.

As for the IMF, except for the case of supporting the country's Covid-19 response, IMF lending did not directly target the health sector but was instead disbursed to address balance of payment issues and support the general adjustment policy programmes in the country (Table 10.4, Appendix 6).

The observation that more ODA for health is disbursed as loans means that direct benefit to the health sector from ODA lending has increased accordingly. This has not occurred at the expense of ODA grants for health, which have gone from US\$155 million in 2015 to US\$163 million in 2022 (constant 2022 US\$), but reflects an overall expansion of ODA for health (Figure 8.3) (83).

## 8.5.4. Lending indirectly benefiting the health sector

Lending to the economic and production sectors could have indirectly supported increases in GHE-S by strengthening the foundation for raising revenue for the public budget, some of which could be spent on health. In part supported by the increased lending seen, Senegalese government revenue has grown from US\$1.4 billion in 2000 to US\$5.5 billion in 2022 (constant 2022 US\$) (Figure 8.7) (82). Senegalese GGE has grown from US\$1.4 billion in 2000 to US\$7.4 billion in 2022, while GDP has grown from \$10 billion in 2000 to \$28 billion in 2022 (constant 2022 US\$) (Figure 8.4 and Figure 8.7) (3, 82). GHE-S has however halved as a share of GGE, from 9.0% in 2003 to 4.4% in 2021 (Figure 8.7) (3, 8). Following steady increases from 2000-2005, GHE-S did not increase in real terms between 2005-2010 despite low PPG external debt servicing levels after HIPC debt relief was granted between 2000-2004 (84) (Figures Figure 8.4 and Figure 8.7). GHE-S was \$290 million in 2006 and \$280 million in 2019 (constant 2022 US\$), corresponding to a real-term decrease from \$26 per capita to \$18

<sup>&</sup>lt;sup>36</sup> Both broader WB lending under multi-sectoral development strategies as well as individual project lending has of course benefited across a multitude of sectors in Senegal (96).

per capita<sup>37</sup> (Figure 8.7) (8). This was followed by a significant, but temporary, increase to respond to Covid-19 in 2020 (Figure 8.7) (8).



Figure 8.7: Panel A: General Government Expenditure (GGE) and government revenue as a proportion of GDP (2000-2022) and Government Health Expenditure as a Source (GHE-S) as a proportion of GGE (2000-2021) in Senegal (8, 82). Panel B: GHE-S from 2000-2021 in Senegal (8). (Data deflated using WB US\$ GDP deflator (3).

There is therefore limited evidence supporting this indirect pathway of influence, although the counterfactual trend in the absence of lending is unknown.

## 8.5.5. Debt and health expenditure

Our findings pointed to a constraining effect of public debt servicing on government health spending. Government prioritisation of servicing its debt obligations was explained as receiving higher priority than funding the health sector, along with making profitable investments to ensure economic growth. The education sector appeared to suffer less from a debt servicing constraint than the health sector, possibly due to receiving higher government priority.

A constraining effect of debt on social spending in Senegal has been acknowledged in official country reports both when debt levels were highest around the turn of the millennium, and again in 2020. The 1998 Enhanced Structural Adjustment Facility (ESAF) policy framework paper recognised that "the fiscal burden [of external debt] remains high, however, with debt service to revenues projected to be 38 percent in 1998. This makes it more difficult to increase expenditures, including for the priority sectors of education and health" (97). In the 2002 Poverty Reduction Strategy Paper (PRSP), it was similarly stated that: "the country's debt burden is a major obstacle to efficient allocation of public resources in favour of the social

<sup>&</sup>lt;sup>37</sup> Numbers deflated using US\$ GDP deflator data for consistency across figures (3).

sectors" (98). In the recent context of Senegal's growing debt burden, a 2020 IMF country report for Senegal stated that: "... debt service continues to absorb a sizable portion of fiscal revenues, thus limiting room for other expenditures in critical areas such as infrastructure investment, health, and education" (99).

Approximately half of participants, representing all stakeholder groups, believed that Senegal's debt constrained the public budget, including the health budget. Some did not believe that this was the case, while others were not knowledgeable on the matter. Several interviewees across all stakeholder groups believed a constraining effect of Senegal's debt to impact across all sectors. A government official explained how the health sector was not exempt, and the government at times had to cut them to meet debt obligations.

"We live it every day... Because we are obliged to pay the debt ... we are obliged to make budgetary regulations on such sensitive sectors as health, education, even though we try to spare them." (Government)

A donor concurred:

"Absolutely, [debt obligations are] going to affect the health sector because we know that these loans ... the government of Senegal will have to respect their engagement in terms of timeline. If during this time ... they have to pay back these loans, they don't have enough resources, they have to cut somewhere. And when they cut, the health sector is also affected." (Donor)

Two CSO leaders and the cited donor representative however believed that the health sector was affected more heavily by debt constraint than other sectors.

"The IMF sets an objective to the government in term of what should be the level of debt and deficit each year. And so ... the government needs to adjust its budget accordingly and so at the end of the day it's the budget for social sectors that are cut." (Donor).

This interviewee did however attribute this to the government and highlighted the change in attitude from the WB toward prioritising social sectors.

When comparing the two main social sectors, total government education expenditure has been consistently higher than total government health expenditure, and increasing in real terms over time (domestic and external source, including capital expenditure)<sup>38</sup> (Figure 8.8)

<sup>&</sup>lt;sup>38</sup> Education and military expenditure data shown in Figure 8.8 also include government capital expenditure, which is not available for the full time series for the health sector. The figure therefore only shows total government current account health expenditure (domestic and external source). In 2020, domestic government capital health expenditure was US\$87 million, and external capital health expenditure was \$124 million, some of which may have been channelled outside of the government (constant 2022 US\$). This leads to the provided range of total government health spending being 1.7-2 times less than total government education spending in 2020.

(3, 8). No pattern was identified between year-on-year changes in PPG external debt servicing and total government current health spending.



Figure 8.8: Panel A: Selected government sectoral spending and Public and Publicly Guaranteed (PPG) external debt service in Senegal, 2000-2022 (constant 2022 US\$) (3, 8). Panel B: Year-onyear changes in government spending for same sectors and PPG external debt service in Senegal, 2001-2022 (3, 8). Figures include external transfers via government. Education and military expenditure also include government capital expenditure, which is not available for the full time series for the health sector, so figure only shows total government current account health expenditure (domestic and external source). For 2013-2020, total capital health expenditure via the government was at least US\$18-87 million (lower bound, excluding external) and maximum US\$34-211 million (upper bound, including all external, some of which may have been channelled outside of the government) (constant 2022 US\$).

In 2020 (latest comparable data), total government education expenditure was 1.7-2 times total government health expenditure (3, 8). ODA levels for health and education have been similar since 2002, except during 2021-22 when ODA for health became predominant due to

Covid-19 (83). Differences in ODA therefore do not explain differences in government spending. Across SSA, education sectors do receive more government funds than health sectors, however at about 15.5% of GGE for education and 11.1-12.1% of GGE for health, corresponding to a factor 1.3-1.4<sup>39</sup> on average (including domestic government current and capital spending, and external transfers via government) (3, 8). The Senegalese government thus appears to have given relative priority to the education sector, and any constraining effects of debt servicing present seem to still have allowed the government to provide significantly more and increasing levels of education spending compared to health spending. An academic explained that the government prioritised the education sector above debt servicing and the health sector below both:

*"Health is* [number] 7 ... *in the hierarchy of priorities in Senegal ... 1 is education, 2<sup>nd</sup> is roads, the third is debt payment."* (Academic)

This would help explain why the education budget would be less likely to be constrained by debt servicing requirements than the health sector as a result of government prioritisation.

A few interviewees highlighted certain government prioritisations made under budget constraint, driving spending away from the health sector, some of which were thought to be exacerbated by debt servicing obligations. This included a focus on sectoral visibility, profitability, a belief in economic growth as first priority before social sectors, and cutting the health sector due to the presence of donors.

An increased focus on sectoral visibility due to a debt servicing constraint on the government budget was explained by a CSO representative as politicians focusing spending where it would help secure re-election, when there was little left to spend:

"If the money mobilized to repay the debt ... returns to the donors, the government which does not have much means will prioritize ... and they are not going to put it into health. They will put it in another much more visible sector to say that "I am building an airport, I am building roads". ... I have never seen a politician during an election campaign say that "I treated so many people" ... They say: ... "I don't have a lot of money. The little I have, I put it in the sectors that allow me to be re-elected in the next elections". (CSO)

Lack of profitability of the health sector was also believed by the same interviewee to be an explanation for why the sector received less government priority and was subject to budget constraint from debt servicing obligations:

<sup>&</sup>lt;sup>39</sup> The lower bound includes all external capital health spending, while the upper bound does not, as it is not specified whether these funds are distributed via government or not.

*Q: "Do you think the health sector is more affected than other sectors in terms of being constrained to pay off debt?"* 

A: "Yes I think so ... As I said, health, the social, is not profitable. This is why we do not prioritize it." (CSO)

A CSO leader also saw health and social protection investments as receiving last priority among government expenses, exacerbated by the weight of the debt:

"When we advocate towards the ministry of the budget to put more money into health and social protection, in general they say that what remains of the budget cannot cope with all those expenses ... The fiscal space in our countries is very small ... From this budgetary space you must first pay the debt, after that the security, the sovereign expenses, the operations, before being able to give water and health to the populations ... The weight of the debt does not allow to invest in basic social services." (CSO).

Another dynamic was identified by a donor in which the presence of donors to step in and cover costs in the health sector rendered this sector more likely to be subject to budget cuts during the financial year to meet debt obligations:

"[Cuts due to debt obligations] can affect any sector, but specifically for the health sector, and the reason it's going to affect more the health sector is the government knows that the health sector will receive more donor resources compared to the others" (Donor).

# 8.5.6. IMF/WB policy recommendations and conditionalities and health financing sources

We approach this section historically, because a key finding was that both IMF and WB health financing policy recommendations and conditionalities in Senegal evolved over time. This was especially the case for the WB from the launch of Senegal's PRSPs in 2002 and onward.

Throughout Senegal's history of engagement with the International Financial Institutions (IFIs), policy recommendations, targets, objectives, and loan conditionalities from the IMF/WB have included austerity measures working to balance the government's books by increasing revenue and controlling public spending. This has had a double-sided influence on the overall scope for government health spending. In the 1980's and 1990's, the WB promoted user fees, hospital autonomisation and private health service delivery, along with increased government health spending. Especially after 2002, an increasing number of targets and safeguards were put in place from the IMF/WB to protect and increase government health spending, accompanied by a mixed health financing approach from the WB, expanding Community-Based Health Insurance (CBHI) while building the role of the private health sector. IMF/WB efforts have however failed to achieve real-term increases in GHE-S between 2006-19, and OOP has prevailed. Our findings pointed to lingering IMF/WB policy influence from the 1980's

and 90's as a possible explanation for this, as well as government priority of economic growth and servicing its debt to the IMF, WB and other creditors before investing in the health sector.

A government official explained how an evolution in IMF conditionalities had occurred:

"The conditionalities have evolved a lot since the 80's ... Now it's the benchmarks ... This question of human capital ... The monetary fund took it up because of the criticism that was made of the adjustment programs of the 80's and 90's" (Government) A CSO representative concurred for both IFIs:

"Our relations with the World Bank ... and IMF have evolved much ... The IMF come just to see if the macroeconomic framework really respects basic principles... For the World Bank [their role] is more to support priority development programmes ... The World Bank had led the government not to finance social sectors, whereas now we have seen that the World Bank supports certain health initiatives." (CSO)

A donor representative explained that while the WB had evolved toward prioritising investing in human capital through social spending, there was still a belief within the government that economic growth through investments in productive sectors and industries was more important:

"There still is that belief that first economic growth with investment. Productivity, industries, and that investing in people is nice to have when ... you can afford it. And even though that's changing a little, and we see that change from the World Bank, still it's here... I would not be surprised if within the government in Senegal, some people believe that investing in people is a waste of money... So they cut social spending." (Donor)

#### 8.5.6.1. IMF/WB influence on health financing sources before year 2002

# 8.5.6.1.1. 1966-1995: Structural adjustment, austerity and health cost sharing

Table 10.4 (Appendix 6) provides a historical overview of the IMF and WB arrangements and instruments in place in Senegal since 1966. Senegal first became a member country of the IMF and the WB in 1962 (68, 100), and the country received its first loan from the WB in 1966 (100, 101). Following economic difficulties in the late 1970's, Senegal's first IMF/WB Structural Adjustment Program (SAP) was initiated in 1979 (102-105), and structural adjustment occurred through the 1980's and 90's (33, 98, 102, 104, 105). IMF/WB prescriptions were centred around public austerity, privatization, decentralization, market liberalization, stimulating growth and exports, increasing domestic revenue mobilisation and reducing the size of the civil service (33, 102-104, 106-116).

In the early/mid 1980's, government adherence to structural adjustment measures was limited (33, 108, 116). Real current health expenditures declined by 21% between the first and second halves of the 1980's, which was scarcely attributable to IMF/WB influence given that government compliance with their recommendations was limited (33, 108, 116, 117). Figure 8.9 shows the composition of total health financing sources in Senegal in 1981 and 1989 (116). Overall little change was seen over this decade, though with a slight relative decrease in government health financing and OOP/health insurance with limited private sector and external financing expansion.



Figure 8.9: Composition of total health financing sources (capital + current) between 1981 and 1989 in Senegal (116).

From initial nonconformity, Senegal's compliance with IMF/WB policy prescriptions increased between 1985 and 1996 (33, 108, 116, 117). Public expenditures were cut, and while government current health spending increased from 4% to 5.5% of total current expenditure from 1986-96, it stagnated as a percent of GDP around 0.5% over this time period (116). In 1984 and 1987, the WB had recommended for the government to consider cost sharing (i.e. user fees) before embarking on projects to improve or expand health service coverage, because the government could not expect to be able to finance the recurrent costs of such projects (104). These recommendations continued from the WB (118), and user fees were introduced from 1985 and onward to help pay for primary health services in the context of public sector austerity (33, 108, 111). This was also facilitated by the UNICEF/WHO Bamako Initiative adopted by African health ministers in 1987 and implemented in Senegal in the 1990's. The initiative involved patients paying into so-called revolving drug funds so that facilities could recover their costs, restock drugs and locally balance their finances in a decentralised fashion (33, 114, 119-121).

A 1991 WB health project in Senegal had as an objective to "*restructure the health sector to enable it to provide improved basic health services*", which included private sector provision of family planning services (90). In 1993, the WB recommended for Senegal to develop health insurance and drug cost recovery, whilst also protecting expenditure on health "*to ensure minimum efficiency of public services*" (114, 116). This however now came with the target of significantly increasing current health expenditure out of total government expenditure from 4.1% in 1989/90 to 6.7% in 1995 (116). An academic interviewee commented that:

"The World Bank published a document, that inspired our government which is "Investing in Health". Otherwise, the money spent in health was perceived as expenditure and not investment ... When they said "investing in health", they started pushing the government to invest more money in primary health care which was good ... Senegal took over from UNICEF the funding of the immunization program, which was for us a big success, started putting more money in programs like malaria". (Academic)

## 8.5.6.1.2. 1995-2002: Increased government health spending, mixed health insurance schemes, health cost sharing and hospital autonomisation

From 1995 up until around 2002, policy recommendations under the IMF Enhanced Structural Adjustment Facility (ESAF), Poverty Reduction and Growth Facilities (PRGF) and WB Country Assistance Strategy (CAS) focused on similar means of public sector austerity and private sector liberalisation. However there was now a new emphasis, particularly from the WB, on equity, poverty reduction, promoting the status of women, environmental and social sustainability (97, 103, 109, 122-126). IMF and WB recommendations now targeted an increase in current budget share allocation to health from 6% in 1996 to 9% in 2002, though with greater use of beneficiaries (i.e. patients) and NGOs in the financing of basic health care services (97, 109, 126). The IMF recommended a mixed health sector financing approach, with increased government financing emphasizing primary health care and improved access of vulnerable groups to health services, combined with decentralized cost sharing, formal sector health insurance, CBHI, and private sector health insurance (97, 122-124, 126).

In 1998, the government enacted a national hospital reform where hospitals were made autonomous and user fees for their services were introduced (127, 128), described by World Bank staff as: "*Senegal has virtually privatized its hospitals*" (128). An academic commented:

"The spirit of the World Bank was to tell governments: "Stop spending for the hospitals and trying to get free services at the hospital level. Let people who can pay, pay. Let people who are poor be organized in community insurance, that is the mutuals." (Academic) IMF/WB promotion of Mutuelles was corroborated in an ESAF policy framework paper from 1998 (97). A hospital management staff member in Dakar believed that past IFI influence on the health service payment pattern in the 1990's had lingered on:

"... The international institutions [mentions IMF] encouraged the state to reform the health system. And it is this health system reform that has precisely recommended payment for healthcare by the population with financial autonomation of hospitals... Still the same rhythm of payment for population health care at their own expenses [is there today] ... in terms of who pays there is no evolution" (hospital management staff member).

Besides the above policy recommendations, no specific IMF/WB conditionality for health financing sources was identified until 2007 (below section).

Senegal's HIPC program launched in 2000, with debt relief criteria consisting of privatisation of public sector enterprises and liberalisation of the private sector, increased DRM, along with social objectives including increasing primary health centre utilization (84, 129).

#### 8.5.6.2. IMF/WB influence on health financing sources from 2002 and onward

## 8.5.6.2.1. 2002-2010: Increased government health spending, debt relief, and mixed health insurance schemes

Starting in 2002, Senegal's PRSPs were launched. PRSPs I and II ran from 2002-2010, supported by IMF loans via the PRGF and WB IDA loans and grants via Poverty Reduction Support Credits (PRSCs). These had a similarly austere fiscal policy stance as previously, but with a series of stipulations to support the health sector. This included increasing budget allocations to health, allocating HIPC funds to the health sector, improving access for vulnerable groups to health insurance schemes, expanding CBHI and formal sector health insurance mechanisms as well as the role of the private sector in health service delivery (98, 130-138). They also involved subsidizing health care and establishing a health risk protection system for vulnerable groups, reducing the financial burden of payment for health services for the poor, and expanding health insurance cover for formal sector employees and dependents (135, 136, 138-144). The overall aim was for 50% of the population to be covered by health insurance by 2015, compared to 8% in 2005. Accompanying conditionalities in loan agreements directly pertaining to health financing included a WB indicative trigger (nonbinding) to extend a pilot of health insurance for transport and agricultural sector workers (136); and triggers (non-binding) and prior actions (binding) to prepare and implement a state financial support mechanism for health mutuals (CBHI) and supporting 50 new health mutuals (144). PRSC I (2004) and II (2005) also contained triggers and prior actions (PAs) to decentralize budget execution in the health sector (135, 138). Between 2003-2006, the budget execution rate for the health sector increased from 73% to 94% (135), which is a direct way

of increasing GHE-S within the existing budgetary framework, and GHE-S did increase substantially in that time period (Figure 8.7).

PRSC III (2007) had as objectives to increase state subsidy of essential drugs and services, and government-enforced stronger controls on drug tariffs, notably in remote regions (136). Prior actions (binding conditionality) under PRSC III required a harmonized tariff system to control tariffs on drugs, in order to lower the burden of payment on the poor, taking steps to expand CBHI, and establishing public subsidies for essential health services for vulnerable groups (136). It also included an indicative trigger (non-binding) requesting an extension of a pilot program providing health insurance to uninsured workers in the transport and agricultural sectors, working to increase prepaid and pooled financing for these groups, and an indicative trigger requesting enhancement of contracting with the private sector in health service delivery (136). A CSO leader was critical of IMF/WB promotion of the private sector in health due to the difficulties for poor people to pay for private health services:

"[The World Bank and IMF], their ideology is first to say less state. The state must reduce its market intervention and guarantee more space for the private sector... [But] it is not the job of the private sector to provide basic social services ... to take care of those who don't have the resources to enter the market... It is to generate profits for its shareholders, and therefore ... there is a risk of not having quality social services or a good geographical distribution" (CSO).

## 8.5.6.2.2. 2010-onward: Austerity, safeguarding health spending, exemption schemes and CBHI

In the 2010's, IMF fiscal policy objectives, articulated through Policy Support Instruments (PSIs, non-financing instruments), were of a similar nature as previously, however the first social spending floor (non-binding) was introduced from 2010 at 35% of total government spending (145). This would remain until 2022 where it was increased to 40% (146). Senegal consistently met and frequently exceeded the social sector spending floors by a few percentage points. However, since 2010, the Senegalese government has spent between 19% and 26% of its total budget on education, so it has met these floors primarily by funding the education sector (3). Fiscal prudency measures were recommended by the IMF to increase fiscal space for social spending (147), and frontline health workers were exempted from a wage bill freeze (148).

The third IMF/WB PRSP running from 2013-17 aimed to establish UHC including through health mutuals, extend free healthcare to vulnerable groups, set up a health solidarity fund, target 95% health budget execution, and increase the health mutual scheme coverage rate to

66% by 2017 (149, 150). WB conditionality gradually moved away from making stipulations for the health (and other) sectors and instead simply required that the WB IDA was satisfied with the existing development programme and macroeconomic policy framework in place (e.g. (151)).

Promotion of a mixed health financing approach in PRSPs was found to align with the government's overall development strategy for the health sector (152-154).

From 2020 onward, the IMF disbursed loans via different facilities for the Covid-19 response, accompanied by a Policy Coordination Instrument (PCI) (non-financial), jointly supporting increased health spending while increasing the social spending floor to 40% of total government spending (99, 146, 155-160). A temporarily increased fiscal deficit was allowed to respond to the pandemic before fiscal consolidation was again recommended in 2023 (161). The WB also supported the Senegalese emergency health system response to Covid-19 and the country's UHC programme, aiming to improve its progressiveness, expand coverage, expand grants/subsidy under CBHI (*"Mutuelles"*) for poor and vulnerable groups, and expand maternal health vouchers (*"Gratuités"*) (67, 93, 94, 162).

In spite of these policy changes over time, an academic explained that governments would still limit investments in the health sector due to the general austerity requirements from the IFIs:

"All the countries that work with [the IMF and WB], countries that are in economic difficulties, that are being asked to cut certain positions... Even if they don't cut for health, they will no longer invest in this sector." (Academic)

## 8.6. Discussion

With this study, we aimed to unpack the ways in which external official lending, debt and associated IMF/WB policy influence have affected the different domestic health financing sources in Senegal. We found that Senegal has gone through a journey in this regard, containing the following phases:

The 1980's and '90's were characterized by relatively high public borrowing, a high debt burden, and IMF/WB austerity prescriptions, user fee promotion, and promotion of hospital autonomisation and private health service delivery, though also with recommendations to increase government health spending. Our findings lended support to the hypothesis that some of the prevailing issues in Senegal with relatively low progressivity in the mix of health financing sources in Senegal in part stem from IMF/WB influence from this era, in spite of an ensuing evolution in policy stance, particularly from the WB.

The 2000's were characterized by debt relief, low public borrowing, ongoing IMF fiscal austerity measures though with a series of measures to support the health sector, increasingly progressive health financing policy influence from the WB, IMF/WB-promoted CBHI expansion, and growing GHE-S in the first half of the decade followed by decreasing budget priority and real-term stagnation. WB binding conditionality for decentralization of health sector budget execution in 2004-05 led to strong increases in health budget execution rates (135, 138). These efficiency gains supported significant real-term increases in GHE-S between 2004-06. The decreasing government budget priority in the latter half of this decade pointed to government prioritisation as an important co-determinant for why real-term GHE-S began to stagnate. The major debt relief granted did not translate into GHE-S increases in real terms when debt servicing levels were low following debt relief, and it seems an opportunity was lost when the country could have made significant strides towards more equitable payment patterns in the health sector.

The 2010's were characterized by marked increases in public borrowing, debt stock and debt servicing, and in spite of a series of policy recommendations and conditionalities from the IMF to protect and increase government social and health spending, and increasingly progressive WB health financing policy influence, no real-term gains were seen in GHE-S, and OOP prevailed. In 2020, a temporary increase in GHE-S was seen to respond to Covid-19, partially reverted in 2021.

While IMF policies to some extent and especially WB policies pertaining to health financing have evolved over the past two decades, it seems they have been somewhat ineffective in transforming the health payment patterns in Senegal. Government priority of profitable investments, economic growth and debt servicing, including to the IMF and WB and following their recommendations, as well as ongoing austerity measures and lingering influence from IMF/WB-promoted health system reforms in the 1980's and 1990's, were identified as possible explanations for why the degree of progressivity in the mix of domestic health financing sources has not improved much over time. The explanation of government priority of the education sector before debt servicing requirements also pointed to a differential impact of debt servicing obligations between the two sectors due to this government prioritisation. While investments in education could help support economic growth and thereby the revenue basis for GHE-S, such synergies were not reflected in trends in GHE-S over time in the period of study. They could however materialise in the future, provided GHE-S receives increasing budget priority out of GGE.

Going forward, if the cycle continues of lending primarily for economic growth and meeting increasing debt obligations before investing in the health sector, the outlook for a revitalisation of the budget priority given to the health sector looks somewhat limited.

Separating the different pathways of influence identified (Figure 8.2), external official lending was associated with a continued expansion of overall government revenue and expenditure, and development loans supported the health sector directly, although to a much lesser extent than economic and productive sectors. However, the majority of participants who held knowledge on the matter, across all stakeholder groups, believed that Senegal's ensuing debt burden constrained the public budget, including the health budget. Official IMF and WB documents from 1998, 2002 and 2020 supported this assessment (97-99). Given the consistent growth of Senegalese GGE (Figure 8.7), these observations should be interpreted as relative considerations of public expenditure growth rates, and the counterfactual of GGE in the absence of external official lending is not known.

Our findings indicating that Senegal's debt obligations had a constraining effect on government health spending extends some previous findings by other authors. Two studies of the period 1970-2000 found that the Senegalese government spent around 20 percent of its resources on debt servicing (163, 164), and that a one percent increase in debt servicing was associated with a reduction in total domestic expenditure by 0.13 percent of GDP (163, 164). Our findings also align well with multi-country econometric studies, generally finding negative associations between debt variables and government health spending (11, 13-16, 18, 165) (one study has had more variable findings (17)).

Whether IMF/WB policy influence has historically constrained or supported government health spending has been a subject of heated debate with academics and civil society on one side and IMF/WB staff on the other (12, 20, 25-43). Our results point to a mixed effect, in part due to lingering influences from past policies, and due to ongoing austerity policies on one side and efforts to increase GHE-S on the other. The ongoing WB promotion of CBHI and private health service delivery may also have limited the scope for progress on UHC in Senegal. Issues with CBHI in Senegal have included voluntary enrolment with limited reach, user copayment, small risk pools, variable financial sustainability and risk protection of poor members, and limited cross subsidy from rich to poor (166-171). In the general literature, private health service delivery has been associated with increased service costs, inequitable availability and access patterns for poor and vulnerable groups, high user co-payments and failure to advance financial risk protection (172-174). This mixed health financing approach is however in line with the government's overall development strategy for the health sector (152-154).

Our finding that WB policy had become more progressive over time in Senegal seems to follow a general evolution within the WB (175-177). A lesser degree of change was seen at the IMF in Senegal, which aligns somewhat with findings across countries from other authors (44). However, our findings of social spending floors being raised over time, funding to respond to Covid-19, and numerous measures to increase and safeguard social and health spending, did show some signs of a transition from traditional austerity policy towards increasing social concern at the IMF.

We found that social spending floors were consistently met or exceeded, which is not the norm across West Africa, though mainly by prioritising the education sector (44, 161). An issue with IMF social spending floors is that the IMF defines social spending in Senegal very broadly, covering a range of sectors and purposes (161). In other countries, IMF definitions of social spending vary widely, and the health sector is not always included or explicitly mentioned (178). This means that countries can focus their spending heavily on variable subsets of social purposes and satisfy this non-binding target, which somewhat dilutes the meaning of a social spending floor from the perspective of individual sectors, including the health sector. In a traditional economic growth logic, investments in education, water supply, electrification and sanitation could receive priority over investing in the health of poor and vulnerable groups.

Finally, it deserves mention that while this paper focused on the modality of development lending, we acknowledge the important EDP contribution of providing grant financing. In spite of its decreasing share over time, it has provided 82% of ODA for the health sector in Senegal on average between 2002-2022 (83). This is included in the assessment of ODA impacts on domestic health financing in Chapter 7.

## 8.6.1. Limitations

Unpacking our macroeconomic and fiscal dynamics of interest is inherently complicated, with complex causal chains, numerous contemporaneous determinants and competing political considerations. This served as the motivation for our multipronged methodological approach and led us to focus more on identifying and discussing different pathways of effect while remaining cautious about drawing causal inferences.

Using descriptive statistics based on relatively short observational time series to quantitatively assess sectoral impacts from cross-sector fiscal level burdens proved challenging. This is an inherent statistical limitation when using single-country observational macroeconomic data, further complicated by the fact that debt impacts the general public budget before sectoral

allocations. Using complementary data sources such as individual stakeholder accounts or written documentation of fiscal policy influences were necessary to discuss possible causal relationships.

This study focused on policy influence from the IMF/WB, however there are of course other external official creditors to Senegal, multilateral and bilateral. The influence mechanisms from the main bilateral EDPs in Senegal have been explored in Chapter 6, however an in-depth exploration of loan conditionalities from e.g. the French Development Agency or the African Development Bank might reveal different results, potentially expanding and amending the framework and understanding presented in this Chapter.

In interviews, five participants did not consent to recording of the interview or quoting them, so instead FF took notes and used these interviews to broadly further his understanding of the research questions and the interviewee's position in regard hereto. Also, potential interviewees in one government agency, two hospitals, two multilateral organisations and two academics did not respond or were unavailable. However, as seen in Table 8.1, our interview group for this study was overall well balanced, with 8 development partners, 9 government representatives, and 8 non-government representatives.

## 8.7. Conclusions and future research

While development lending has benefited the health sector in Senegal, the health sector should be protected from unprecedented public external debt servicing levels. This responsibility falls both on domestic policy makers, EDPs and the IMF. Enough flexibility should be made in repayment timelines to avoid the government having to consider cutting social sectors in order to make repayments. Exploring options for debt relief to address the rapidly growing debt burden of Senegal may become necessary, however our findings indicate that lower debt servicing levels do not equate increased health spending without political priority. Mechanisms such as debt-to-health swaps could be considered to ensure the health sector would benefit from any funds made available through debt relief [23]. This could also help level out the persistent and growing imbalance between the health and education sectors seen and prevent the health sector from losing out behind more economically profitable sectors.

In spite of increasing efforts and ambitions, non-binding targets under IMF/WB programs have failed to achieve adequate levels of GHE-S. While especially WB health financing policy has become more progressive over time and clearly supports GHE-S increases, impacts of past policies of cost recovery and hospital autonomisation seem to linger and shape the inequitable

payment patterns seen in Senegal. As long as economic growth and servicing its now historical debt levels are top priorities for the government to meet its obligations to the IFIs and other creditors before investing in the health sector, substantive real-term GHE-S gains and expansion of public-sector driven UHC expansion appear unlikely. To break this cycle, along with considering debt relief mechanisms, a legally binding agreement within existing broader fiscal and development frameworks between the Senegalese government, the IMF, WB and other partners could be considered to boost health spending toward internationally agreed upon targets such as the Abuja pledge of 15% of the government budget being allocated to the health sector (179), ideally from domestic revenue. Within the highly variable IMF social spending floors, sectoral floors for the health and education sectors of at least 10% of the government expenditure budget each, from domestic revenue, could be considered. If the Abuja pledge was to be taken seriously, it would be closer to 15% for the health sector, while a more pragmatic floor would lie somewhere in between. Country ownership and buy-in would be essential in this discussion, to avoid such policy changes becoming coercive. Such measures should hopefully allow for an expansion of well targeted state subsidy of health services through exemption schemes and CBHI premium coverage, or alternatively a move toward broader, integrated risk protection schemes and decreased emphasis on the private health sector with its reliance on co-payments and inequitable access patterns (172-174).

Future research avenues include exploring the lending and debt implications for the health sector in other countries using qualitative and mixed methods to understand individual country realities. Our framework and discussion of influence pathways could be used as a basis for this in other contexts, with modifications as necessary. Such case studies could focus on countries at high risk of or in debt distress, also exploring how implications of debt translate at different levels of the health system. Importantly though, keeping in mind the range of beneficial effects of concessional development lending, which had also benefited the health sector in Senegal, is essential for a nuanced and complete understanding of the health financing implications of development lending.

Another research pathway would be to model impacts of policy measures such as debt relief, debt-to-health swaps and setting health spending floors on health financing and service coverage. The proposed research should aim to identify the most effective development finance policy levers for expanding equitable health financing and making progress toward UHC.

## References

<sup>1.</sup> World Bank. Heavily Indebted Poor Countries (HIPC) Initiative. 2018 [cited 2022 February 3]. Available from: <u>https://www.worldbank.org/en/topic/debt/brief/hipc</u>.

Abbasi K. The World Bank and world health: Changing sides. Bmj. 1999;318(7187):865-9. 2.

World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: 3. https://data.worldbank.org/.

IMF. Debt Relief Under the Heavily Indebted Poor Countries (HIPC) Initiative. 2021 [cited 2021 October 4. 12]. Available from: https://www.imf.org/en/About/Factsheets/Sheets/2016/08/01/16/11/Debt-Relief-Under-the-Heavily-Indebted-Poor-Countries-Initiative.

IMF. Multilateral Debt Relief Initiative — Questions and Answers. 2017 [cited 2021 December 20]. 5. Available from: https://www.imf.org/external/np/exr/mdri/eng/mdrians.htm#q01.

World Bank. International Debt Statistics. 2024 [cited 2024 April 24]. Available from: 6. https://databank.worldbank.org/source/international-debt-statistics#

World Bank. World Bank Country and Lending Groups. 2024 [cited 2024 April 23]. Available from: 7. https://datahelpdesk.worldbank.org/knowledgebase/articles/906519-world-bank-country-and-lending-groups.

WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from: 8. http://apps.who.int/nha/database.

Birungi C, Azcona JA, Munevar D. A pandemic triad: HIV, COVID-19 and debt in low- and middle-9. income countries. Afr J AIDS Res. 2022;21(2):110-22.

10. Federspiel F, Borghi J, Martinez-Alvarez M. Growing debt burden in low- and middle-income countries

during COVID-19 may constrain health financing. Glob Health Action. 2022;15(1):2072461. 11. Fosu AK. The External Debt-Servicing Constraint and Public Expenditure Composition. Helsinki, Finland: UNU-WIDER; 2007.

12 Stubbs T, Kentikelenis A. Targeted social safeguards in the age of universal social protection: the IMF and health systems of low-income countries. Critical Public Health. 2017;28(2):132-9.

Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 13. 2007:8(4):299-310.

14 Shabbir S, Yasin HM. Implications of Public External Debt for Social Spending: A Case Study of Selected Asian Developing Countries. Lahore Journal of Economics. 2015;20(1):71-103.

Lahiani A, Mtibaa A, Gabsi F. Fiscal Consolidation, Social Sector Expenditures and Twin Deficit 15. Hypothesis: Evidence from Emerging and Middle-Income Countries. Comp Econ Stud. 2022;64(4):710-47. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical 16.

determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

Behera DK, Dash U. Impact of macro-fiscal determinants on health financing: empirical evidence from 17. low-and middle-income countries. Glob Health Res Policy. 2019;4:21.

18. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

Kimalu PK. Debt relief and health care in Kenya. World Institute for Development Economics (UNU-19. WIDER). 2002;WIDER Discussion Papers(No. 2002/65, ISBN 9291902551, UNU-WIDER, Helsinki).

20. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

Akazili J, McIntyre D, Kanmiki EW, Gyapong J, Oduro A, Sankoh O, et al. Assessing the catastrophic 21. effects of out-of-pocket healthcare payments prior to the uptake of a nationwide health insurance scheme in Ghana. Glob Health Action. 2017;10(1):1289735.

Mhazo AT, Maponga CC. The political economy of health financing reforms in Zimbabwe: a scoping 22. review. Int J Equity Health. 2022;21(1):42.

Okuonzi SA, Macrae J. Whose policy is it anyway? International and national influences on health policy 23 development in Uganda. Health Policy Plan. 1995;10(2):122-32.

Kusi-Ampofo O, Church J, Conteh C, Heinmiller BT. Resistance and change: a multiple streams 24. approach to understanding health policy making in Ghana. J Health Polit Policy Law. 2015;40(1):195-219. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government 25. health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

26. Ooms G, Schrecker T. Expenditure ceilings, multilateral financial institutions, and the health of poor populations. The Lancet. 2005;365(9473):1821-3.

Center for Global Development. Does the IMF Constrain Health Spending in Poor Countries? . 2007 27. [cited 2024 March 19]. Available from: https://www.cgdev.org/sites/default/files/14103\_file\_IMF\_report.pdf.

Brunswijck G. Unhealthy Conditions. Eurodad; 2018. 28.

29. Stuckler D, Basu S. The International Monetary Fund's effects on global health: before and after the 2008 financial crisis. Int J Health Serv. 2009;39(4):771-81.

Stuckler D, Basu S. The Body Economic: Why Austerity Kills: Basic Books; 2013. ISBN: 978-30 0465063987

31. Abocejo F. The Impact of International Monetary Fund (IMF) Structural Adjustment Policies (SAP) on the Philippines. Recoletos Multidisciplinary Research Journal. 2014;2(1):19-24.

Isiani MC, Anthonia Obi-Ani N, Obi-Ani P, Chidume CG, Okoye-Ugwu S. Interrogating the International 32. Monetary Fund (IMF) Policies in Nigeria, 1986–2018. Cogent Arts & Humanities. 2021;8(1).

33. Foley EE. Your pocket is what cures you: The Politics of health in Senegal: Rutgers University Press; 2010. ISBN: 978-0-8135-4667-4

34. Meurs M, Seidelmann L, Koutsoumpa M. How healthy is a 'healthy economy'? Incompatibility between current pathways towards SDG3 and SDG8. Global Health. 2019;15(1):83.

35. Birn AE, Nervi L, Siqueira E. Neoliberalism Redux: The Global Health Policy Agenda and the Politics of Cooptation in Latin America and Beyond. Development and Change. 2016;47(4):734-59.

36. Clements B, Gupta S, Nozaki M. What happens to social spending in IMF-supported programmes? Applied Economics. 2013;45(28):4022-33.

37. Gupta S. Can a causal link be drawn? A commentary on "the impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations". Social science & medicine (1982). 2017;181:199-201.

38. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs. International Monetary Fund; 2004.

39. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

40. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

41. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

42. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

43. Boachie MK, Agyemang J, Immurana M. Health sector funding in Ghana: The effect of IMF conditionalities. Dialogues Health. 2022;1:100045.

44. Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A, Stubbs T, editors: Oxford University Press; 2023. 0 p. ISBN: 9780190637736

Millet D, Toussaint E. Who Owes Who? 1 ed: Zed Books; 2004. 185 p. ISBN: 1842774271
 Cornia GA, Jolly R, Stewart F. Adjustment with a human face, Vol. 1. Oxford [Oxfordshire]; Oxford

[Oxfordshire]; New York: Clarendon Press; Oxford University Press; 1987. ISBN: 0198286104

47. IMF. What is the IMF? 2024 [cited 2024 May 20]. Available from:

https://www.imf.org/en/About/Factsheets/IMF-at-a-Glance.

48. (IMF) IMF. Articles of agreement. 2020 [cited 2024 June 22]. Available from: <u>https://www.imf.org/external/pubs/ft/aa/pdf/aa.pdf</u>.

49. International Development Association. Articles of Agreement. 1960 [cited 2024 June 22]. Available from: https://thedocs.worldbank.org/en/doc/2a209939e876fdcd0d957036daebff6e-0410011960/original/IDA-Articles-of-Agreement-English.pdf.

50. International Bank for Reconstruction and Development (IBRD). Articles of Agreement. 2012 [cited 2024 June 22]. Available from: <u>https://thedocs.worldbank.org/en/doc/722361541184234501-</u>

0330022018/original/IBRDArticlesOfAgreementEnglish.pdf.

51. OECD. DAC Glossary of Key Terms and Concepts. 2023 [cited 2023 December 18]. Available from: https://www.oecd.org/dac/dac-glossary.htm.

52. IMF. IMF conditionality. 2023 [cited 2023 August 22]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/IMF-Conditionality.

53. World Bank. Advice and Analytics. 2023 [cited 2021 November 21]. Available from:

https://www.worldbank.org/en/what-we-do/products-and-services/advisory-services.

54. Ataguba JE, Asante AD, Limwattananon S, Wiseman V. How to do (or not to do) ... a health financing incidence analysis. Health Policy Plan. 2018;33(3):436-44.

55. Makinen M, Waters H, Rauch M, Almagambetova N, Bitran R, Gilson L, et al. Inequalities in health care use and expenditures: Empirical data from eight developing countries and countries in transition. Bulletin of the World Health Organization. 2000;78(1):55-65.

56. Xu K, Evans DB, Carrin G, Águilar-Rivera AM, Musgrove P, Evans T. Protecting households from catastrophic health spending. Health Aff (Millwood). 2007;26(4):972-83.

57. WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available from: https://www.who.int/publications/i/item/9789241564021.

58. Chuma J, Okungu V. Viewing the Kenyan health system through an equity lens: implications for universal coverage. Int J Equity Health. 2011;10:22.

59. Moreno-Serra R, Smith PC. Broader health coverage is good for the nation's health: evidence from country level panel data. J R Stat Soc Ser A Stat Soc. 2015;178(1):101-24.

60. Moreno-Serra R, Smith P. The Effects of Health Coverage on Population Outcomes: A Country-Level Panel Data Analysis. Results for Development Institute. 2011.

61. Qin VM, Hone T, Millett C, Moreno-Serra R, McPake B, Atun R, et al. The impact of user charges on health outcomes in low-income and middle-income countries: a systematic review. BMJ Glob Health. 2018;3(Suppl 3):e001087.

62. Martinez-Alvarez M, Federspiel F, Singh NS, Schaferhoff M, Lewis Sabin M, Onoka C, et al. Equity of resource flows for reproductive, maternal, newborn, and child health: are those most in need being left behind? BMJ. 2020;368:m305.

63. CIA. Senegal. 2022 [cited 2022 January 31]. Available from: <u>https://www.cia.gov/the-world-factbook/countries/senegal/</u>.

64. Amy Niang. 2024 Senegal election crisis points to deeper issues with Macky Sall and his preferred successor. 2024 [cited 2024 April 24]. Available from: <u>https://theconversation.com/2024-senegal-election-crisis-points-to-deeper-issues-with-macky-sall-and-his-preferred-successor-223035</u>.

65. Agence nationale de la statistique et de la demographie (ANSD). Le Sénégal En Bref. 2023 [cited 2023 August 11]. Available from: <u>https://www.ansd.sn/</u>.

66. Transparency International. Corruption Perception Index. 2024 [cited 2024 April 24]. Available from: https://www.transparency.org/en/cpi/2023/index/sen.

67. World Bank. Country partnership framework for the Republic of Senegal for the period FY20-FY24. 2020 [cited 2023 September 18]. Available from: <a href="https://documents.worldbank.org/en/publication/documents-reports/documentdetail/608781583719225540/senegal-country-partnership-framework-for-the-period-fy20-fy24">https://documents.worldbank.org/en/publication/documents-reports/documentdetail/608781583719225540/senegal-country-partnership-framework-for-the-period-fy20-fy24</a>.

IMF. Senegal. 2023 [cited 2023 September 25]. Available from: <u>https://www.imf.org/en/Countries/SEN</u>.
 IMF. IMF eLIBRARY: Senegal. 2023 [cited 2023 October 10]. Available from:

https://www.elibrary.imf.org/subject/SN.

70. World Bank. Open Knowledge Repository. 2023 [cited 2023 October 10]. Available from: https://openknowledge.worldbank.org/.

71. Agence nationale de la statistique et de la demographie (ANSD). ansd.sn. 2024 [cited 2024 July 25]. Available from: <u>https://www.ansd.sn/index.php</u>.

72. Ministère de l'Economie du Plan et de la Coopération. economie.gouv.sn. 2023 [cited 2023 October 10]. Available from: <u>https://www.economie.gouv.sn/</u>.

73. Ministère des Finances et du Budget. Direction de la Dette Publique (DDP). 2023 [cited 2023 October 10]. Available from: <u>https://www.sentresor.org/structures-du-tresor/directions-administratives-da/direction-de-la-dette-publique-ddp/presentation-de-la-ddp/</u>.

74. Ministère des Finances et du Budget. senfinances.org. 2023 [cited 2023 October 10]. Available from: https://www.senfinances.org/.

75. Agence de la Couverture Maladie Universelle. agencecmu.sn. 2024 [cited 2024 July 25]. Available from: https://agencecmu.sn/.

76. Ministère de la Santé et de l'Action Sociale. <u>https://sante.gouv.sn/</u>. 2024 [cited 2024 September 19]. Available from: <u>https://sante.gouv.sn/</u>.

77. London School of Hygiene and Tropical Medicine. LSHTM Data Compass. 2023 [cited 2023 November 14]. Available from: <u>https://datacompass.lshtm.ac.uk/</u>.

78. Ritchie J, Spencer L. Qualitative Data Analysis for Applied Policy Research. In: Bryman A, Burgess R, editors. Analyzing Qualitative Data. 1st ed: Routledge; 1994.

79. QSR International Pty Ltd. NVivo qualitative data analysis software. Version 12, 2020.

80. Ward DJ, Furber C, Tierney S, Swallow V. Using Framework Analysis in nursing research: a worked example. J Adv Nurs. 2013;69(11):2423-31.

81. Microsoft® 2024. Microsoft® Excel for Mac, version 16.81.

82. IMF. Public Finances in Modern History. 2024 [cited 2024 June 10]. Available from: https://www.imf.org/external/datamapper/datasets/FPP.

83. OECD. Creditor Reporting System (CRS). 2024 [cited 2023 February 16]. Available from: https://stats.oecd.org/Index.aspx?DataSetCode=CRS1.

84. IMF, World Bank. Senegal: Enhanced Heavily Indebted Poor Countries (HIPC) Initiative Completion Point Document. 2004 [cited 2023 September 11]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Enhanced-Heavily-Indebted-Poor-Countries-HIPC-Initiative-Completion-Point-Document-17380.

85. OECD Development Co-operation Directorate. A Recipient Perspective on TOSSD: The Case of Senegal. 2017 [cited 2020 November 13]. Available from: <u>http://www.oecd.org/dac/financing-sustainable-</u>

development/development-finance-standards/A%20recipient%20perspective%20on%20TOSSD\_Senegal.pdf.
 86. Wallace P, Pronina L, Ciss M. Senegal Attracts Almost \$10 Billion in Eurobond Bids. 2018 [cited 2020 November 13]. Available from: <a href="https://www.bloomberg.com/news/articles/2018-03-06/senegal-is-said-to-attract-almost-10-billion-in-eurobond-bids">https://www.bloomberg.com/news/articles/2018-03-06/senegal-is-said-to-attract-almost-10-billion-in-eurobond-bids</a>.

87. Reuters. Senegal issues 775 million euros in Eurobonds. 2021 [cited 2023 August 23]. Available from: https://www.reuters.com/article/ozabs-uk-senegal-eurobond-idAFKCN2DF0L7-OZABS.

88. World Bank. Development Credit Agreement (Rural Health Project). 1983 [cited 2024 July 17]. Available from: <a href="https://documents1.worldbank.org/curated/en/754991468058791015/pdf/Credit-1310-Senegal-Rural-Health-Project-Credit-Agreement.pdf">https://documents1.worldbank.org/curated/en/754991468058791015/pdf/Credit-1310-Senegal-Rural-Health-Project-Credit-Agreement.pdf</a>.

89. World Bank. Development Credit Agreement (Integrated Health Sector Development Program). 1997 [cited 2024 July 17]. Available from:

https://documents1.worldbank.org/curated/en/532951468304777075/pdf/3A5F3840AED9BD9C85256F0200824F 92.pdf

90. World Bank. Human Resources Development Project. 1991 [cited 2024 July 16]. Available from: https://documents1.worldbank.org/curated/en/933341468107951172/pdf/10B0ADAAC91EB7F685256F02008245 19.pdf.

91. World Bank. Development Credit Agreement (HIV/AIDS Prevention and Control Project). 2002 [cited 2024 July 17]. Available from:

https://documents1.worldbank.org/curated/en/151421468106481346/pdf/B74DF895D60239A785256F03001850 14.pdf 92. World Bank. Health and Nutrition Financing Project. 2014 [cited 2024 July 15]. Available from: https://documents1.worldbank.org/curated/en/435611468168568721/pdf/RAD2073336869.pdf.

93. World Bank. Financing agreement: Investing in Maternal, Child, and Adolescent Health Project). 2020 [cited 2024 July 15]. Available from:

https://documents1.worldbank.org/curated/en/549231586273646943/pdf/Official-Documents-Financing-Agreement-for-Credit-6495-SN.pdf.

94. World Bank. Financing Agreement: Additional Financing for the Senegal COVID-19 Response Project. 2021 [cited 2024 July 15]. Available from:

https://documents1.worldbank.org/curated/en/745311624664202803/pdf/Official-Documents-Financing-Agreement-for-Additional-Financing-Credit-No-6964-SN-and-Grant-No-D8890-SN.pdf.

95. Ministère des Finances et du Budget. Projet de loi de finances rectificative pour l'annee 2021. 2021 [cited 2024 September 20]. Available from: <u>https://budget.sec.gouv.sn/recherche-par-types-documents/loi-de-finances-rectificatives</u>.

96. World Bank. Documents & Reports. 2024 [cited 2024 July 19]. Available from:

https://documents.worldbank.org/en/publication/documents-

<u>reports/documentlist?countcode=SN&srt=docdt&order=desc&docty\_exact=Credit+Agreement</u>^Loan+Agreement^ Project+Agreement^Financing+Agreement.

97. Republic of Senegal, IMF, World Bank. Senegal: Enhanced Structural Adjustment Facility Policy Framework Paper, 1998-2000. 1998 [cited 2023 September 14]. Available from: https://www.imf.org/external/np/pfp/senegal/senge00.htm.

98. Republic of Senegal. Poverty Reduction Strategy Paper. 2002 [cited 2023 September 12]. Available from: <u>https://www.imf.org/external/np/prsp/2002/sen/01/100502.pdf</u>.

99. IMF. Request for a three-year policy coordination instrument—press release; staff report; and statement by the executive director for senegal. 2020 [cited 2023 August 23]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2020/01/18/Senegal-Request-for-a-Three-Year-Policy-Coordination-Instrument-Press-Release-Staff-Report-48958

100. World Bank. The World Bank and Senegal, 1960-87. 1989 [cited 2023 September 25]. Available from: https://documents1.worldbank.org/curated/pt/922521533559327885/pdf/The-World-Bank-and-Senegal-1960-87.pdf.

101. IMF. Senegal : First Review Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility, and Requests for Waiver of Performance Criteria and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2004 [cited 2020 November 10]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-First-Review-Under-the-Three-Year-Arrangement-Under-the-Poverty-Reduction-and-Growth-17381.</u>

102. Agajelu AC, Aniefiok-Hanson IC. The international monetary fund and Senegal relations from 1979-2004. PREORCJAH. 2018;3(2).

103. World Bank. Memorandum of the president of the International Development Association
to the executive directors on a country assistance strategy of the World Bank Group for the Republic of Senegal.
1995 [cited 2023 September 20]. Available from:

https://documents1.worldbank.org/curated/en/972171468169158047/pdf/multi-page.pdf.

104. World Bank. Senegal - Country economic memorandum. 1984 [cited 2023 September 21]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/901191468335438265/senegal-country-economic-memorandum. 105. IMF. Senegal: Background Papers and Statistical Appendix. 1995 [cited 2023 September 14]. Available

from: https://www.elibrary.imf.org/view/journals/002/1995/071/article-A001-en.xml.

Tahari A, Vrijer Jd, Fouad M. Adjustment for Growth: The African Experience. 1996 [cited 2022
 February 3]. Available from: <u>https://www.elibrary.imf.org/view/books/084/00149-9781557755667-en/C4.xml</u>.
 IMF. Senegal: Selected Issues and Statistical Appendix. 1997 [cited 2023 September 14]. Available

from: https://www.elibrary.imf.org/view/journals/002/1997/094/article-A001-en.xml.

108. World Bank. Senegal - Stabilization, partial adjustment and stagnation. 1993 [cited 2023 September 20]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/document detail/744251468335438596/senegal-stabilization-partial-adjustment-and-stagnation.

109. World Bank. Memorandum of the president of the International Development Association to the executive directors on a Country Assistance Strategy of the World Bank Group for the Republic of Senegal. 1997 [cited 2023 September 20]. Available from: <a href="https://documents.worldbank.org/en/publication/documents-reports/documentdetail/624001468307173324/">https://documents.worldbank.org/en/publication/documents-reports/documentdetail/624001468307173324/</a>.

110. World Bank. Loan Agreement (Structural Adjustment Loan) between Republic of Senegal and International Bank for Reconstruction and Development. 1981 [cited 2023 September 22]. Available from: https://documents1.worldbank.org/curated/en/845781468058508059/pdf/Loan-1931-Senegal-Structural-Adjustment-Loan-And-Credit-Loan-Agreement-1.pdf.

111. World Bank. Senegal - Second Structural Adjustment Program Project. 1986 [cited 2023 September 21]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/867321468105873397/senegal-second-structural-adjustment-program-project. 112. IMF. The Composition of Fiscal Adjustment and Growth: Lessons from Fiscal Reforms in Eight

Economies. 1997 [cited September 15 2023]. Available from: https://www.elibrary.imf.org/display/book/9781557756299/9781557756299.xml. 113. IMF. Economic Adjustment and Reform in Low-Income Countries. 1997 [cited 2023 September 15]. Available from: <u>https://www.elibrary.imf.org/display/book/9781557757159/ch08.xml</u>.

114. World Bank. Senegal - Macroeconomic update report. 1993 [cited 2023 September 20]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/145871468306282080/senegal-macroeconomic-update-report.

115. World Bank. Senegal - An economy under adjustment (Vol. 2) : Main report. 1987 [cited 2023 September 21]. Available from: <u>https://documents.worldbank.org/en/publication/documents-reports/documentdetail/744271468113087188/main-report.</u>

116. World Bank. Senegal - Public expenditure review. 1993 [cited 2023 September 20]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/825081468103462556/senegal-public-expenditure-review.

117. Rouis M. Adjustment performance in Senegal : a review of the design, implementation and outcome since 1980. 1993 [cited 2023 September 21]. Available from:

https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/352991468915316247/adjustment-performance-in-senegal-a-review-of-the-design-implementation-and-outcome-since-1980.

118. World Bank. Senegal - A review of the three year public investment program : 1987 / 88-1989 / 90. 1987 [cited 2023 September 21]. Available from: <u>https://documents.worldbank.org/en/publication/documents-reports/documentdetail/715181468103461080/senegal-a-review-of-the-three-year-public-investment-program-1987-88-1989-90</u>.

119. IMF. Staff Report for the 2012 Article IV Consultation, Fourth Review Under the Policy Support Instrument, and Request for Modification of an Assessment Criterion—Staff Report; Staff Supplements; Public Information Notice and Press Release on the Executive Board Discussion; and Statement by the Executive Director for Senegal. 2013 [cited 2023 August 31]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2012-Article-IV-Consultation-Fourth-Review-Under-the-Policy-40177.

120. Keshavjee S. Blind Spot

How Neoliberalism Infiltrated Global Health. 1 ed: University of California Press; 2014. ISBN: 9780520282834 121. Hanson K, McPake B. The Bamako Initiative: where is it going. Health Policy and Planning. 1993;8(3):267-74.

122. IMF. Senegal: 2002 Article IV Consultation and Requests for a Three-Year Arrangement Under the Poverty Reduction and Growth Facility and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2003 [cited 2023 September 11]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-2002-Article-IV-Consultation-and-Requestsfor-a-Three-Year-Arrangement-Under-the-16631.

123. IMF. Senegal: Second Review Under the Third Annual Arrangement Under the Poverty Reduction and Growth Facility and Requests for Waiver of Performance Criteria, and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2002 [cited 2023 September 12]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Second-Review-Under-the-Third-Annual-Arrangement-Under-the-Poverty-Reduction-and-15777</u>.

124. IMF. Senegal: Staff Report for the 2001 Article IV Consultation, First Review Under the Third Annual Arrangement Under the Poverty Reduction and Growth Facility, and Request for Waiver of Performance Criteria. 2001 [cited 2023 September 12]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Staff-Report-for-the-2001-Article-IV-Consultation-First-Review-Under-the-Third-15430.

125. IMF. Republic of Senegal—Letter of Intent, Memorandum on Economic and Financial Policies for 2001, Technical Memorandum of Understanding. 2001 [cited 2023 September 12]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Staff-Report-for-the-2001-Article-IV-Consultation-First-Review-Under-the-Third-15430</u>.

126. Republic of Senegal. Senegal: Enhanced Structural Adjustment Facility Economic and Financial Policy Framework Paper (1999-2001). 1999 [cited 2023 September 13]. Available from: https://www.imf.org/external/np/pfp/1999/senegal/index.htm.

127. Jean Hugues Caffin. L'aide au développement et le financement basé sur la performance : quelle performativité ? : analyse du processus de conceptualisation et de diffusion du financement basé sur la performance dans la gestion des systèmes de santé africains par la Banque Mondiale et l'USAID : étude du cas du Programme national de financement basé sur les résultats du Ministère de la Santé du Sénégal: Ecole Doctorale de Management Panthéon-Sorbonne; 2018.

128. Lemiere C, Puret J, Turbat V. Senegal - A tale of excessive hospital autonomy An evaluation of the hospital reform in Senegal. 2012 [cited 2023 September 18]. Available from:

https://documents1.worldbank.org/curated/en/467871468105835321/pdf/688060ESW0P1070l0reform0in0Seneg al.pdf.

129. IMF, World Bank. Decision Point Document under the Enhanced Heavily Indebted Poor Countries(HIPC) Initiative. 2000 [cited 2024 July 7]. Available from:

https://documents1.worldbank.org/curated/en/433661583532267858/pdf/Senegal-Enhanced-Heavily-Indebted-Poor-Countries-HIPC-Debt-Initiative.pdf.

130. IMF. Senegal: 2004 Article IV Consultation and Second Review Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility and Requests for

Waiver of Nonobservance of Performance Criteria and Rephasing of the

Arrangement—Staff Report; Press Releases on the Executive Board Discussion; and

Statement by the Executive Director for Senegal. 2006 [cited 2023 September 8]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2004-Article-IV-</u> <u>Consultation-and-Second-Review-Under-the-Three-19483</u>.

131. IMF. Third and Fourth Reviews Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility and Request for Waiver of Performance Criteria. 2005 [cited 2023 September 8]. Available from: https://www.imf.org/-/media/Websites/IMF/imported-full-text-pdf/external/pubs/ft/scr/2006/\_cr06127.ashx.

132. IMF. Senegal: First Review Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility, and Requests for Waiver of Performance Criteria and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2004 [cited 2023 September 11]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-First-Review-Under-the-Three-Year-Arrangement-Under-the-Poverty-Reduction-and-Growth-17381.

133. Republic of Senegal. Senegal: Poverty Reduction Strategy Paper—Second Annual Progress Report. 2004 [cited 2023 September 8]. Available from: <u>https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/pubs/ft/scr/2006/cr0668.ashx</u>.

134. IMF, World Bank. Poverty Reduction Strategy Paper Joint Staff Assessment. 2002 [cited 2023 September 11]. Available from: <u>https://www.imf.org/external/np/jsa/2002/sen/eng/120202.pdf</u>.

135. World Bank. Senegal - First, Second, and Third Poverty Reduction Support Credits. 2008 [cited 2024 June 13]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/106551468305075940/senegal-first-second-and-third-poverty-reduction-support-credits. 136. World Bank. Senegal - Third Poverty Reduction Support Credit. 2007 [cited 2024 June 13]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/100271468103766953/senegal-third-poverty-reduction-support-credit.

137. World Bank. https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/979831468308065665/senegal-country-assistance-strategy. 2007 [cited 2023 September 19]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/979831468308065665/senegal-country-assistance-strategy.

138. World Bank. First Poverty Reduction Support Credit. 2004 [cited 2024 July 2]. Available from: https://documents1.worldbank.org/curated/en/449131468760175785/text/283320SN0original.txt.

139. World Bank. Implementation completion and results report. 2008 [cited 2024 July 2]. Available from: https://documents1.worldbank.org/curated/en/106551468305075940/text/ICR6690Multipl1IC010disclosed081251 .txt.

140. IMF. Senegal: Poverty Reduction Strategy Paper Annual Progress Report. 2010 [cited 2023 September 6]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Annual-Progress-Report-24516</u>.

141. Republic of Senegal. Senegal: Poverty Reduction Strategy Paper. 2006 [cited 2023 September 7]. Available from: <u>https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/pubs/ft/scr/2007/cr07316.ashx</u>.

142. IMF, World Bank. Senegal: Poverty Reduction Strategy Paper—Joint Staff Advisory Note. 2007 [cited 2023 September 8]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Joint-Staff-Advisory-Note-21334</u>.

143. IMF, World Bank. Senegal - Joint IDA-IMF staff advisory note and the second poverty reduction strategy paper progress report. 2008 [cited 2023 September 19]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/842141468106140771/senegal-joint-ida-imf-staff-advisory-note-and-the-second-povertyreduction-strategy-paper-progress-report.

144. World Bank. Fourth Poverty Reduction Support Credit. 2010 [cited 2024 July 3]. Available from: https://documents1.worldbank.org/curated/en/241821468303664169/pdf/527480PGD0Box31ly10IDA1R2010101 2511.pdf.

145. IMF. Senegal: Sixth Review Under the Policy Support Instrument, Request for a Three-Year Policy Support Instrument and Cancellation of Current Policy Support Instrument—Staff Report; Debt Sustainability Analysis; Press Release; Executive Director Statement. 2010 [cited 2023 September 6]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Sixth-Review-Under-the-Policy-Support-Instrument-Request-for-a-Three-Year-Policy-24502.

146. IMF. Senegal: Sixth Review Under the Policy Coordination Instrument and Third Reviews Under the Stand-By Arrangement and the Arrangement Under the Stand-By Credit Facility-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2023 [cited 2023 August 22]. Available from: https://www.imf.org/en/Publications/CR/Issues/2023/01/18/Senegal-Sixth-Review-Under-the-Policy-Coordination-Instrument-and-Third-Reviews-Under-the-528312.

147. IMF. 2014 article IV consultation and eighth review under the policy support instrument—staff report; press release; and statement by the executive director for Senegal. 2015 [cited 2023 August 28]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2014-Article-IV-Consultation-and-Eighth-Review-Under-the-Policy-42578.

148. IMF. Request for a three-year policy support instrument—press release; staff report; and statement by the executive director for Senegal. 2015 [cited 2023 August 28]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Request-for-a-Three-Year-Policy-Support-Instrument-Press-Release-Staff-Report-and-43308.

149. IMF. Senegal: Poverty Reduction Strategy Paper. 2013 [cited 2023 August 28]. Available from: https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/pubs/ft/scr/2013/cr13194.ashx.

150. IMF, World Bank. Poverty reduction strategy paper—joint staff advisory note on the national strategy for economic and social development. 2013 [cited 2023 August 29]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Joint-Staff-Advisory-Note-on-the-National-Strategy-40740.

151. World Bank. Senegal - Fifth Poverty Reduction Support Credit. 2011 [cited 2024 July 3]. Available from: https://www.worldbank.org/en/news/loans-credits/2011/04/26/senegal-fifth-poverty-reduction-support-credit.

152. Ministère de la Santé et de l'Action Sociale (MSAS). Strategie nationale de financement de la santé pour tendre vers la couverture sanitaire universelle. 2017 [cited 2021 October 19]. Available from: https://www.sante.gouv.sn/sites/default/files/2%20MSAS%20Strat%C3%A9gie%20Nationale%20de%20Finance ment%20de%20Ia%20Sant%C3%A9.pdf.

153. Ministère de la Santé et de l'Action Sociale (MSAS). Plan National de Développement Sanitaire et Social (PNDSS). 2019 [cited 2024 July 29]. Available from:

https://www.sante.gouv.sn/sites/default/files/1%20MSAS%20PNDSS%202019%202028%20Version%20Finale.pdf.

154. Government of Senegal. Plan Sénégal Émergent 2014 [cited 2024 March 27]. Available from: https://www.au-senegal.com/IMG/pdf/plan-senegal-emergent-2014.pdf.

155. IMF. IMF Executive Board Approves for Senegal a US\$442 Million Disbursement Under the Rapid Credit Facility and Purchase Under the Rapid Financing Instrument. 2020 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2020/04/16/Senegal-Request-for-Disbursement-Under-the-Rapid-Credit-Facility-and-Purchase-Under-the-49335.

156. IMF. Fifth review under the policy coordination instrument, second reviews under the standby arrangement and the arrangement under the standby credit facility, and requests for augmentation of access, waiver of the nonobservance of a performance criterion, and modification of a performance criterion and quantitative targets—press release; staff report; and statement by the executive director for senegal. 2023 [cited 2023 August 23]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2022/06/27/Senegal-Fifth-Review-Under-the-Policy-Coordination-Instrument-Second-Reviews-Under-the-520104">https://www.imf.org/en/Publications/CR/Issues/2022/06/27/Senegal-Fifth-Reviews-Under-the-Policy-Coordination-Instrument-Second-Reviews-Under-the-520104</a>.

157. IMF. 2021 article IV consultation, fourth review under the policy coordination instrument, first reviews under the stand-by arrangement and the arrangement under the standby credit facility, and request for modification of performance criteria and quantitative targets. 2021 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2022/01/14/Senegal-2021-Article-IV-Consultation-Fourth-Review-Under-the-Policy-Coordination-Instrument-511932.

158. IMF. Senegal: Third Review Under the Policy Coordination Instrument and Request for Modification of Quantitative Targets, and Requests for a Stand-By Arrangement and an Arrangement Under the Standby Credit Facility-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2021 [cited 2023 August 23]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2021/06/17/Senegal-Third-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-460831">https://www.imf.org/en/Publications/CR/Issues/2021/06/17/Senegal-Third-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-460831</a>.

159. IMF. Second review under the policy coordination instrument and request for modification of quantitative targets—press release; and staff report. 2021 [cited 2023 August 23]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2021/01/19/Senegal-Second-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-50026

160. IMF. First review under the policy coordination instrument and request for modification of quantitative targets—press release; staff report; and statement by the executive director for Senegal. 2020 [cited 2023 August 23]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2020/07/24/Senegal-First-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-49608">https://www.imf.org/en/Publications/CR/Issues/2020/07/24/Senegal-First-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-49608</a>.

161. IMF. Senegal: Requests for an Extended Arrangement Under the Extended Fund Facility, an Arrangement Under the Extended Credit Facility, and an Arrangement Under the Resilience and Sustainability Facility-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2023 [cited 2023 August 22]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2023/07/07/Senegal-Requests-for-an-Extended-Arrangement-Under-the-Extended-Fund-Facility-an-535844">https://www.imf.org/en/Publications/CR/Issues/2023/07/07/Senegal-Requests-for-an-Extended-Arrangement-Under-the-Extended-Fund-Facility-an-535844</a>.

162. World Bank. Second Equitable and Resilient Recovery in Senegal Development Policy Financing. 2023 [cited 2024 July 15]. Available from:

https://documents1.worldbank.org/curated/en/099062023122065708/pdf/P17529301be541076098d00fab699502 811.pdf.

163. Ouattara B. Fiscal Effects of Aid flows in Senegal. 2004 [cited 2023 March 15]. Available from: http://hummedia.manchester.ac.uk/schools/soss/economics/discussionpapers/EDP-0420.pdf.

164. Ouattara B. Aid, Debt and Fiscal Policies in Senegal. Journal of International Development. 2006;18(8):1105-22.

165. Fosu AK. The External Debt-Servicing Constraint and Public-Expenditure Composition in Sub-Saharan Africa. African Development Review. 2010;22(3):378-93.

166. Daff BM, Diouf S, Diop ESM, Mano Y, Nakamura R, Sy MM, et al. Reforms for financial protection schemes towards universal health coverage, Senegal. Bull World Health Organ. 2020;98(2):100-8.

167. Wood A. Patronage, partnership, voluntarism: Community-based health insurance and the improvisation of universal health coverage in Senegal. Social science & medicine (1982). 2023;319:115491.

168. Mladovsky P, Ndiaye P. Solidarity in community-based health insurance in Senegal: rhetoric or reality? African Health Monitor. 2015.

169. Ridde V, Caffin JH, Hane F. External influences over Senegalese health financing policy: delaying universal health coverage? Health Policy Plan. 2024;39(1):80-3.

170. Ly MS, Faye A, Ba MF. Impact of community-based health insurance on healthcare utilisation and outof-pocket expenditures for the poor in Senegal. BMJ Open. 2022;12(12):e063035.

171. Ridde V, Gaye I, Ventelou B, Paul E, Faye A. Mandatory membership of community-based mutual health insurance in Senegal: A national survey. PLOS Glob Public Health. 2023;3(9):e0001859.

172. Homedes N, Ugalde A. Why neoliberal health reforms have failed in Latin America. Health Policy. 2005;71(1):83-96.

173. Chapman A. The impact of reliance on private sector health services on the right to health. Health Hum Rights. 2014;16(1):122-33.

174. Coveney L, Musoke D, Russo G. Do private health providers help achieve Universal Health Coverage? A scoping review of the evidence from low-income countries. Health Policy Plan. 2023;38(9):1050-63.

175. Cormier B, Manger MS. The Evolution of World Bank Conditionality: A Quantitative Text Analysis. 2020.
176. Vetterlein A. Economic Growth, Poverty Reduction, and the Role of Social Policies: The Evolution of the World Bank's Social Development Approach. Global Governance. 2007;13(4):513-33.

177. Tichenor M, Winters J, Storeng KT, Bump J, Gaudillière JP, Gorsky M, et al. Interrogating the World Bank's role in global health knowledge production, governance, and finance. Global Health. 2021;17(1):110.

178. Kentikelenis A, Stubbs T. Social protection and the International Monetary Fund: promise versus performance. Global Health. 2024;20(1):41.

179. Organisation of African Unity. Abuja Declaration on HIV/AIDS, Tuberculosis and Other Related Infectious Diseases. Abuja, Nigeria; 2001.

## 9. Chapter 9: Discussion

This PhD project aimed to investigate how External Development Partners (EDPs) and the International Monetary Fund (IMF) affect the mix of domestic health financing sources in recipient/borrower countries. The research objectives pursuing this aim are repeated here for clarity before discussing the study findings:

- To investigate the relationships between development assistance, public external debt and IMF loan conditionalities and the levels and balance of Government Health Expenditure as a Source (GHE-S) and Out Of Pocket payments (OOP) among 105 Low- and Middle-Income Countries (LMICs) from 2005-2019 (Chapter 6).
- 2. To explore mechanisms of EDP influence on the mix of domestic health financing sources present in Senegal (Chapter 7).
- 3. To explore how external official lending, public external debt obligations and IMF/WB policy recommendations and loan conditionalities affect the mix of domestic health financing sources in Senegal (Chapter 8).
- 4. To develop recommendations for policy and for future research based on the study findings (Chapters 6, 7, 8 and 9)

This final chapter summarizes the main findings from these chapters. This first section of the chapter is organised around the three main health financing sources investigated: GHE-S, OOP and Voluntary Health Insurance (VHI).

The Discussion then discusses how the main findings contribute to the existing literature, identifies and discusses some key limitations present in the methods used, provides reflections on my own role and influence on the research process, findings and interpretation, and on alternative conceptual framings and methodological approaches to the explored research questions. The policy and research recommendations arising from this thesis are then provided jointly at the end of this chapter, before a final conclusion is presented.

## 9.1. Discussion of main findings

Figure 1 provides an overview of the integrated results from Chapters 6-8. The colour-coding shows which hypotheses were confirmed, rejected, or where more nuanced/mixed conclusions were reached. The different pathways of influence are discussed under the respective sub-sections for health financing sources below.



Figure 9.1: Integrated results from chapters 6-8. GHE-S / GDP correlated positively with OOP / GDP and negatively with OOP / CHE. GHE-S / CHE correlated negatively with OOP / CHE.

## 9.1.1. GHE-S

## 9.1.1.1. Official development assistance and GHE-S

This thesis mainly hypothesized a displacing effect of development assistance to the health sector on GHE-S (fungibility). A secondary alternative hypothesis of a synergistic co-financing effect was also proposed, i.e. that external injection of funds crowds in GHE-S, with much less overall support in the literature.

Chapter 6 found that, when channelled through the government, Official Development Assistance<sup>+</sup> (ODA<sup>+</sup>) for health was associated with reduced GHE-S, consistent with fungibility. Evidence of fungibility was also found at the country level in Chapter 7. The co-financing mechanism was employed by some EDPs in Senegal, however seemingly with limited overall effect on GHE-S.

These findings indicating fungibility of development assistance to the health sector align with the majority of the literature on the topic (1-15), in particular when channelled via the government (6, 7, 9, 10, 15).

EDPs should thus expect that governments will redistribute some of their own funds for health to other purposes when receiving on-budget external assistance for health. While EDPs may view this as undesirable, the role of recipient countries' ministries of finance is to distribute their budgets from all revenue sources to meet the needs across all sectors. Given that health is a sector with many partners, it is a predictable phenomenon that some domestic funds will be allocated to other sectors that receive less international attention.

#### 9.1.1.2. Debt and GHE-S

In this thesis, it was hypothesized that external official lending from EDPs + IMF would work to support increases in GHE-S, but that the resulting debt obligations would constrain GHE-S. Chapter 8 found empirical support for both these relationships. Chapter 6 also confirmed that increases in Public and Publicly Guaranteed (PPG) external debt servicing led to decreases in GHE-S/Current Health Expenditure (CHE), though not per Gross Domestic Product (GDP) or per capita. The latter dampens the scope for policy inference, however the identification of the same effect on domestic government health spending in Senegal in Chapter 8 is encouraging in terms of the potential for drawing policy conclusions.

Chapter 8 specifically identified the influence pathway of external official lending working to expand the general government budget and thereby the overall fiscal space available for the health sector, as hypothesized. The temporal aspect of this effect likely works both within the immediate 0–1-year term when funds are received, and then on a much longer term as funds are invested and start generating economic activity and revenue for the government. During this longer term is also when any grace periods will run out and loans will start having to be repaid. These temporal considerations make policy inferences more complicated, because as the immediate effect of having to service an outstanding debt may be negative on part of the health sector, what are the benefits to the health sector today from development loans received 20 years ago? This consideration is included as a point of nuance in the policy recommendations at the end of this chapter.

Chapter 8 also identified the possible dynamic that spending cuts during the fiscal year to meet loan repayment timelines were more likely to occur in the health sector due to the presence of donors. This is similar but not equal to the fungibility mechanism. It points to a possible interaction between debt, development assistance for the health sector, and government health spending. If faced with debt repayment obligations, will a government cut where it expects that donors will cover costs? This question nuances the fungibility debate by adding the perspective that governments may reduce funding to a sector not as a mechanistic

response to having received external funds, but out of a future expectation of donor support where cuts are made.

Chapter 8 further identified how political economy factors such as sectoral visibility and profitability could become more prominent under budget constraint from debt obligations, driving government funds away from the health sector. It also showed how debt constraint seemed to affect the health sector more than the education sector due to political prioritisation.

Together, this leads to the following understanding: Indebted, aid-recipient governments first receive external official loans and grants, some of which will be ear-marked, which expand their budget along with all other revenue sources. From the available revenue, they allocate a proportion to debt servicing before making sectoral allocations. What they have left is allocated where it is politically and economically rational to do so, which may involve considerations of profitability and visibility of spending. The amount allocated to the health sector may also be determined by donor presence. During the financial year, debt repayment timelines, along with all other expenses incurred and pending during the year, may necessitate amendment of the government expenditure budget, where political priority as well as donor presence may again play a role in deciding where to make adjustments. These fiscal considerations and decisions result in the observed phenomena of fungibility and debt constraint of the health budget.

The findings for debt generally align with the majority of the identified econometric literature (11, 14, 16-20), finding constraining effects of debt on government health spending more often than not, and with available single-country reports from SSA also identifying this (21-23). No literature was identified studying the specific interactions between debt obligations, political prioritisations under budget constraint, and government health spending. The political prioritisation identified has variable support in the literature (24-27), with some authors finding more visible and profitable spending made during election years.

Our findings for aid, debt and GHE-S and those of the bulk of the existing literature are somewhat discouraging for EDPs, in the sense that their ODA for health may limit GHE-S due to financing fungibility, and their overall development lending may also help limit the degree to which health systems are funded by domestic government funds in the long term.

#### 9.1.1.3. Non-financial influence mechanisms on GHE-S

This thesis hypothesized that EDPs would use their non-financial development cooperation activities to promote GHE-S, which was confirmed in Senegal. The IMF was hypothesized to promote austerity at the expense of GHE-S. In Senegal, the IMF was however found to have

the double-sided influence of promoting both revenue generation as well as increases in GHE-S, while at the same time limiting overall government spending to balance the fiscal deficit, with mechanisms in place to safeguard the health sector. Identified mechanisms of influence included aim- and standard setting, policy and technical advice/recommendations, lobbying and negotiating<sup>40</sup>, and loan conditionalities.

Based on the reviewed literature, the identified influence mechanisms of technical advice and financing are well established mechanisms in the health policy literature (28-36), including for health financing reform (37-39). The external imposition of norms, aims and standards have been found in other contexts as well (40-44), while lobbying/negotiation has been less studied in a health policy context (45), possibly because it is harder to gauge as an outside researcher. These non-financial mechanisms generally promoted GHE-S and VHI and sought to reduce OOP, though for OOP only from the early 2000's after initial promotion under structural adjustment, the Bamako Initiative and hospital reform in the 1980's and 1990's. From the early 2000's onward, this had an opposite (positive) direction of effect on GHE-S in Senegal to those of the identified financial mechanisms of aid and debt, identified at both the country and global level.

A tension thus arises: On one hand EDPs tell governments to spend more on health, while on the other the provision of ODA for health leads to reduced GHE-S and as does the debt servicing obligations arising from loans, while development loans themselves can directly benefit the health sector and expand the revenue base underlying GHE-S.

In this balance it appears that the health sector in Senegal had lost out, receiving decreasing priority in the government budget over time (Chapter 8). This suggests stronger negative influences through fungibility, debt constraint and IMF austerity policies than positive influences from development loans and EDP+IMF promotion of increased GHE-S. However, the consistent increases in the budget for the education sector, suggest that government prioritisation may play a bigger role in determining government funding levels than external influences, as the education sector received high levels of aid and is also subject to an overall fiscal level debt constraint. However, it is unclear whether this would be the case in more fragile contexts with weaker governance.

<sup>&</sup>lt;sup>40</sup> The results in Chapter 7 did not specifically identify the IMF using lobbying and negotiation, but this is likely a limitation/ omission, as the IMF by default negotiates with their member country governments when developing country programs.

In Chapter 6, no evidence was found for an influence on GHE-S from IMF programs or conditionalities at the global level. This may reflect, A: the double-sided mandate of the IMF in balancing the books, and B: complex individual country-level policy prescriptions being suboptimally captured in simple quantitative indicators across countries, as shown and discussed in Chapter 8. This negative finding at the global level aligned mostly with those of the IMF and its Independent Evaluation Office (46-50) (although with some exceptions (51)), and some previous research (52-54).

### 9.1.2. OOP

#### 9.1.2.1. Official development assistance and OOP

ODA<sup>+</sup> for health was hypothesized to generally increase OOP/CHE through a main displacement effect on GHE-S, possibly counteracted by a direct effect of reducing OOP through covering health costs (OOP displacement/subsidy).

Chapter 6 found that OOP measured per GDP reduced with both on- and off-budget ODA<sup>+</sup> for health. No overall effect was however found on OOP/CHE (nor on GHE-S/CHE). EDPs were found to displace OOP/GDP about as much as they displaced GHE-S/GDP with their on-budget support for health, with a smaller displacing effect of their off-budget support for health added on top. Together, this indicates that EDPs serve as a source of revenue in the health sector, covering some costs for both the government and for users at the point of care, shifting the burden of payment onto themselves rather than shifting the balance between GHE-S and OOP. From an equity perspective, these are ameliorating findings, as it appears that EDPs apart from displacing/subsidising GHE-S also partially cover health expenses from the most inequitable source of financing in LMICs. The existing literature on this relationship was limited and divergent (4, 13, 14, 55-57).

#### 9.1.2.2. Debt and OOP

Some limited evidence was found in the global level analysis of an increased reliance on OOP associated with public external debt servicing (Chapter 6).

The hypothesis was that an increase in OOP/CHE would occur through a decrease in GHE-S, which was seen, but only when measured per CHE, not per GDP. This is interpreted as public external debt servicing reducing the budget envelope for government health spending, resulting in a shift in the burden of payment toward the user. The magnitude of these effects may not have been identifiable relative to the underlying growth in GDP and population.

The existing literature is scarce, with one study finding respectively no significant relationship between general government debt per GDP and OOP (14), and others having found a negative relationship in Sub-Saharan Africa (SSA) (exact variables not reported) (18).
Overall, EDP non-financial influence mechanisms were found to decrease OOP in Senegal after 2000, and ODA<sup>+</sup> for health was found to reduce OOP at the global level. However, public external debt servicing obligations increased reliance on OOP at the global level.

#### 9.1.2.3. Non-financial influence mechanisms on OOP

The quantitative global analysis in Chapter 6 found no effect of IMF programmes or conditionalities on OOP. At the country level, we found evidence of EDP efforts to lower OOP, through setting aims and standards, as hypothesized (Chapter 7). WB policy recommendations and conditionalities also evolved from around the turn of the millennium to clearly seek to reduce OOP. However, their past promotion of OOP may have lingered in the payment patterns present in Senegal, and their ongoing recommendations and conditionalities to expand the role of the private sector in health service provision may also not be conducive to a transition away from OOP (Chapter 8).

The work in Senegal was consistent with the negative findings for IMF influence on OOP at the global level, in finding evidence of support for GHE-S and VHI while promoting reduced health expenditures for the poor, coupled on the other hand with general austerity measures on the public budget limiting the overall scope for an expansion of GHE-S and accompanying transition away from relying on OOP.

A historical evolution of WB policy from more traditionally neoliberal during the era of structural adjustment toward an increasing focus on pro-poor social policy has been identified by other authors (58, 59).

No other studies were identified investigating effects of IMF programmes or their conditionalities on reliance on OOP.

#### 9.1.3. VHI

This thesis hypothesized that EDPs + IMF would directly promote Community-Based Health Insurance (CBHI) through aid and non-financial mechanisms as identified in the literature. Chapters 7 and 8 found evidence of support for the expansion of *Mutuelles* in Senegal via EDP financing, policy/technical advice, and binding conditionalities from the WB. The initial disagreement around preferred models between EDPs seen in Chapter 7 highlighted the importance of donor coordination and alignment in this regard.

The general role of EDPs funding experimentation and scale up of alternative health financing schemes in LMICs, including CBHI, has been identified elsewhere (37-39, 60).

## 9.2. Main thesis contributions

Chapter 8 is the only empirical single-country case study of the impacts of debt on health financing identified outside of Kenya. In Kenya, a previous discussion paper has explored trends of public debt and health financing, while an Oxfam case study has used mixedmethods to explore debt impacts on the health sector including government health spending, both concluding a constraining effect (21, 23). By taking a mixed methods approach with new primary data generation in interviews combined with in-depth secondary data exploration in a new country context, we were able to significantly expand on the existing literature in particular in understanding pathways of influence and by providing a full view from initial lending over loan conditionalities to debt servicing obligations. Debt has typically been studied in fields outside of health, and the performed study highlights its relevance to both health policy and economics researchers. Using the case of the WB, the chapter went beyond existing case studies and econometric literature that tends to only look at the effects of debt in isolation from the prior influence of loans, and empirically established the essential point that loans may benefit the health sector in the first place. By showing how government education spending kept on growing in spite of debt obligations, it provided an original empirical basis for the interpretation that a constraining effect of debt on spending in a given social sector may be modified or even fully negated by political priority of that sector. These nuances are important in current and coming debates on debt relief and lead to policy recommendations as per Section 9.4.1.

Chapter 8 also provided the first dedicated, in-depth historical overview of IMF/WB policy recommendations and conditionality from a health financing perspective. In a field heavily dominated by panel data studies and desktop reviews, no other studies were identified that directly asked borrower country stakeholders of their experiences with how IMF/WB policy conditionalities affect the health sector in their country. With policy documents, the devil is often in the detail, and by critically examining these and asking key stakeholders what they have seen and experienced in Senegal, Chapter 8 uncovered, analysed and presented new and important knowledge. The evolution seen at the WB over time with increasingly progressive and equitable policy making, and the different efforts from the IMF to support and safeguard health and social spending – amidst austerity measures - adds moderation and nuance to an otherwise somewhat heated and polarised debate on the role of the IFIs in health financing.

The negative finding of no association between IMF programmes or conditionalities and OOP at the global level is the first identified attempt at exploring this relationship. Although the

finding of no association with GHE-S at the global level is not unique, Chapter 8 added insight on how these relationships are shaped by complex, multipronged policy recommendations and conditionalities at the country level.

The thesis made some important conceptual/analytical contributions. The frameworks generated in Chapters 7 and 8, integrated in Figure 9.1 in combination with the results from Chapter 6, provide a novel and comprehensive overview of the ways in which EDPs + IMF can influence the different domestic health financing sources in LMICs. This new conceptual knowledge should be relevant to a range of stakeholders from EDPs, the IMF, LMIC governments and civil society stakeholders in health financing.

The analytical framework derived in Chapter 7 in particular helped fill a literature gap between existing health financing policy analysis frameworks that do not specifically emphasize mechanisms of EDP policy influence, and the policy influence analysis frameworks from the broader development literature that do focus on mechanisms of influence but not specifically for health financing policy (45, 61-76). It forms a novel, re-usable analysis tool ready to be applied in other contexts for exploring the different pathways or mechanisms of EDP influence on different health financing sources, with any local contextual amendments as appropriate. The research recommendations for its real-world application are discussed in Chapter 7 and Section 9.4.2 below.

The literature on debt and health financing is still quite modest, and Chapter 6 expands a relatively small field of econometric studies on debt and government health spending (14, 16, 17, 19, 77, 78) and on debt and OOP (14, 18). Specifically, the negative relationships between PPG external debt servicing and GHE-S/CHE and positive with OOP/CHE are novel findings. They provide an important contribution to our understanding of the equity implications of public external debt servicing. The study used more robust econometric methods compared to much of the existing literature, adjusting for both disaggregated aid- and IMF-variables as appropriate as well as a large set of covariates, with an extensive battery of sensitivity and diagnostic testing.

Chapter 6 was the first study to compare the effects of both aid, debt and IMF variables on health financing in the same model. While the resulting significance patterns turned out with limit scope for comparing effect sizes between independent variables, these findings were still the product of a novel exploration that provides value in and of itself for other researchers seeking to model these relationships going forward.

Chapter 6 also used its strengthened estimation approach to identify a relationship between lagged ODA<sup>+</sup> for health per GDP, both on- and off-budget, and OOP/GDP at the global level. The particular model specification and robustness checking of this relationship has not been performed before, and the general relationship between development assistance for the health sector and OOP is rather scarcely studied in the existing literature, with diverging findings, and with some methodological issues as discussed in Chapter 6 (4, 13, 14, 55-57).

While the results for non-health ODA<sup>+</sup> produced in Chapter 6 were negative, they were a product of novel hypothesis testing and model specification, providing an early exploration of the potential displacement of government funds by sector-allocable ODA to non-health sectors to the benefit of the health sector.

The research in this thesis confirmed the presence of fungibility of on-budget development assistance to the health sector at the global level (1-15). It did so using more robust econometric methods than much of the existing literature as described above, for the first time adjusting for both non-health ODA<sup>+</sup>, debt and IMF-variables. It also added a unique country level perspective in Chapter 7, with first-hand observations from key informants confirming this relationship and explaining underlying political considerations.

Chapter 5, published as a short communication in Global Health Action in 2022, provided a warning of the worrying trends in LMIC debt burdens and the risk of constraining public health sector financing, visualising how many countries spend more on debt servicing than health. The chapter set out a research agenda that has been partly addressed in Chapter 6 and 8, providing evidence in support of a constraining effect of debt on health financing in Senegal and at the global level in the form of shifting CHE payment patterns from GHE-S toward OOP. These findings thus provide added empirical justification for the published warning. The UN, WB, Center for Global Development (CGD) and The Lancet have since raised the alarm on this matter (79-82), and the short communication can thus, at a much humbler level, be seen as a small contribution to a growing global awareness of the potential social implications of unsustainable debt trajectories in many heavily indebted Global South countries.

# 9.3. Methodological limitations and reflections

#### 9.3.1. General limitations

#### 9.3.1.1. Measurement error

The thesis relied heavily on statistics from generally underresourced LMIC national statistical offices, which may face a series of financial, logistical and capacity constraints, limiting their

ability to produce precise and reliable data (83). International datasets from High Income Country (HIC) institutions should in principle suffer less from such constraints. However, even different international datasets estimating aid levels to Reproductive, Maternal, Neonatal and Child Health (RMNCH) based mainly on Organisation for Economic Cooperation and Development (OECD) data and reports from major multilateral organizations, have been shown to provide greatly varying estimates (84). However, the datasets used across the thesis are the best available and are widely used by other researchers, and with the range of sensitivity testing and data triangulation performed, its findings should remain valid.

#### 9.3.2. Panel study limitations

In chapters 4 and 6, the issues of causal inference from observational data, endogeneity and confounding, selection bias and instrument proliferation were discussed, motivating the chosen estimator, model selection, and sensitivity testing. Some issues remain, as outlined in the following sections.

#### 9.3.2.1. Subjectivity in panel data analysis

By conducting the analysis in Chapter 8, I have learned that econometric regression analysis using secondary, observational data, is not a fully objective science. The entire process from study design, data compilation, data cleaning, initial exploratory testing, model building and refining, final analysis, sensitivity testing, interpretation and discussion, are full of decisions. Every decision requires weighing up pros and cons, exploring alternatives and assessing the consequences of going down different paths, which involves an unavoidable degree of subjectivity. Leamer (1983) described this issue as follows:

"All the concepts of traditional theory, utterly lose their meaning by the time an applied researcher pulls from the bramble of computer output the one thorn of a model he likes best, the one he chooses to portray as a rose" (85).

Leamer recommended honesty about one's assumptions and the fragility of one's estimates through extensive sensitivity testing; conservative inference; and viewing inferences for what they are, namely opinions based on one's findings, not to be mistaken with facts (85). I have tried to the best of my ability to heed this advice throughout Chapter 6, Appendix 4 and this discussion chapter.

#### 9.3.2.2. VHI as an outcome variable

Chapter 6 did not include VHI/Voluntary Health care Payment Schemes (VHPS) as an outcome variable. This choice was made for three main reasons. A: to limit hypothesis proliferation, which was already somewhat present with three sets of input variables of interest and two sets of outcome variables, along with covariates. B: Due to challenges in economic interpretation and inference. VHI/VHPS captures everything from CBHI for the rural poor to private health insurance for the urban rich. Increases in one country may thus reflect entirely

different changes in payment patterns, progressivity, pooling and cross-subsidy than in another country. C: VHI only forms 5% of CHE across LMICs, versus 45% from GHE-S and 34% from OOP (86). The approach for Chapter 6 to focus on the two major sources making up nearly 80% of health financing in LMICs was therefore preferred, to then focus on submitting the results for these to rigorous sensitivity testing.

The choice was therefore made to only delve into this matter at the country level, where the particular CBHI arrangement of *Mutuelles* could be understood and explored, with its specific terms, conditions and coverage. This does not mean that a panel data study of VHI would not be feasible, but its design, economic interpretations and policy inferences would need to be very careful. The choice made limits the findings for this financing source compared to the other two, but given the precarity of studying this variable with the used global-level panel data approach, this may not necessarily have been a limitation for this thesis in terms of its ability to produce valid and meaningful results and interpretations, rather than a higher number of results and interpretations.

#### 9.3.2.3. Missing data

The approach to missing data was described in Chapters 4 and 6. Quality and validity issues were viewed as too substantial when more than 5 years were missing, corresponding to more than 33% of the data for a given country time series for a key variable of interest, and those countries were therefore removed. This choice was partly arbitrary by default, and a different threshold could have been chosen, however as explained in Chapter 4, the chosen threshold removed mainly small island nations and countries with severe conflict, where the capacity to collect national statistics may be challenged and there may be considerable risk of measurement error for any datapoints that were available. By analogous reasoning, the years years 2002-2004 were removed, because the OECD Creditor Reporting System (CRS) ODA data separated by channel of assistance was strongly fragmented for these years, with missing data for 28 to 131 countries for individual variables in a given year, a fragmentation that greatly improved in 2005.

While these choices came at the expense of statistical power and increased the risk of type II error, the findings in Chapter 6 are based on the higher quality data available, rather than all the data available, with the excluded part deemed likely to suffer from some internal validity issues.

An alternative approach would have been multiple imputation. There are many different approaches to this (87), however examples of the perils of doing so from the Development Assistance for Health (DAH) fungibility debate (15, 88), and the principal notion of artificially generating data, led me to avoid this course of action.

#### 9.3.3. Case study limitations

The main limitations for the case study in Senegal are discussed in Chapter 7 and 8, while additional limitations are discussed in this section.

#### 9.3.3.1. Alternative conceptual framings

#### 9.3.3.1.1. Wider political economy factors

While Chapters 7 and 8 focused on external partner influence on health financing, a host of other political economy factors may co-determine health financing. As examples, government health expenditure has been positively associated with tax revenue (78), GDP (4), GDP per capita (78), aging of the population (78), democratic accountability (89) and government stability (89), and negatively associated with fiscal deficit (78) and corruption (89). While some of these factors were included as covariates in Chapter 6, and some political considerations were captured in Chapter 8, it was beyond the scope of Chapter 7 and 8 to fully lay out this web of co-determining factors in Senegal. It must therefore be kept in mind, that the case study in Senegal only captures a few dimensions of the political economy of health financing but not all, and the findings need to be held against those of others working on the political, social and economic determinants of health financing.

As discussed in Chapter 3, a number of different theoretical frameworks exist for analysing the political economy of health policymaking, however with few that capture the mechanisms or pathways of stakeholder influence, which was the focus of Chapter 7. The ones that most encapsulated influence mechanisms were used and adapted for that chapter, however alternative conceptual framings would have been conceivable, if the research questions were shifted to focus more for example on the policy process or on domestic political, social and economic factors.

A general challenge was that the research in this thesis was strongly question-driven (pragmatist), needing to generate novel analytical frameworks, while integrating and amending existing ones. It also had no well-established economic theories available in the existing literature to rely upon for exploring its research questions. An approach designing the research questions in direct alignment with existing conceptual frameworks and economic theories, ready to be applied, would likely have been easier. These were however the questions that I was and am passionate about, and I therefore chose the path necessary to address them, although more challenging. The resulting theoretical/conceptual contributions unpacking influence pathways can hopefully benefit other researchers going forward.

#### 9.3.3.1.2. Framing influence mechanisms as power dynamics

An alternative framing that could also potentially have been feasible would have been a power framing, provided the associated necessary modifications to the research questions and methods were made.

Figure 9.2 shows a photo of the placement of the IMF resident representative office in the same building as the Ministry of Finance in Senegal at the time interviews were conducted. What is the extent and nature of exchanges that occur with this level of proximity to the most powerful ministry in the country?



Figure 9.2: Photo of the IMF resident representative office and Ministry of Finance in Senegal, December, 2019 (author's own photo).

External influence on domestic policy and finance is undoubtedly a question of power, and the identified influence mechanisms could alternatively be conceptualised as power dynamics. Both the Senegalese government and its EDPs were seen to exercise "agenda-setting" or "structural" power (90-92) - the government by setting out its national strategies, and EDPs by defining their own areas of interest and accompanying scope for support, which again are shaped by a range of national and political interests (e.g.(93-96)). Examples of "discursive" power from development partners were also seen, i.e. shaping the development discourse, and the ideation, understanding and interpretation of issues and solutions by stakeholders (90, 91, 97-100), and of course financial power (36). There are a number of different theoretical approaches to power analysis, including within health policy, e.g. (40, 91, 92, 101). Analyses of power dynamics in development cooperation including for health have been performed at the global level (102), for external donors and local governments (36, 103), at the project level (104), for North-South NGO collaborations (e.g. (105-107)), and regarding discursive power in international development cooperation (98), generally cementing the power and influence

of development partners and external NGOs. Applying this analytical lens to EDP influence on domestic health financing would be a potential area of future research.

#### 9.3.3.2. Uniform description of EDPs

With the exception of clearly distinguishing between IMF and WB influence in Chapter 8, EDPs were otherwise generally described as one in Senegal, while of course mentioning individual EDP findings from the document review. This occurred as a result of the analysis in Chapter 7 showing no clear-cut delineations between the modality or direction of influence between individual EDPs, nor between bilaterals and multilaterals. Several interviewees gave the impression that this was due to a high degree of sovereignty, agency, and EDP alignment. This may well not be the case elsewhere, and in that case, the frameworks could be applied individually to different EDPs, which would help identify areas of mutual policy incoherence and poor EDP alignment.

An in-depth investigation into the lending agreements from bilateral creditors or other IFIs, might also have elicited different results. However, the international mandate of addressing fiscal and macroeconomic issues concerning balance of payments, inflation, fiscal deficit, debt distress and more rests with the IMF, supported by the WB. Requirements for a comprehensive national development strategy within an externally derived framework as e.g. a Poverty Reduction Strategy Paper (PRSP) is also a unique phenomenon from the WB, and studies of many other creditors than the IMF/WB could potentially yield results showing a narrower scope for external influence contained within the identified influence mechanisms. The main exemption from this general remark would likely be France, which has a long and extensive history of development cooperation with Senegal due to their colonial legacy, and further in-depth studies into influence from country creditor lending agreements could start with France.

#### 9.3.3.3. Interview limitations

Interviewer bias inevitably affected this part of the research (108). There naturally was an asymmetrical power relationship between me as the researcher who set the scene and dictated the topics to be discussed, and the contents of the questions were in one way or another colored by my preconceived notions and hypotheses for these topics (108). The main way I attempted to mitigate this was through reflexivity (Section 9.3.4), i.e. reflecting on the relationship between me as an interviewer and the interviewee and the personal biases I brought to interviews. The loose structure of the semi-structured interviews did however allow the interviewee to take control for parts of the interview, which was often the case and helped mitigate the asymmetry in my relationship with the interviewee.

My French was both somewhat limited and rusty upon arriving in Senegal. I therefore completed a 6-week course in intermediate French (B1-B2 level) aimed at preparing participants for studies at university level in France, and I immersed myself as much as possible in the language while in Senegal. After a few months, I became proficient enough to conduct interviews on my own without an interpreter. Throughout interviews, both with and without an interpreter, certain nuances may however have been lost in translation. Everything said in French by participants was however transcribed in French by my transcriptionist, so I have been able to analyse the original words of my participants 1:1. That said, some Senegalese interviewees could still have been more comfortable in Wolof.

The fact that I am white and foreign were also unchangeable limitations to my ability to build trust and report with Senegalese participants, who might possibly have provided more reserved responses to me than to a Senegalese researcher. The reliance on an interpreter for most interviews further reinforced this position as an outsider and led to a more formulaic interview style. Additionally, there may have been linguistic and cultural nuances that I have not been attuned to, that could have carried over to the analysis and interpretation stage as well.

#### 9.3.3.4. External validity

The external validity of findings in the case study is limited when it comes to relationships, policies and events specific to Senegal. Findings of a more principal nature, such as the mechanisms of EDP influence, the general policy positions of the IMF and EDPs present in Senegal, the impacts of a large debt burden, and the challenge in reducing OOP and progressing toward UHC, should however remain pertinent in other contexts. By attempting to inductively look for rather high-level, principal dynamics present in Senegal and use the observations made to generate analytical frameworks and conceptualisations of the dynamics in question, the scope for using the findings to explore similar dynamics in other contexts should hopefully have been expanded.

One factor that improves the applicability of the case study findings to other contexts is the fact that the interviewed external stakeholders are present across many LMICs, including in SSA, and other countries may have had similar experiences with these organizations. By anonymising interviews, this factor has however been attenuated, but it was done out of a primary concern with protecting the confidentiality of study participants.

#### 9.3.3.5. Changing political context in Senegal

Sadly, the regional beacon of political stability I conducted interviews in in 2019/20 temporarily destabilised. Following a series of corruption scandals, arrests of political opponents and

banning of their parties, ex-president Macky Sall attempted to run for a third term against the two-term presidency constitutional limit, triggering a constitutional crisis in 2023/24 (109-111). Violent protests erupted resulting in the deaths of at least 60 demonstrators (110, 112). Twenty days after being released from prison, opposition candidate Diomaye Faye was elected president on April 2, 2024 (111), after which conditions have been more stable. At the time of interviews and ensuing document and quantitative review and analysis, political conditions were much more stable. The interpretation that the case study findings reflect EDP-government interactions under political stability should therefore hold, and the temporary political instability in Senegal should not have affected the validity or relevance of the case study findings.

#### 9.3.4. Reflexivity

Every person is shaped by their socio-historical position, and as can no social scientist claim to be removed from their subject matter and bias-free (113). My positionality as a Western, middle-class, cis-gendered, heterosexual, white male in good health from the stable and rich welfare country of Denmark, has led to a range of privileges throughout the course of my life, including the opportunity to pursue this PhD. How this positionality led to interview limitations in Senegal, being an outsider or "etic" (113, 114), has been discussed above in Section 9.3.3.3. My positionality meant that I came to Senegal embodying historical and present systems of exploitation and injustice, asking high-level key stakeholders to give me their time and information for free. Undeservingly, people generally opened their doors for me, and I owe a debt of gratitude to interview participants for not outright rejecting me.

In terms of my overall approach to the general research topic, in my particular case, a selfawareness of my privileged positionality has motivated an ongoing journey of learning, critical self-reflection and importantly exposure and immersion in other cultures and social contexts. My clinical work in a number of different settings in different parts of the world has shown me, first-hand, the human consequences of social and economic inequities, from cradle to grave. These experiences have led to a lasting engagement in issues of social justice and social equity. This may have led to a partiality in my hypothesis generation, study design and execution, *a priori* favouring a view of the beneficiary country as a "victim" of Western wrongdoings, perhaps out of feelings of guilt, moral injustice and resulting indignation. This may have rendered me less attuned to evidence of alternative explanations of findings on the part of beneficiary countries, such as political and institutional inefficiency, self-serving or inequitable political prioritisations, corruption, etc. I tried to mitigate this bias in my interviews by attempting to give each interviewee equally "firm" treatment, asking probing questions to elicit the reasoning behind their positions and pursuing statements needing further justification. For the panel study, I made sure to include a comprehensive set of control variables to capture alternative explanations of observed trends, including governance and corruption factors. I leave this PhD with a more nuanced world-view, now seeing more shades of grey, where it was more black and white when I started.

In interview studies, where data is generated as a dialogue, reflexivity is an essential part of being able to maintain a certain level of scientificity. Being aware of this, I tried to prepare myself better for interviews by conducting a mock interview to receive critical feedback both on the content of my questions but also on how I came across, my style of enquiry etc. After each of the interviews, I would reflect on questions that had not gone as intended, situations where the interviewee and I had misunderstood each other, my degree of probing for further information, whether my questions were perceived as sensitive, the general report between the interviewee and I, and more. I would also ask my interpreter for feedback on my interview style and cultural awareness.

Epistemologically, I have experienced a personal development from more positivist at the beginning of the research process, in part due to having a background in medicine and public health with emphasis on quantitative subjects, towards much more pragmatist. This has allowed me to see the strengths and limitations of different methodologies more clearly and understand the value of letting the core research problem drive one's choice of methods.

Finally, conducting PhD studies as a father of first one, then two, now three, while being selffunded for the last year of my studies, has been challenging. It has taught me the value and necessity of strict time management and developing and maintaining an efficient work process. I can only thank my wife for having made the daily sacrifices, in particular during the last year, to allow me to give undivided attention and immerse myself fully in the research and writing process.

#### 9.3.5. How Covid-19 affected the research

I was personally fortunate enough to have completed the interviews in Senegal just before the Covid-19 pandemic. The research process was therefore not impacted. However, my overall study subject was dramatically affected. Both ODA, development assistance for the health sector and GHE-S levels spiked (86, 115), a global financial recession unfolded (116), and debt service suspension was offered to many countries by the IMF and WB (117). However, the research questions were at a relatively high level, in a long time perspective, the interview data was collected before Covid-19, and the last year of our panel study is 2019. I was

therefore able to pursue the research more or less as intended, while discussing the impacts of Covid-19 on the financial indicators of interest in the papers where this was relevant.

#### 9.4. Recommendations

#### 9.4.1. Policy recommendations

#### 9.4.1.1. Displacement of GHE-S and OOP by development assistance

Chapter 7 showed how the co-financing mechanism is an available policy tool for counteracting displacement of domestic government health spending when disbursing onbudget ODA<sup>+</sup> for health. However, government redistribution decisions can arguably be considered as rational, sovereign decision making under budget constraint (118, 119) - as long as funds retain some kind of social/development purpose. Co-financing requirements might run counter to the idea of country ownership. Aid fungibility/displacement might also be an issue in one country but not in another. As such, any use of co-financing requirements to avert aid displacement needs to be used A: in a tailored, individual country manner, B: with strong reference to the Paris and Accra principles for development cooperation (120), and C: with the broader fiscal budget dynamics upon receiving ODA in mind, which could in principle benefit the health sector in a given country, although this was not confirmed at the global level.

Importantly, EDPs were also found to subsidise/displace OOP with their ODA<sup>+</sup> for health, both on- and off-budget. This should soften the debate on fungibility: While EDPs to some extent subsidise GHE-S, they also seem to subsidise OOP, in Chapter 6 with comparable effect sizes. The latter is in direct support of UHC and should be an encouraging finding for EDPs in health.

#### 9.4.1.2. Debt relief

The evidence provided at the global level for public external debt servicing shifting the composition of CHE (though no significant association between debt servicing and GHE-S/GDP was found) (Chapter 6), the findings that debt appeared to constrain government health spending in Senegal (Chapter 8), contextualised by the available literature linking increased debt servicing to decreased health spending, warrant the following recommendation.

In some countries, particularly the more indebted such as Senegal, it may be necessary to address the country's debt burden to generate the necessary fiscal space for health to make progress toward UHC, provided governments prioritise freed up funds to the health sector. The findings in this thesis do not support a general interpretation and recommendation that all external debt is bad for the health sector. Rather, to help generate the necessary fiscal space

for health, discussions and solutions to a country's debt burden may need to happen on a case-by-case basis, which could start with the most indebted countries. This is based on the assumption that the effects identified in Senegal may be worse in countries that are more indebted, an assumption that likely depends on a number of political and economic factors and may well not apply in countries that successfully manage to insulate their health sectors from the fiscal impacts of debt repayment obligations. Being able to do so could partly depend on the flexibility of debt repayment timelines, and a role for international creditors could be to make loan repayment timelines more flexible to avoid sensitive budget cuts during the financial year.

The findings in this thesis do not provide grounds for any broad recommendations in terms of choosing between alternative mechanisms for addressing countries' debt burdens where needed, but they do support the interpretation that doing so may help enable a shift toward increased GHE-S and in turn less reliance on OOP, if the health sector is prioritised by the government. Generally speaking, relief of a country's debt burden can be instituted through a variety of mechanisms including debt restructuring, debt cancellation, debt-to-health swaps, expansion of concessional lending to repay more expensive loans and reduce the need for new ones, and more (121, 122). There are different lines of thought about these different alternatives, some advocating debt cancellation from a justice and anticolonial perspective (123, 124), others praising the benefits of debt-to-health swaps as has been facilitated e.g. by the Global Fund to fight Aids, Tuberculosis and Malaria (GFATM), where freed up funds from bilateral debt cancellation are channelled via their organisation to their designated purposes (125, 126). In 2020, the Covid-19 pandemic led to the institution of the G20 Debt Service Suspension Initiative, temporarily pausing debt servicing requirements to help countries cope with the Covid-19 pandemic. This was followed by The G20 Common Framework for Debt Treatments, intended to serve a longer term function of helping restructure the debts for countries facing insolvency and protracted liquidity issues (127). This scheme has been heavily criticised for not providing any debt relief nearly four years after its establishment, for example by the chief economist at the World Bank, Indermit Gill (128). With debt burdens growing, this debate is intensifying, and the lack of agreement reflects entrenched interests on all sides: Bilateral creditor governments, multilateral creditor institutions, private creditors and borrower governments. Eventually, debt servicing levels in many countries have reached or will reach Heavily Indebted Poor Countries (HIPC)/ Multilateral Debt Relief Initiative (MDRI)levels, and resolution will be needed again. The case study of Senegal showed how HIPC debt relief did not clearly lead to increases in GHE-S, and that government education spending could continue to grow in spite of debt pressures, while government health spending showed periods of stagnation and received decreasing budget priority over time. In the case of Senegal, this points to debt-to-health swaps as an important policy tool to ensure the health sector benefits, if the country's debt sustainability worsens over time and debt relief is to be considered. To what extent this recommendation can be extrapolated to other countries, would have to rely on the specific national development agenda and the budget priority given to different sectors. An equity principle would target debt-swaps toward the social and environmental sectors most in need.

Chapter 8 also identified the positive mechanism for government health spending arising from having received external official loans in the first place. The downstream positive effects for the health sector of such loans years later are much harder to quantify but need to be kept in mind as a tempering consideration when debating the present-day issues associated with debt obligations. Importantly, the negative debt implications discussed in this thesis are also expected to be proportional to the degree of concessionality of a loan. The more concessional lending EDPs can make available, the less likely governments are to resort to borrowing money on non-concessional terms, and the less likely negative debt implications are to be seen.

#### 9.4.1.3. Dedicated IMF health spending floor

The finding of arbitrarity and inadequacy of an overall social spending floor in Chapter 8, led to the consideration of the option of a separate spending floor for the health sector instituted by the IMF, in Senegal and potentially elsewhere. If the Abuja target was to be taken seriously, this would be at 15% of General Government Expenditure (GGE). A more pragmatic and incrementalist recommendation would be at 10 or 12% of GGE. In Senegal's case, this would constitute a large expansion of GHE-S, but would still be nowhere near existing targets such as the Chatham House goal set by Di Mcintyre and Filip Meheus of 5% of GDP to make progress towards UHC (86, 129, 130), or their US\$86 per capita target just to provide core primary health care services in LICs (131). Country ownership and buy-in would be essential for the success and viability of a designated health spending floor. To avoid health exceptionalism, it could be combined with a floor on education spending. Starting both at 10% each could be a politically viable starting point.

The IMF could further consider offering improved terms of lending for countries spending more on health, education, the environment and other social purposes. This would incentivise and reward increased social spending, rather than the current practice of a general, non-binding floor on social spending, which has been found to often act more as ceilings on the ambitions of governments, rather than incentives for increased spending (132).

#### 9.4.1.4. Policy recommendations in Senegal

A key policy recommendation for the Senegalese government arises from the following observations in this thesis. A: GHE-S in Senegal has increased very modestly and at times stagnated since 2000. B: GHE-S/General Government Expenditure (GGE) has steadily declined. C: GGE has grown substantially, more than five-doubling in real terms between 2000-2022 (133, 134). D: government spending on education has grown strongly, in spite of any constraining effects from debt on fiscal space or IMF austerity prescriptions. E: EDPs, including the WB, plus the IMF, have promoted increases in GHE-S. Viewed together, and as supported by key informant observations, the logical conclusion is that the government has not given sufficient budget priority to the health sector. The policy recommendation that follows is that the political decision needs to be made to increase the real-term health expenditure budget substantially to meet population needs, honouring the Abuja declaration and Senegal's past and present national health sectoral development strategies and supported by its EDPs. In light of domestic and externally derived non-binding targets having been largely ineffective, a legally binding agreement within existing broader fiscal and development frameworks between the Senegalese government, the IMF, WB and other partners could be considered to boost health spending toward internationally agreed upon targets such as the Abuja pledge (135), ideally from domestic revenue. This would transcend changing governments, priorities and economic circumstance, and ensure that 10-15% of the Senegalese expenditure budget was earmarked to the health sector, providing sustainable and reliable financing for the health sector. Hopefully this would translate into increases in GHE-S/CHE and corresponding decreases in VHI and OOP per CHE, with ensuing improvements to progressivity in the mix of domestic health financing sources and health service access for poor and vulnerable groups. National ownership and motivation would be essential for a such process to avoid being tainted by elements of coercion.

The latest Senegalese government expenditure budget for 2024 allocated 4.6% to the Ministry of Health and Social Action, 10.0% to debt servicing, 5.4% to military spending, 5.5% to the energy sector, of which 3.8 percentage points were allocated to fossil fuels, and 22.7% across three education, training and university/research ministries (136). Addressing the debt servicing share through debt relief and debt-to-health swap mechanisms has been discussed above. Ministerial spending on military and fossil fuels making up twice the amount of health spending invites the question of whether some redistribution from these areas could occur. It is not tempting to recommend a redistribution from education to health, but at present, there is a factor 5 between the budgets of Senegal's three education-related ministries and the Ministry of Health and Social Action. Perhaps the imbalance observed in this particular case invites for this, otherwise uncomfortable, recommendation.

As for the EDPs in Senegal, the conflict seen around decentralised versus departmental-level CBHI has not been conducive to progress in Senegal's ongoing journey toward UHC. The high degree of health financing scheme proliferation and fragmentation seen in Senegal, combined with decentralized CBHI pools, speaks for a need for integration (137). EDPs should show their ongoing support for scheme integration, steered by equity principles of progressive revenue raising, large pools, and cross-subsidy from rich to poor (138). The ongoing integration efforts for departmental-level CBHI align with such principles, and from an equity perspective, EDPs should harmonize and align behind this goal. Further UHC expansion beyond this may likely necessitate a larger government health budget as discussed above, and EDP promotion of this should continue.

#### 9.4.2. Recommendations for future research

A number of different avenues for future research can be identified from the work in the individual chapters and jointly from the thesis as a whole, as outlined in the following sections.

9.4.2.1. Extending and adapting analytical frameworks and regression models A high degree of EDP alignment with clearly formulated government policy was found in Senegal, leading to the uniform application of the analytical framework in Chapter 7. In contexts with poorer governance, donor coordination and alignment, the framework could however be applied individually across multiple EDPs in a country. This would allow one to map out health financing policy incoherence, separated by EDP and by mechanism. This greater granularity might help increase the specificity and utility of resulting policy recommendations as pertaining to individual EDPs. Next steps could also be to integrate political economy co-factors, following e.g. conceptual frameworks by Sparkes et al. (75) or Fox and Reich (2015) (45, 76). While analytically more extensive and complex, it would help identify interactions between local political, social and economic factors, the government and EDPs, allowing for an even more comprehensive understanding of the web of determining factors in domestic health financing policy. One could also integrate the "3-i's" (66) analogous to (45), exploring the motivations for why a given EDP influence is seen.

Attempts at reproduction of existing econometric studies have been very rare in the examined literature, only for a few panel data studies. Reproduction of the panel data study in this thesis to test and challenge the validity of its findings would be warmly welcomed, and the underlying data will be made available upon request.

Specific extensions of the econometric study could include a deeper exploration of the mirrorhypothesis to fungibility, namely the displacement by non-health ODA of domestic government funds from non-health sectors toward the health sector. Questions of government prioritisations under budget constraint from debt servicing obligations could also be further investigated in a set of regression models with sectoral allocations each per GGE as dependent variables, and PPG external debt servicing (or general government debt servicing) as the main independent variable. This would empirically explore which sectors are most affected by debt obligations at the global level.

#### 9.4.2.2. More country case studies of debt impacts on health sector financing

As discussed in Chapters 3 and 8 and Section 9.2 in this chapter, only two reports from Kenya were identified that sought to empirically investigate the impacts of debt on government health spending at the single-country level. As discussed in Chapters 5, 6 and 8, with debt burdens growing across LMICs, more country-level studies are strongly needed to understand fiscal consequences, what these mean for health system financing, equity and population and individual-level health service provision. For countries that have managed to insulate their health spending from debt obligations, what are the explanatory factors that have allowed them to do so, and what lessons can be learned from this for other indebted LMICs, and for the IFIs? Studies of these different relationships could start with the most heavily indebted countries. As an alternative to the in-depth single country lens, they could consist of surveys of a set of heavily indebted countries, at the expense of some contextual depth.

#### 9.4.2.3. Modelling debt relief effects on social spending

As governments, development agencies, International Financial Institutions (IFIs), civil society and populations look for solutions to the looming debt crises in many LMICs, an important role for researchers could be to simulate different policy options. Systematically modelling different debt relief scenarios under a number of transparent assumptions, in particular about how much of relieved debt is allocated to health spending (which could be pre-determined in debtto-health swaps) could help inform decision-making to optimise social benefits and make these more tangible to decisions-makers. They could be combined with modelling of domestic climate funding gains, domestic education funding gains etc., which could all be linked to potential impacts on relevant Sustainable Development Goals (SDG) indicators. This type of study would benefit from an improved empirical understanding of under which circumstances, debt relief translates into increased social spending. Statistically, this would be a question of either effect modification or disaggregated analysis by the political and economic variables deemed to determine differential policy responses to debt relief.

#### 9.4.2.4. Effects of private lending on government health spending in LMICs

This research focused on the roles of external official creditors, i.e. bilateral and multilateral lenders representing single or multiple nation states, however, as seen in Senegal, private lenders were playing an increasing role. The 2023 WB international debt report found among

International Development Association (IDA)-eligible countries that 21% of long-term PPG external debt was owed to private creditors in 2022 compared to 10% in 2012 (80). In 2022, when high inflation rates led to tight monetary policy in HICs increasing the cost of borrowing, private bondholders sold their bonds in LMICs and bought in HICs at higher interest rates, causing the repayment of US\$127 billion from LMIC governments to private bondholders (80). Together with outflows to banks and other private creditors, LMICs ended up paying a total US\$185 billion more as principal repayments to private creditors than they received from them in loans in 2022 (80). These macroeconomic fluctuations illustrate the volatility of private financing, that are bound to have some level of impact during the fiscal year among borrower governments. There are additional moral issues around wealthy, private entities owning the debt of governments with limited resources, as exemplified by the case of BlackRock, the world's largest asset manager, owning 7% of Zambia's government bonds and demanding repayment in full while the country is in a debt crisis (139-141).

When different foreign investment companies own the debt of struggling LMIC governments, debt suspension and restructuring processes are also rendered more challenging than is already the case for an official multilateral debt restructuring process. Multilateral creditors such as the IMF and WB at times end up having to effectively bail out private creditors, issuing new loans to repay old ones (142). Research into and reports of these issues and their social sector impacts have until recently been somewhat confined to civil society (e.g. (23, 143-145)), though with some recent attention from the UN (146). The social and health impacts of private lending to LMIC governments should be the subject of academic investigation. Such research would have high policy relevance in building an understanding of the workings and human consequences of a looming Global South debt crisis (146), and could hopefully even help inform a way out of it.

#### 9.4.2.5. Future research in Senegal

The finding that the Senegalese education budget increased steadily over the past two decades while health budget growth was more limited and at times absent, invites for an exploration of the underlying causes of this discrepancy through a broad political economy lens. Similar methods as the ones used for the case study in this thesis could be used to pursue this question, however a broader framing bringing in more contextual determinants might elicit a more comprehensive understanding of the different drivers.

Further in-depth studies into external influences on domestic health financing from other country creditor lending agreements could also be conducted, beginning with France due to their historically influential position in Senegal.

#### 9.4.2.6. Alternative methodological approaches

#### 9.4.2.6.1. Complex systems modelling

By focusing on mapping mechanisms of effect and evaluating a number of determining variables in regression models, an emergent finding of the research is the complexity of the relationships between EDP+IMF activities, recipient governments, and health financing, between EDPs themselves, and the host of political economy factors that co-determine the central dynamics investigated. In realising and acknowledging this complexity, a complex systems modelling approach to the same or similar research questions appears as a natural extension of the work. Mapping the entire political economy of the health financing sources in a country for instance, including contextual factors, stakeholders, influence pathways with directions of effect (and sizes if possible), and feedback loops, would likely provide a more exhaustive representation. Its policy utility might however be challenged by its sheer complexity, and some degree of balance between complexity and utility would likely need to be sought.

#### 9.5. Conclusion

The evidence provided in this thesis has shown how external financing both for health and non-health purposes is not a neutral injection of funds without domestic repercussions, but rather financing flows that trigger a number of fiscal and macroeconomic dynamics both arising from the financing itself, and from the non-financial exchanges and requirements that follow. This set of relationships help co-determine, along with a range of political and socioeconomic cofactors, to what degree a recipient country relies on pooled and prepaid sources of financing such as government health expenditures, different health insurance mechanisms, or unpooled user fees paid at the point of care. Effects can be delayed and can last over time, stemming from past policy influence. The direction of effect can vary between EDPs and change over time as EDP policies evolve, which could give rise to health financing policy incoherence. These relationships expectedly differ between countries, some may not be present or may not be identifiable at the global level, while they may be deeply influential at the single-country level. Methodologically, the thesis has illustrated how global level observational studies are useful for gaining a broad macro-level understanding of a research question and testing principal economic hypotheses, and how a country-level, case-by-case approach is necessary to be able to make specific observations and inferences, and better justified policy recommendations applicable at the country level.

In Senegal, EDPs have generally supported an expansion of more progressive domestic health financing, promoting the expansion of GHE-S and exemption schemes, SHI and VHI,

and seeking to reduce OOP. Debt obligations, WB promotion of the role of the private health sector, austerity measures from the IMF, past structural adjustment policies promoting user fees and hospital autonomisation, as well as inadequate political priority of the health budget, have been limiting factors in Senegal's journey toward UHC. Supported by its partners, expansion of GHE-S and integration of health financing schemes including consolidation of CBHI will be important steps to increase financial risk sharing and cross subsidy in Senegal.

Finally, the growing indebtedness of many LMICs is worrisome, and debt resolution may be necessary for countries at high risk of or in debt distress. However as illustrated by the case of Senegal, creating fiscal space does not by default translate into increased health spending, and bilateral and multilateral agreements, such as debt-to-health swaps, may be necessary to ensure the health sector benefits from increased fiscal space.

## References

1. Mishra P, Newhouse D. Does health aid matter? J Health Econ. 2009;28(4):855-72.

2. Farag M, Nandakumar AK, Wallack SS, Gaumer G, Hodgkin D. Does funding from donors displace government spending for health in developing countries? Health Aff (Millwood). 2009;28(4):1045-55.

3. Stuckler D, Basu S, McKee M. International Monetary Fund and aid displacement. Int J Health Serv. 2011;41(1):67-76.

4. Xu K, Saksena P, Holly A. The determinants of health expenditure: A country-level panel data analysis. Results for Development Institute WHO; 2011.

5. Gebrehanna S, Upadhyay MP. Fungibility of Foreign Aid in Selected African Countries. Indian Development Review. 2012;10(1):63-80.

6. Van de Sijpe N. The fungibility of health aid reconsidered. 2013.

7. Van de Sijpe N. Is Foreign Aid Fungible? Evidence from the Education and Health Sectors. World Bank Economic Review. 2013;27(2):320-56.

8. Fernandes Antunes Á, Xu K, James CD, Saksena P, Van de Maele N, Carrin G, et al. General budget support: has it benefited the health sector? Health economics. 2013;22(12):1440-51.

9. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

10. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

11. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

12. Barkat K, Mrabet Z, Alsamara M. Does Official Development Assistance for Health from Developed Countries Displace Government Health Expenditure in Sub-Saharan Countries? Economics Bulletin. 2016;36(3):1616-35.

13. Younsi M, Chakroun M, Nafla A. Robust analysis of the determinants of healthcare expenditure growth: evidence from panel data for low-, middle- and high-income countries. International Journal of Health Planning and Management. 2016;31(4):580-601.

14. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

15. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJ. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

16. Shabbir S, Yasin HM. Implications of Public External Debt for Social Spending: A Case Study of Selected Asian Developing Countries. Lahore Journal of Economics. 2015;20(1):71-103.

17. Fosu AK. The External Debt-Servicing Constraint and Public Expenditure Composition. Helsinki, Finland: UNU-WIDER; 2007.

18. Said R, Morai AS. Relationship between Public Debt Burden and Health Expenditure in Sub-Saharan African Countries: The Role of Institutional Quality. Journal of Business and Social Review in Emerging Economies. 2020;6(2):493-502.

19. Fosu AK. The External Debt-Servicing Constraint and Public-Expenditure Composition in Sub-Saharan Africa. African Development Review. 2010;22(3):378-93.

 Lahiani A, Mtibaa A, Gabsi F. Fiscal Consolidation, Social Sector Expenditures and Twin Deficit Hypothesis: Evidence from Emerging and Middle-Income Countries. Comp Econ Stud. 2022;64(4):710-47.
 Kimalu PK, Debt relief and health care in Kenya, World Institute for Development Economics (UNU-

Kimalu PK. Debt relief and health care in Kenya. World Institute for Development Economics (UNU-WIDER). 2002;WIDER Discussion Papers(No. 2002/65, ISBN 9291902551, UNU-WIDER, Helsinki).
 Yiega V. The Impact of Foreign Debt on Economic and Social Rights: A Case Study on the Right to

22. Yiega V. The Impact of Foreign Debt on Economic and Social Rights: A Case Study on the Right to Health in Kenya, Nigeria and Zambia. ADHR (CFS). 2022; (Research Paper 06/06).

23. OXFAM. Policy report: Impact of debt servicing on social spending and wellbeing of low- income households in Kenya: a case study of Nairobi city and Mombasa counties. 2022 [cited 2024 March 26]. Available from: <a href="https://kenya.oxfam.org/latest/publications/impact-debt-servicing-social-spending-and-wellbeing-low-income-households-kenya">https://kenya.oxfam.org/latest/publications/impact-debt-servicing-social-spending-and-wellbeing-low-income-households-kenya</a>.

24. Nguyen TC, Tran TL. The political budget cycles in emerging and developing countries. Journal of Economics and Development. 2023;25(3):205-25.

25. de Haan J, Ohnsorge F, Yu S. Election-Induced Fiscal Policy Cycles in Emerging Market and Developing Economies. SSRN Electronic Journal. 2024.

26. Ebeke C, Ölçer D. Fiscal Policy over the Election Cycle in Low-Income Countries. In: Gaspar V, Gupta S, Mulas-Granados C, editors. Fiscal Politics: International Monetary Fund; 2017.

27. Nguyen TC, Castro V, Wood J. The impact of financial crises on social spending: Delving into the effects in developed and developing countries. International Journal of Finance & Economics. 2023.

28. Fischer SE, Strandberg-Larsen M. Power and Agenda-Setting in Tanzanian Health Policy: An Analysis of Stakeholder Perspectives. Int J Health Policy Manag. 2016;5(6):355-63.

29. Razavi SD, Kapiriri L, Abelson J, Wilson M. Who is in and who is out? A qualitative analysis of stakeholder participation in priority setting for health in three districts in Uganda. Health Policy Plan. 2019;34(5):358-69.

30. Nabyonga Orem J, Marchal B, Mafigiri D, Ssengooba F, Macq J, Da Silveira VC, et al. Perspectives on the role of stakeholders in knowledge translation in health policy development in Uganda. BMC Health Serv Res. 2013;13:324.

31. Khan MS, Meghani A, Liverani M, Roychowdhury I, Parkhurst J. How Do External Donors Influence National Health Policy Processes? Experiences of Domestic Policy Actors in Cambodia and Pakistan. Health Policy and Planning. 2018;33(2):215-23.

32. Koduah A, van Dijk H, Agyepong IA. The role of policy actors and contextual factors in policy agenda setting and formulation: maternal fee exemption policies in Ghana over four and a half decades. Health Res Policy Syst. 2015;13:27.

33. Onoka CA, Hanson K, Hanefeld J. Towards universal coverage: a policy analysis of the development of the National Health Insurance Scheme in Nigeria. Health Policy Plan. 2015;30(9):1105-17.

34. Alawode G, Adewoyin AB, Abdulsalam AO, Ilika F, Chukwu C, Mohammed Z, et al. The Political Economy of the Design of the Basic Health Care Provision Fund (BHCPF) in Nigeria: A Retrospective Analysis for Prospective Action. Health Syst Reform. 2022;8(1):2124601.

35. Khalid F, Afzal S, Shirazi A, Quick I, Mataria A. Coordinating external assistance for UHC: Pakistan's early experience of the SDG3 GAP. Health Policy and Planning. 2024;39(2):247-51.

36. Herberholz C. The role of external actors in shaping migrant health insurance in Thailand. PLoS One. 2020;15(7):e0234642.

37. Ir P, Bigdeli M, Meessen B, Van Damme W. Translating knowledge into policy and action to promote health equity: The Health Equity Fund policy process in Cambodia 2000-2008. Health Policy. 2010;96(3):200-9.
38. Mulat AK, Mao W, Bharali I, Balkew RB, Yamey G. Scaling up community-based health insurance in

Ethiopia: a qualitative study of the benefits and challenges. BMC Health Serv Res. 2022;22(1):473.
39. Kiendrébéogo JA, Meessen B. Ownership of health financing policies in low-income countries: a journey

with more than one pathway. BMJ Glob Health. 2019;4(5):e001762. 40. Shiffman J. Knowledge, moral claims and the exercise of power in global health. Int J Health Policy Manag. 2014;3(6):297-9.

41. Bazbauers AR. World Bank Technical Assistance: The Relational Dynamics of Policy Movement. Public Administration and Development. 2017;37(4):246-59.

42. Palier B, Mandin L. Le rôle des organisations internationales dans les réformes des systèmes nationaux de protection sociale. In: Schemeil Y, Wolf-Dieter E, editors. Normer le monde: L'Harmattan; 2009. p. 177-210.
43. Eboko F. Vers une matrice de l'action publique en Afrique ? Approche trans-sectorielle de l'action

publique en Afrique contemporaine. 2015 [cited 2024 March 27 ]. Available from: https://journals.openedition.org/lectures/17120.

44. Nagemi C, Mwesigwa C. Donor Power and Prioritization in Development Assistance for Health Policies: The Case of Uganda. Journal of Developing Economies. 2021;2(1):54-74.

45. Fox AM, Reich MR. The Politics of Universal Health Coverage in Low- and Middle-Income Countries: A Framework for Evaluation and Action. Journal of Health Politics, Policy and Law. 2015;40(5):1023-60.

46. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs. International Monetary Fund; 2004.

47. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

48. IMF Independent Evaluation Office. Fiscal Adjustment in IMF-Supported Programs. 2003 [cited 2024 March 19]. Available from: <u>https://www.imf.org/external/np/ieo/2003/fis/pdf/main.pdf</u>.

49. Clements B, Gupta S, Nozaki M. What Happens to Social Spending in IMF-Supported Programs? . 2011. Available from: <u>http://www.imf.org/external/pubs/ft/sdn/2011/sdn1115.pdf</u>.

50. IMF. Social safeguards and program design in PRGT and PSI-supported programs. 2017 [cited 2024 March 20]. Available from: <u>https://www.imf.org/en/Publications/Policy-Papers/Issues/2017/06/01/pp042117social-safeguards-and-program-design-in-prgt-and-psi</u>.

51. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

52. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

53. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

54. Boachie MK, Agyemang J, Immurana M. Health sector funding in Ghana: The effect of IMF conditionalities. Dialogues Health. 2022;1:100045.

55. Frimpong AÖ, Amporfu E, Arthur E. Effects of public and external health spending on out-of-pocket payments for healthcare in sub-Saharan Africa. Health Policy Plan. 2022;37(9):1129-37.

56. Ali MEM, Ebaidalla M, Rizk R. Foreign aid and out-of-pocket health expenditure in Sub-Saharan Africa: does institutional development matter. International Journal of Sustainable Economy. 2020;12(3).

57. Gabani J, Suhrcke M, Neelsen S, Eozenou PH, Smitz MF. Does health aid matter to financial risk protection? A regression analysis across 159 household surveys, 2000-2016. Social science & medicine (1982). 2024;356:117148.

Cormier B, Manger MS. The Evolution of World Bank Conditionality: A Quantitative Text Analysis. 2020.
 Vetterlein A. Economic Growth, Poverty Reduction, and the Role of Social Policies: The Evolution of the World Bank's Social Development Approach. Global Governance. 2007;13(4):513-33.

60. Jacobs E. The politics of the basic benefit package health reforms in Tajikistan. Glob Health Res Policy. 2019;4:14.

61. Kingdon JW. Agendas, alternatives, and public policies: Boston : Little, Brown, [1984] ©1984; 1984. ISBN:

62. Walt G, Gilson L. Reforming the health sector in developing countries: the central role of policy analysis. Health Policy Plan. 1994;9(4):353-70.

63. Grindle M, Thomas JW. Public choices and policy change : the political economy of reform in developing countries. Baltimore: Johns Hopkins University Press; 1991. ISBN: 0801841550

64. Gautier L, Ridde V. Health financing policies in Sub-Saharan Africa: government ownership or donors' influence? A scoping review of policymaking processes. Glob Health Res Policy. 2017;2:23.

65. Lindquist EA. Discerning Policy Influence: Framework for a Strategic Evaluation of IDRC-Supported Research. 2001 [cited April 9 2024]. Available from:

https://www.betterevaluation.org/sites/default/files/discerning\_policy.pdf.

66. Hall PA. The Role of Interests, Institutions, and Ideas in the Comparative Political Economy of the Industrialized Nations Cambridge University Press 1997. ISBN: 0521586682

67. National Collaborating Centre for Healthy Public Policy. Understanding Policy Developments and Choices Through the "3-i" Framework: Interests, Ideas and Institutions. 2014 [cited 2024 July 29]. Available from: https://ccnpps-ncchpp.ca/docs/2014\_ProcPP\_3iFramework\_EN.pdf.

68. Mulvale G, McRae SA, Milicic S. Teasing apart "the tangled web" of influence of policy dialogues: lessons from a case study of dialogues about healthcare reform options for Canada. Implement Sci. 2017;12(1):96.

69. Overseas Development Institute (ODI). ROMA: A guide to policy engagement and influence. 2014 [cited April 9 2024]. Available from: <u>https://odi.cdn.ngo/media/documents/odi roma guide.pdf</u>.

70. Start D, Hovland I. Tools for Policy Impact: A Handbook for Researchers. 2004 [cited 2024 July 29]. Available from: <u>https://cdn.odi.org/media/documents/194.pdf</u>.

71. Crewe E, Young J. Bridging Research and Policy: Context, Evidence and Links. 2002 [cited 2022 July 6]. Available from: <u>https://cdn.odi.org/media/documents/184.pdf</u>.

72. Cathexis Consulting. A Guide to Policy-Influence Evaluation: Selected Resources and Case Studies. 2015 [cited April 9 2024]. Available from:

https://www.nccmt.ca/uploads/media/0001/01/c3374b8be4b35e1340385f1b593d3bb9f50f6a38.pdf.

73. De Raeve P, Bolzonella F, Davidson PM. A Theoretically Derived Approach to Impact: Implementing Policy Influence Strategies. Policy Polit Nurs Pract. 2022:15271544221100164.

74. Steinberg PF. Understanding Policy Change in Developing Countries: The Spheres of Influence Framework. Global Environmental Politics. 2003;3(1):11-32.

75. Sparkes SP, Bump JB, Ozcelik EA, Kutzin J, Reich MR. Political Economy Analysis for Health Financing Reform. Health Syst Reform. 2019;5(3):183-94.

76. Rizvi SS, Douglas R, Williams OD, Hill PS. The Political Economy of Universal Health Coverage: A Systematic Narrative Review. Health Policy and Planning. 2020;35(3):364-72.

77. Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 2007;8(4):299-310.

78. Behera DK, Dash U. Impact of macro-fiscal determinants on health financing: empirical evidence from low-and middle-income countries. Glob Health Res Policy. 2019;4:21.

79. The Lancet. Economic storms threaten to cast health adrift. Lancet. 2024;403(10424):325.

80. Bank W. International Debt Report 2023. 2023 [cited 2024 February 22]. Available from: <u>https://www.worldbank.org/en/programs/debt-statistics/idr/products</u>.

81. Fan V, Gupta S. What's Rising Debt Got to Do With Health Spending? 2023 [cited 2023 September 27]. Available from: <u>https://www.cgdev.org/blog/whats-rising-debt-got-do-health-spending</u>.

82. United Nations DESA FSDO. Financing for Sustainable Development Report 2024. 2024 [cited 2024 April 17]. Available from: <a href="https://desapublications.un.org/publications/financing-sustainable-development-report-2024">https://desapublications.un.org/publications/financing-sustainable-development-report-2024</a>.

83. Jerven M. Poor Numbers: How we are misled by African development statistics and what to do about it. 1 ed: Cornell University Press; 2013. 187 p. ISBN: 978-0-8014-7860-4

84. Pitt C, Grollman C, Martinez-Alvarez M, Arregoces L, Borghi J. Tracking aid for global health goals: a systematic comparison of four approaches applied to reproductive, maternal, newborn, and child health. The Lancet Global Health. 2018;6(8):e859-e74.

85. Learner EE. Let's Take the Con Out of Econometrics. The American Economic Review. 1983;73:31-43.
86. WHO. Global Health Expenditure Database. 2024 [cited 2024 March 28]. Available from: http://apps.who.int/nha/database.

87. Young R, Johnson DR. Handling Missing Values in Longitudinal Panel Data With Multiple Imputation. J Marriage Fam. 2015;77(1):277-94.

88. Roodman D. Doubts about the evidence that foreign aid for health is displaced into non-health uses. The Lancet. 2012;380(9846):972-3.

89. Liang L-L, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

90. Milsom P, Smith R, Baker P, Walls H. Corporate power and the international trade regime preventing progressive policy action on non-communicable diseases: a realist review. Health Policy Plan. 2021;36(4):493-508.

91. Lukes S. Power: A Radical Viiew. 2nd ed: Palgrave Macmillan; 2005. ISBN: 978-0-333-42092-8 978-0-230-80257-5

92. Bachrach P, Baratz MS. Two Faces of Power. The American Political Science Review. 1962;56(4):947-52.

93. Alesina A, Dollar D. Who Gives Foreign Aid to Whom and Why? Journal of Economic Growth. 2000;5(1):33-63.

94. Matteis AD. Whose poverty really matters when deciding aid volumes? International Journal of Public Policy. 2016;12(1/2).

95. Steele CA. Public Goods and Donor Priorities: The Political Economy of Development Aid for Infectious Disease Control. Foreign Policy Analysis. 2017;13(4):986-1002.

96. Rye Olsen G. European public opinion and aid to Africa: is there a link? The Journal of Modern African Studies. 2002;39(4):645-74.

97. Gaventa J. Finding the Spaces for Change: A Power Analysis. IDS Bulletin. 2006;37(6):23-33.

98. Naylor T. Deconstructing Development: The Use of Power and Pity in the International Development Discourse1. International Studies Quarterly. 2011;55(1):177-97.

99. Preker AS, Carrin G. Health Financing for Poor People: The World Bank; 2004. 480 p. ISBN: 978-0-8213-5525-1

100. World Bank. Investing in Health. 1993 [cited 2024 July 29]. Available from:

https://openknowledge.worldbank.org/handle/10986/5976.

101. Erasmus E, Gilson L. How to start thinking about investigating power in the organizational settings of policy implementation. Health Policy Plan. 2008;23(5):361-8.

102. Fergus CA. Power across the global health landscape: a network analysis of development assistance 1990-2015. Health Policy Plan. 2022;37(6):779-90.

103. Swedlund HJ. From Donorship to Ownership? Budget Support and Donor Influence in Rwanda and Tanzania. Public Administration and Development. 2013;33(5):357-70.

104. Rebien CC. Participatory Evaluation of Development Assistance. Evaluation. 2016;2(2):151-71.

105. Elbers W, Schulpen L. Corridors of Power: The Institutional Design of North-South NGO Partnerships. Voluntas: International Journal of Voluntary and Nonprofit Organizations. 2013;24(1):48-67.

106. Ashman D. Strengthening North-South Partnerships for Sustainable Development. Nonprofit and Voluntary Sector Quarterly. 2016;30(1):74-98.

107. Kruckenberg LJ. North–South partnerships for sustainable energy: Knowledge–power relations in development assistance for renewable energy. Energy for Sustainable Development. 2015;29:91-9.
108. Roller M. Interviewer bias and reflexivity in gualitative research. 2018 [cited 2018 September 10].

Available from: <u>https://researchdesignreview.com/2012/11/14/interviewer-bias-reflexivity-in-qualitative-research/</u>. 109. Amy Niang. 2024 Senegal election crisis points to deeper issues with Macky Sall and his preferred successor. 2024 [cited 2024 April 24]. Available from: <u>https://theconversation.com/2024-senegal-election-crisis-</u>

points-to-deeper-issues-with-macky-sall-and-his-preferred-successor-223035.
 110. Elodie Toto. How Macky Sall Provoked a Constitutional Crisis in Senegal. 2024 [cited 2024 May 5].
 Available from: https://foreignpolicy.com/2024/03/01/senegal-election-macky-sall-sonko-faye-constitutional-coup/.

Haque N. From prisoner to president in 20 days, Senegal's Diomaye Faye takes office. 2024 [cited 2024 May 5]. Available from: <u>https://www.aljazeera.com/features/2024/4/2/from-prisoner-to-president-in-20-days-senegals-diomaye-faye-takes-office</u>.

Amnesty International. Sénégal. Il faut enquêter sur les homicides et les violences policières à 112 l'encontre des anifestant e s. 2024 [cited 2024 May 5]. Available from:

https://www.amnesty.org/fr/latest/news/2024/02/senegal-investigate-killings-and-police-brutality-againstprotesters/.

Hammersley M, Atkinson P. Ethnography. 3rd ed: Routledge (Taylor & Francis Group); 2007. ISBN: 113.

Green J, Thorogood N. Qualitative Methods for Health Research: SAGE Publications; 2004. ISBN: 114.

115. OECD. Creditor Reporting System (CRS). 2024 [cited 2023 February 16]. Available from: https:// stats.oecd.org/Index.aspx?DataSetCode=CRS1.

World Bank. From Double Shock to Double Recovery – Implications and Options for Health Financing in 116. the Time of COVID-19. 2021 [cited 2022 March 22]. Available from:

https://openknowledge.worldbank.org/bitstream/handle/10986/35298/From Double Shock to Double Recovery %20Revision%2020210916.pdf?sequence=7&isAllowed=y.

World Bank. Debt Service Suspension Initiative. 2022 [cited 2024 May 7]. Available from: 117. https://www.worldbank.org/en/topic/debt/brief/covid-19-debt-service-suspension-initiative

Rana Z, Koch D-J. Is it time to 'decolonise' the fungibility debate? Third World Quarterly. 2019;41(1):42-118. 57.

119. Martinez Alvarez M, Borghi J, Acharya A, Vassall A. Is Development Assistance for Health fungible? Findings from a mixed methods case study in Tanzania. Social science & medicine (1982). 2016;159:161-9. OECD. The Paris Declaration on Aid Effectiveness and the Accra Agenda for Action. 2005, 2008 [cited 120.

2024 July 29]. Available from: https://web-archive.oecd.org/temp/2021-08-02/73869parisdeclarationandaccraagendaforaction.htm.

121. United Nations. Our Common Agenda - Policy Brief 6: Reforms to the International Financial Architecture. 2023 [cited 2024 May 6]. Available from: https://www.un.org/sites/un2.un.org/files/our-commonagenda-policy-brief-international-finance-architecture-en.pdf.

122 Wuennenberg L. Debt-To-Health Swaps: Financing health system resilience beyond the COVID-19 pandemic. 2022 [cited 2022 March 14]. Available from: https://www.iisd.org/sustainable-recovery/debt-to-healthswaps-financing-health-system-resilience-beyond-the-covid-19-pandemic/

Debt Justice. Global South Debt. 2024 [cited 2024 May 8]. Available from: 123.

https://debtjustice.org.uk/campaigns/no-new-debt-trap.

Justice D. Colonialism and Debt. 2024 [cited 2024 May 8]. Available from: 124.

https://debtjustice.org.uk/campaigns/colonialism. 125. Hu Y, Wang Z, Zhou S, Yang J, Chen Y, Wang Y, et al. Redefining Debt-to-Health, a triple-win health financing instrument in global health. Global Health. 2024;20(1):39.

126. The Global Fund to Fight AIDS TaM. Debt2Health: Collaboration Through Financial Innovation. Year not specified [cited 2024 May 8]. Available from:

https://www.theglobalfund.org/media/12284/publication\_debt2health\_overview\_en.pdf.

Georgieva K, Pazarbasioglu C. The G20 Common Framework for Debt Treatments Must Be Stepped 127. Up. 2021 [cited 2023 September 28]. Available from:

https://www.imf.org/en/Blogs/Articles/2021/12/02/blog120221the-g20-common-framework-for-debt-treatmentsmust-be-stepped-up.

Elliott L. World Bank official calls for shake-up of G20 debt relief scheme. 2024 [cited 2024 May 8]. 128 Available from: https://www.theguardian.com/business/2024/apr/21/world-bank-chief-economist-indermit-gill-g20debt-relief-mechanism-common-framework.

McIntyre D, Meheus F, Rottingen JA. What level of domestic government health expenditure should we 129. aspire to for universal health coverage? Health Econ Policy Law. 2017;12(2):125-37.

McIntyre D, Meheus F. Fiscal Space for Domestic Funding of Health and Other Social Services. 2014 130. [cited 2019 September 17].

McIntyre D, Meheus F, Røttingen JA. What level of domestic government health expenditure should we 131. aspire to for universal health coverage? Health Econ Policy Law. 2017;12(2):125-37.

Kentikelenis A, Stubbs T. IMF Social Spending Floors: A fig leaf for austerity? 2023 [cited 2023 132. September 29]. Available from: https://oxfamilibrary.openrepository.com/bitstream/handle/10546/621495/bp-imfsocial-spending-floors-130423-en.pdf?sequence=4.

133. IMF. Public Finances in Modern History. 2024 [cited 2024 June 10]. Available from: https://www.imf.org/external/datamapper/datasets/FPP.

World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: 134. https://data.worldbank.org/.

Organisation of African Unity. Abuja Declaration on HIV/AIDS, Tuberculosis and Other Related 135. Infectious Diseases. Abuja, Nigeria; 2001.

Direction Générale du Budget. Loi n°2023-18 du 15 decembre 2023 portant loi de finances pour l'annee 136. 2024. 2024 [cited 2024 August 11]. Available from: https://budget.sec.gouv.sn/recherche-par-typesdocuments/lois-de-finances-initiales

Daff BM, Diouf S, Diop ESM, Mano Y, Nakamura R, Sy MM, et al. Reforms for financial protection 137. schemes towards universal health coverage. Senegal. Bull World Health Organ. 2020;98(2):100-8.

WHO. Health systems financing: the path to universal coverage. 2010 [cited 2024 May 29]. Available 138. from: https://www.who.int/publications/i/item/9789241564021.

139. Investopedia. Vulture Fund: What it is, How it Works, Examples. 2022 [cited 2024 May 3]. Available from: https://www.investopedia.com/terms/v/vulturefund.asp.

140. The People's Forum. Press release: Demand BlackRock Cancel Zambia's Debt. 2024 [cited 2024 May 3]. Available from: <u>https://peoplesforum.org/blog\_post/press-release-demand-blackrock-cancel-zambias-debt/</u>.

141. Debt Justice. BlackRock could make 110% profit out of Zambia's debt crisis. 2022 [cited 2024 May 3]. Available from: <u>https://debtjustice.org.uk/press-release/blackrock-could-make-110-profit-out-of-zambias-debt-crisis</u>.

142. Jubilee Debt Campaign. IMF loans bailing out private lenders during the Covid-19 crisis. 2020 [cited 2024 May 3]. Available from: <u>https://debtjustice.org.uk/wp-content/uploads/2020/07/IMF-bailouts-</u>briefing 07.20.pdf.

143. Development Finance International. The worst ever global debt crisis. 2023 [cited 2024 March 26]. Available from:

https://assets.nationbuilder.com/eurodad/pages/3195/attachments/original/1696947958/Debt Service Watch Bri efing Final Word EN 0910.pdf?1696947958.

144. Jubilee Debt Campaign. Sixty-four countries spend more on debt payments than health. 2020 [cited 2022 March 14]. Available from: <u>https://jubileedebt.org.uk/press-release/sixty-four-countries-spend-more-on-debt-payments-than-health</u>.

145. OXFAM. No country should be forced to choose between paying back debts or providing health care. 2024 [cited 2024 March 26]. Available from: <u>https://www.oxfam.org/en/no-country-should-be-forced-choose-between-paying-back-debts-or-providing-health-care</u>.

146. United Nations Global Crisis Response Group. A world of debt: A growing burden to global prosperity. 2023 [cited 2024 March 25]. Available from: <u>https://www.un.org/regionalcommissionsnyoffice/news/launch-un-report-%E2%80%9C-world-debt%E2%80%9D</u>.

# 10. Appendices

# 10.1. Appendix 1: Literature review methods and detailed literature summaries

#### 10.1.1. Literature review methods

The literature collected in this thesis is the product of a series of literature reviews conducted at the different stages of the research. A formal, systematic literature review was performed during the initial design of the study in 2018 as described below. Over the course of the study, two purposive document reviews were conducted for Chapters 7 and 8, and ad hoc background literature reviews for Chapters 5-8. A final updated literature review was then conducted when writing up the thesis.

The initial systematic literature review in 2018 involved the following methods. Econlit, Medline, Embase and Global Health databases were searched for relevant English language literature describing relationships between development assistance for health, debt and loan conditionalities on one side and different sources of financing for health on the other - government health expenditure, OOP and voluntary health financing contributions. No time period constraints were applied to this search. The different search strings applied are presented in Table 10.1. Studies were included for review if the title or abstract mentioned any of these dynamics and excluded if any of these were not a focus of the full-text article. Article references were also reviewed for further relevant literature, some articles were referred from supervisors, and Google was also searched ad hoc for grey literature.

Database(s)	Search String	Results after duplicate	Of which					
		removal	relevant					
Development Assistance for Health and Government Health Expenditure								
Medline, Embase, Global Health, Econlit	(DAH or "development assistance" or fungibility or displacement) and ((government\$ or public or domestic) adj2 (expenditure\$ or spending or financ\$ or investment\$)) and health	161	21					
Development Assistance for Health and Out Of Pocket Expenditures								
Medline, Embase, Global Health, Econlit	(DAH or "development assistance" or fungibility or displacement) and ("out-of-pocket" or OOP or OPP or "out 307 of pocket") and health		0					
Development Ass	sistance for Health and Voluntary Health care Payn	nent Schemes						
Medline, Embase, Global Health, Econlit	(DAH or "development assistance") and (((voluntary or private or pool\$) adj3 (expenditure\$ or spending or financ\$ or contribution\$ or fund\$)) or (insurance or prepaid)) and health	22	0					
Public External D	Debt and Government Health Expenditure							
Medline, Embase, Global Health, Econlit	((public\$ or PPG or government\$) and (debt or loan\$ or lending)) and (("government" or public or domestic) adj1 health adj1 (spending or expenditure\$) or GHE or TGHE or GHE-\$ or \$GGHE\$)	19	4					
Public External D	Debt and Out Of Pocket Expenditures							
Medline, Embase, Global Health, Econlit	"(public\$ or PPG or government\$) and (debt or loan\$ or lending) and (OOP or \$pocket\$))"	91	0					
Google Scholar	1: Same as above. 2: ("external public debt" OR "public external debt" OR "PPG debt" OR "PPG external debt" OR "external PPG debt") AND ("out-of-pocket" OR OOP OR OPP OR "out of pocket")	1: 141 2: 89	1: 0 2: 0					
Public External Debt and Voluntary Health care Payment Schemes								
Medline, Embase, Global Health, Econlit	"(public\$ or PPG or government\$) and (debt or loan\$ or lending) and (((voluntary or private or pool\$) adj3 (expenditure\$ or spending or financ\$ or contribution\$ or fund\$)) or (insurance or prepaid)) and health"	295	0					
Senegal's aid and debt history								
Medline, Embase, Global Health, Econlit	Senegal and (DAH or "development assistance" or "foreign aid" or debt or loan\$ or lending or "structural adjustment" or SAP or SAF or "poverty reduction" or PRSP or PRGF or "heavily indebted" or HIPC or MDRI	118	1					

Table 10.1: Search strategies applied at the design stage of the research in 2018.

Slightly modified versions of the above search strings in Table 10.1 were run again on PubMed and Econlit between April 5-9, 2024, for 2018 and onward, to capture any new relevant literature missed in the preceding reviews. A new search string was also added<sup>41</sup> homing more

<sup>&</sup>lt;sup>41</sup> ("development partner" or "development partners" or EDP or EDPs or "financial partner" or "financial partners" or TFP or TFPs or donor or creditor\* or lender\*) and (impact\* or influence\* or effect\* or affect\* or power\* or determin\*) and health and (financ\* or fund\* or contribut\* or expenditur\* or spending\* or OOP or "out-of-pocket" or "out of pocket") NOT (transplant\* or transfus\* or blood\* or milk\* or methyl\* or tissue\* or heart\* or cardiac\* or eyes\* or kidney\* or liver\* or eggs\* or oocyte\* or teeth\* or cancer\* or syndrom\* or neuro\* or metabol\* or molecul\* or serol\* or microbio\* or nutrit\* or biotech\* or vaccin\* or workforc\* or nurs\*)

directly in on EDP influence on domestic health financing contributions, and one<sup>42</sup> on IMF/WB's influence on the same, both without time limitations. This added 3402 new records, where titles and abstracts were screened for potential relevance to the research questions, identifying 60 additional papers for full-text review. References were also reviewed to identify more relevant papers. Papers that directly pertained to the research questions were integrated into the corresponding sections of Chapter 3.

To be able to create the analytical framework used in Chapter 7, a review of the literature on existing frameworks for analysing development partner influence was performed. This review was performed using the LSHTM Discover platform, which cross-searches a range of different databases, as well as Google Scholar, using combinations of keywords such as development, partner, assistance, framework, theoretical, conceptual, influence, impact, health financing and more. Frameworks cited in identified papers were also reviewed. This literature is described in the corresponding section in Chapter 3.

The two purposive document reviews in Chapters 7 and 8 were part of the research methods, and the relevant identified literature is described in the results and discussion sections of those chapters, while the purposive document review methods are described in the methods chapter of this thesis and Chapters 7 and 8.

Literature on HICs was excluded from all the searches as the research questions pertained to LMICs, and HICs do not have external development partners in the traditional sense of the term, nor do they receive ODA.

# 10.1.2. Multi-country panel data studies of associations between development assistance for the health sector and government health expenditure

Table 10.2 summarises the identified studies of associations between development assistance for the health sector and government health expenditure:

Study	Time span	No. of countries	Development assistance for health sector data source	GHE data source	Main method of estimation	GHE change from \$1 increase in development assistance for the health sector
Mishra and Newhouse (2009) (1)	1985- 2004	118	OECD CRS	IMF (GHE-S/A not stated)	OLS	+\$1.50

<sup>&</sup>lt;sup>42</sup> (IMF or "International Monetary Fund" or "World Bank" or SAP or "structural adjustment" or "loan conditionalities") and ("health financing" or "health expenditure" or "health spending" or "fiscal space for health")

Farag et al. (2009) (2)	1995- 2006	144	WHO NHA (channel not specified)	NHA (GHE-S/A not stated)	two-way, fixed-effect GLS	LICs: -27 cents (95% CI: 15–40) MICs: -63 cents (95% CI: 32–96)
Lu et al. (2010) (3)	1995- 2006	111	OECD CRS and major donor reports (government and non- government channel)	IMF, NHA ( <i>GHE-S</i> )	GMM	-\$0.43 to -\$1.14. (government aid channel)
Stuckler et al. (2011) (4)	1996- 2006	119	WHO NHA (channel not specified)	NHA (GHE-S/A not stated)	Not stated	IMF-borrowing: -\$0.998 (SE: 0.22) Non-IMF-borrowing: -\$0.55 (SE: 0.11)
Xu et al. (2011) (5)	1995- 2008	143	WHO NHA (channel not specified)	NHA ( <i>GHE-S</i> )	FE + dynamic models	Percent elasticity: 1% increase in external health financing → LICs: -0.26 to -0.19% Lower-MICs: -0.033% or no sign. change UMICs: no sign. change in GHE-S
Gebrehanna and Upadhyay (2012) (6)	1980- 2005	9 (SSA)	OECD (concessional loans to health sector)	IMF <i>(GHE-A)</i>	Iterative GLS	Percent elasticity: 1 percentage point increase in concessional loans to health sector as % of GDP → +1.22 percentage points
						increase in GHE-A as % of GDP (SE: 0.278)
Van de Sijpe (2013a) (7)	1995- 2006	111	OECD CRS (channelled via government)	IMF <i>(GHE-S)</i>	FE model	"Little or no displacement"
Van de Sijpe (2013b) (8)	1990- 2003	108	OECD CRS (general)	IMF <i>(GHE-A)</i>	FE model	Percent elasticity: 1 percentage point increase in aid for health/GDP → +0.26 (0.12) percentage points increase in GHE-A, however " <i>little if any</i> <i>fungibility</i> " when disaggregating by "on- and off-budget" aid.
Fernandes Antunes et al. (2013) (9)	2002- 2007	82	WHO NHA + OECD-DAC (channel not specified)	NHA (GHE-S)	FE + dynamic models	FE: -\$0.41 (SE: 0.069). Insignificant in dynamic model
Dieleman et al. (2013) (10)	1995- 2010	134	IHME (DAH-G and DAH-NG)	NHA (GHE-S)	Two-step system GMM	-\$0.86 (95% CI: 0.63-1.08) (DAH-G)
Dieleman et al. (2014) (11)	1995- 2010	119	IHME ( <i>DAH-G</i> )	NHA (GHE-S)	Two-step system GMM	-\$0.62 (90% CI: 0.15-1.09) (DAH-G)
Liang et al. (2014) (12)	1995- 2010	120	IHME (DAH-G and DAH-NG)	NHA (GHE-S)	FE-2SLS	Percent elasticity: 1% increase in DAH-G → -0.03 to -0.04% change in GHE-S
Barkat et al. (2016) (13)	1995- 2012	45 SSA countries	OECD CRS	NHA (GHE-S/A not stated)	GMM, FE-IV	Percent elasticity: $1\%$ increase in development assistance for health sector $\rightarrow$

						+0.04% to 0.1% change in GHE
Younsi et al. (2016) (14)	1993- 2013	113 LMICs	World Bank <i>(general)</i>	World Bank <i>(GHE-S)</i>	Static and dynamic models, GMM	Percent elasticity: 1% increase in development assistance for health sector → -0.19 to -0.25% change in GHE-S among LICs (no sizeable effect among MICs)
Patenaude (2021) (15)	2000- 2015	134 LMICs	IHME (DAH-G and DAH-NG)	<i>GHE-S,</i> source unclear (NHA?)	System GMM	Percent elasticity: 1% increase in DAH-G → 0.001-0.002% increase (SE: 0.001-0.002%)

Table 10.2: Summary of multi-country panel data studies of associations between development assistance for the health sector and Government Health Expenditure (GHE). DAH-G: Development Assistance for Health (DAH) channelled via government. DAH-NG: DAH via non-governmental channel. GHE-S: Government Health Expenditure as A Source. GHE-A: Government Health Expenditure as Agent. GLS: Generalized Least Squares, GMM: Generalized Method of Moments, GHE-S: Government Health Expenditure as Source, GHE-A: Government Health Expenditure as Source, GHE-A: Government Health Expenditure as Agent. FE-2SLS: Fixed-effects two-stage least squares regression, FE-IV: Fixed Effects Instrumental Variables.

The first panel data study of displacement effects of development assistance for the health sector on GHE, was by Mishra and Newhouse (2009) (1). Based on OECD CRS and IMF data between 1985 and 2004, the authors found a \$1 increase in development assistance for the health sector to be associated with a \$1.50 increase in GHE. The authors did not include development assistance to reproductive health and sexually transmitted infection control including HIV/AIDS. Farag et al. (2009) included 144 countries from 1995-2006 and found that a \$1 increase in development assistance for the health sector was associated with a 27 cent decrease in GHE in LICs 63 cents decrease in MICs on average (2). Lu et al. (2010) found an elasticity of -\$0.43 to -\$1.14 Government Health Expenditure as Source (GHE-S) for every dollar of development assistance for the health sector received among 111 LMICs over the same time period (3). This study led to serious debate:

Sanjeev Gupta from the IMF argued that this finding should not cause too much concern, as substitution of spending to other sectors e.g. education and sanitation may improve health, higher spending does not necessarily translate into better outcomes, and development assistance for the health sector used to promote macroeconomic stability could lead to higher growth, which in turn would increase a country's revenue base for future spending and could itself improve social indicators (16). Ooms et al. (2010) discussed aid volatility and unpredictability as potential determinants of the observed fungibility, which would thereby function as a financial buffering mechanism, and argued for the establishment of a shared global fund for health to address this (17). David Roodman (2012) raised methodological concerns regarding multiple imputations, erroneous instrumenting and instrument proliferation (18). Batniji and Bendavid (2013) claimed that the conclusions were driven by outliers and highlighted inconsistencies between IMF and WHO GHE-S data, however this paper was

retracted due to statistical errors (19). In a following perspective piece, they argued that the heterogeneity of ODA displacement and the inaccuracy of public health financing data renders these aggregate results inappropriate as a basis for health aid prioritization, and that fungibility is not an inherently adverse phenomenon in that it may merely reflect rational government prioritization across multiple sectors (20) (also discussed by Ooms et al. (2010) (17)). A series of other comments were made that were addressed by Lu et al. (21-29). The debate appeared to settle with Murray, Dieleman, Lu and Hanlon (2013) arguing for the need for improvements in sector-level public expenditure data and statistical methods to be able to investigate the topic of aid fungibility in further detail (30), however this conclusion was followed up with further studies by Dieleman, Hanlon et al, in part in response to a reanalysis of Lu et al.'s data by Van de Sijpe (2013a) (7). Van de Sijpe performed a series of sensitivity tests on the data taking into account "off-budget" aid and found "little or no displacement of health expenditure from own resources" (7). Dieleman et al. responded to the raised concerns and analysed an expanded panel of 134 countries between 1995-2010, and found that a \$1 increase in Development Assistance for Health (DAH) to Government (DAH-G, as opposed to DAH to non-government recipients: DAH-NG) was still associated with a \$0.86 decrease (95% CI: 0.63-1.08) in GHE-S (10). In 2014, analysing a smaller panel of 119 countries between 1995-2010, the authors found that a \$1 increase in DAH-G was associated with a \$0.62 decrease (90% CI: 0.15-1.09) in GHE-S, whereas a \$1 decrease was not significantly associated with GHE-S (11). Providing another aspect to the above debate, Stuckler et al. (2011) found that among IMF-borrowing countries, development assistance for the health sector was completely fungible, whereas in non-borrowing countries, 55 cents on the dollar of was displaced (4). Further studies of aid fungibility in the health sector were since performed:

In a working paper from the Results for Development Institute, Xu et al. (2011) found a 1% increase in external health financing to correlate with a 0.26%<sup>43</sup> decrease in Domestic General Government Health Expenditure (GGHE-D, which is exactly the same as GHE-S) in LICs up to no significant change in UMICs on average (5).

Based on an analysis of concessional loans disbursed to the health sector in nine SSA countries between 1980-2005, Gebrehanna and Upadhyay (2012) found that these loans were "fully non-fungible": A 1 percentage point increase in loan amounts as a percent of GDP was associated with a 1.22 percentage point increase in government health spending (i.e. GHE-A) as a percent of GDP (6).

<sup>&</sup>lt;sup>43</sup> lowest average estimate between models reported for LICs (FE model), other estimate was -0.19% (dynamic model).

Van de Sijpe (2013b) analyzed a panel of 108 countries from 1990-2003 and found a 1 percentage point increase in development assistance for the health sector per GDP to be associated with a 0.26 percentage point increase in overall GHE. However, he found little if any evidence of displacement when separating aid into its different channels and argued for the disaggregation of aid by its "on- and off-budget" components when conducting fungibility analyses (8).

Fernandes Antunes et al. (2013) analyzed a panel of 82 countries between 2002-2007 and found, in an FE-model, that a \$1 increase in "health-specific aid" was associated with a \$0.41 decrease in GHE-S. They also specified a dynamic panel model in which their findings were insignificant (9).

Liang et al. (2014) found a percent elasticity of 0.03 to 0.04% decrease in GHE-S for every 1% increase in DAH on average (panel of 120 countries from 1995 to 2010) (12).

In 2016, Barkat et al. found an increase in development assistance for the health sector of 1% to be associated with an increase in GHE of 0.04 to 0.1%, among 45 SSA countries from 1995-2012 (13).

Younsi et al. (2016) analyzed panel data from 113 LMICs between 1993-2013 and found a 0.19% - 0.25% decrease in GHE-S (depending on model specification) for every 1% increase in development assistance for the health sector among LICs, and no sizeable effect for MICs (14).

Patenaude (2021) found no fungibility effect of DAH, nor DAH-G on GHE-S in 134 LMICs between 2000-2015 (15).

By descriptively comparing trends in DAH and public spending on health as a share of all public spending over time in LICs, Nonvignon et al. (2024) concluded that DAH tends to crowd out public spending on health in LICs<sup>44</sup> (32).

# 10.1.3. Single-country case studies of fungibility of development assistance for the health sector

In Vietnam, Dodd et al. (2010) found indication of a negative elasticity between donor health project funding and GGHE excluding donor health projects (elasticity  $\epsilon$  = -0.186 (p= 0.09; n=13; time-period: 1995-2007) (33). The authors noted that this tendential association was less important than the significant positive association observed between the rapid growth in Vietnam and total government health spending ( $\epsilon$  = 1.875; p<0.001), and that aid was playing a diminishing role in financing the country's health system (3.4% of GGHE in 2007) (33). In another analysis of the Vietnamese health system by Wagstaff (2011), no evidence was found

<sup>&</sup>lt;sup>44</sup> Same caveat as previously: A descriptive evaluation of (negatively) correlating trends is inadequate for claiming causation (31).

of *intersectoral* fungibility<sup>45</sup> resulting from a primary care HSS project across approx. half of all provinces in Vietnam funded by a WB loan, however there was clear evidence of *intrasectoral* fungibility<sup>46</sup>. The latter was interpreted as consistent with a stated mutual understanding between WB and the Vietnamese government that the government would increase its funding in provinces not supported by WB (34).

Lu et al. (2017) found evidence of additionality between development assistance and government investments at the district level in rural health centres in Rwanda: A \$1 increase in ODA per capita allocated to these health centres was associated with a \$0.17 increase in government support in the same year (35).

In a 2016 mixed-methods study from Tanzania, Martinez-Alvarez et al. found both quantitative and qualitative evidence of fungibility. DAH was found to be displaced to other sectors as a deliberate policy in the budget allocation process, and also due to macro-economic policies such as debt servicing (36).

In Nigeria, domestic government HIV/AIDS funding in the 2000's and early 2010's was characterized as "*abysmal, as the government has practically handed over financing of HIV services to donors*" (37).

Lupieri and Doetter (2020) have also found descriptive statistical evidence for fungibility of development assistance for the health sector in Jordan<sup>47</sup> (38).

Described as "reverse fungibility", Jackson et al. (2020) found evidence that aid volatility and withdrawal following a corruption scandal led the Zambian government to increase its own budget allocation to primary health care<sup>48</sup> (39).

#### 10.1.4. Econometric studies of IMF effects on government health

#### spending

In a panel of 92 countries from 1980-2000, Noorudin and Simmons (2006) found that IMF programme participation was associated with decreased government share allocation to health and per capita health spending in democracies, and increases in share allocation to health in non-democracies (40).

Using Ordinary Least Squares (OLS) regression on a panel of 63 LICs from 1985-2009, Kentikelenis et al. (2015) found that IMF programs were associated with increased health

<sup>&</sup>lt;sup>45</sup> Intersector fungibility refers to the movement of domestic funds from a sector that receives external funding to another sector.

<sup>&</sup>lt;sup>46</sup> Intrasector fungibility refers to the movement of domestic funds from one area within a sector to another area within that sector, when the sector receives external funding.

<sup>&</sup>lt;sup>47</sup> Same caveats for descriptive statistical evidence apply as in previous sections.

<sup>&</sup>lt;sup>48</sup> Same as above.

spending in SSA LICs but lowered health spending in non-SSA LICs<sup>49</sup> (41). A follow up study found a negative correlation with health spending across LICs<sup>50</sup> (42).

Stubbs et al. (2017) used OLS on a panel of 16 West-African countries from 1995-2014<sup>51</sup> and found no overall effect of IMF programme participation on government health spending (43), but instead found that "*each additional binding IMF policy reform reduces government health expenditure per capita by 0.248 percent…*" (43).

Daoud and Reinsberg (2019) used Seemingly Unrelated Regression (SUR) on an instrumental variable model of 128 LMICs from 1980-2014, and found that IMF programme participation but not conditionalities were associated with increases in the general government expenditure budget share allocation to public health (44).

The previously mentioned single-country case study of Ghana from 1980-2014 using 3SLS found no effect of IMF conditionalities on government health spending (45).

By simply describing trends over time<sup>52</sup> from 1970-1993, Van der Gaag and Barham (1998) (WB staff) found that for "*countries that started the adjustment process early and took it seriously* … Government spending on health care increased on average" and concluded that "*it appears that the adjustment process can produce the favorable economic environment in which the system-wide health care problems can be addressed adequately*" (46).

Using ARIMA models on data from 146 countries from 1985-2000, the IMF IEO (2003 and 2004) found that government health spending was either maintained or increased under IMF programmes compared to when having no programme (47, 48). An IMF report from 2006 similarly found that LICs had insulated social expenditures from austerity measures (49), and Clements et al. (2011) (IMF staff) also had similar findings using fixed effects and system GMM on data from 1985-2009<sup>53</sup> (50). A 2017 IMF report again found that IMF-supported programs in LICs protected health spending, using a mix of methods reviewing Article IV reports, descriptive statistics and unspecified regression methods (possibly the same as Clements et al. (50-52)<sup>54</sup>). A 2019 IMF report using inverse probability weighted regression adjustment estimation and description of levels and trends reached similar conclusions (53).

#### References

1. Mishra P, Newhouse D. Does health aid matter? J Health Econ. 2009;28(4):855-72.

2. Farag M, Nandakumar AK, Wallack SS, Gaumer G, Hodgkin D. Does funding from donors displace government spending for health in developing countries? Health Aff (Millwood). 2009;28(4):1045-55.

<sup>&</sup>lt;sup>49</sup> Caveat: While using OLS on observational data across 63 units of observation provides correlational evidence, drawing causal inferences may be more difficult.

<sup>&</sup>lt;sup>50</sup> Estimation method not stated but deduced to be OLS.

<sup>&</sup>lt;sup>51</sup> Caveat: Same caveats for OLS as above apply to a larger extent as this panel was much smaller.

<sup>&</sup>lt;sup>52</sup> Same caveat for drawing causal inference from a description of trends apply as that described in Section 3.5.1.

<sup>&</sup>lt;sup>53</sup> All Sargan test p-values 1.00, please see previous descriptions of this issue.

<sup>&</sup>lt;sup>54</sup> I was unable to access the published article by Clements et al. (2013) (51).

3. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJL. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

4. Stuckler D, Basu S, McKee M. International Monetary Fund and aid displacement. Int J Health Serv. 2011;41(1):67-76.

5. Xu K, Saksena P, Holly A. The determinants of health expenditure: A country-level panel data analysis. Results for Development Institute WHO; 2011.

6. Gebrehanna S, Upadhyay MP. Fungibility of Foreign Aid in Selected African Countries. Indian Development Review. 2012;10(1):63-80.

7. Van de Sijpe N. The fungibility of health aid reconsidered. 2013.

8. Van de Sijpe N. Is Foreign Aid Fungible? Evidence from the Education and Health Sectors. World Bank Economic Review. 2013;27(2):320-56.

9. Fernandes Antunes Á, Xu K, James CD, Saksena P, Van de Maele N, Carrin G, et al. General budget support: has it benefited the health sector? Health economics. 2013;22(12):1440-51.

10. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

11. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

12. Liang LL, Mirelman AJ. Why do some countries spend more for health? An assessment of sociopolitical determinants and international aid for government health expenditures. Social science & medicine (1982). 2014;114:161-8.

13. Barkat K, Mrabet Z, Alsamara M. Does Official Development Assistance for Health from Developed Countries Displace Government Health Expenditure in Sub-Saharan Countries? Economics Bulletin. 2016;36(3):1616-35.

14. Younsi M, Chakroun M, Nafla A. Robust analysis of the determinants of healthcare expenditure growth: evidence from panel data for low-, middle- and high-income countries. International Journal of Health Planning and Management. 2016;31(4):580-601.

15. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

16. Gupta S. Comment: Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;376:593-94.

17. Ooms G, Decoster K, Miti K, Rens S, Van Leemput L, Vermeiren P, et al. Crowding out: are relations between international health aid and government health funding too complex to be captured in averages only? The Lancet. 2010;375(9723):1403-5.

18. Roodman D. Doubts about the evidence that foreign aid for health is displaced into non-health uses. The Lancet. 2012;380(9846):972-3.

19. Batniji R, Bendavid E. Does development assistance for health really displace government health spending? Reassessing the evidence. PLoS Med. 2012;9(5):e1001214.

20. Batniji R, Bendavid E. Considerations in assessing the evidence and implications of aid displacement from the health sector. PLoS Med. 2013;10(1):e1001364.

21. Murray CJL. Doubts about the evidence that foreign aid for health is displaced into non-health uses – Author's reply. The Lancet. 2012;380(9846).

22. Baker B. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):589-3.

23. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJL. Effect of development assistance on domestic health expenditures - Authors' reply. The Lancet. 2010;376(9741):592-3.

24. Bustreo F, Requejo J, Johnston A. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):591-3.

25. Chen B, Fan VY, Manne J. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):592-3.

26. deRiel E, Fort M, Barry D. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):590-3.

27. Moatti J-P, Abu-Zaineh M, Teyssier LS. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):591-3.

28. Gwatkin DR. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):592-3.

29. Wright S. Effect of development assistance on domestic health expenditures. Lancet (London, England). 2010;376(9741):590-3.

Murray CJ, Dieleman JL, Lu C, Hanlon M. More data and appropriate statistical methods needed to fully measure the displacement effects of development assistance for health. PLoS Med. 2013;10(1):e1001365.
 Tyler Vigen. Spurious correlations. 2024 [cited 2024 April 22]. Available from:

https://www.tylervigen.com/spurious-correlations.

32. Nonvignon J, Soucat A, Ofori-Adu P, Adeyi O. Making development assistance work for Africa: from aiddependent disease control to the new public health order. Health Policy and Planning. 2024;39:179-192.

33. Dodd R, James CD, Phuong NTK. External aid for health in Viet Nam: additional or fungible? World Health Report 2010: Background Paper, WHO 2010;40.

34. Wagstaff A. Fungibility and the impact of development assistance: evidence from Vietnam's health sector. Journal of Development Economics. 2011;94(1):62-73.

35. Lu C, Cook B, Desmond C. Does foreign aid crowd out government investments? Evidence from rural health centres in Rwanda. BMJ Glob Health. 2017;2(3):e000364.

36. Martinez Alvarez M, Borghi J, Acharya A, Vassall A. Is Development Assistance for Health fungible?
Findings from a mixed methods case study in Tanzania. Social science & medicine (1982). 2016;159:161-9.
37. Chima CC, Homedes N. Impact of global health governance on country health systems: the case of HIV initiatives in Nigeria. J Glob Health. 2015;5(1):010407.

38. Lupieri S, Doetter LF. Transnational Interdependency and Healthcare System Change: The Role of Humanitarian and Development Aid in Shaping Health Policy in Jordan. Global Social Policy. 2020;20(2):192-214.

39. Jackson A, Forsberg B, Chansa C, Sundewall J. Responding to aid volatility: government spending on district health care in Zambia 2006-2017. Glob Health Action. 2020;13(1):1724672.

40. Nooruddin I, Simmons JW. The Politics of Hard Choices: IMF Programs and Government Spending. International Organization. 2006;60(04).

41. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

42. Stubbs T, Kentikelenis A. Targeted social safeguards in the age of universal social protection: the IMF and health systems of low-income countries. Critical Public Health. 2017;28(2):132-9.

43. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

44. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

45. Boachie MK, Agyemang J, Immurana M. Health sector funding in Ghana: The effect of IMF conditionalities. Dialogues Health. 2022;1:100045.

46. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

47. IMF Independent Evaluation Office. Fiscal Adjustment in IMF-Supported Programs. 2003 [cited 2024 March 19]. Available from: https://www.imf.org/external/np/ieo/2003/fis/pdf/main.pdf.

48. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs 2004. Available from: http://www.ieo-imf.org/ieo/files/completedevaluations/BP041.pdf.

49. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

50. Clements B, Gupta S, Nozaki M. What Happens to Social Spending in IMF-Supported Programs? . 2011. Available from: <u>http://www.imf.org/external/pubs/ft/sdn/2011/sdn1115.pdf</u>.

51. Clements B, Gupta S, Nozaki M. What happens to social spending in IMF-supported programmes? Applied Economics. 2013;45(28):4022-33.

52. IMF. Social safeguards and program design in PRGT and PSI-supported programs. 2017 [cited 2024 March 20]. Available from: <u>https://www.imf.org/en/Publications/Policy-Papers/Issues/2017/06/01/pp042117social-safeguards-and-program-design-in-prgt-and-psi</u>.

53. IMF. A Strategy for IMF Engagement on Social Spending. 2019 [cited 2023 August 22]. Available from: https://www.imf.org/en/Publications/Policy-Papers/Issues/2019/06/10/A-Strategy-for-IMF-Engagement-on-Social-Spending-46975.

# 10.2. Appendix 2: Key health indicators in Senegal

Senegal is performing better than the SSA average on a number of key health indicators, and these have improved over time. Since 2000, the total life expectancy at birth in Senegal had improved significantly to 69 years in 2019 (WHO African region average: 65 years) (1), the under-5 mortality rate to 39 deaths per 1000 live births in 2021 (WHO African region average: 72 deaths per 1000) (1), and the maternal mortality ratio to 261 deaths per 100,000 live births in 2020 (SSA average: 536 deaths per 100,000 live births) (1, 2). In 2022, 88% of Senegalese one-year-olds had received their third Diphtheria-Tetanus-Polio (DTP) immunisation (WHO African region average: 72%) (3). In 2019, 26% of women of reproductive age, married or in-union, used modern methods of contraception (SSA average: 29%) (1, 2), and the HIV prevalence rate among adults aged 15-49 was 0.3% in 2022 (WHO African region average: 3.2%) (1). Tuberculosis deaths among HIV-negative people had declined to 15 per 100,000
in 2022 (WHO African region average: 26 per 100,000) (1), and deaths due to malaria to 26 per 100,000 population in 2021 (WHO African region average: 58 per 100,000) (1). The agestandardised Non-Communicable Disease (NCD) mortality rate had decreased to 551 deaths per 100,000 population in 2019 (WHO African region average: 587 per 100,000), of which 61% were estimated to be premature deaths (WHO African region average: 64%) (1).

# References

1. WHO. Global Health Observatory Data Repository. 2024 [cited 2024 February 17]. Available from: https://www.who.int/data/gho.

2. World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: https://data.worldbank.org/.

3. WHO Immunization Data portal. Diphtheria tetanus toxoid and pertussis (DTP) vaccination coverage. 2024 [cited 2024 February 19]. Available from: <u>https://immunizationdata.who.int/pages/coverage/dtp.html</u>.

## Appendix 3: Evidence of retention of copyright (Chapter 10.3. 5)

ZGHA 2072461 Growing Debt Burden in Low- and Middle-Income Countries During Covid-19 Ma...

## CC BY OPEN ACCESS LICENCE

This is a licence agreement under which you, the author, retain copyright in your article, and grant Umeå University, 901 87 Umeå, Sweden (hereinafter 'the Society') to allow us and our publisher Informa UK Limited registered in England under no. 1072954 trading as Taylor & Francis Group, Registered Office: Mortimer House, 37-41 Mortimer Street, London WIT 3JH (hereinafter 'Taylor & Francis') a non-exclusive licence to publish your article, including abstract, tables, figures, data, and supplemental material hosted by our publisher, as the Version of Record in the Journal on an Open Access basis under a Creative Commons Attribution License (CC BY) http://creativecommons.org/licenses/by/4.0/ subject to the Terms & Conditions set out below.

ARTICLE TITL	E: Growing Debt Burden in Low- and Middle-Income Countries During Covid-19 May Constrain Health Financing	('Article')
AUTHOR(S):	Frederik Federspiel, Josephine Borghi, Melisa Martinez-Alvarez	

JOURNAL TITLE: **Global Health Action**  ('Journal')

#### Please complete and sign below:

X I confirm that I have read and accept the full terms of the Journal's Article Publishing Agreement including the Terms & Conditions.

imes I confirm the article will be made available under the following access and use licence, and that I have read and understood the terms of this licence: Creative Commons Attribution License (CC BY).

X I confirm that I agree to assume responsibility for any applicable payment of the Article Publishing Charge.

X I confirm I grant the Society the rights to publish my article on an Open Access basis, in all forms and all media (whether known at this time or developed at any time in the future) throughout the world, including the right to translate the article into other languages, create adaptations, summaries or extracts of the article or other derivative works based on the article and the right to sub-license all such rights to others subject to the Terms & Conditions set out below, to take effect if and when the article is accepted for publication.



X

I confirm that I have read and accept my author warranties.

I confirm that I have read and agree to comply with Taylor & Francis' policy on publishing ethics

#### Please tick either box A or box B, BUT NOT BOTH

A I own copyright, and I am granting the Society a non-exclusive licence to publish my article as the Version of Record on an Open Access basis, with the copyright statement '@ <YEAR> The Author(s). Published by <Taylor &Francis>'. In the case of a multi-authored article, | confirm that | am authorized by my co-authors to enter into this licence as their agent on their behalf. The co-authors have agreed the priority of the assertion of copyright and the order of names in the publication of the article.

B I am a civil servant or an employee of a Government, Government Agency, International Organization, or Commercial Corporation which is granting a non-exclusive licence to publish the article and which hereby recognizes the Society as the sole licensee for the publication of the final, definitive, and citable Version of Record. In the case of a multi-authored article, I confirm that I am authorized by my coauthors to enter into this licence as their agent on their behalf. The co-authors have agreed the priority of the assertion of copyright and the order of names in the publication of the article.

#### If you have ticked B, please indicate which of the statements below apply to you (and your co-authors):

- I am an employee of the UK, Canadian, Australian, or another Commonwealth Realm Government, and the Crown retains and asserts copyright.
- I am a US Government (including NIH) employee and there is no copyright to transfer.
- I am a contractor of the US Government (includes NIH contractors) under contract number:
- I am an employee of the European Commission and copyright is asserted and retained by the European Union. I am an employee of the World Bank and copyright is asserted and retained by that entity.
- I am an employee of the Food & Agricultural Organization and copyright is asserted and retained by that entity. I am an employee of a Government, Agency, or International Organization and copyright is retained by that entity. Name of
- entity:
- I am employed and the copyright belongs to my employer (or is a 'work made for hire' under US law). Name of corporation

ZGHA 2072461 Growing Debt Burden in Low- and Middle-Income Countries During Covid-19 Ma...

#### GRANT OF PUBLISHING RIGHTS

 Signed:
 Name (Print):
 Frederik Federspiel

 Position
 PhD Student
 Date:
 April 28, 2022

#### THIS FORM IS A LEGALLY BINDING DOCUMENT. WE RECOMMEND THAT YOU RETAIN A COPY OF IT AND CONSULT A LEGAL ADVISOR IF YOU HAVE ANY QUESTIONS.

#### LICENCE TO PUBLISH: TERMS & CONDITIONS

#### DEFINITION

1. Your article is defined as comprising (a) your Accepted Manuscript (AM) in its final form; (b) the final, definitive, and citable Version of Record (VoR) including the abstract, text, bibliography, and all accompanying tables, illustrations, data, and media; and (c) any supplemental material hosted by our publisher. This licence and these Terms & Conditions constitute the entire agreement and the sole understanding between you and us ('agreement'); no amendment, addendum, or other communication will be taken into account when interpreting your and our rights and obligations under this agreement, unless amended by a written document signed by both of us.

#### **TAYLOR & FRANCIS' RESPONSIBILITIES**

- 2. If deemed acceptable by the Editors of the Journal, we shall prepare and publish your article in the Journal. We may post your accepted manuscript as free-to-access in advance of the formal publication of the Version of Record (VoR). We shall publish the VoR in the Journal on an Open Access basis, viz., to be made freely available online with no subscription fee or article-pay-to-view fee or any other form of access fee or any publication embargo being applied. We reserve the right to make such editorial changes as may be necessary to make the article suitable for publication or as we reasonably consider necessary to avoid infringing third-party rights or breaching any laws; and we reserve the right not to proceed with publication for whatever reason.
- If before publication we reasonably consider that the article should not be published, on the advice of our legal advisors, we may decline to publish the article, in which case we will refund you any Article Publishing Charge you have paid.
- 4. If after publication we reasonably consider that the article should be retracted or removed from our website, on the advice of our legal advisors, for example, because of a breach in your Author Warranties, we may retract and withdraw it, and in such case shall be under no obligation to refund you any Article Publishing Charge you have paid.
- 5. You hereby expressly grant us the right to bring an action for infringement of copyright in relation to your article as defined above pursuant to section 101A of the Copyright, Designs and Patents Act 1988 (as amended from time to time) and any other equivalent legislation as may exist in other territories.
- 6. If we do not receive payment of the applicable Article Publishing Charge after six (6) weeks, we reserve the right to rescind the Open Access status of your article and to publish it on an alternative licence basis.

#### YOUR RIGHTS AS AUTHOR

- 7. These rights are personal to you, and your co-authors, and cannot be transferred by you to anyone else. You assert and retain the following rights as author(s):
  - i. The right to re-use your own work on a commercial or non-commercial basis, and in any way permitted under the <u>Creative Commons Attribution License</u> (CC BY), including but not limited to, translation, adaptation, and resale.
  - ii. The right to be identified as the author of your article, whenever and wherever the article is published, such rights including moral rights arising under § 77, Copyright, Designs & Patents Act 1988, and, so far as is legally possible, any corresponding rights we may have in any territory of the world.
  - iii. The right to retain patent rights, trademark rights, or rights to any process, product or procedure described in your article.
  - iv. The right to post and maintain at any time your 'Author's Original Manuscript (AOM), i.e., the unpublished version of the article created by you prior to peer review; and the article in its published form as supplied by us as a <u>digital eprint</u> on your own website for personal or professional use, or

on your institution's network or intranet or website, or in a subject repository or network, with the acknowledgement: 'The Version of Record of this manuscript has been published and is freely available in <JOURNAL TITLE> <date of publication> http://www.tandfonline.com/ <Article DOI>.'

#### WARRANTIES MADE BY YOU AS AUTHOR

- 7. You warrant that:
  - i. All persons who have a reasonable claim to authorship are named in the article as co-authors including yourself, and you have not fabricated or misappropriated anyone's identity, including your own.
  - ii. You have been authorized by all such co-authors to sign this agreement as agent on their behalf, and to agree on their behalf the priority of the assertion of copyright and the order of names in the publication of the article.
  - iii. The article is your original work, apart from any permitted third-party copyright material you include, and does not infringe any intellectual property rights of any other person or entity and cannot be construed as plagiarizing any other published work, including your own published work.
  - iv. The article is not currently under submission to, nor is under consideration by, nor has been accepted by any other journal or publication, nor has been previously published by any other journal or publication, nor has been assigned or licensed by you to any third party.
  - v. The article contains no content that is abusive, defamatory, libellous, obscene, fraudulent, nor in any way infringes the rights of others, nor is in any other way unlawful or in violation of applicable laws.
  - vi. Research reported in the article has been conducted in an ethical and responsible manner, in full compliance with all relevant codes of experimentation and legislation. All articles which report *in vivo* experiments or clinical trials on humans or animals must include a written statement in the Methods section that such work was conducted with the formal approval of the local human subject or animal care committees, and that clinical trials have been registered as applicable legislation requires.
  - vii. Any patient, service user, or participant (or that person's parent or legal guardian) in any research or clinical experiment or study who is described in the article has given written consent to the inclusion of material, text or image, pertaining to themselves, and that they acknowledge that they cannot be identified via the article and that you have anonymized them and that you do not identify them in any way. Where such a person is deceased, you warrant you have obtained the written consent of the deceased person's family or estate.
- viii. You have complied with all mandatory laboratory health and safety procedures in the course of conducting any experimental work reported in your article; your article contains all appropriate warnings concerning any specific and particular hazards that may be involved in carrying out experiments or procedures described in the article or involved in instructions, materials, or formulae in the article; your article includes explicitly relevant safety precautions; and cites, if an accepted Standard or Code of Practice is relevant, a reference to the relevant Standard or Code.
- ix. You have acknowledged all sources of research funding, as required by your research funder, and disclosed any financial interest or benefit you have arising from the direct applications of your research.
- x. You have obtained the <u>necessary written permission</u> to include material in your article that is owned and held in copyright by a third party, which shall include but is not limited to any proprietary text, illustration, table, or other material, including data, audio, video, film stills, and screenshots, and any supplemental material.
- xi. You have read and complied with our policy on publishing ethics
- xii. You have read and complied with the Journal's Instructions for Authors.
- xiii. You will keep us and our affiliates indemnified in full against all loss, damages, injury, costs and expenses (including legal and other professional fees and expenses) awarded against or incurred or paid by us as a result of your breach of the warranties given in this agreement.
- xiv. You consent to allowing us to use your article for marketing and promotional purposes.

#### **GOVERNING LAW**

8. This agreement (and any dispute, proceeding, claim or controversy in relation to it) is subject to English law and the parties hereby submit to the exclusive jurisdiction of the Courts of England and Wales.

#### THIRD PARTY ACCESS & USAGE TERMS & CONDITIONS FOR OPEN ACCESS CONTENT RIGHTS ARE GIVEN AT http://www.tandfonline.com/page/terms-and-conditions

# 10.4. Appendix 4: Supplementary methods description, results and metadata for Chapter 6

# 10.4.1. Supplementary methods description

	10.4.1.	1.	Variable	names
--	---------	----	----------	-------

Variable	Full name
oop_che	Out-Of-Pocket Spending per Current Health Expenditure
oop_gdp	Out-Of-Pocket Spending per Gross Domestic Product
ghes_che	Government Health Expenditure as Source per Current Health Expenditure
ghes_gdp	Government Health Expenditure as Source per Gross Domestic Product
odaplus_health_pub_gdp	Official Development Assistance+ for health channeled via recipient country
	public sector per Gross Domestic Product
odaplus_health_civpriv_gdp	Official Development Assistance+ for health channeled via NGO's, CSO's
	and private sector per Gross Domestic Product
odaplus_nonhealth_pub_gdp	Official Development Assistance+ for non-health purposes channeled via
	recipient country public sector per Gross Domestic Product
odaplus_nonhealth_civpriv_gdp	Official Development Assistance+ for non-health purposes channeled via
	NGO's, CSO's and private sector per Gross Domestic Product
debt_serv_gdp	Debt Service per Gross Domestic Product (external Public and Publicly
	Guaranteed)
debt_stock_gdp	Debt Stock per Gross Domestic Product (external Public and Publicly
	Guaranteed)
imf_cond_ba1tot	International Monetary Fund Conditionalities Burden of Adjustment
	Indicator 1, Total (all policy areas)
imf_part	International Monetary Fund programme Participation
gdp_cap_c	Gross Domestic Product per Capita (constant 2020 US\$)
imr	Infant Mortality Rate
gov_eff	Government Effectiveness
corr_control	Corruption Control
battle_dum	Battle Dummy
col_indfrom_uk	Colonial Independence from United Kingdom
col_indfrom_fra	Colonial Independence from France
col_indfrom_spa	Colonial Independence from Spain

Please see Metadata (Section 10.4.5) for further details on variables in dataset.

## 10.4.1.2. Country list

Albania Algeria Angola Argentina Armenia Azerbaijan Bangladesh Belarus Belize Benin Bolivia Bosnia and Herzegovina Botswana Brazil Burkina Faso Burundi Cambodia Cameroon Cape Verde Central African Republic Chad China Colombia Comoros Congo Costa Rica Cote d'Ivoire Democratic Republic of Congo Djibouti Dominican Republic Ecuador Egypt El Salvador Eswatini Ethiopia Fiji Gabon Gambia Georgia Ghana Guatemala Guinea Guinea-Bissau Guyana Haiti Honduras India Indonesia Iran Jamaica Jordan Kazakhstan Kenya Kyrgyz Republic

Laos Lebanon Lesotho Liberia Madagascar Malawi Mali Mauritania Mauritius Mexico Moldova Mongolia Morocco Mozambique Myanmar Nepal Nicaragua Niger Nigeria North Macedonia Pakistan Papua New Guinea Paraguay Peru Philippines Rwanda Samoa Sao Tome and Principe Senegal Serbia Sierra Leone Solomon Islands South Africa Sri Lanka Sudan Tajikistan Tanzania Thailand Togo Tonga Tunisia Turkey Turkmenistan Uganda Ukraine Uzbekistan Vanuatu Vietnam Yemen Zambia Zimbabwe

## 10.4.1.3. Variable description, motivation and hypotheses

Table 6.1 (main manuscript) shows summary descriptive statistics for the variables used in our models.

## 10.4.1.3.1. Dependent variables

Our dependent variables are described and motivated in the main manuscript.

## 10.4.1.3.2. Independent variables of interest

We used the following set of main input variables of interest. We examined Official Development Assistance Plus (ODA<sup>+</sup>) for health from the OECD CRS database, which includes disbursements from Development Assistance Committee (DAC) countries, non-DAC countries, multilaterals and the BMGF<sup>55</sup> due to their substantial development assistance contributions (1, 2). The categories "Health, Total" and "Population Policies/Programmes and Reproductive Health, Total" were included in this variable, while other categories were counted as non-health ODA. Following the majority of the available literature, we mainly hypothesized a displacement/ crowding out effect of ODA<sup>+</sup> for health on GHE-S (i.e. fungibility) as well as on OOP. Also following the literature on government health spending (3-8), we hypothesized that a fungibility dynamic would depend on whether the development assistance is "on-budget" or "off-budget", which determines whether or not the government can see and predict development assistance coming in and adjust its own budget accordingly. We therefore disaggregated our ODA<sup>+</sup>-variables into development assistance channelled via the recipient country public sector, and channelled via Non-Governmental Organisations (NGO's), Civil Society Organisations (CSO's) or Private Sector Institutions (PSI's), jointly referred to as the civil/private sector. For OOP as a dependent variable, we hypothesized that ODA<sup>+</sup> for health via either channel would have a displacing effect, as either could work to subsidise or cover user fees. Other studies have found respectively no association (disaggregated by channel) (8) and a negative association (not disaggregated by channel) (9). As Government Health Expenditure as Agent in LMICs (GHE-A, i.e. all government health spending from domestic and external revenue) ≈ GHE-S + on-budget development assistance to the health sector (10), fungibility implies a negative relationship between on-budget ODA<sup>+</sup> for health and GHE-S, reflecting that a \$1 external injection of funds into the health sector did not raise GHE-A by \$1 due to the government lowering its own financing (GHE-S) (7).

<sup>&</sup>lt;sup>55</sup> We counted all BMGF development finance on the OECD CRS database as grants, even though this is not specified in the database. This was based on the conservative assumption that the vast majority if not all of these transfers are indeed grants, since the BMGF had given 34523 total grants with a substantial proportion to health, and only 10 loans within health as of November 10, 2023 (https://www.gatesfoundation.org/about/committed-grants; https://sif.gatesfoundation.org/portfolio/).

We also explored non-health ODA<sup>+</sup> (2), to explore whether development assistance outside the health sector influences levels of government and household health financing (11, 12). The main hypothesis tested was that of non-health sectoral aid benefiting the health sector by allowing governments to shift some extra funds toward the health sector, i.e. that on-budget non-health ODA<sup>+</sup> would correlate positively with GHE-S per GDP and/or per CHE. However, if GDP growth effects are equal to or stronger than any positive effects on GHE-S, no correlation/negative correlation would be seen for GHE-S/GDP (we also run a GHE-S/capita analysis as sensitivity analysis in this appendix). Importantly, the variables also control for the macroeconomic effects of non-health aid flows. We deducted debt relief, administrative expenses, other in-donor country expenses and promotion of development awareness from ODA<sup>+</sup> variables to count only real financial transfers with a development purpose. Using twosided T-tests, we allowed for the alternative hypotheses that ODA+ for health could crowd in GHE-S and ODA+ for non-health sectors could draw domestic government funds away from the health sector, e.g. through co-financing requirements.

We examined the effects of PPG external debt service from the WB International Debt Statistics database (13). In theory, increased debt servicing should take away money from the public budget, reducing the fiscal space for all government expenses including the health sector, leading to our main hypotheses of a negative association between PPG external debt servicing and GHE-S/GDP and/or GHE-S/CHE. Alternatively, the ability to service one's debt could reflect favourable fiscal conditions and/or occur at a manageable level posing no risk of "debt distress", with enough revenue to finance both debt and health, which would show as the confirmation of our null-hypothesis of no effect. We further take into consideration the PPG external debt stock per GDP (13), similar to Lora and Olivera (2007) (14). This allows us to adjust for the level of indebtedness, i.e. adjusted for the size of the debt relative to the size of the economy, are increases in debt servicing per GDP associated with decreases in GHE-S/GDP or GHE-S/CHE, and associated increases in OOP/GDP or OOP/CHE? It also allows us to examine any effects of the debt stock itself on our outcome variables (14).

We included a variable capturing the level of adjustment due to IMF conditions using the IMF Monitor conditionality dataset (15). The variable counts the number of conditions in place in a given country-year. More conditions could lead to less GHE-S generally through what can be referred to as an austerity effect (16-27), which is our main hypothesis. IMF/WB authors have conversely found that they could lead to more GHE-S (28-32), meaning that conditions lead to growth and more fiscal space for health, and that health spending is protected from any austerity measures. As for OOP, we hypothesize that privatization and decentralisation measures resulting from IMF conditionalities, as well as an indirect effect via the above

austerity effect where the state funds less health services and users have to take over payment, could lead to increases in OOP. To isolate the marginal effect of an additional conditionality within an IMF programme, we included a dummy-variable for IMF programme participation, generated from the conditionality variable ("on" when conditions observed) (33). This also captures any effects of IMF programme participation separate from those acting through conditionalities, such as catalysing development assistance and investment, or effects of technical assistance or other means of policy influence (33-35).

## 10.4.1.3.3. Covariates

We adjusted our models using a set of covariates. We included GDP per capita (constant 2020 US\$) (36), as this may codetermine both our outcome variables, independent variables of interest and covariates. We included Infant Mortality Rate (IMR) as a proxy for unmet population health need (11, 37, 38), from the United Nations Inter-Agency Group for Child Mortality Estimation (39). Such unmet need may reflect a low GHE-S and a high resulting reliance on OOP, as well as health need driving OOP health expenditures. IMR has also been identified as the strongest health need determinant of development assistance for health allocations, above Under-5 Mortality Rate (U5MR) and Human Immunodeficiency Virus (HIV) prevalence (37).

We included a variable capturing government effectiveness and a variable measuring control of corruption from the Worldwide Governance Indicators database (40). These are commonly used in the development assistance literature as they are found to codetermine aid flows and aid effectiveness, including development lending (e.g. (41-47)). A country's governance also results from and determines IMF programme participation, policy recommendations and conditionalities (48). More corrupt governments have been found to spend less on health (49, 50), and the same is true for less effective governments (10, 40). These variables might also help explain why development assistance and government health expenditure fail to translate into reduced dependency on OOP health spending.

We adjusted for the effects of a country being in armed conflict, by including a dummy variable for when the number of deaths in battle<sup>56</sup> was larger than or equal to 1 per million population in a country-year, using the Uppsala Conflict Data Program / Peace Research Institute Oslo Battle-Related Deaths Dataset (51). Development assistance and lending flows have by some been found to differ (52, 53), and by others not to differ (54), between countries depending on conflict status. Receiving development assistance has been found to be associated with a

<sup>&</sup>lt;sup>56</sup> We replaced missing values for battle-related deaths with 0's, as this dataset only report recorded battle-related deaths and assign missing values to all other country-years.

higher likelihood of conflict escalation (55), and GHE has been found to suffer in war-ridden countries (56). We further hypothesize that conflict might cause an increased reliance on OOP health spending in a destabilised health economy.

Finally, we examined for differences in the effects of colonial legacy between former British, French and Spanish colonies. Bilateral donors tend to give more development assistance to their former colonies (examples: (57-60)), and the economies and institutions of formerly colonized nations are partly shaped by their colonial legacy (61-63), including their health system models (64). We hypothesize that this could also co-determine health system payment patterns, both by system inheritance and by ongoing policy influence from the former coloniser. We used the Issue Correlates Of War dataset for this, using a variable categorizing from which colonial ruler independence was gained from (65).

## 10.4.1.4. Approach to issue of serial correlation in dependent variables

The issue of serial correlation in dependent variables is commonly seen in the literature (e.g. (24, 96, 97), including for GHE-S/GDP as dependent variable (24)). By including lagged dependent variables, the amount of explanatory power contained in last year's variable level is accounted for, and our independent variable estimates and error terms contain the remaining explanatory power beyond this. The use of instrumentation in System GMM is the preferred approach when faced with serial correlation in panel data series (85). The Arellano-Bond test for first-order serial correlation of the error terms in the levels equation is the guiding test in this situation (85, 86). We made sure that all p-values were above 0.15, including by adding 2<sup>nd</sup>-order lags where necessary, thus safely accepting the null-hypothesis that lagged GMM-style instruments were not made invalid by serial correlation of residuals (85). By not breaking up our dataset into subsets, we ensured our dataset was large enough to ensure validity of this test. Our unit root tests further firmly confirmed the absence of a unit root (below). These estimation choices and testing should thus have adequately addressed potential inference issues present when examining dependent variables that are serially correlated.

# 10.4.2. Supplementary findings

## 10.4.2.1. Results from full model regressions

## Two-step system GMM results for full model specifications

Dependent variable	1) OOP/CHE	2) Ln(OOP / GDP)	3) GHE-S / CHE	4) Ln(GHE-S / GDP)
L. dependent variable	0.708*** [0.053]	0.903*** [0.052]	0.823*** [0.077]	0.819*** [0.061]
L. ln(ODA <sup>+</sup> for health (public) / GDP)	-0.389 [0.301]	-0.018* [0.010]	-0.311 [0.282]	-0.022** [0.011]
L. ln(ODA <sup>+</sup> for health (civ./priv.) / GDP)	-0.456** [0.195]	-0.014* [0.007]	-0.159 [0.260]	0.003 [0.009]
L. ln(ODA <sup>+</sup> for non-health (public) / GDP)	0.086 [0.337]	-0.008 [0.014]	-0.018 [0.377]	0.004 [0.016]
L. $\ln(ODA^+ \text{ for non-health})$	0.149 [0.533]	-0.026 [0.019]	0.311 [0.425]	0.019 [0.018]
L. ln(debt service/GDP)	0.563* [0.292]	0.019 [0.017]	-0.401 [0.294]	0.004 [0.011]
IMF conditionalities	-0.015 [0.088]	-0.026 [0.021] -0.000 [0.000]	-0.159 [0.821] 0.016 [0.014]	-0.018 [0.026]
IMF participation Ln(GHE-S/GDP)	0.375 [0.462] -7.530***	0.012 [0.022] 0.063* [0.037]	-1.131* [0.640] -	0.006 [0.022]
Le(CDD/corr)	[1.664]	0 155** [0 070]	2 206 [2 226]	0.055 [0.071]
IMR	-0.106**	-0.002 [0.002]	-0.015 [0.060]	-0.000 [0.002]
Gov. effectiveness	-0.488 [1.480]	-0.049 [0.053]	-0.961 [1.715]	0.012 [0.053]
Corruption control	-1.607 [1.391]	-0.016 [0.060]	2.603[2.210]	0.052 [0.059]
Conflict	0.559 [0.745]	0.019 [0.023]	0.147 [0.606]	0.031 [0.027]
Ind. from UK	-1.933 [1.216]	-0.015 [0.043]	-0.447 [1.286]	0.007 [0.054]
Ind. from France	-0.220 [1.629]	0.031 [0.054]	-0.154 [1.069]	-0.054 [0.037]
Ind. from Spain	-0.532 [1.480]	0.005 [0.059]	0.524 [1.678]	0.037 [0.028]
Constant	16.382 [14.770]	1.193** [0.578]	-8.906 [18.296]	0.016 [0.047]
Observations	1429	1429	1429	1429
Countries	105	105	105	105
Instruments	92	92	73	73
Lag limits (years)	4	4	3	3
Country fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
AB-test AR(2) (p-level)	0.996	0.756	0.155	0.513
Hansen test (p-level)	0.262	0.366	0.198	0.544
Diffin-Hansen test (p-level)	0.229	0.305	0.847	0.190
F-statistic	520.66***	165.84***	886.84***	260.43***

Windmeijer-corrected robust standard errors in brackets; \*p < 0.1, \*\*p<0.05, \*\*\*p<0.01

Civ./priv: ODA+ channelled via NGO's & civil society organisations or private sector institutions.

AB-test AR(2) = Arellano-Bond test for first-order autocorrelation in the levels equation. Diff.-in-Hansen test: Difference-in-Hansen test for exogeneity of GMM instruments in levels equation. Govt.= Government. IMR= Infant Mortality Rate. Ind. = Independence (colonial). L. = 1-year lag. Ln= natural logarithm.

## 10.4.2.2. Other findings for ODA variables

ODA<sup>+</sup> for health purposes disbursed to the civil/private sector measured per GDP, correlated negatively with OOP/CHE in our full model (semi-elasticity: -0.456) though not in our reduced model. This association was not present at other lag-levels, it only tolerated increasing the GMM-style lag limit by 1, and it was robust to most but not all covariate swaps (below). There was a near-significant negative correlation with OOP/GDP in the full model (p=0.050), which was just significant in our reduced model (p=0.047). This effect was also present at the 2<sup>nd</sup> order lag but not when unlagged, at GMM-style lag limits 5, 7 and 8, and with all covariate swaps. We found no association with GHE-S/CHE or with GHE-S/GDP.

We also found a significant negative association between ODA<sup>+</sup> for non-health purposes channelled via the civil/private sector and OOP/GDP in our reduced model (elasticity: -0.04), but not our full model. This finding was not robust to changes in the lag limit. The effect was also present in a 2<sup>nd</sup> order lag version of our full model but not of the reduced model, and not in a contemporaneous version. It was robust to all covariate swaps.

## 10.4.2.3. Covariate findings

We made the following findings among our included covariates:

A 1% increase in GHES/GDP was associated with a -0.08 percentage point decrease in OOP/CHE, both in full ([-0.11: -0.04], p=0.000) and reduced ([-0.11: -0.05], p=0.000) models. This association remained significant when chancing variable lag-structures, GMM-style instrument lag limits (below) and covariates. A 1% increase in GHES/GDP was also associated with a 0.07% ([0.001: 0.14], p=0.048) increase in OOP/GDP in the reduced model, which was also robust to the same changes.

We found a significant negative association between GDP per capita and OOP/GDP (full model: elasticity: -0.15 ([-0.29: -0.02], p=0.02); reduced model: elasticity -0.25 ([-0.40: -0.11], p=0.001). This finding only remained in reduced model versions with 2<sup>nd</sup> order lags of the independent variables of interest, but not other variable lag variations. It was however robust to changes in GMM-style lag limits and to many but not all covariate swaps.

One additional infant dying before reaching one year of age per 1,000 live births was associated with a -0.1 ([-0.20: -0.008], p=0.038) percentage point decrease in OOP/CHE in our full model, but this was insignificant in our reduced model (p=0.058). The finding was not robust to changes in lag-levels of independent variables of interest, but it was robust to changing GMM lag limits and to some, but not all covariate swaps.

A one standard-deviation increase in the level of control of corruption as measured in the Worldwide Governance Indicators (40) was associated with a 3.2 ([0.61: 5.75], p=0.017) percentage point increase in GHES/CHE in our reduced model. This finding was robust when independent variables of interest were lagged two years, but not when they were contemporaneous. The finding remained when increasing, but not decreasing lag limits, and it was robust to swapping mortality indicators but not colonial history covariates.

# 10.4.3. Diagnostic test results

# 10.4.3.1. Correlation matrices

## Pairwise correlations, significance level p<0.05

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19) (20)
(1) oop_che	1.00																		
(2) oop_gdp	-0.65*	1.00																	
(3) ghes_che	0.78*	-0.46*	1.00																
(4) ghes_gdp	-0.56*	0.82*	-0.13*	1.00															
(5) ln_odaplus_health_pub_gdp	-0.10*	-0.39*	-0.17*	-0.32*	1.00														
(6) ln_odaplus_health_civpriv_gdp	-0.06*	-0.47*	-0.08*	-0.35*	0.82*	1.00													
(7) ln_odaplus_nonhealth_pub_gdp	-0.07*	-0.29*	-0.08*	-0.21*	0.81*	0.70*	1.00												
(8) ln_odaplus_nonhealth_civpriv_gdp	-0.03	-0.38*	0.01	-0.25*	0.77*	0.79*	0.82*	1.00											
(9) ln_debt_serv_gdp	-0.08*	0.29*	0.02	0.29*	-0.18*	-0.20*	-0.05	-0.11*	1.00										
(10) ln_debt_stock_gdp	0.00	-0.11*	0.06*	-0.05*	0.36*	0.32*	0.47*	0.48*	0.54*	1.00									
(11) imf_cond_ba1tot	0.09*	-0.22*	0.15*	-0.12*	0.25*	0.23*	0.30*	0.29*	0.01	0.17*	1.00								
(12) imf_part	0.10*	-0.26*	0.17*	-0.13*	0.29*	0.28*	0.37*	0.34*	0.01	0.20*	0.84*	1.00							
(13) gdp_cap_c	-0.22*	0.68*	-0.13*	0.58*	-0.69*	-0.70*	-0.59*	-0.68*	0.31*	-0.27*	-0.27*	-0.31*	1.00						
(14) imr	0.24*	-0.64*	0.04	-0.66*	0.49*	0.52*	0.31*	0.40*	-0.33*	0.08*	0.22*	0.23*	-0.72*	1.00					
(15) gov_eff	-0.33*	0.56*	-0.19*	0.56*	-0.36*	-0.46*	-0.26*	-0.45*	0.29*	-0.10*	-0.16*	-0.19*	0.60*	-0.61*	1.00				
(16) corr_control	-0.49*	0.51*	-0.33*	0.53*	-0.08*	-0.18*	0.03	-0.16*	0.19*	-0.01	-0.09*	-0.07*	0.40*	-0.41*	0.74*	1.00			
(17) battle_dum	0.19*	-0.13*	0.08*	-0.18*	-0.11*	-0.05*	-0.07*	0.03	-0.06*	-0.09*	0.02	0.03	-0.13*	0.17*	-0.13*	-0.19*	1.00		
(18) col_indfrom_uk	-0.21*	0.00	-0.27*	0.00	0.23*	0.18*	0.05*	0.06*	-0.06*	-0.02	-0.07*	-0.08*	-0.05	0.12*	0.12*	0.18*	-0.01	1.00	
(19) col_indfrom_fra	0.16*	-0.27*	0.01	-0.36*	0.17*	0.15*	0.23*	0.18*	-0.11*	0.08*	0.14*	0.16*	-0.24*	0.38*	-0.34*	-0.23*	0.04 -	0.31*	1.00
(20) col_indfrom_spa	-0.08*	0.21*	0.05	0.23*	-0.32*	-0.22*	-0.27*	-0.20*	0.11*	-0.06*	-0.10*	-0.11*	0.26*	-0.19*	0.12*	0.03	0.00 -	0.14* -	0.13* 1.00

Pairwise correlations, significance le	evel p<	<0.001																	
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19) (20)
(1) oop_che	1.00	)																	
(2) oop_gdp	-0.65*	1.00																	
(3) ghes_che	0.78*	-0.46*	1.00																
(4) ghes_gdp	-0.56*	0.82*	-0.13*	1.00															
(5) ln_odaplus_health_pub_gdp	-0.10*	-0.39*	-0.17*	-0.32*	1.00														
(6) ln_odaplus_health_civpriv_gdp	-0.06	-0.47*	-0.08	-0.35*	0.82*	1.00													
(7) ln_odaplus_nonhealth_pub_gdp	-0.07	-0.29*	-0.08	-0.21*	0.81*	0.70*	1.00												
(8) ln_odaplus_nonhealth_civpriv_gdp	-0.03	-0.38*	0.01	-0.25*	0.77*	0.79*	0.82*	1.00											
(9) ln_debt_serv_gdp	-0.08	0.29*	0.02	0.29*	-0.18*	-0.20*	-0.05	-0.11*	1.00										
(10) ln_debt_stock_gdp	0.00	-0.11*	0.06	-0.05	0.36*	0.32*	0.47*	0.48*	0.54*	1.00									
(11) imf_cond_ba1tot	0.09*	-0.22*	0.15*	-0.12*	0.25*	0.23*	0.30*	0.29*	0.01	0.17*	1.00								
(12) imf_part	0.10*	-0.26*	0.17*	-0.13*	0.29*	0.28*	0.37*	0.34*	0.01	0.20*	0.84*	1.00							
(13) gdp_cap_c	-0.22*	0.68*	-0.13*	0.58*	-0.69*	-0.70*	-0.59*	-0.68*	0.31*	-0.27*	-0.27*	-0.31*	1.00						
(14) imr	0.24*	-0.64*	0.04	-0.66*	0.49*	0.52*	0.31*	0.40*	-0.33*	0.08	0.22*	0.23*	-0.72*	1.00					
(15) gov_eff	-0.33*	0.56*	-0.19*	0.56*	-0.36*	-0.46*	-0.26*	-0.45*	0.29*	-0.10*	-0.16*	-0.19*	0.60*	-0.61*	1.00				
(16) corr_control	-0.49*	0.51*	-0.33*	0.53*	-0.08	-0.18*	0.03	-0.16*	0.19*	-0.01	-0.09*	-0.07	0.40*	-0.41*	0.74*	1.00			
(17) battle_dum	0.19*	-0.13*	0.08	-0.18*	-0.11*	-0.05	-0.07	0.03	-0.06	-0.09*	0.02	0.03	-0.13*	0.17*	-0.13*	-0.19*	1.00		
(18) col_indfrom_uk	-0.21*	0.00	-0.27*	0.00	0.23*	0.18*	0.05	0.06	-0.06	-0.02	-0.07	-0.08	-0.05	0.12*	0.12*	0.18*	-0.01	1.00	
(19) col_indfrom_fra	0.16*	-0.27*	0.01	-0.36*	0.17*	0.15*	0.23*	0.18*	-0.11*	0.08	0.14*	0.16*	-0.24*	0.38*	-0.34*	-0.23*	0.04 -	-0.31*	1.00
(20) col_indfrom_spa	-0.08	0.21*	0.05	0.23*	-0.32*	-0.22*	-0.27*	-0.20*	0.11*	-0.06	-0.10*	-0.11*	0.26*	-0.19*	0.12*	0.03	0.00 ·	-0.14* -	0.13* 1.00

#### п. .

## 10.4.3.2. Variance Inflation Factors for multicollinearity

The below tables show variables with moderate/high degrees of multicollinearity, having Variance Inflation Factors (VIF) above 5, for all models:

## Model 1)

Variable	VIF
L. ln_odaplus_nonhealth_pub_gdp	6.42
L. ln_odaplus_health_pub_gdp	6.01
L. ln_odaplus_nonhealth_civpriv_gdp	5.86
ln_gdp_cap_c	5.04

## Model 2)

Variable	VIF
L. ln_odaplus_nonhealth_pub_gdp	6.40
L. ln_odaplus_health_pub_gdp	6.11
L. ln_odaplus_nonhealth_civpriv_gdp	5.88
ln_gdp_cap_c	5.17

## Model 3)

Variable	VIF
L. ln_odaplus_nonhealth_pub_gdp	6.40
L. ln_odaplus_health_pub_gdp	5.88
L. ln_odaplus_nonhealth_civpriv_gdp	5.72
ln_gdp_cap_c	5.42

## Model 4)

Variable	VIF
L. ln_odaplus_nonhealth_pub_gdp	6.40
L. ln_odaplus_nonhealth_civpriv_gdp	5.86
L. ln_odaplus_health_pub_gdp	5.85
ln_gdp_cap_c	5.03

#### Model 1a)

None

## Model 2a)

None

## Model 3a)

Variable	VIF
L. ghes_che	17.84
L2. ghes_che	17.37

### Model 4a)

None

# 10.4.3.3. Unit root testing of dependent variables

AIC= Akaike Information Criterion. Null hypothesis: Panels contain unit root. For Im-Pesaran Shin unit-root test: T-bar needs to have a higher absolute value than cut-off to reject null.

## OOP/CHE

Test type	ADF lags	Trend?	Demean?	Test statistic	Value	5% alpha-level cutoff
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	*
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	No	T-bar	-2.3920	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	No	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	Yes	T-bar	-2.3511	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	Yes	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	2.00 (chosen by AIC)	Yes	No	W-t-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	2.07 (chosen by AIC)	Yes	Yes	W-t-bar (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	Yes	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	2.00 (chosen by AIC)	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	2.07 (chosen by AIC)	Yes	Yes	Adjusted t (p-value)	0.0000	

## Ln(OOP/GDP)

Test type	Lags	Trend?	Demean?	Test statistic	Value	5% alpha-level cutoff
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	No	T-bar	-2.3856	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	No	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	Yes	T-bar	-2.3941	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	Yes	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	2.13 (chosen by AIC)	Yes	No	W-t-bar (p-value)	0.0004	
Im-Pesaran-Shin unit-root test	2.21 (chosen by AIC)	Yes	Yes	W-t-bar (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	Yes	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	2.13 (chosen by AIC)	Yes	No	Adjusted t (p-value)	0.0223	
Levin-Lin-Chu	1.21 (chosen by AIC)	Yes	Yes	Adjusted t (p-value)	0.0000	

## GHES/CHE

Test type	Lags	Trend?	Demean?	Test statistic	Value	5% alpha-level cutoff
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	No	T-bar	-2.5662	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	No	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	Yes	T-bar	-2.5712	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	Yes	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	1.78 (chosen by AIC)	Yes	No	W-t-bar (p-value)	0.0247	
Im-Pesaran-Shin unit-root test	1.85 (chosen by AIC)	Yes	Yes	W-t-bar (p-value)	0.0019	
Levin-Lin-Chu	1	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	Yes	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1.78 (chosen by AIC)	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1.85 (chosen by AIC)	Yes	Yes	Adjusted t (p-value)	0.0000	

## Ln(GHES/GDP)

Test type	Lags	Trend?	Demean?	Test statistic	Value	5% alpha-level cutoff
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Augmented Dickey-Fuller)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	No	Inverse chi-squared (p-value)	0.0000	
Fisher-type unit-root test (Phillips-Perron)	1	Yes	Yes	Inverse chi-squared (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	No	T-bar	-2.4913	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	No	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	0	Yes	Yes	T-bar	-2.5598	-2.320
Im-Pesaran-Shin unit-root test	0	Yes	Yes	Z-t-tilde-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	1.98 (chosen by AIC)	Yes	No	W-t-bar (p-value)	0.0000	
Im-Pesaran-Shin unit-root test	2.25 (chosen by AIC)	Yes	Yes	W-t-bar (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1	Yes	Yes	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	1.98 (chosen by AIC)	Yes	No	Adjusted t (p-value)	0.0000	
Levin-Lin-Chu	2.25 (chosen by AIC)	Yes	Yes	Adjusted t (p-value)	0.0000	

## 10.4.4. Sensitivity analysis

## 10.4.4.1. Alternative lag-specifications

(Year dummies were included in all models but not presented here for brevity. GMM-style instrument generation limited to 5-year lags to show isolated effect of changing lag structures).

## No lags on independent variables of interest:

- Full models:

## Model 1

oop_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig	
L. oop_che	.72	.06	13.12	0	.61	.83	***	
ln_odaplus_health_pub_gdp	27	.29	-0.93	.35	83	.3		
ln_odaplus_health_civpriv_gdp	37	.21	-1.80	.08	78	.04	*	
ln_odaplus_nonhealth_pub_gdp	.53	.55	0.97	.33	55	1.62		
ln_odaplus_nonhealth_civpriv_gdp	.11	.37	0.28	.78	64	.85		
ln_debt_serv_gdp	18	.54	-0.34	.74	-1.26	.9		
ln_debt_stock_gdp	.19	.78	0.25	.8	-1.36	1.74		
imf_cond_ba1tot	01	.01	-0.57	.57	03	.02		
imf_part	.43	.54	0.80	.43	64	1.49		
ln_ghes_gdp	-6.94	1.49	-4.67	0	-9.89	-3.99	***	
ln_gdp_cap_c	19	2.09	-0.09	.93	-4.34	3.95		
imr	11	.06	-1.91	.06	23	0	*	
gov_eff	39	1.76	-0.22	.83	-3.88	3.1		
corr_control	-1.19	2.27	-0.53	.6	-5.7	3.31		
battle_dum	.84	.83	1.00	.32	82	2.49		
col_indfrom_uk	-1.99	1.68	-1.18	.24	-5.33	1.35		
col_indfrom_fra	87	1.72	-0.51	.61	-4.29	2.55		
col_indfrom_spa	.34	1.36	0.25	.8	-2.37	3.05		
Mean dependent var		40.	102 SD	dependent va	ar 19	19.365		
Number of obs		1	443 F-te	st	766	766.482		

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 92 Lag limit (years): 4 AB test (AR2) (p-value): 0.916 Hansen test (p-value): 0.220 Difference-in-Hansen test (p-value): 0.525

Μ	od	el	2
	~ ~		

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_oop_gdp	.94	.05	17.69	0	.83	1.04	***
ln_odaplus_health_pub_gdp	01	.01	-1.29	.2	03	.01	
ln_odaplus_health_civpriv_gdp	01	.01	-1.21	.23	03	.01	
ln_odaplus_nonhealth_pub_gdp	.01	.02	0.33	.74	03	.05	
ln_odaplus_nonhealth_civpriv_gdp	.01	.01	0.59	.56	02	.04	
ln_debt_serv_gdp	01	.01	-0.86	.39	03	.01	
ln_debt_stock_gdp	.03	.03	0.92	.36	03	.08	
imf_cond_ba1tot	0	0	-0.75	.46	0	0	
imf_part	.01	.02	0.43	.67	03	.05	
ln_ghes_gdp	.07	.04	1.71	.09	01	.14	*
ln_gdp_cap_c	07	.07	-1.00	.32	19	.06	
imr	0	0	0.13	.9	0	0	
gov_eff	0	.05	0.10	.92	09	.1	

corr_control	02	.05	-0.45	.65	12	.08	
battle_dum	.03	.02	1.66	.1	01	.07	*
col_indfrom_uk	03	.04	-0.72	.47	11	.05	
col_indfrom_fra	01	.05	-0.14	.89	1	.08	
col_indfrom_spa	0	.06	0.05	.96	12	.13	
Mean dependent var		0.5	0.73	0.733			
Number of obs	1443 F-test 261.474						

Countries: 105

Instrument count: 92

Lag limit (years): 4

AB test (AR2) (p-value): 0.458 Hansen test (p-value): 0.275

Difference-in-Hansen test (p-value): 0.170

#### Model 3

ghes_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ghes_che	.83	.08	9.84	0	.66	.99	***
ln_odaplus_health_pub_gdp	.03	.25	0.11	.92	48	.53	
ln_odaplus_health_civpriv_gdp	16	.19	-0.82	.41	54	.22	
ln_odaplus_nonhealth_pub_gdp	09	.56	-0.17	.87	-1.21	1.02	
ln_odaplus_nonhealth_civpriv_gdp	.6	.5	1.19	.24	4	1.6	
ln_debt_serv_gdp	.28	.3	0.93	.35	32	.88	
ln_debt_stock_gdp	45	.78	-0.58	.56	-1.99	1.09	
imf_cond_ba1tot	.01	.01	1.15	.25	01	.04	
imf_part	-1.13	.51	-2.21	.03	-2.14	12	**
ln_gdp_cap_c	2.23	2.4	0.93	.36	-2.53	7	
imr	05	.07	-0.73	.47	19	.09	
gov_eff	-2.29	1.5	-1.53	.13	-5.27	.68	
corr_control	2.52	1.85	1.36	.18	-1.15	6.2	
battle_dum	.21	.58	0.37	.71	93	1.35	
col_indfrom_uk	.72	1.18	0.61	.54	-1.61	3.06	
col_indfrom_fra	.24	1.03	0.23	.82	-1.8	2.28	
_col_indfrom_spa	1.61	1.86	0.86	.39	-2.09	5.3	
Mean dependent var		40.	225 SD	ar 19	19.033		
Number of obs		1	443 F-te	985	985.136		
*** <i>p</i> <.01, ** <i>p</i> <.05, * <i>p</i> <.1							

Countries: 105 Instrument count: 73 Lag limit (years): 3

AB test (AR2) (p-value): 0.192 Hansen test (p-value): 0.430

Difference-in-Hansen test (p-value): 0.768

Model 4							
ln_ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ghes_dp	.82	.07	12.32	0	.69	.95	***
ln_odaplus_health_pub_gdp	.01	.01	1.35	.18	0	.02	
ln_odaplus_health_civpriv_gdp	0	.01	0.03	.97	02	.02	
ln_odaplus_nonhealth_pub_gdp	0	.02	-0.09	.93	04	.03	
ln_odaplus_nonhealth_civpriv_gdp	.01	.02	0.75	.45	02	.05	
ln_debt_serv_gdp	.02	.01	1.60	.11	0	.04	
ln_debt_stock_gdp	0	.03	-0.17	.87	06	.05	
imf_cond_ba1tot	0	0	0.02	.98	0	0	
imf_part	01	.02	-0.56	.58	05	.03	
ln_gdp_cap_c	.09	.07	1.20	.23	06	.24	

imr	0	0	-1.18	.24	01	0	
gov_eff	04	.05	-0.83	.41	14	.06	
corr_control	.01	.06	0.21	.83	11	.14	
battle_dum	.03	.03	1.26	.21	02	.09	
col_indfrom_uk	01	.04	-0.16	.87	09	.07	
col_indfrom_fra	04	.04	-0.92	.36	13	.05	
col_indfrom_spa	.03	.04	0.69	.49	05	.11	
Mean dependent var		0.5	0.721				
Number of obs	1443 F-test 377.266						

Countries: 105 Instrument count: 73 Lag limit (years): 3 AB test (AR2) (p-value): 0.458 Hansen test (p-value): 0.357 Difference-in-Hansen test (p-value): 0.397

## - Reduced models:

Model 1a

oop_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. oop_che	.68	.06	10.98	0	.55	.8	***
ln_odaplus_health_pub_gdp	03	.26	-0.13	.89	55	.48	
ln_odaplus_health_civpriv_gdp	09	.24	-0.39	.7	57	.38	
ln_debt_serv_gdp	.18	.57	0.31	.76	96	1.31	
ln_ghes_gdp	-7.9	1.49	-5.31	0	-10.85	-4.95	***
ln_gdp_cap_c	.29	2.05	0.14	.89	-3.79	4.36	
imr	13	.05	-2.36	.02	23	02	**
gov_eff	.36	1.2	0.30	.76	-2.02	2.75	
corr_control	73	2.05	-0.35	.72	-4.79	3.34	
battle_dum	.51	.81	0.63	.53	-1.1	2.12	
col_indfrom_uk	-2.72	1.91	-1.42	.16	-6.5	1.07	
col_indfrom_fra	36	1.82	-0.20	.84	-3.97	3.25	
col_indfrom_spa	.59	1.45	0.41	.68	-2.29	3.47	
Mean dependent var		40.	102 SD	dependent va	ar 19	.365	
Number of obs		1	443 F-te	st	483	.073	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 87 Lag limit (years): 6 AB test (AR2) (p-value): 0.926 Hansen test (p-value): 0.301 Difference-in-Hansen test (p-value): 0.228

### Model 2a

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_oop_gdp	.92	.06	14.83	0	.8	1.05	***
ln_odaplus_health_pub_gdp	01	.01	-1.22	.22	03	.01	
ln_odaplus_health_civpriv_gdp	01	.01	-0.71	.48	02	.01	
ln_odaplus_nonhealth_civpriv_gdp	.02	.02	1.30	.2	01	.06	
ln_ghes_gdp	.11	.04	2.47	.02	.02	.19	**
ln_gdp_cap_c	14	.07	-1.87	.06	28	.01	*
imr	0	0	-0.88	.38	01	0	

gov_eff	.05	.06	0.97	.34	06	.17
corr_control	03	.06	-0.41	.68	15	.1
battle_dum	.03	.02	1.42	.16	01	.08
col_indfrom_uk	04	.06	-0.74	.46	15	.07
col_indfrom_fra	.03	.06	0.59	.56	08	.15
col_indfrom_spa	.05	.08	0.70	.48	1	.21
Mean dependent var	0.550 SD dependent var 0.733					
Number of obs	1443 F-test 128.934					

Countries: 105 Instrument count: 87 Lag limit (years): 6 AB test (AR2) (p-value): 0.518 Hansen test (p-value): 0.182 Difference-in-Hansen test (p-value): 0.375

## Model 3a

Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
.71	.08	9.12	0	.56	.87	***
.22	.08	2.92	0	.07	.37	***
28	.52	-0.53	.6	-1.32	.76	
.05	.34	0.14	.89	62	.72	
18	2.32	-0.08	.94	-4.78	4.42	
03	.06	-0.51	.61	15	.09	
-1.28	1.31	-0.98	.33	-3.88	1.31	
2.82	1.55	1.81	.07	26	5.89	*
21	.54	-0.39	.7	-1.28	.86	
.22	.89	0.24	.81	-1.55	1.99	
.35	.93	0.38	.71	-1.5	2.2	
1.42	1.03	1.37	.17	63	3.46	
40.566 SD dependent var 19.169						
1356 F-test 2714.					.024	
	Coef.           .71           .22          28           .05          18          03           -1.28           2.82          21           .22           .35           1.42	Coef.         St.Err.           .71         .08           .22         .08          28         .52           .05         .34          18         2.32          03         .06           -1.28         1.31           2.82         1.55          21         .54           .22         .89           .35         .93           1.42         1.03	Coef.St.Err.t-value.71.08 $9.12$ .22.08 $2.92$ 28.52 $-0.53$ .05.34 $0.14$ 18 $2.32$ $-0.08$ 03.06 $-0.51$ -1.28 $1.31$ $-0.98$ 2.82 $1.55$ $1.81$ 21.54 $-0.39$ .22.89 $0.24$ .35.93 $0.38$ $1.42$ $1.03$ $1.37$ 40.566SD1356F-te	Coet.         St.Err.         t-value         p-value           .71         .08         9.12         0           .22         .08         2.92         0          28         .52         -0.53         .6           .05         .34         0.14         .89          18         2.32         -0.08         .94          03         .06         -0.51         .61           -1.28         1.31         -0.98         .33           2.82         1.55         1.81         .07          21         .54         -0.39         .7           .22         .89         0.24         .81           .35         .93         0.38         .71           1.42         1.03         1.37         .17           40.566         SD dependent v         1356         F-test	Coet.St.Err.t-valuep-value $95\%$ Cont.71.089.120.56.22.082.920.0728.52-0.53.6-1.32.05.340.14.8962182.32-0.08.94-4.7803.06-0.51.6115-1.281.31-0.98.33-3.882.821.551.81.072621.54-0.39.7-1.28.22.890.24.81-1.55.35.930.38.71-1.51.421.031.37.176340.566SD dependent var191356F-test2714	Coef.St.Err.t-valuep-value[95% ContInterval].71.089.120.56.87.22.082.920.07.3728.52-0.53.6-1.32.76.05.340.14.8962.72182.32-0.08.94-4.784.4203.06-0.51.6115.09-1.281.31-0.98.33-3.881.312.821.551.81.07265.8921.54-0.39.7-1.28.86.22.890.24.81-1.551.99.35.930.38.71-1.633.4640.566SD dependent var19.1691356F-test2714.024

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 88 Lag limit (years): 8 AB test (AR2) (p-value): 0.647 Hansen test (p-value): 0.225 Difference-in-Hansen test (p-value): 0.721

#### Model 4a

ln_ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_ghes_gdp	.74	.09	7.92	0	.55	.92	***
ln_odaplus_health	.01	.01	0.64	.52	01	.02	
_pub_gdp							
ln_gdp_cap_c	.21	.1	1.99	.05	0	.41	**
imr	0	0	0.42	.67	0	.01	
gov_eff	04	.06	-0.61	.54	15	.08	
corr_control	.03	.07	0.42	.68	12	.18	
battle_dum	.03	.03	1.03	.31	03	.09	
col_indfrom_uk	04	.06	-0.73	.47	15	.07	
col_indfrom_fra	11	.07	-1.49	.14	26	.04	
col_indfrom_spa	02	.07	-0.35	.73	16	.11	
Mean dependent var		0.563	SD depen	ident var		0.721	
Number of obs		1460	F-test			241.976	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 73 Lag limit (years): 7 AB test (AR2) (p-value): 0.200 Hansen test (p-value): 0.230 Difference-in-Hansen test (p-value): 0.317

## Two lags on independent variables of interest:

## - Full models (only significant variations shown):

## Model 2

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_oop_gdp	.93	.07	12.62	0	.78	1.08	***
L2. ln_odaplus_health_pub_gdp	02	.01	-1.95	.05	04	0	*
L2. ln_odaplus_health_civpriv_gdp	01	.01	-1.56	.12	03	0	
L2. ln_odaplus_nonhealth_pub_gdp	.01	.01	0.91	.37	02	.04	
L2. ln_odaplus_nonhealth_civpriv_gdp	03	.01	-2.52	.01	06	01	**
ln_debt_serv_gdp	0	.02	-0.03	.98	03	.03	
ln_debt_stock_gdp	.02	.04	0.52	.6	06	.1	
imf_cond_ba1tot	0	0	-0.68	.5	0	0	
imf_part	.02	.03	0.96	.34	03	.07	
ln_ghes_gdp	.07	.06	1.16	.25	05	.18	
ln_gdp_cap_c	22	.11	-1.97	.05	45	0	*
imr	0	0	-1.31	.19	01	0	
gov_eff	.03	.06	0.50	.62	09	.14	
corr_control	01	.08	-0.16	.87	16	.14	
battle_dum	.02	.02	0.85	.4	03	.07	
col_indfrom_uk	.03	.06	0.49	.63	09	.15	
col_indfrom_fra	.05	.06	0.79	.43	07	.16	
col_indfrom_spa	.06	.06	0.93	.36	06	.18	
Mean dependent var		0.	547 SD	dependent v	ar 0	0.733	
Number of obs		1.	326 F-te	est	132	.998	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 76 Lag limit (years): 3 AB test (AR2) (p-value): 0.567 Hansen test (p-value): 0.151 Difference-in-Hansen test (p-value): 0.330

## - Reduced models (only significant variations shown):

Mo	del	2a

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_oop_gdp	.94	.05	17.51	0	.84	1.05	***
L2. ln_odaplus_health_pub_gdp	02	.01	-2.29	.02	04	0	**
L2. ln_odaplus_health_civpriv_gdp	02	.01	-2.31	.02	03	0	**
ln_odaplus_nonhealth_civpriv_gdp	.01	.02	0.54	.59	02	.04	

ln_ghes_gdp	.08	.04	2.08	.04	0	.16	**
ln_gdp_cap_c	2	.07	-2.97	0	33	07	***
imr	0	0	-1.39	.17	01	0	
gov_eff	.06	.05	1.15	.25	04	.16	
corr_control	03	.05	-0.64	.53	12	.06	
battle_dum	.04	.02	1.55	.12	01	.08	
col_indfrom_uk	0	.05	0.07	.94	1	.11	
col_indfrom_fra	.02	.05	0.44	.66	08	.12	
col_indfrom_spa	.04	.05	0.86	.39	05	.14	
Mean dependent var		0.5	0.73	33			
Number of obs	1326 F-test 194.769						

Countries: 105 Instrument count: 86 Lag limit (years): 6 AB test (AR2) (p-value): 0.595 Hansen test (p-value): 0.337 Difference-in-Hansen test (p-value): 0.687

#### Model 4a

ln_ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_ghes_gdp	.7	.09	8.02	0	.53	.87	***
L2. ln_odaplus_health_pub_gdp	02	.01	-2.21	.03	04	0	**
ln_gdp_cap_c	.14	.1	1.44	.15	05	.34	
imr	0	0	0.13	.9	01	.01	
gov_eff	0	.06	-0.06	.95	12	.11	
corr_control	.03	.09	0.29	.77	16	.21	
battle_dum	.06	.03	1.70	.09	01	.13	*
col_indfrom_uk	01	.08	-0.18	.86	17	.14	
col_indfrom_fra	13	.09	-1.35	.18	31	.06	
col_indfrom_spa	02	.09	-0.24	.81	21	.16	
Mean dependent var		0.	571 SD	dependent v	ar 0.	.720	
Number of obs		1	356 F-te	st	111.	.931	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 72 Lag limit (years): 7 AB test (AR2) (p-value): 0.340 Hansen test (p-value): 0.470 Difference-in-Hansen test (p-value): 0.664

## 10.4.4.2. Alternative lag limits

The below table shows how changing model lag limits affects the p-value of significant variables identified in our model specifications. P-values less than 0.05 are highlighted in **bold**. Insignificant variables are not shown. Dependent variable lags were significant in all models but not shown. The instrument matrix is collapsed for all specifications:

## Model 1)

oop_che		]	Direction of association			
Independent variable	2	3	4	5	6	
L. ln_odaplus_health_civpriv_gdp (p-value)	0.078	0.085	0.021	0.012	0.076	-
ln_ghes_gdp (p-value)	0.000	0.000	0.000	0.000	0.000	-
imr (p-value)	0.024	0.010	0.038	0.006	0.011	-
Hansen-test p-value	0.206	0.506	0.262	0.558	0.856	
Instrument count	62	77	92	107	122	

## Model 2)

ln_oop_gdp	Lag limits					Direction of association
Independent variable	2	3	4	5	6	
L. ln_odaplus_health_pub_gdp (p-value)	0.227	0.082	0.086	0.099	0.027	-
L. ln_odaplus_health_civpriv_gdp (p-value)	0.107	0.078	0.050	0.005	0.028	-
ln_ghes_gdp (p-value)	0.018	0.063	0.092	0.105	0.067	+
_ln_gdp_cap_c (p-value)	0.057	0.019	0.029	0.012	0.002	-
Hansen-test p-value	0.052	0.113	0.366	0.546	0.869	
Instrument count	62	77	92	107	122	

## Model 3)

## None significant

## Model 4)

ln_ghes_gdp		Direction of association				
Independent variable	2	3	4	5	6	
ln_odaplus_health_pub_gdp (p-value)	0.128	0.062	0.075	0.008	0.035	-
Hansen-test p-value	0.153	0.592	0.615	0.480	0.670	
Instrument count	59	73	87	101	115	

# Model 1a)

oop_che		Direction of association				
Independent variable	3	4	5	6	7	
L. ln_debt_serv_gdp (p-value)	0.066	0.060	0.041	0.065	0.060	+
ln_ghes_gdp (p-value)	0.000	0.000	0.000	0.000	0.000	-
imr (p-value)	0.011	0.013	0.008	0.005	0.006	-
Hansen-test p-value	0.224	0.194	0.402	0.624	0.493	
Instrument count	57	67	77	78	97	

# Model 2a)

ln_oop_gdp			Direction of association			
Independent variable	5	6	7	8	9	
L. ln_odaplus_health_pub_gdp (p-value)	0.107	0.061	0.017	0.007	0.002	-
L. ln_odaplus_health_civpriv_gdp (p-value)	0.048	0.092	0.047	0.036	0.067	-
L. ln_odaplus_nonhealth_civpriv_gdp (p-value)	0.121	0.050	0.037	0.061	0.097	-
ln_ghes_gdp (p-value)	0.085	0.030	0.048	0.039	0.016	+
_ln_gdp_cap_c (p-value)	0.030	0.004	0.001	0.000	0.001	-
Hansen-test p-value	0.263	0.330	0.469	0.403	0.803	
Instrument count	77	87	97	107	117	

# Model 3a)

ghes_che		]	Direction of association			
Independent variable	6	7	8	9	10	
L. ln_debt_serv_gdp (p-value)	0.061	0.101	0.023	0.038	0.042	-
corr_control (p-value)	0.075	0.111	0.017	0.010	0.027	+
Hansen-test p-value	0.224	0.296	0.242	0.223	0.211	
Instrument count	72	80	88	96	109	

# Model 4a)

ln_ghes_gdp		Direction of association				
Independent variable	5	6	7	8	9	
ln_odaplus_health_pub_gdp (p-value)	0.082	0.065	0.045	0.037	0.143	-
col_indfrom_fra (p-value)	0.370	0.247	0.191	0.107	0.049	-
Hansen-test p-value	0.334	0.283	0.346	0.302	0.309	
Instrument count	59	66	77	80	87	

We also examined the sensitivity of our findings for IMF participation in a contemporaneous version of model 3:

ln_ghes_che		Direction of association				
Independent variable	2	3	4	5	6	
imf_part (p-value)	0.389	0.029	0.036	0.042	0.085	-
Hansen-test p-value	0.359	0.430	0.302	0.450	0.829	
Instrument count	59	73	87	101	115	

# Model 3, contemporaneous specification

## 10.4.4.3. Alternative variables

The below table displays the sensitivity of our findings to swapping variables with alternative variables. Only significant variables shown. Lagged dependent variables significant in all models but not shown for brevity. Year dummies included in all models but not shown for brevity. Per GDP to per cap: All financial variables measured per GDP swapped to per capita, including dependent variables. IMR: Infant Mortality Rate. U5MR: Under-5-Mortality Rate. Cond: Conditionality variable (imf\_cond\_ba1tot swapped for imf\_cond\_ba2tot and imf\_cond\_ba3tot). Colony: Country colonial independence was obtained from. SSA: Sub-Saharan Africa. WB Inc: World Bank Income Group.

Italic used to highlight problematically high instrument counts necessary to maintain acceptable test statistics in alternative model, or problematically high Hansen-test p-values.

#### Model 1

oop_che									Direction of
									association
Original variable:	IMR	IMR	Cond. 1	Cond. 1	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	Cond. 2	Cond. 3	-	WB Inc.	SSA	Per cap.*	
L. ln_odaplus_health_civpriv_gdp (p-value)	0.009	0.013	0.039	0.028	0.027	0.069	0.034	0.023	-
ln_ghes_gdp (p-value)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	-
ln_gdp_cap_c (p-value)	0.611	0.281	0.943	0.957	0.762	0.269	0.849	0.005	+
imr (p-value)	0.219	0.713	0.041	0.038	0.029	0.033	0.076	0.088	-
Hansen-test p-value	0.172	0.124	0.205	0.242	0.158	0.154	0.163	0.294	
Instrument count	92	92	92	92	89	91	90	92	

\*Financial independent variables per capita.

## Model 2

ln_oop_gdp									Direction of association
Original variable:	IMR	IMR	Cond. 1	Cond. 1	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	Cond. 2	Cond. 3	-	WB Inc.	SSA	Per cap.*	
L. ln_odaplus_health_civpriv_gdp	0.080	0.081	0.043	0.046	0.028	0.068	0.032	0.210	-
ln_gdp_cap_c (p-value)	0.045	0.049	0.031	0.029	0.014	0.102	0.019	0.037	+
imr (p-value)	0.639	0.862	0.335	0.344	0.198	0.389	0.076	0.027	-
corr_control	0.746	0.523	0.772	0.789	0.814	0.807	0.673	0.036	+
Hansen-test p-value	0.342	0.124	0.368	0.371	0.358	0.289	0.371	0.180	
Instrument count	92	92	92	92	89	91	90	107	

\*Both dependent variable and financial independent variables per capita.

# Model 3)

# None significant.

## Model 4

ln_ghes_gdp									Direction of
									association
Original variable:	IMR	IMR	Cond. 1	Cond. 1	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	Cond. 2	Cond. 3	-	WB Inc.	SSA	Per cap.*	
L. ln_odaplus_health_pub_gdp	0.040	0.049	0.041	0.042	0.053	0.072	0.018	0.071	-
L. ln_odaplus_nonhealth_pub_gdp	0.756	0.567	0.870	0.805	0.786	0.778	0.762	0.019	+
ln_gdp_cap_c (p-value)	0.299	0.104	0.341	0.393	0.238	0.366	0.333	0.000	+
Hansen-test p-value	0.600	0.575	0.565	0.553	0.631	0.619	0.651	0.166	
Instrument count	73	73	73	73	70	72	71	73	

\*Both dependent variable and financial independent variables per capita.

## Model 1a

oop_che							Direction of
Original variable	IMR	IMR	Colony	Colony	Colony	Per GDP	association
New variable:	U5MR	MMR	-	WB Inc.	SSA	Per cap.*	
L. ln debt serv gdp (p-value)	0.062	0.076	0.036	0.017	0.037	0.151	+
ln_ghes_gdp (p-value)	0.000	0.000	0.000	0.000	0.000	0.000	-
ln_gdp_cap_c (p-value)	0.893	0.874	0.528	0.604	0.485	0.018	+
imr (p-value)	0.138	0.638	0.005	0.013	0.015	0.021	-
Hansen-test p-value	0.367	0.238	0.400	0.310	0.404	0.396	
Instrument count	97	97	97	74	75	77	

\*Financial independent variables per capita.

#### Model 2a

ln_oop_gdp							Direction of association
Original variable:	IMR	IMR	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	-	WB Inc.	SSA	Per cap.*	
L. ln_odaplus_health_pub_gdp	0.022	0.031	0.020	0.062	0.023	0.759	-
L. ln_odaplus_health_civpriv_gdp	0.018	0.024	0.024	0.015	0.040	0.717	-
L. ln_odaplus_nonhealth_civpriv_gdp	0.040	0.031	0.033	0.011	0.040	0.334	-
ln_ghes_gdp	0.044	0.071	0.082	0.386	0.067	0.440	+
ln_gdp_cap_c	0.006	0.008	0.001	0.094	0.001	0.139	-
imr (p-value)	0.138	0.638	0.072	0.013	0.179	0.002	-
Hansen-test p-value	0.473	0.467	0.438	0.194	0.465	0.195	
Instrument count	97	97	94	96	95	107	

\*Both dependent variable and financial independent variables per capita.

## Model 3a

ghes_che							Direction of
							association
Original variable:	IMR	IMR	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	-	WB Inc.	SSA	Per cap.*	
L. ln_debt_serv_gdp	0.022	0.014	0.019	0.012	0.028	0.020	-
corr_control	0.015	0.009	0.037	0.269	0.077	0.017	+

\*Financial independent variables per capita.

## Model 4a

ln_ghes_gdp							Direction of
							association
Original variable:	IMR	IMR	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	-	WB Inc.	SSA	Per cap.*	
L. ln_odaplus_health_pub_gdp	0.034	0.025	0.037	0.035	0.017	0.229	-
ln_gdp_cap_c	0.157	0.113	0.158	0.182	0.220	0.000	+
Hansen-test p-value	0.268	0.413	0.290	0.265	0.323	0.341	
Instrument count	73	73	70	72	71	108	

\*Both dependent variable and financial independent variables per capita.

# We also examined the sensitivity of our findings for IMF participation in a contemporaneous version of model 3:

## Model 3, contemporaneous specification

ln_ghes_che							Direction of
							association
Original variable:	IMR	IMR	Colony	Colony	Colony	Per GDP	
New variable:	U5MR	MMR	-	WB Inc.	SSA	Per cap.*	
imf_part	0.025	0.080	0.024	0.030	0.025	0.033	-
Hansen-test p-value	0.474	0.619	0.437	0.441	0.397	0.480	
Instrument count	73	73	70	72	71	73	

\*Financial independent variables per capita.

## 10.4.4.4. Interaction terms

Only models with significant interaction terms displayed. Year dummies included in all models but not shown for brevity. Problematic diagnostic test statistics highlighted in italic.

## - Region being Sub-Saharan Africa:

Model 3a
----------

ghes_che			t-	p-	[95%		
-	Coef.	St.Err.	value	value	Conf	Interval]	Sig
L. ghes_che	.72	.07	10.13	0	.58	.86	***
L2. ghes_che	.24	.07	3.35	0	.1	.38	***
L. ln_odaplus_nonhealth_civpriv_gdp	06	.43	-0.14	.89	91	.79	
reg_ssf*L.ln_odaplus_nonhealth_civpriv_gdp	2.37	.75	3.15	0	.88	3.86	***
L. ln_debt_serv_gdp	56	.48	-1.17	.24	-1.5	.38	
reg_ssf*ln_debt_serv_gdp	.02	.56	0.04	.97	-1.08	1.12	
ln_gdp_cap_c	.88	2.6	0.34	.74	-4.27	6.03	
imr	04	.06	-0.57	.57	16	.09	
gov_eff	-1.02	1.26	-0.81	.42	-3.53	1.48	
corr_control	2.81	1.64	1.72	.09	43	6.06	*
battle_dum	36	.67	-0.54	.59	-1.7	.97	
reg_ssf	3.26	1.93	1.69	.09	56	7.08	*
Mean dependent var		40.5	66 SD	dependent	19.1	.69	
			var				
Number of obs		13	56 F-tes	st	2080.1	.97	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 94 Lag limit (years): 7 AB test (AR2) (p-value): 0.693 Hansen test (p-value): 0.303 Difference-in-Hansen test (p-value): 0.804

## - World Bank income group being low-income:

This model is possibly overspecified, and the two significant interaction terms become insignificant with any lag limit change.

Model 4	1
---------	---

ln_ghes_gdp			t-	p-	[95%	Interval]	
	Coef.	St.Err.	value	value	Conf		Sig
L. ln_ghes_gdp	.97	.05	18.03	0	.87	1.08	***
L. ln_odaplus_health_pub_gdp	01	.01	-1.49	.14	03	0	
inc_lic* L.ln_odaplus_health_pub_gdp	07	.03	-2.13	.04	13	0	**
L. ln_odaplus_health_civpriv_gdp	0	.01	-0.14	.89	02	.01	
inc_lic*	.06	.02	2.93	0	.02	.1	***
L.ln_odaplus_health_civpriv_gdp							
L. ln_odaplus_nonhealth_pub_gdp	0	.02	0.04	.97	03	.03	
L. ln_odaplus_nonhealth_civpriv_gdp	.02	.02	1.11	.27	02	.06	
L. ln_debt_serv_gdp	01	.01	-0.49	.63	03	.02	
L. ln_debt_stock_gdp	02	.03	-0.63	.53	07	.04	
imf_cond_ba1tot	0	0	0.08	.94	0	0	
imf_part	0	.02	-0.10	.92	05	.04	

ln_gdp_cap_c	05	.11	-0.47	.64	27	.16
imr	0	0	0.10	.92	01	.01
gov_eff	.06	.05	1.19	.24	04	.15
corr_control	.05	.05	1.02	.31	05	.16
battle_dum	.03	.03	1.08	.28	02	.08
inc_lic	05	.1	-0.49	.62	24	.15
Mean dependent var		0.5	57 SD de	ependent var	0.72	23
Number of obs		14	29 F-test	-	571.20	00
*** p<.01, ** p<.05, * p<.1						

Countries: 105 Instrument count: 77 Lag limit (years): 3 AB test (AR2) (p-value): 0.418 Hansen test (p-value): 0.679 Difference-in-Hansen test (p-value): 0.306

## 10.4.4.5. No log-transformation

Year dummies included in all models but not shown for brevity. Problematic diagnostic tests and instrument counts highlighted in italic. Shown models have "cherry-picked" lag-structures, looking for significance (confirmation bias).

## Model 1

We were unable to produce a specification that convincingly passes all diagnostic tests and where instruments do not outnumber groups. Below is for "cherry-picked" variable lag-specification, looking for significance (confirmation bias). Model is overspecified and becomes invalid at lower instrument counts. Findings for debt variables disappear at different ODA+ lag-specifications.

oop cha	Coof	St Err	t value	o voluo	[05% Conf	Intervall	Sig
oop_cne	Coel.	St.E11.	t-value	p-value	[9570 COIII	Interval	Sig
L. oop_che	.72	.06	11.54	0	.59	.84	***
L. odaplus_health_pub_gdp	51	.8	-0.64	.53	-2.1	1.08	
L. odaplus_health_civpriv_gdp	-2.38	2.08	-1.15	.25	-6.5	1.74	
odaplus_nonhealth_pub_gdp	1	.14	-0.71	.48	37	.17	
odaplus_nonhealth_civpriv_gdp	44	.81	-0.55	.59	-2.05	1.16	
debt_serv_gdp	19	.09	-2.07	.04	37	01	**
debt_stock_gdp	.06	.03	2.32	.02	.01	.12	**
imf_cond_ba1tot	0	.01	-0.34	.74	03	.02	
imf_part	.83	.62	1.33	.19	4	2.05	
ghes_gdp	-3.38	.83	-4.07	0	-5.02	-1.73	***
gdp_cap_c	0	0	-0.53	.6	0	0	
imr	13	.05	-2.49	.01	23	03	**
gov_eff	-1.25	1.56	-0.80	.42	-4.33	1.84	
corr_control	99	1.77	-0.56	.58	-4.51	2.52	
battle_dum	.67	.85	0.79	.43	-1.01	2.35	
col_indfrom_uk	-1.97	1.56	-1.26	.21	-5.06	1.12	
col_indfrom_fra	-1.24	1.72	-0.72	.47	-4.65	2.17	
col_indfrom_spa	.66	1.65	0.40	.69	-2.61	3.93	
Mean dependent var		40.	183 SD	dependent v	ar 19	.385	
Number of obs		1	433 F-te	est	391	.839	
***							

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 107 Lag limit (years): 5 AB test (AR2) (p-value): 1.000 Hansen test (p-value): 0.186 Difference-in-Hansen test (p-value): 0.230

#### Model 2

oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. oop_gdp	.79	.09	8.97	0	.62	.97	***
L. odaplus_health_pub_gdp	09	.04	-2.11	.04	17	0	**
odaplus_health_civpriv_gdp	02	.17	-0.12	.9	36	.32	
L. odaplus_nonhealth_pub_gdp	.01	.01	0.47	.64	02	.03	
odaplus_nonhealth_civpriv_gdp	.11	.06	1.90	.06	0	.23	*
L. debt_serv_gdp	01	.02	-0.58	.56	04	.02	
L. debt_serv_gdp	0	0	-0.70	.49	0	0	
L2. imf_cond_ba1tot	0	0	-1.51	.13	0	0	
L2. imf_part	.17	.05	3.46	0	.07	.27	***
ghes_gdp	.02	.05	0.41	.68	07	.11	

gdp_cap_c	0	0	-1.14	.26	0	0	
imr	0	0	-1.10	.27	01	0	
gov_eff	.17	.14	1.26	.21	1	.45	
corr_control	26	.22	-1.16	.25	7	.18	
battle_dum	.13	.06	2.15	.03	.01	.24	**
col_indfrom_uk	08	.1	-0.79	.43	28	.12	
col_indfrom_fra	06	.1	-0.59	.56	26	.14	
col_indfrom_spa	06	.17	-0.35	.72	4	.28	
Mean dependent var		2.1	46 SD de	ependent var	1.342	2	
Number of obs		13	47 F-test		287.660	6	

Countries: 105 Instrument count: 91 Lag limit (years): 4 AB test (AR2) (p-value): 0.541 Hansen test (p-value): 0.235 Difference-in-Hansen test (p-value): 0.714

## Model 3

ghes_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ghes_che	.84	.08	11.00	0	.69	.99	***
L. odaplus_health_pub_gdp	86	.66	-1.30	.2	-2.16	.45	
odaplus_health_civ_priv_gdp	.75	1.1	0.68	.5	-1.44	2.93	
L. odaplus_nonhealth_pub_gdp	03	.08	-0.39	.7	2	.13	
odaplus_nonhealth_civpriv_gdp	28	.47	-0.59	.55	-1.2	.65	
L. debt_serv_gdp	04	.1	-0.41	.68	25	.16	
L. debt_stock_gdp	06	.02	-2.46	.02	1	01	**
imf_cond_ba1tot	.01	.01	1.33	.19	01	.04	
imf_part	99	.58	-1.70	.09	-2.14	.17	*
gdp_cap_c	0	0	1.02	.31	0	0	
imr	05	.05	-0.84	.4	15	.06	
gov_eff	-1.21	1.3	-0.92	.36	-3.79	1.38	
corr_control	2.21	2.03	1.09	.28	-1.81	6.24	
battle_dum	.4	.6	0.67	.5	78	1.58	
col_indfrom_uk	03	1.38	-0.02	.98	-2.78	2.71	
col_indfrom_fra	3	1.38	-0.22	.83	-3.04	2.43	
col_indfrom_spa	.56	1.67	0.33	.74	-2.76	3.88	
Mean dependent var		40.	261 SD	dependent va	ar 19	19.024	
Number of obs		1	447 F-te	est	667	.835	
XXXX							

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 73 Lag limit (years): 3 AB test (AR2) (p-value): 0.190 Hansen test (p-value): 0.378 Difference-in-Hansen test (p-value): 0.680

## Model 4

ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ghes_gdp	.89	.04	21.29	0	.81	.98	***
L. odaplus_health_pub_gdp	05	.03	-1.44	.15	11	.02	
odaplus_health_civpriv_gdp	.17	.08	2.12	.04	.01	.32	**
L2. odaplus_nonhealth_pub_gdp	0	.01	-0.50	.62	01	.01	
odaplus_nonhealth_civpriv_gdp	.01	.03	0.47	.64	05	.08	
L. debt_serv_gdp	0	.01	0.13	.89	01	.02	
L. debt_stock_gdp	0	0	0.23	.82	0	0	
L. imf_cond_ba1tot	0	0	-0.90	.37	0	0	
L. imf_part	.05	.05	1.03	.31	05	.15	
gdp_cap_c	0	0	-0.03	.98	0	0	
imr	0	0	-0.91	.37	01	0	
gov_eff	.07	.09	0.79	.43	11	.26	
corr_control	.13	.1	1.23	.22	08	.33	
battle_dum	.01	.04	0.38	.71	06	.09	
col_indfrom_uk	01	.06	-0.18	.86	14	.12	
col_indfrom_fra	0	.05	0.06	.95	1	.11	
col_indfrom_spa	.17	.1	1.61	.11	04	.37	
Mean dependent var		2.	225 SD	dependent va	ar 1	.429	
Number of obs		1	<u>347 F</u> -te	st	1114	.232	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 86 Lag limit (years): 4 AB test (AR2) (p-value): 0.535 Hansen test (p-value): 0.204 Difference-in-Hansen test (p-value): 0.779

#### Model 1a

oon che	Coef	St Err	t-value	n-value	[95% Conf	Intervall	Sig
	70	01.111.	12.00	p-value	[) 5 /0 COM	intervarj	1018
L. oop_che	.72	.05	13.28	0	.62	.83	ተተተ
L2. odaplus_health_pub_gdp	.05	.7	0.08	.94	-1.34	1.45	
L. odaplus_health_civpriv_gdp	-3.68	2.76	-1.33	.19	-9.16	1.79	
L2. debt_serv_gdp	.14	.13	1.08	.28	11	.39	
ghes_gdp	-3.5	.83	-4.19	0	-5.15	-1.84	***
gdp_cap_c	0	0	0.06	.95	0	0	
imr	15	.05	-2.97	0	25	05	***
gov_eff	2	1.5	-0.13	.89	-3.18	2.78	
corr_control	-3.95	1.97	-2.01	.05	-7.85	06	**
battle_dum	.29	.64	0.45	.65	98	1.55	
col_indfrom_uk	-1.85	1.36	-1.36	.18	-4.55	.86	
col_indfrom_fra	81	1.45	-0.56	.58	-3.69	2.07	
col_indfrom_spa	91	1.77	-0.51	.61	-4.42	2.6	
Mean dependent var		39.	858 SD	dependent va	ar 19	.368	
Number of obs		1	343 F-te	est	553	.256	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 96 Lag limit (years): 7 AB test (AR2) (p-value): 0.732 Hansen test (p-value): 0.332 Difference-in-Hansen test (p-value): 0.832

## Model 2a

oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig	
L. oop_gdp	.74	.11	6.43	0	.51	.96	***	
L2. odaplus_health_pub_gdp	12	.06	-1.99	.05	25	0	**	
L2. odaplus_health_civpriv_gdp	02	.13	-0.15	.88	27	.23		
odaplus_nonhealth_civpriv_gdp	.11	.05	2.12	.04	.01	.21	**	
ghes_gdp	03	.05	-0.60	.55	12	.06		
gdp_cap_c	0	0	-1.96	.05	0	0	*	
imr	0	0	-1.29	.2	01	0		
gov_eff	.1	.1	1.00	.32	1	.31		
corr_control	12	.17	-0.71	.48	47	.22		
battle_dum	.1	.07	1.53	.13	03	.24		
col_indfrom_uk	21	.14	-1.48	.14	48	.07		
col_indfrom_fra	13	.14	-0.94	.35	4	.14		
col_indfrom_spa	05	.21	-0.25	.81	47	.36		
Mean dependent var		2.157 SD dependent var			ar 1	1.347		
Number of obs		1330 F-test			225	225.319		

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 86 Lag limit (years): 6 AB test (AR2) (p-value): 0.590 Hansen test (p-value): 0.337 Difference-in-Hansen test (p-value): 0.384

#### Model 3a

Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig	
.74	.08	9.15	0	.58	.89	***	
.2	.07	2.69	.01	.05	.34	***	
-1.13	.51	-2.20	.03	-2.14	11	**	
02	.09	-0.17	.86	2	.17		
0	0	1.35	.18	0	0		
.01	.06	0.08	.93	11	.12		
-1.4	1.12	-1.25	.21	-3.61	.82		
1.5	1.84	0.82	.42	-2.14	5.14		
19	.54	-0.35	.73	-1.25	.88		
.02	.87	0.02	.98	-1.71	1.75		
15	.69	-0.22	.83	-1.51	1.21		
.06	1.24	0.05	.96	-2.41	2.53		
	40.566 SD dependent var			ar 19	19.169		
	1356 F-test				2750.879		
	Coef. .74 .2 -1.13 02 0 .01 -1.4 1.5 19 .02 15 .06	Coef.         St.Err.           .74         .08           .2         .07           -1.13         .51          02         .09           0         0           .01         .06           -1.4         1.12           1.5         1.84          19         .54           .02         .87          15         .69           .06         1.24	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 64 Lag limit (years): 5 AB test (AR2) (p-value): 0.558 Hansen test (p-value): 0.191 Difference-in-Hansen test (p-value): 0.787
#### Model 4a

ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ghes_gdp	.95	.05	20.53	0	.86	1.04	***
L. odaplus_health_pub_gdp	.07	.09	0.81	.42	11	.25	
gdp_cap_c	0	0	-0.26	.79	0	0	
imr	0	0	-1.16	.25	01	0	
gov_eff	.05	.08	0.65	.51	1	.2	
corr_control	.06	.09	0.67	.51	12	.25	
battle_dum	.03	.04	0.83	.41	05	.12	
col_indfrom_uk	0	.05	-0.01	.99	1	.1	
col_indfrom_fra	.02	.05	0.44	.66	08	.13	
col_indfrom_spa	.09	.08	1.11	.27	07	.25	
Mean dependent var		2.	210 SD	dependent v	ar 1	.420	
Number of obs		1	460 F-te	st	1705	.671	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 73 Lag limit (years): 7 AB test (AR2) (p-value): 0.085 Hansen test (p-value): 0.173 Difference-in-Hansen test (p-value): 0.602

#### 10.4.4.6. Aggregate ODA-variables

#### Year dummies included in all models but not shown for brevity.

#### Model 1

oop_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. oop_che	.7	.05	14.15	0	.6	.8	***
L. ln_odaplus_health_gdp	73	.39	-1.89	.06	-1.5	.04	*
L. ln_odaplus_nonhealth_gdp	.36	.57	0.63	.53	77	1.49	
L. ln_debt_serv_gdp	.37	.27	1.34	.18	18	.91	
L. ln_debt_stock_gdp	.62	.77	0.80	.42	91	2.14	
imf_cond_ba1tot	02	.01	-1.33	.19	04	.01	
imf_part	.71	.52	1.37	.17	32	1.74	
ln_ghes_gdp	-7.55	1.39	-5.43	0	-10.3	-4.79	***
ln_gdp_cap_c	-1.15	1.81	-0.64	.52	-4.74	2.43	
imr	12	.05	-2.20	.03	22	01	**
gov_eff	1.41	1.5	0.94	.35	-1.57	4.39	
corr_control	86	1.56	-0.55	.58	-3.95	2.23	
battle_dum	.3	.75	0.40	.69	-1.19	1.79	
col_indfrom_uk	-2.2	1.51	-1.46	.15	-5.21	.8	
col_indfrom_fra	73	1.74	-0.42	.68	-4.19	2.73	
col_indfrom_spa	.36	1.68	0.22	.83	-2.97	3.7	
Mean dependent var		39.	952 SD	dependent va	ar 19	.411	
Number of obs		1	461 F-te	st	468	.132	
*** <i>p</i> <.01, ** <i>p</i> <.05, * <i>p</i> <.1							

Countries: 105 Instrument count: 95 Lag limit (years): 5 AB test (AR2) (p-value): 0.998 Hansen test (p-value): 0.257 Difference-in-Hansen test (p-value): 0.168

#### Model 2

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_oop_gdp	.88	.05	16.69	0	.78	.98	***
L. ln_odaplus_health_gdp	03	.01	-2.61	.01	06	01	**
L. ln_odaplus_nonhealth_gdp	0	.02	-0.08	.94	05	.04	
L. ln_debt_serv_gdp	.02	.01	1.17	.25	01	.05	
L. ln_debt_stock_gdp	02	.02	-0.69	.49	06	.03	
imf_cond_ba1tot	0	0	-0.91	.36	0	0	
imf_part	.03	.02	1.52	.13	01	.07	
ln_ghes_gdp	.07	.04	1.74	.08	01	.14	*
ln_gdp_cap_c	17	.07	-2.45	.02	32	03	**
imr	0	0	-0.78	.44	0	0	
gov_eff	.02	.06	0.43	.67	09	.14	
corr_control	.01	.05	0.19	.85	1	.12	
battle_dum	.04	.02	1.56	.12	01	.08	
col_indfrom_uk	04	.05	-0.79	.43	14	.06	
col_indfrom_fra	0	.06	0.06	.95	11	.12	
_col_indfrom_spa	.01	.05	0.16	.88	09	.1	
Mean dependent var		0.	543 SD	dependent v	ar 0	.737	
Number of obs		1	461 F-te	st	235	.018	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 95 Lag limit (years): 5 AB test (AR2) (p-value): 0.626 Hansen test (p-value): 0.266 Difference-in-Hansen test (p-value): 0.360

#### Model 3

Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
.69	.08	8.26	0	.52	.85	***
.2	.07	2.93	0	.06	.33	***
26	.33	-0.78	.44	92	.4	
83	.51	-1.65	.1	-1.84	.17	
23	.27	-0.82	.41	77	.32	
.43	.73	0.59	.56	-1.02	1.88	
0	.01	0.03	.98	02	.03	
75	.63	-1.19	.24	-2	.5	
43	2.43	-0.18	.86	-5.26	4.39	
03	.06	-0.52	.61	15	.09	
63	1.36	-0.47	.64	-3.32	2.06	
2.67	2.21	1.21	.23	-1.72	7.06	
39	.68	-0.57	.57	-1.74	.96	
.25	1.05	0.24	.81	-1.84	2.34	
.65	.95	0.68	.5	-1.24	2.54	
.3	1.7	0.18	.86	-3.07	3.67	
	40.	566 SD	dependent v	ar 19.	.169	
	1	356 F-te	est	1856.	.911	
	Coef. .69 .2 26 83 23 .43 0 75 43 03 63 2.67 39 .25 .65 .3	$\begin{array}{c ccccc} Coef. & St.Err.\\ .69 & .08\\ .2 & .07\\26 & .33\\83 & .51\\23 & .27\\ .43 & .73\\ 0 & .01\\75 & .63\\43 & 2.43\\03 & .06\\63 & 1.36\\ 2.67 & 2.21\\39 & .68\\ .25 & 1.05\\ .65 & .95\\ .3 & 1.7\\ -40.\\ 1\end{array}$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Coef.     St.Err.     t-value     p-value       .69     .08 $8.26$ 0       .2     .07     2.93     0      26     .33     -0.78     .44      83     .51     -1.65     .1      23     .27     -0.82     .41       .43     .73     0.59     .56       0     .01     0.03     .98      75     .63     -1.19     .24      43     2.43     -0.18     .86      03     .06     -0.52     .61      63     1.36     -0.47     .64       2.67     2.21     1.21     .23      39     .68     -0.57     .57       .25     1.05     0.24     .81       .65     .95     0.68     .5       .3     1.7     0.18     .86       40.566     SD dependent value     1356     F-test	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 100 Lag limit (years): 6 AB test (AR2) (p-value): 0.593 Hansen test (p-value): 0.301 Difference-in-Hansen test (p-value): 0.450

#### Model 4

ln_ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_ghes_gdp	.76	.08	9.10	0	.59	.92	***
L. ln_odaplus_health_gdp	02	.02	-0.94	.35	06	.02	
L. ln_odaplus_nonhealth_gdp	.01	.03	0.40	.69	04	.06	
L. ln_debt_serv_gdp	01	.01	-0.48	.63	04	.02	
L. ln_debt_stock_gdp	0	.03	-0.14	.89	06	.05	
imf_cond_ba1tot	0	0	0.22	.83	0	0	
imf_part	01	.02	-0.23	.82	05	.04	
ln_gdp_cap_c	.08	.09	0.95	.34	09	.26	
imr	0	0	-0.96	.34	01	0	
gov_eff	03	.07	-0.42	.67	16	.11	
corr_control	.02	.06	0.26	.79	1	.13	
battle_dum	.03	.03	0.86	.39	04	.09	
col_indfrom_uk	.05	.05	0.93	.36	06	.16	
col_indfrom_fra	05	.05	-0.99	.32	14	.05	
col_indfrom_spa	.04	.06	0.64	.53	08	.15	
Mean dependent var		0.	563 SD o	dependent va	ur 0	.721	
Number of obs		1.	461 F-te	st	222	.786	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105

Instrument count: 89 Lag limit (years): 5 AB test (AR2) (p-value): 0.179 Hansen test (p-value): 0.246 Difference-in-Hansen test (p-value): 0.204

#### Model 1a

oop_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. oop_che	.68	.06	11.56	0	.56	.8	***
L. ln_odaplus_health_gdp	83	.44	-1.87	.06	-1.71	.05	*
L. ln_odaplus_nonhealth_gdp	21	.86	-0.25	.8	-1.92	1.49	
L. ln_debt_serv_gdp	.71	.34	2.08	.04	.03	1.39	**
ln_ghes_gdp	-7.59	1.43	-5.29	0	-10.43	-4.74	***
ln_gdp_cap_c	-2.71	1.94	-1.39	.17	-6.56	1.14	
imr	12	.06	-1.96	.05	24	0	*
gov_eff	1.8	1.37	1.31	.19	92	4.52	
corr_control	81	1.67	-0.48	.63	-4.12	2.51	
battle_dum	.01	.72	0.02	.99	-1.42	1.44	
col_indfrom_uk	-2.16	1.8	-1.20	.23	-5.72	1.41	
col_indfrom_fra	43	1.85	-0.23	.82	-4.09	3.24	
col_indfrom_spa	.03	1.66	0.02	.98	-3.26	3.33	
Mean dependent var		39.	952 SD	dependent v	ar 19	.411	
Number of obs		1	461 F-te	est	354	.401	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 71 Lag limit (years): 5 AB test (AR2) (p-value): 0.965 Hansen test (p-value): 0.507 Difference-in-Hansen test (p-value): 0.313

#### Model 2a

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_oop_gdp	.94	.05	17.38	0	.83	1.05	***
L. ln_odaplus_health_gdp	05	.02	-2.86	.01	08	01	***
L. ln_odaplus_nonhealth_gdp	03	.03	-1.20	.23	08	.02	
ln_ghes_gdp	.08	.04	1.92	.06	0	.16	*
ln_gdp_cap_c	27	.07	-3.93	0	41	13	***
imr	0	0	-2.15	.03	01	0	**
gov_eff	.02	.05	0.37	.71	08	.11	
corr_control	.04	.06	0.74	.46	07	.15	
battle_dum	.05	.03	2.02	.05	0	.11	**
col_indfrom_uk	.03	.06	0.54	.59	08	.14	
col_indfrom_fra	.06	.06	1.06	.29	05	.17	
col_indfrom_spa	01	.07	-0.15	.88	15	.13	
Mean dependent var		0.	543 SD	dependent va	ar 0.	.737	
Number of obs		1	461 F-te	st	179.	.523	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 98 Lag limit (years): 8 AB test (AR2) (p-value): 0.695 Hansen test (p-value): 0.240 Difference-in-Hansen test (p-value): 0.485

#### Model 3a

L. In_	odaplus	_health_	_gdp added to	explore f	ungibility	hypothe:	sis in this	mod	el.
2 <sup>nd</sup> -0	rder lag c	on depe	ndent variable	necessar	ry to con	vincingly	pass AB	test (	(AR2)

ghes_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ghes_che	.66	.09	7.78	0	.49	.83	***
L2. ghes_che	.2	.07	3.14	0	.08	.33	***
L. ln_odaplus_health_gdp	33	.36	-0.91	.36	-1.06	.39	
L. ln_odaplus_nonhealth_gdp	21	.58	-0.36	.72	-1.36	.94	
L. ln_debt_serv_gdp	41	.32	-1.30	.2	-1.05	.22	
ln_gdp_cap_c	1.69	2.33	0.73	.47	-2.92	6.3	
imr	.01	.05	0.28	.78	09	.12	
gov_eff	-1.34	1.28	-1.04	.3	-3.88	1.21	
corr_control	2.75	1.78	1.55	.13	78	6.28	
battle_dum	37	.69	-0.55	.59	-1.74	.99	
col_indfrom_uk	.14	.89	0.16	.88	-1.62	1.9	
col_indfrom_fra	1	1.02	-0.10	.92	-2.12	1.92	
col_indfrom_spa	.86	1.69	0.51	.61	-2.5	4.21	
Mean dependent var		40.	566 SD	dependent va	ar 19	.169	
Number of obs		1	356 F-te	st	1262	.818	

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 105 Instrument count: 97 Lag limit (years): 8 AB test (AR2) (p-value): 0.625 Hansen test (p-value): 0.184 Difference-in-Hansen test (p-value): 0.713

#### Model 4a

ln_ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_ghes_gdp	.7	.08	8.60	0	.54	.86	***
L. ln_odaplus_health_gdp	05	.03	-1.61	.11	11	.01	
ln_gdp_cap_c	.07	.09	0.79	.43	11	.25	
imr	0	0	-0.43	.67	01	0	
gov_eff	05	.07	-0.72	.48	18	.08	
corr_control	.05	.08	0.68	.5	11	.22	
battle_dum	.04	.03	1.10	.27	03	.1	
col_indfrom_uk	.06	.09	0.65	.52	11	.22	
col_indfrom_fra	08	.06	-1.18	.24	2	.05	
col_indfrom_spa	03	.08	-0.37	.71	2	.14	
Mean dependent var		0.	563 SD	dependent va	ar 0	.721	
Number of obs		1-	461 F-te	est	108	.486	

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 105 Instrument count: 73 Lag limit (years): 7 AB test (AR2) (p-value): 0.181 Hansen test (p-value): 0.343 Difference-in-Hansen test (p-value): 0.650

#### 10.4.4.7. With Inverse Mills Ratios

Below is shown the results of the used Probit regression of known factors predicting IMF programme participation following the available literature (16, 66-69). Alternative specifications were tested as well, however this model proved superior when weighing predictive capacity, parsimoniousness and data availability.

imf_part	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. imf_part	2.04	.1	20.59	0	1.84	2.23	***
L. cur_acc_bal_gdp	02	.01	-3.52	0	03	01	***
L. res_imp	07	.02	-3.25	0	12	03	***
L. excrate	0	0	-1.79	.07	0	0	*
L. gdp_growth	02	.01	-1.77	.08	05	0	*
L. ln_gdp_cap_c	18	.06	-2.91	0	3	06	***
L. voice_acc	.1	.09	1.21	.22	06	.27	
Constant	.47	.53	0.90	.37	56	1.51	
Mean dependent var		0.	311 SD	dependent v	ar 0	.463	
Pseudo r-squared		0.	458 Nur	nber of obs	1	1245	
Chi-square		707.	303 Pro	b > chi2	0	.000	
Akaike crit. (AIC)		851.	934 Bay	esian crit. (B	IC) 892	.949	
*** ~ 01 ** ~ 05	k h < 1						

\* p<.01, \*\* p<.05, \* p<.1

Probit regression

imf part is a binary variable capturing IMF programme participation, based on (15), cur acc bal gdp is the current account balance in % of GDP (36, 70); res imp is the total reserves in months of imports (36, 70); excrate is the official exchange rate in local currency units per US\$ (36, 70); gdp growth is GDP growth in % (36); voice acc is Voice and Accountability from the Worldwide Governance Indicators (WGI) dataset (40), measured in standard deviations over a normal distribution. It captures perceptions of to what extent citizens are able to select their government, as well as freedoms of expression, association and media. Other authors (16, 66, 68) have used a Democracy estimate from the Polity5 dataset, however support for this data ended in 2020, and the dataset does not cover 2019 (71). For these reasons, we proxied democracy using the WGI variable, which showed the strongest correlation with IMF programme participation among the six main variables in the dataset.

Similar to Kentikelenis, Stubbs et al. (16, 66), below is shown a classification table illustrating the predictive capacity of the above Probit regression model.

			True						
		+	-	Total					
p	+	311	79	390					
lassifie	-	76	779	855					
Ö	Total	387	858	1245					

This model correctly classified 87.6% of country-year observations:

 $\frac{true \ positives + true \ negatives}{total \ observations} * 100\% = \frac{311 + 779}{1245} * 100\% = 87.6\%$ 

Below is shown regression results from models 1-4 when including Inverse Mills Ratios. Year dummies included in all models but not shown for brevity. Please note: The inclusion of predictive variables for IMF participation lowered the number of countries with data from 105 to 92, and observations from 1429 to 1215. Instrument lag limits were adjusted accordingly to avoid overspecification. Potentially problematic diagnostic test statistics are highlighted in *italic*.

Adding Inverse Mills Ratios to our full models did not cause any substantive changes to our overall results or conclusions. Changes to coefficients and p-values seen were most likely due to the loss of degrees of freedom resulting from the inclusion of new explanatory variables with incomplete data for our Heckman selection model (214 observations and 13 units of observation lost), and the accompanying need to adjust some model lag limits to avoid overspecification. This resulted in the p-value for the relationship between ODA+ for health via civil/private sector per GDP and OOP/CHE being pushed from 0.02 to 0.12, and that for the relationship with OOP/GDP from 0.05 to 0.02. If the two explanatory variables with significant missing data were dropped (current account balance and total reserves), the relationship with OOP/CHE became significant again, with similar predictive capacity of the probit model. The p-value for the relationship between ODA+ for health channelled via the public sector per GDP and GHE-S/GDP went from 0.04 to 0.07, however it regained its significance at 1 deeper lag limit on GMM-style instruments. The negative relationship between IMF participation and GHE-S/CHE in a contemporaneous model became stronger, from -1.13 percentage points to -1.57 percentage points, though still with an insignificant Inverse Mills Ratio. The finding remained sensitive to the lag structure of other variables.

#### Model 1

oop_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig	
L. oop_che	.71	.05	15.53	0	.62	.8	***	
L. ln_odaplus_health_pub_gdp	4	.35	-1.13	.26	-1.09	.3		
L. ln_odaplus_health_civpriv_gdp	3	.19	-1.57	.12	69	.08		
L. ln_odaplus_nonhealth_pub_gdp	19	.38	-0.49	.62	94	.56		
L. ln_odaplus_nonhealth_civpriv_gdp	33	.49	-0.66	.51	-1.31	.65		
L. ln_debt_serv_gdp	.39	.31	1.23	.22	24	1.01		
L. ln_debt_stock_gdp	.49	.75	0.65	.52	-1.01	1.99		
imf_cond_ba1tot	.01	.01	0.38	.7	02	.03		
imf_part	.39	.55	0.70	.48	71	1.48		
ln_ghes_gdp	-6.83	1.71	-3.99	0	-10.24	-3.43	***	
ln_gdp_cap_c	78	1.55	-0.51	.61	-3.86	2.29		
imr	15	.06	-2.44	.02	27	03	**	
gov_eff	33	1.47	-0.22	.82	-3.25	2.59		
corr_control	-3.2	2.22	-1.44	.15	-7.62	1.21		
battle_dum	.47	.82	0.58	.57	-1.15	2.09		
col_indfrom_uk	-1.65	1.35	-1.22	.22	-4.32	1.03		
col_indfrom_fra	-2.08	1.68	-1.24	.22	-5.41	1.25		
col_indfrom_spa	-2.07	1.72	-1.20	.23	-5.49	1.35		
Inverse Mills Ratio	.06	.26	0.24	.81	45	.57		
Mean dependent var		38.	611 SD	dependent v	ar 19	19.623		
Number of obs		1	215 F-te	est	497	.347		

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 92 Instrument count: 81 Lag limit (years): 3 AB test (AR2) (p-value): .701 Hansen test (p-value): 0.418 Difference-in-Hansen test (p-value): 0.148

Model	2
mouci	_

ln_oop_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig	
L. ln_oop_gdp	.89	.06	13.97	0	.77	1.02	***	
L. ln_odaplus_health_pub_gdp	01	.01	-0.99	.32	03	.01		
L. ln_odaplus_health_civpriv_gdp	02	.01	-2.32	.02	04	0	**	
L. ln_odaplus_nonhealth_pub_gdp	01	.02	-0.47	.64	04	.03		
L. ln_odaplus_nonhealth_civpriv_gdp	03	.02	-1.24	.22	08	.02		
L. ln_debt_serv_gdp	.03	.02	1.60	.11	01	.08		
L. ln_debt_stock_gdp	04	.03	-1.45	.15	1	.02		
imf_cond_ba1tot	0	0	-0.08	.94	0	0		
imf_part	0	.02	0.11	.92	05	.05		
ln_ghes_gdp	.04	.05	0.75	.46	06	.13		
ln_gdp_cap_c	16	.06	-2.40	.02	28	03	**	
imr	0	0	-0.53	.6	0	0		
gov_eff	03	.05	-0.74	.46	13	.06		
corr_control	.03	.07	0.38	.71	11	.16		
battle_dum	.03	.03	0.93	.35	03	.09		
col_indfrom_uk	02	.04	-0.65	.52	1	.05		
col_indfrom_fra	.02	.06	0.37	.72	1	.14		
col_indfrom_spa	02	.07	-0.26	.8	15	.12		
Inverse Mills Ratio	0	.01	0.03	.97	02	.02		
Mean dependent var		0.	508 SD	dependent v	var 0	0.755		
Number of obs		1	215 F-te	est	108	.790		

\*\*\* p<.01, \*\* p<.05, \* p<.1

Countries: 92 Instrument count: 81 Lag limit (years): 3 AB test (AR2) (p-value): 0.853 Hansen test (p-value): 0.278 Difference-in-Hansen test (p-value): 0.865

#### Model 3

ghes_che	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig	
L. ghes_che	.83	.09	9.38	0	.65	1	***	
L. ln_odaplus_health_pub_gdp	45	.27	-1.69	.1	97	.08	*	
L. ln_odaplus_health_civpriv_gdp	.06	.25	0.22	.82	45	.56		
L. ln_odaplus_nonhealth_pub_gdp	07	.37	-0.20	.84	81	.66		
L. ln_odaplus_nonhealth_civpriv_gdp	15	.39	-0.37	.71	93	.63		
L. ln_debt_serv_gdp	1	.42	-0.23	.82	94	.75		
L. ln_debt_stock_gdp	98	.86	-1.14	.26	-2.68	.73		
imf_cond_ba1tot	.02	.02	0.96	.34	02	.05		
imf_part	-1.42	.74	-1.92	.06	-2.89	.05	*	
ln_gdp_cap_c	37	1.87	-0.20	.84	-4.08	3.35		
imr	05	.06	-0.84	.4	17	.07		
gov_eff	.6	1.98	0.30	.76	-3.33	4.53		
corr_control	2.38	2.49	0.96	.34	-2.57	7.33		
battle_dum	.23	.78	0.30	.77	-1.32	1.79		
col_indfrom_uk	.17	1.56	0.11	.92	-2.92	3.26		
col_indfrom_fra	.21	1.05	0.20	.84	-1.88	2.31		
col_indfrom_spa	.54	1.66	0.33	.74	-2.75	3.84		
Inverse Mills Ratio	05	.32	-0.16	.87	7	.59		
Mean dependent var		42	422 SD	dependent v	var 19	19.049		
Number of obs		1	215 F-te	st	840	.373		
****								

\*\*\* *p*<.01, \*\* *p*<.05, \* *p*<.1

Countries: 92 Instrument count: 77 Lag limit (years): 3 AB test (AR2) (p-value): 0.151 Hansen test (p-value): 0.278 Difference-in-Hansen test (p-value): 0.354

#### Model 4

ln_ghes_gdp	Coef.	St.Err.	t-value	p-value	[95% Conf	Interval]	Sig
L. ln_ghes_gdp	.83	.06	13.16	0	.7	.95	***
L. ln_odaplus_health_pub_gdp	02	.01	-1.82	.07	05	0	*
L. ln_odaplus_health_civpriv_gdp	.01	.01	0.66	.51	01	.02	
L. ln_odaplus_nonhealth_pub_gdp	.01	.02	0.48	.63	03	.04	
L. ln_odaplus_nonhealth_civpriv_gdp	01	.02	-0.58	.56	05	.03	
L. ln_debt_serv_gdp	0	.02	0.26	.8	03	.04	
L. ln_debt_stock_gdp	05	.03	-1.54	.13	12	.02	
imf_cond_ba1tot	0	0	0.02	.99	0	0	
imf_part	01	.03	-0.54	.59	07	.04	
ln_gdp_cap_c	06	.07	-0.78	.44	2	.09	
imr	0	0	-1.00	.32	01	0	
gov_eff	.04	.07	0.56	.58	1	.18	
corr_control	.08	.06	1.34	.18	04	.2	
battle_dum	.04	.03	1.29	.2	02	.11	
col_indfrom_uk	01	.06	-0.11	.91	12	.11	
col_indfrom_fra	02	.05	-0.43	.67	11	.07	
col_indfrom_spa	.04	.05	0.79	.43	05	.13	
Inverse Mills Ratio	01	.01	-0.68	.5	03	.02	

Mean dependent var	0.638	SD dependent var	0.721
Number of obs	1215	F-test	213.828
*** p<.01, ** p<.05, * p<.1			

Countries: 92 Instrument count: 77 Lag limit (years): 3 AB test (AR2) (p-value): 0.271 Hansen test (p-value): 0.426 Difference-in-Hansen test (p-value): 0.312

### 10.4.5. Metadata for variables in dataset

Variable name	Description	Unit	Туре	Original variable name in database	WB opendata code	Original source	URL	Comment
General								
countryname	Country name; Recipient country of development assistance; debtor country.	-	Text	-	-	-	-	
iso3c	3-letter isocode of country	-	Text	Country Code	-	Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on	Obtained with "wbopendata" command in Stata
iso3n	3-digit isocode of country	-	Numerical	-	-	Raciborski, R. (2008). "kountry: A Stata utility for merging cross- country data from multiple sources," The Stata Journal, 8(3), 390-400.	<u>https://databank.worldb</u> <u>ank.org/data/source/wo</u> <u>rld-development-</u> <u>indicators/preview/on</u>	Obtained with "kountry" command in STATA
wb_regionco de	3-letter code of World Bank Region	-	Text	region	-	Reference: Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on	Obtained with "wbopendata" command in Stata
wb_region	World Bank region name	-	Text	regionname	-	Reference: Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on	Obtained with "wbopendata" command in Stata
un_reg_det	United Nations region name (detailed)	-	Text	-	-	Raciborski, R. (2008). "kountry: A Stata utility for merging cross- country data from multiple sources," The Stata Journal, 8(3), 390-400; United Nations Statistics Division, Methodology, Standard country or area codes for statistical use (M49)	<u>https://unstats.un.org/u</u> nsd/methodology/m49/	Obtained with "kountry" command in STATA

incomelevel	World Bank income group (abbreviation)	-	Text	incomelevel	-	Reference: Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on	Obtained with "wbopendata" command in Stata
incomeleveln ame	World Bank income group (full name)	-	Text	incomelevelna me	-	Reference: Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on	Obtained with "wbopendata" command in Stata
lendingtype	World Bank lending group code	-	Text	lendingtype	-	Reference: Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	<u>https://databank.worldb</u> <u>ank.org/data/source/wo</u> <u>rld-development-</u> <u>indicators/preview/on</u>	Obtained with "wbopendata" command in Stata
lendingtypen ame	World Bank lending group name	-	Text	lendingtypena me	-	Reference: Azevedo, J.P. (2011) "wbopendata: Stata module to access World Bank databases," Statistical Software Components S457234, Boston College Department of Economics. http://ideas.repec.org/c/boc/boco de/s457234.html	<u>https://databank.worldb</u> <u>ank.org/data/source/wo</u> <u>rld-development-</u> jndicators/preview/on	Obtained with "wbopendata" command in Stata
year	Calendar year of development assistance disbursed, debt repayment made, health expenditures made, etc.	Number	Integer	-	-	-	-	
Health financing								
ghes_che	"Domestic General Government Health Expenditure (GGHE-D) as % Current Health Expenditure (CHE)"	%	continuous	"Domestic General Government Health Expenditure (GGHE-D) as % Current Health Expenditure (CHE)"	SH.XPD. GHED.C H.ZS	WHO Global Health Expenditure Database	<u>http://apps.who.int/nha/</u> database/Select/Indicat <u>ors/en</u>	

ghes_cap	"Domestic General Government Health Expenditure (GGHE-D) per Capita in US\$"	current US\$	continuous	"Domestic General Government Health Expenditure (GGHE-D) per Capita in US\$"	SH.XPD. GHED.P C.CD	WHO Global Health Expenditure Database	<u>http://apps.who.int/nha/</u> database/Select/Indicat <u>ors/en</u>	
ghes_gdp	"Domestic General Government Health Expenditure (GGHE-D) as % Gross Domestic Product (GDP)"	%	continuous	"Domestic General Government Health Expenditure (GGHE-D) as % Gross Domestic Product (GDP)"	SH.XPD. CHEX.G D.ZS	WHO Global Health Expenditure Database	<u>http://apps.who.int/nha/</u> database/Select/Indicat ors/en	
oop_che	"Out-of-pocket (OOPS) as % of Current Health Expenditure (CHE)"	%	continuous	"Out-of-pocket (OOPS) as % of Current Health Expenditure (CHE)"	SH.XPD. OOPC.C H.ZS	WHO Global Health Expenditure Database	<u>http://apps.who.int/nha/</u> database/Select/Indicat <u>ors/en</u>	
oop_cap	"Out-of-Pocket Expenditure (OOPS) per Capita in US\$"	current US\$	continuous	"Out-of-Pocket Expenditure (OOPS) per Capita in US\$"	SH.XPD. OOPC.P C.CD	WHO Global Health Expenditure Database	http://apps.who.int/nha/ database/Select/Indicat ors/en	
Development assistance								
odaplus_heal th	Official Development Assistance (ODA) and Bill and Melinda Gates Foundation (BMGF) funding for health	current US\$	continuous	-	-	Organisation for Economic Co- operation and Development (OECD) Creditor Reporting System (CRS)	https://stats.oecd.org/	For sectoral categories 120: "Health, Total" and 130: "Population Policies/Programmes & Reproductive Health, Total" - negative debt relief, administrative expenses and other in-donor country expenses
odaplus_heal th_pub	Official Development Assistance (ODA) and Bill and Melinda Gates Foundation (BMGF) funding for health channeled via the public sector	current US\$	continuous	-	-	Organisation for Economic Co- operation and Development (OECD) Creditor Reporting System (CRS)	https://stats.oecd.org/	For sectoral categories 120: "Health, Total" and 130: "Population Policies/Programmes & Reproductive Health, Total" - negative debt relief, administrative expenses and other in-donor country expenses.
odaplus_heal th_civpriv	Official Development Assistance (ODA) and Bill and Melinda Gates Foundation (BMGF) funding for health channeled via NGO's & civil socity, and private sector institutions	current US\$	continuous	-	-	Organisation for Economic Co- operation and Development (OECD) Creditor Reporting System (CRS)	https://stats.oecd.org/	For sectoral categories 120: "Health, Total" and 130: "Population Policies/Programmes & Reproductive Health, Total" -

								negative debt relief, administrative expenses and other in-donor country expenses.
odaplus_non health	Official Development Assistance (ODA) and Bill and Melinda Gates Foundation (BMGF) funding not for health	current US\$	continuous	-	-	Organisation for Economic Co- operation and Development (OECD) Creditor Reporting System (CRS)	https://stats.oecd.org/	For all other categories - negative debt relief, administrative expenses and other in-donor country expenses
odaplus_non health_pub	Official Development Assistance (ODA) and Bill and Melinda Gates Foundation (BMGF) funding not for health channelled via public sector	current US\$	continuous	-	-	Organisation for Economic Co- operation and Development (OECD) Creditor Reporting System (CRS)	https://stats.oecd.org/	For all other categories - negative debt relief, administrative expenses and other in-donor country expenses
odaplus_non health_civpri v	Official Development Assistance (ODA) and Bill and Melinda Gates Foundation (BMGF) financial assistance not for health channelled via NGO's & civil society and private sector institutions	current US\$	continuous	-	-	Organisation for Economic Co- operation and Development (OECD) Creditor Reporting System (CRS)	https://stats.oecd.org/	For all other categories - negative debt relief, administrative expenses and other in-donor country expenses
Debt								
debt_stock_e ppg	"External debt stocks, public and publicly guaranteed (PPG) (DOD, current US\$)"	current US\$	continuous	"External debt stocks, public and publicly guaranteed (PPG) (DOD, current US\$)"	DT.DOD. DPPG.C D	World Bank Databank: International Debt Statistics	https://databank.worldb ank.org/source/internati onal-debt-statistics	
debt_serv_ep pg	"Debt service on external debt, public and publicly guaranteed (PPG) (TDS, current US\$)"	current US\$	continuous	"Debt service on external debt, public and publicly guaranteed (PPG) (TDS, current US\$)"	DT.TDS. DPPG.C D	World Bank Databank: International Debt Statistics	<u>https://databank.worldb</u> <u>ank.org/source/internati</u> <u>onal-debt-statistics</u>	
IMF conditionaliti es								
imf_cond_ba 1tot	"The main burden of adjustment indicator (BA1) measures the total number of conditions applicable in a given policy area for a given year in each country." (Total is for all policy areas combined. Includes both binding and non-binding conditions)	number	integer	Burden of Adjustment Indicator 1	-	IMF Monitor	https://imfmonitor.org/c onditionality/	
imf_cond_ba 2tot	"An alternative burden of adjustment indicator (BA2) considers only binding conditions. The IMF attaches greater importance to their implementation and requires waivers if they are missed in order for loan disbursement to occur, thereby carrying greater coercive power than non-binding conditions." (Total is for all policy areas combined).	number	integer	Burden of Adjustment Indicator 2	-	IMF Monitor	<u>https://imfmonitor.org/c</u> onditionality/	

imf_cond_ba 3tot	"Another burden of adjustment indicator (BA3) reflects the greater coercive power of binding conditions while also incorporating non-binding conditions. In this indicator, conditions are assigned a weighting according to the importance that the IMF attaches to their implementation. Binding conditions are assigned a weight of 2, while non-binding conditions are assigned a weight of 1." (Total is for all policy areas combined).	number	integer	Burden of Adjustment Indicator 3	-	IMF Monitor	<u>https://imfmonitor.org/c</u> onditionality/
Covariates							
рор	"Total population is based on the de facto definition of population, which counts all residents regardless of legal status or citizenship. The values shown are midyear estimates."	number	continuous	"Population, total"	SP.POP. TOTL	World Bank Databank: World Development Indicators	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on
gdp	"GDP at purchaser's prices is the sum of gross value added by all resident producers in the economy plus any product taxes and minus any subsidies not included in the value of the products. It is calculated without making deductions for depreciation of fabricated assets or for depletion and degradation of natural resources. Data are in current U.S. dollars. Dollar figures for GDP are converted from domestic currencies using single year official exchange rates. For a few countries where the official exchange rate does not reflect the rate effectively applied to actual foreign exchange transactions, an alternative conversion factor is used."	current US\$	continuous	"GDP (current US\$)"	NY.GDP. MKTP.CD	World Bank Databank: World Development Indicators	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on
gdp_cap	"GDP per capita is gross domestic product divided by midyear population. GDP is the sum of gross value added by all resident producers in the economy plus any product taxes and minus any subsidies not included in the value of the products. It is calculated without making deductions for depreciation of fabricated assets or for depletion and degradation of natural resources. Data are in current U.S. dollars."	current US\$	continuous	"GDP per capita (current US\$)"	NY.GDP. PCAP.CD	World Bank Databank: World Development Indicators	https://databank.worldb ank.org/data/source/wo rld-development- indicators/preview/on
mmr_me	"Maternal mortality ratio is the number of women who die from pregnancy-related causes while pregnant or within 42 days of pregnancy termination per 100,000 live births. The data are estimated with a regression model using information on the proportion of maternal deaths among non-AIDS deaths in women ages 15-49, fertility, birth attendants, and GDP measured using purchasing power parities (PPPs)."	# per 100,000 live births	continuous	"Maternal mortality ratio (modelled estimate, per 100,000 live births)"	SH.STA. MMRT	WHO, UNICEF, UNFPA, World Bank Group, and UNDESA/Population Division. Trends in Maternal Mortality 2000 to 2020. Geneva, World Health Organization, 2023	<u>https://www.who.int/pub</u> lications/i/item/9789240 068759
imr	"Infant mortality rate is the number of infants dying before reaching one year of age, per 1,000 live births in a given year."	# per 1000 live births	continuous	Mortality rate, infant (per 1,000 live births)	SP.DYN.I MRT.IN	Estimates developed by the UN Inter-agency Group for Child Mortality Estimation ( UNICEF,	https://childmortality.org

					WHO, World Bank, UN DESA Population Division)		
"Under-five mortality rate is the probability per 1,000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."	# per 1000 live births	continuous	Mortality rate, under-5 (per 1,000 live births)	SH.DYN. MORT	Estimates developed by the UN Inter-agency Group for Child Mortality Estimation ( UNICEF, WHO, World Bank, UN DESA Population Division )	https://childmortality.org	
"Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."	Number (standard deviations )	continuous	"Government Effectiveness: Estimate"	GE.EST	Worldwide Governance Indicators (WGI). Source: "Daniel Kaufmann and Aart Kraay (2023). Worldwide Governance Indicators, 2023 Update"	<u>https://www.govindicato</u> <u>rs.org/</u>	
"Control of Corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "capture" of the state by elites and private interests. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."	Number (standard deviations )	continuous	"Control of Corruption: Estimate"	CC.EST	Worldwide Governance Indicators (WGI). Source: "Daniel Kaufmann and Aart Kraay (2023). Worldwide Governance Indicators, 2023 Update"	<u>https://www.govindicato</u> <u>rs.org/</u>	
"Battle-related deaths are deaths in battle-related conflicts between warring parties in the conflict dyad (two conflict units that are parties to a conflict). Typically, battle-related deaths occur in warfare involving the armed forces of the warring parties. This includes traditional battlefield fighting, guerrilla activities, and all kinds of bombardments of military units, cities, and villages, etc. The targets are usually the military itself and its installations or state institutions and state representatives, but there is often substantial collateral damage in the form of civilians being killed in crossfire, in indiscriminate bombings, etc. All deathsmilitary as well as civilianincurred in such situations, are counted as battle-related deaths."	Number	Integer	"Battle-related deaths (number of people)"	VC.BTL.D ETH	Uppsala Conflict Data Program (UCDP) / Peace Research Institute Oslo (PRIO) Battle- Related Deaths Dataset version 23.1: - Davies, Shawn, Therese Pettersson & Magnus Öberg (2023). Organized violence 1989-2022 and the return of conflicts between states?. Journal of Peace Research 60(4).	<u>https://ucdp.uu.se/down</u> loads/	
battledeaths_orig with 0's imputed for country- years without data, as the Battle-Related Deaths dataset only includes country-years with deaths	Number	Integer	-	-	Uppsala Conflict Data Program (UCDP) / Peace Research Institute Oslo (PRIO) Battle- Related Deaths Dataset version 23.1: - Davies, Shawn, Therese Pettersson & Magnus Öberg (2023). Organized violence 1989-2022 and the return of conflicts between states?.	<u>https://ucdp.uu.se/down</u> loads/	Generated variable
	"Under-five mortality rate is the probability per 1,000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year." "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5." "Control of Corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "capture" of the state by elites and private interests. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5." "Battle-related deaths are deaths in battle-related conflicts between warring parties in the conflict dyad (two conflict units that are parties to a conflict). Typically, battle-related deaths occur in warfare involving the armed forces of the warring parties. This includes traditional battlefield fighting, guerrilla activities, and all kinds of bombardments of military units, cities, and villages, etc. The targets are usually the military itself and its installations or state institutions and state representatives, but there is often substantial collateral damage in the form of civilians being killed in crossfire, in indiscriminate bombings, etc. All deathsmilitary as well as civilianincurred in such situations, are counted as battle-related deaths."	"Under-five mortality rate is the probability per 1,000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."   # per 1000 live biths     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )     "Control of Corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "capture" of the state by elites and private interests. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )     "Battle-related deaths are deaths in battle-related conflicts between warring parties in the conflict dyad (two conflict units that are parties to a conflict). Typically, battle-related deaths occur in warfare involving the armed forces of the warring parties. This includes traditional battlefield fighting, guerrilla activities, and all kinds of bombardments of military units, cities, and villages, etc. The targets are usually the military itself and its installations or state institutions and state representatives, but there is offen substantial collateral damage in the form of civilians being killed in crossfire, in indiscriminate bombings, etc. All deaths-military as well as civilianincurred in such situations, are counted as battle-related deaths."	"Under-five mortality rate is the probability per 1,000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."   # per 1000 live births   continuous     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   continuous     "Control of Corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "capture" of the state by elites and private interests. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   continuous     "Battle-related deaths are deaths in battle-related conflict. Typically, battle-related deaths core on the warfare involving the armed forces of the warning parties. This includes traditional battlefield fighting, guerrilla activities, and all kinds of bombardments of military units, cities, and villages, etc. The targets are usually the military listelf and its installations or state institutions and state representatives, but there is often substantial collateral damage in the form of civilians being killed in crossfire, in indiscriminate bombings, etc. All deathes_miltary as well as civilians-incurred in such situations, are counted as battle-related deaths."   Number </th <th>"Under-five mortality rate is the probability per 1,000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."   # per 1000 live births   continuous   Mortality rate, under-5 (per 1,000 live births)     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   continuous   "Government Effectiveness: Estimate"     "Control of Corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "Capture" of the state by ellets and private interests. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately-2.5 to 2.5."   Number (standard deviations )   "Control of Corruption: Estimate"     "Battle-related deaths are deaths in battle-related conflict voling the armed forces of the warning parties. This includes traditional battlefield fighting, guerrilia activities, and all kinds of bomardments of military units, clies, and willages, etc. The targets are usually the military itself and its installations or state institutions and state representatives, but there is ofne substantial collateral damage in the form of civilians being killed in crossfire, in indiscriminate bombings, etc. All deathsmilitary as well as civilianincurred in su</th> <th>"Under-five mortality rate is the probability per 1.000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."   # per 1000 live births   continuous   Mortality rate, under-5 (per 1,000 live births)   SH.DYN. MORT     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from adjumptementation, and the credibility of the deviations and implementation, and the credibility of the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard orruption, as well as "capture" of the state by elites and normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard continuous   "Control of Corruption continuous   "Control of Corruption: Estimate"   CC.EST     "Tettle-related deaths are deaths in battle-related conflicts buryounds, cites, and willages, etc. The targets are usually the military itself and state representatives, but there is often substantial collateral damage in the form of civilian-incurred in such situations or state in often substantial collateral damage in the form of civilian-hourred in such situations, are country-years with deaths deaths."   Number   "Battle-related deaths (number of people)"   "Battle-related deaths (number of people)"     battledeaths_orig with 0's imputed for country- years without data, as the Battle-Related Deaths dataset only includes country-years with deaths   Number   Integer   -</th> <th>"Under-five mortality rate is the probability per 1.000 that a newtom baby will die before reaching age five, if subject to age-specific mortality rates in the specified year."   # per 1000 live biths   Mortality rate, under-5 (per 1.000 live biths   SHDYL NUMER   SHDYL MORT     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil gerogene metric commitment in such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   "Government Effectiveness: Estimate"   GE.EST   Workbide Governance indicators (WGI) Source: "Daniel Kaufmann and Aart (aray (2023) Workbide Governance indicators, 2023 update"     "Control of Corruption captures perceptions of the standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   Continuous   "Control of Corruption: geroen the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )     "Battle-related deaths are deaths in battle-related conflict between warning parties in the conflict dydat (two collect and take approximately of standard normal distributions and state representatives, but there is often substantial collaters (Maignes, etc. The advisitions or state institute colloce (PRIO) Battle- Related Deaths_0.03 and the conflict between warning parties. The discorring the area dored to warning parties. The discorring the area parties to a conflict. Typically, battle-related deaths are counted as battle-related conflicts between warning parties in the conflict dydate (two collect and</th> <th>Tunder-five mortality rate is the probability per 1000 that a newtom baby will die before reaching age five, # subject to age-specific mortality rates in the specified year.   mortality rate, under-5 (per bitm)   SH DYN, MORT   Estimates diverged by the UN threagency (Group for Chult Mortality Fare-gency Group for Chult Mortality Earlier agency Group for Chult Mortality and public services, the quality of the cut service and the degree of its independence from political pressures, the quality of public services, the quality of public services in the cut publices. Estimate gives the courty's score on the government is communent to such polices. Estimate gives the courty is score on the government services and the degree of its independence from of compution, services is concernities. 1)   Number (Source intruces 1)   Gerefit (Source intruces)   Worldwide Governance inflications (WG). Source: 70 anticle Kautinan and Aat Compution: 1.5.1</th>	"Under-five mortality rate is the probability per 1,000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."   # per 1000 live births   continuous   Mortality rate, under-5 (per 1,000 live births)     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from political pressures, the quality of policy formulation and implementation, and the credibility of the government's commitment to such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   continuous   "Government Effectiveness: Estimate"     "Control of Corruption captures perceptions of the extent to which public power is exercised for private gain, including both petty and grand forms of corruption, as well as "Capture" of the state by ellets and private interests. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately-2.5 to 2.5."   Number (standard deviations )   "Control of Corruption: Estimate"     "Battle-related deaths are deaths in battle-related conflict voling the armed forces of the warning parties. This includes traditional battlefield fighting, guerrilia activities, and all kinds of bomardments of military units, clies, and willages, etc. The targets are usually the military itself and its installations or state institutions and state representatives, but there is ofne substantial collateral damage in the form of civilians being killed in crossfire, in indiscriminate bombings, etc. All deathsmilitary as well as civilianincurred in su	"Under-five mortality rate is the probability per 1.000 that a newborn baby will die before reaching age five, if subject to age-specific mortality rates of the specified year."   # per 1000 live births   continuous   Mortality rate, under-5 (per 1,000 live births)   SH.DYN. MORT     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil service and the degree of its independence from adjumptementation, and the credibility of the deviations and implementation, and the credibility of the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard orruption, as well as "capture" of the state by elites and normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard continuous   "Control of Corruption continuous   "Control of Corruption: Estimate"   CC.EST     "Tettle-related deaths are deaths in battle-related conflicts buryounds, cites, and willages, etc. The targets are usually the military itself and state representatives, but there is often substantial collateral damage in the form of civilian-incurred in such situations or state in often substantial collateral damage in the form of civilian-hourred in such situations, are country-years with deaths deaths."   Number   "Battle-related deaths (number of people)"   "Battle-related deaths (number of people)"     battledeaths_orig with 0's imputed for country- years without data, as the Battle-Related Deaths dataset only includes country-years with deaths   Number   Integer   -	"Under-five mortality rate is the probability per 1.000 that a newtom baby will die before reaching age five, if subject to age-specific mortality rates in the specified year."   # per 1000 live biths   Mortality rate, under-5 (per 1.000 live biths   SHDYL NUMER   SHDYL MORT     "Government Effectiveness captures perceptions of the quality of public services, the quality of the civil gerogene metric commitment in such policies. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   "Government Effectiveness: Estimate"   GE.EST   Workbide Governance indicators (WGI) Source: "Daniel Kaufmann and Aart (aray (2023) Workbide Governance indicators, 2023 update"     "Control of Corruption captures perceptions of the standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )   Continuous   "Control of Corruption: geroen the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."   Number (standard deviations )     "Battle-related deaths are deaths in battle-related conflict between warning parties in the conflict dydat (two collect and take approximately of standard normal distributions and state representatives, but there is often substantial collaters (Maignes, etc. The advisitions or state institute colloce (PRIO) Battle- Related Deaths_0.03 and the conflict between warning parties. The discorring the area dored to warning parties. The discorring the area parties to a conflict. Typically, battle-related deaths are counted as battle-related conflicts between warning parties in the conflict dydate (two collect and	Tunder-five mortality rate is the probability per 1000 that a newtom baby will die before reaching age five, # subject to age-specific mortality rates in the specified year.   mortality rate, under-5 (per bitm)   SH DYN, MORT   Estimates diverged by the UN threagency (Group for Chult Mortality Fare-gency Group for Chult Mortality Earlier agency Group for Chult Mortality and public services, the quality of the cut service and the degree of its independence from political pressures, the quality of public services, the quality of public services in the cut publices. Estimate gives the courty's score on the government is communent to such polices. Estimate gives the courty is score on the government services and the degree of its independence from of compution, services is concernities. 1)   Number (Source intruces 1)   Gerefit (Source intruces)   Worldwide Governance inflications (WG). Source: 70 anticle Kautinan and Aat Compution: 1.5.1

						Journal of Peace Research 60(4).		
battle_dum	1 if number of battle-related deaths is higher than 1 per 1 million population; 0 if less (generated from battledeaths and pop-variables)	Number	Binary	-	-	Uppsala Conflict Data Program (UCDP) / Peace Research Institute Oslo (PRIO) Battle- Related Deaths Dataset version 23.1: - Davies, Shawn, Therese Pettersson & Magnus Öberg (2023). Organized violence 1989-2022 and the return of conflicts between states?. Journal of Peace Research 60(4).	<u>https://ucdp.uu.se/down</u> loads/	Generated variable
col_indfrom	"The COW country code for the state, empire, or other entity from which this state gained independence. This variable is coded as missing where the state did not gain independence from a COW system member (e.g., by unifying local units into a new state)."	Correlate s Of War (COW)- country code	Numerical	IndFrom	-	Paul R. Hensel (2018). "ICOW Colonial History Data Set, version 1.1."	http://www.paulhensel. org/icowcol.html	
col_indfrom_ uk	Independence was gained from United Kingdom. 1= true, 0= false.	Number	Binary	-		Paul R. Hensel (2018). "ICOW Colonial History Data Set, version 1.1."	http://www.paulhensel. org/icowcol.html	Generated variable
col_indfrom_ fra	Independence was gained from France. 1= true, 0= false.	Number	Binary	-		Paul R. Hensel (2018). "ICOW Colonial History Data Set, version 1.1."	http://www.paulhensel. org/icowcol.html	Generated variable
col_indfrom_ spa	Independence was gained from Spain. 1= true, 0= false.	Number	Binary	-		Paul R. Hensel (2018). "ICOW Colonial History Data Set, version 1.1."	<u>http://www.paulhensel.</u> org/icowcol.html	Generated variable
срі	"Consumer price index reflects changes in the cost to the average consumer of acquiring a basket of goods and services that may be fixed or changed at specified intervals, such as yearly. The Laspeyres formula is generally used. Data are period averages." 2010 = 100	Number	Continuous	"Consumer price index (2010 = 100)"	FP.CPI.T OTL	International Monetary Fund (IMF)	https://data.imf.org/?sk =4ffb52b2-3653-409a- b471-d47b46d904b5	
cpi_us_2020	Same as above with base-year recalculated to 2020.	Number	Continuous	"Consumer price index" (2020 = 100)	-	International Monetary Fund (IMF)	https://data.imf.org/?sk =4ffb52b2-3653-409a- b471-d47b46d904b5	Generated variable
gdpdefl_us_2 020	"The GDP implicit deflator is the ratio of GDP in current local currency to GDP in constant local currency. The base year varies by country." 2020 = 100	Number	Continuous	GDP deflator	NY.GDP. DEFL.ZS	World Bank	https://databank.worldb ank.org/metadatagloss ary/world-development- indicators/series/NY.G DP.DEFL.ZS	Generated variable (recalculated from base year 2015)
cur_acc_bal_ gdp	"Current account balance is the sum of net exports of goods and services, net primary income, and net secondary income."	%	continuous	Current account balance (% of GDP)	BN.CAB. XOKA.G D.ZS	International Monetary Fund, Balance of Payments Statistics Yearbook and data files, and	https://data.imf.org/?sk =4c514d48-b6ba-49ed- 8ab9-	

						World Bank and OECD GDP estimates.	<u>52b0c1a0179b&amp;sid=13</u> 90030341854
							https://data.worldbank. org/indicator/BN.CAB.X OKA.GD.ZS
res_imp	"Total reserves comprise holdings of monetary gold, special drawing rights, reserves of IMF members held by the IMF, and holdings of foreign exchange under the control of monetary authorities. The gold component of these reserves is valued at year-end (December 31) London prices. This item shows reserves expressed in terms of the number of months of imports of goods and services they could pay for [Reserves/(Imports/12)]."	months	continuous	Total reserves in months of imports	FI.RES.T OTL.MO	International Monetary Fund, International Financial Statistics and data files.	<u>https://data.imf.org/?sk</u> =4c514d48-b6ba-49ed- <u>8ab9-</u> 52b0c1a0179b&sid=13 <u>90030341854</u> <u>https://data.worldbank.</u> <u>org/indicator/FI.RES.T</u> <u>OTL.MO</u>
excrate	"Official exchange rate refers to the exchange rate determined by national authorities or to the rate determined in the legally sanctioned exchange market. It is calculated as an annual average based on monthly averages (local currency units relative to the U.S. dollar)."	Local currency units per US\$	continuous	Official exchange rate (LCU per US\$, period average)	PA.NUS. FCRF	International Monetary Fund, International Financial Statistics.	https://data.imf.org/?sk =4c514d48-b6ba-49ed- <u>&amp;ab9-</u> <u>52b0c1a0179b&amp;sid=13</u> <u>90030341854</u> https://data.worldbank. org/indicator/FI.RES.T <u>OTL.MO</u>
gdp_growth	"Annual percentage growth rate of GDP at market prices based on constant local currency. Aggregates are based on constant 2015 prices, expressed in U.S. dollars. GDP is the sum of gross value added by all resident producers in the economy plus any product taxes and minus any subsidies not included in the value of the products. It is calculated without making deductions for depreciation of fabricated assets or for depletion and degradation of natural resources."	%	Continuous	GDP growth (annual %)	NY.GDP. MKTP.KD .ZG	World Bank national accounts data.	<u>https://data.worldbank.</u> org/indicator/NY.GDP. <u>MKTP.KD.ZG</u>
voice_acc	"Voice and Accountability captures perceptions of the extent to which a country's citizens are able to participate in selecting their government, as well as freedom of expression, freedom of association, and a free media. Estimate gives the country's score on the aggregate indicator, in units of a standard normal distribution, i.e. ranging from approximately -2.5 to 2.5."	Number (standard deviations )	Continuous	Voice and Accountability: Estimate	VA.EST	Worldwide Governance Indicators (WGI). Source: "Daniel Kaufmann and Aart Kraay (2023). Worldwide Governance Indicators, 2023 Update"	<u>https://www.govindicato</u> <u>rs.org/</u>

### References

1. Arregoces L, Daly F, Pitt C, Hsu J, Martinez-Alvarez M, Greco G, et al. Countdown to 2015: changes in official development assistance to reproductive, maternal, newborn, and child health, and assessment of progress between 2003 and 2012. The Lancet Global Health. 2015;3(7):e410-e21.

2. OECD. Creditor Reporting System (CRS) 2024 [cited 2023 February 16]. Available from: <u>https://stats.oecd.org/Index.aspx?DataSetCode=CRS1</u>.

3. Van de Sijpe N. Is Foreign Aid Fungible? Evidence from the Education and Health Sectors. World Bank Economic Review. 2013;27(2):320-56.

4. Van de Sijpe N. The fungibility of health aid reconsidered. 2013.

5. Lu C, Schneider MT, Gubbins P, Leach-Kemon K, Jamison D, Murray CJ. Public financing of health in developing countries: a cross-national systematic analysis. The Lancet. 2010;375(9723):1375-87.

6. Dieleman JL, Graves CM, Hanlon M. The Fungibility of Health Aid: Reconsidering the Reconsidered. Journal of Development Studies. 2013;49(12):1755-62.

7. Dieleman JL, Hanlon M. Measuring the displacement and replacement of government health expenditure. Health Econ. 2014;23(2):129-40.

8. Patenaude BN. The relationship between development assistance for health and public health financing in 134 countries between 2000 and 2015. Health Policy Plan. 2021;36(4):369-83.

9. Frimpong AO, Amporfu E, Arthur E. Effects of public and external health spending on out-of-pocket payments for healthcare in sub-Saharan Africa. Health Policy Plan. 2022;37(9):1129-37.

10. WHO. Global Health Expenditure Database 2024 [cited 2024 March 28]. Available from: http://apps.who.int/nha/database.

11. Feyzioglu T, Swaroop V, Zhu M. A Panel Data Analysis of the Fungibility of Foreign Aid. The World Bank Economic Review. 1998;12(1):29-58.

12. Ali MEM, Ebaidalla M, Rizk R. Foreign aid and out-of-pocket health expenditure in Sub-Saharan Africa: does institutional development matter. International Journal of Sustainable Economy. 2020;12(3).

13. World Bank. International Debt Statistics 2024 [cited 2024 April 24]. Available from:

https://databank.worldbank.org/source/international-debt-statistics#.

14. Lora E, Olivera M. Public debt and social expenditure: Friends or foes? Emerging Markets Review. 2007;8(4):299-310.

15. IMF Monitor. IMF Conditionality Dataset, 1980-2019 2023 [cited 2023 November 2]. Available from: https://imfmonitor.org/conditionality/.

16. Stubbs T, Kentikelenis A, Stuckler D, McKee M, King L. The impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations. Social science & medicine (1982). 2017;174:220-7.

17. Ooms G, Schrecker T. Expenditure ceilings, multilateral financial institutions, and the health of poor populations. The Lancet. 2005;365(9473):1821-3.

 Center for Global Development. Does the IMF Constrain Health Spending in Poor Countries? 2007 [cited 2024 March 19]. Available from: <u>https://www.cgdev.org/sites/default/files/14103\_file\_IMF\_report.pdf</u>.
Brunswijck G. Unhealthy Conditions. Eurodad; 2018.

20. Stuckler D, Basu S. The International Monetary Fund's effects on global health: before and after the 2008 financial crisis. Int J Health Serv. 2009;39(4):771-81.

21. Stuckler D, Basu S. The Body Economic: Why Austerity Kills: Basic Books; 2013.

22. Abocejo F. The Impact of International Monetary Fund (IMF) Structural Adjustment Policies (SAP) on the Philippines. Recoletos Multidisciplinary Research Journal. 2014;2(1):19-24.

23. Isiani MC, Anthonia Obi-Ani N, Obi-Ani P, Chidume CG, Okoye-Ugwu S. Interrogating the International Monetary Fund (IMF) Policies in Nigeria, 1986–2018. Cogent Arts & Humanities. 2021;8(1).

24. Kingston K. The Impacts of the World Bank and IMF Structural Adjustment Programmes on Africa: The Case Study of Cote D'Ivoire, Senegal, Uganda, and Zimbabwe. Sacha Journal of Policy and Strategic Studies, Vol 1, No 2, pp 110-130, 2011. 2011.

25. Foley EE. Your pocket is what cures you: The Politics of health in Senegal: Rutgers University Press; 2010.

26. Meurs M, Seidelmann L, Koutsoumpa M. How healthy is a 'healthy economy'? Incompatibility between current pathways towards SDG3 and SDG8. Global Health. 2019;15(1):83.

27. Birn AE, Nervi L, Siqueira E. Neoliberalism Redux: The Global Health Policy Agenda and the Politics of Cooptation in Latin America and Beyond. Development and Change. 2016;47(4):734-59.

28. Clements B, Gupta S, Nozaki M. What happens to social spending in IMF-supported programmes? Applied Economics. 2013;45(28):4022-33.

29. Gupta S. Can a causal link be drawn? A commentary on "the impact of IMF conditionality on government health expenditure: A cross-national analysis of 16 West African nations". Social science & medicine (1982). 2017;181:199-201.

30. Martin R, Segura-Ubiergo A. Social Spending in IMF-Supported Programs. International Monetary Fund; 2004.

31. van der Gaag J, Barham T. Health and health expenditures in adjusting and non-adjusting countries. Social Science & Medicine. 1998;46(8):995-1009.

32. Thomas AH. Do Debt-Service Savings and Grants Boost Social Expenditures? : International Monetary Fund; 2006.

 Forster T, Kentikelenis AE, Stubbs TH, King LP. Globalization and health equity: The impact of structural adjustment programs on developing countries. Social science & medicine (1982). 2020;267:112496.
Kentikelenis A, Stubbs T. A Thousand Cuts: Social Protection in the Age of Austerity. Kentikelenis A, Stubbs T, editors: Oxford University Press; 2023. 0 p.

35. Stubbs TH, Kentikelenis AE, King LP. Catalyzing Aid? The IMF and Donor Behavior in Aid Allocation. World Development. 2016;78:511-28.

36. World Bank. World Bank Open Data 2024 [cited 2024 March 28]. Available from: https://data.worldbank.org/.

37. Lee SA, Lim J-Y. Does International Health Aid Follow Recipients' Needs? Extensive and Intensive Margins of Health Aid Allocation. World Development. 2014;64:104-20.

38. Reidpath DD, Allotey P. Infant mortality rate as an indicator of population health. J Epidemiol Community Health. 2003;57(5):344-6.

39. UN Inter-agency Group for Child Mortality Estimation. Child Mortality, Stillbirth, and Causes of Death Estimates 2024 [cited February 9 2024]. Available from: <u>https://childmortality.org/</u>.

40. Kaufmann D, Kraay A. Worldwide Governance Indicators: World Bank; 2023 [cited 2024 March 21]. Available from: <u>https://www.govindicators.org/</u>.

41. Oryema JB, Gyimah-Brempong K, Picone G. The impact of debt relief on under five mortality rate in Sub-Saharan Africa. Glob Public Health. 2017;12(6):795-807.

42. In'airat M. Aid allocation, selectivity, and the quality of governance. Journal of Economics Finance and Administrative Science. 2014;19(36):63-8.

43. Sterck O, Roser M, Ncube M, Thewissen S. Allocation of development assistance for health: is the predominance of national income justified? Health Policy Plan. 2018;33(suppl\_1):i14-i23.

44. Rahim M. Quazi. Effects of Foreign Aid on GDP Growh and Fiscal Behavior: An Econometric Case Study of Bangladesh. The Journal of Developing Areas. 2005;38:95-117.

45. Craig Burnside, David Dollar. Aid, Policies, and Growth: Revisiting the Evidence. World Bank Policy Research Working Paper 3251. 2004.

46. Jung Y, Kim J, Kim K. Whom is economic aid meant for? The push vs. pull determinant factors of official development assistance. International Review of Economics & Finance. 2024;89:173-95.

47. Nanda VP. The "Good Governance" Concept Revisited. The ANNALS of the American Academy of Political and Social Science. 2016;603(1):269-83.

48. IMF. IMF and Good Governance 2024 [cited 2024 February 9]. Available from:

https://www.imf.org/en/About/Factsheets/Sheets/2023/The-IMF-and-Good-Governance.

49. Jajkowicz O, Drobiszová A. The Effect of Corruption on Government Expenditure Allocation in OECD Countries. Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis. 2015;63(4):1251-9.

50. Hashem EA. The Effects of Corruption on Government Expenditures: Arab

Countries Experience. Journal of Economics and Sustainable Development. 2014;5(4).

51. Davies S, Pettersson T, Öberg M. Organized violence 1989–2022, and the return of conflict between states. Journal of Peace Research. 2023;60(4):691-708.

52. World Bank. World Bank Group Assistance to Low-Income Fragile and Conflict-Affected States: An Indepedent Evaluation. 2014:166.

53. Yogo UT, Mallaye D. Health Aid and Health Improvement in Sub Saharan Africa. Munich Personal RePEc Archive. 2012(44938).

54. Collier P, Hoeffler A. Aid, Policy, and Growth in Post-Conflict Societies. World Bank Policy Research Working Paper 2902. October 2002.

55. Bluhm R, Gassebner M, Langlotz S, Schaudt P. Fueling Conflict? (De)Escalation and Bilateral Aid. AidData Working Paper. 2016;31.

56. Mishra P, Newhouse D. Does health aid matter? J Health Econ. 2009;28(4):855-72.

57. Dietrich S. The Politics of Public Health Aid: Why Corrupt Governments Have Incentives to Implement Aid Effectively. World Development. 2011;39(1):55-63.

58. Alesina A, Dollar D. Who Gives Foreign Aid to Whom and Why? Journal of Economic Growth. 2000;5(1):33-63.

59. Matteis AD. Whose poverty really matters when deciding aid volumes? International Journal of Public Policy. 2016;12(1/2).

60. Neumayer E. The Determinants of Aid Allocation by Regional Multilateral Development Banks and United Nations Agencies. International Studies Quarterly. 2003;47(1):101-22.

61. Jones P. History matters: New evidence on the long run impact of colonial rule on institutions. Journal of Comparative Economics. 2013;41(1):181-200.

62. Robin M. Grier. Colonial Legacies and Economic Growth. Public Choice. 1999;98(3/4):317-35.

63. Acemoglu D, Robinson JA, Johnson S. The Colonial Origins of Comparative Development: An Empirical Investigation. American Economic Review 2001;91(5):1369-401.

64. Azevedo MJ. The State of Health System(s) in Africa: Challenges and Opportunities. In: Azevedo MJ, editor. Historical Perspectives on the State of Health and Health Systems in Africa, Volume II: The Modern Era. Cham: Springer International Publishing; 2017. p. 1-73.

65. Hensel PR. ICOW Colonial History Data Set, version 1.1 2023 [cited 2023 November 17]. Available from: <u>http://www.paulhensel.org/icowcol.html</u>.

66. Kentikelenis AE, Stubbs TH, King LP. Structural adjustment and public spending on health: evidence from IMF programs in low-income countries. Social science & medicine (1982). 2015;126:169-76.

67. Clements B, Gupta S, Nozaki M. What Happens to Social Spending in IMF-Supported Programs? 2011 [Available from: <u>http://www.imf.org/external/pubs/ft/sdn/2011/sdn1115.pdf</u>.

68. Nooruddin I, Simmons JW. The Politics of Hard Choices: IMF Programs and Government Spending. International Organization. 2006;60(04).

69. Daoud A, Reinsberg B. Structural adjustment, state capacity and child health: evidence from IMF programmes. Int J Epidemiol. 2019;48(2):445-54.

70. IMF. International Finance Statistics (IFS) 2024 [cited 2024 March 21]. Available from: https://data.imf.org/?sk=4c514d48-b6ba-49ed-8ab9-52b0c1a0179b&sid=1390030341854.

71. Center for Systemic Peace. Polity5 dataset 2020 [cited 2024 March 21]. Available from: https://www.systemicpeace.org/polityproject.html.

## 10.5. Appendix 5: Interview supplementary materials









#### PARTICIPANT INFORMATION SHEET

# Study title: Official Development Assistance, External Public Debt and Equity in Domestic Health Financing in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal

Primary funder: Economic & Social Research Council (ESRC)

**Principal Investigator (PI) :** Dr Frederik Federspiel, MD, MPH, PhD Candidate, Dept. Of Global Health and Development, Faculty of Public Health and Policy, London School of Hygiene and Tropical Medicine.

Phone : 78 193 42 08 Email : frederik.federspiel@lshtm.ac.uk

**Co-PI:** Elhadji Mamadou Mbaye, PhD, Responsable Unité Sciences Sociales et Santé - Institut de Recherche en Santé, de Surveillance Epidémiologique et de Formation (IRESSEF)

Phone : 76 497 72 11 Email : elhadjimamadou.mbaye@iressef.org

#### Background and aims

The composition of domestic financing for health in Senegal is shaped by a broader political and economic context. This research aims to study the role of international donors and creditors in determining this financing in Senegal. We intend to publish the results of this research.

#### Methods

This case study of Senegal consists of interviews with representatives and document reviews at ministries and health policy institutions at the regional and district level, international official donor and creditor institutions, academics and health professionals, and civil society organizations in Senegal.

#### Participation

You have been invited to participate in this study because we believe you may be knowledgeable about some of the dynamics between international donors and creditors and the Senegalese health sector. By participating and sharing your knowledge, you will be able to contribute to our understanding of this topic. By participating in this study, you will not be putting yourself or anyone else at any risk. Participation is completely voluntary. If you agree to participate, you may still withdraw at any time without giving a reason. The interviewer will take notes during the interview. We would like to record the interview and transcribe the interview for our analysis, however, you may ask for the interview not to be recorded. In that case, the interviewer will only take hand-written notes without recording during the interview.

#### Confidentiality

You have the option of being anonymous or to have your identity reported in the publications we will write from the research. If you choose to be anonymous, you may choose from different levels of anonymity: You can choose to only have your name anonymized, to also have your position in the organization you work in anonymized, and to have the organization you work in anonymized as well. You can further ask that we don't quote you at all, not even anonymously, and in that case we will only use the information you provide us to inform our analyses but with no direct citation or reference. Frederik Federspiel (the principal investigator) and a transcriptionist will be the only persons who will have access to the recordings produced from the interview. The transcriptionist will be bound by a confidentiality agreement that will make her/him unable to disclose any information about you to third parties. All raw interview recordings will be destroyed at the end of the project.

Please consider whether you would like to remain anonymous, and if so the degree of anonymity you would prefer.

You will also have the opportunity to decide whether the transcripts themselves, i.e. your words on paper, can be made publicly available in anonymized form in our secure online data repository at the end of the project. If you accept this, we will anonymize the transcript according to your choice in the participant consent form.

Any questions you may have will be answered during the interview. If you have any questions in the meantime, please feel free to contact Frederik using the information below.

#### **Further information**

If you have any questions, please contact:

Frederik Federspiel Unité sciences sociales et santé Institut de Recherche en Santé, de Surveillance Epidémiologique et de Formation, IRESSEF Arrondissement 4, Rue 2 D1 - Pôle Urbain de Diamniadio - BP 7325, Dakar Phones : 221-33 872 2691, 33 872 2588, 33 872 2664 Fax : 221-33 850 2017 Email : <u>frederik.federspiel@lshtm.ac.uk</u>

#### 10.5.2. Ethical consent form







#### ETHICAL CONSENT FORM

# Study title: Official Development Assistance, External Public Debt and Equity in Domestic Health Financing in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal

Primary funder: Economic & Social Research Council (ESRC)

**Principal Investigator (PI):** Dr Frederik Federspiel, MD, MPH, PhD Candidate, Dept. Of Global Health and Development, Faculty of Public Health and Policy, London School of Hygiene and Tropical Medicine.

Phone: 78 193 42 08 Email: frederik.federspiel@lshtm.ac.uk

**Co-PI:** Elhadji Mamadou Mbaye, PhD, Responsable Unité Sciences Sociales et Santé - Institut de Recherche en Santé, de Surveillance Epidémiologique et de Formation (IRESSEF)

Phone: 76 497 72 11 Email: elhadjimamadou.mbaye@iressef.org

Participant identification number:

\_\_\_\_\_ (Participant name)

1. a.  $\hfill\square$  I have read the information sheet for this study and understood the information provided

#### OR

1. b. The study personnel have informed me in a language I understand

#### AND I

- confirm that my choice to participate is entirely voluntary,
- understand that I may withdraw from this study at any time without giving a reason,
- confirm that I have had the opportunity to ask questions about this study and that I am satisfied with the answers provided,
- understand that I authorize the persons described in the information sheet to access the interview recording and transcript (in the case that I authorize recording of the interview),
- have had time to think about whether I would like to participate in this study,
- agree to participate in this study.
- 2.  $\Box$  I agree that this interview can be recorded.

3. Please read the following options carefully and tick **ONE**:

a.  $\Box$  I agree that the contents of this interview can be quoted and that these quotations can be attributed to me

b. I agree that the contents of this interview can be quoted, but I would like my name to be anonymized, although you can mention my organization and my position within it

c. I agree that the contents of this interview can be quoted, but I would like my name and my position within my organization to be anonymized, although you can mention my organization

d. I agree that the contents of this interview can be quoted, but I would like my name, position and organization to be anonymized. However, you may refer to me as "a representative of Ministry X", "a representative of a donor agency", "a representative of a creditor agency" or "a representative of an NGO", as the case may be

(if other, please indicate: .....)

e. I agree that the contents of this interview can be quoted, but I would like my name to be anonymized, as well as any information that can be used to identify me, including my organization and my position within it

f. I do not agree that the contents of my interview can be quoted, but the researchers can use the information from my interview to inform their analyses

4. Do you allow the transcript of your interview to be made available in a secure online data repository? If so, we will follow the level of anonymity you have selected above, for example, if you have selected option "c" above, we will delete any information that allows others to identify your name and your position within your organization.

a.	Yes
۱.	NI-

b.		No
----	--	----

Full name: .....

Date (	(dd/mm/yy):	Time (24h	) :
	······································	······ (_ ···	

Signature: .....

To be completed by the person obtaining the consent:

Name of person obtaining consent:

.....

I confirm that I have explained the study information accurately in

..... and that the participant has understood it to the best of my knowledge. He/she has freely consented to participate.

Name: .....

Date (dd/mm/yy): ..... Time (24h) :....

Signature: .....

#### 10.5.3. Example interview topic guide

#### Example interview guide for donors in Senegal

Study title: Official Development Assistance, External Public Debt and Equity in Domestic Health Financing in Low- and Middle-Income Countries: Panel Data Analysis and Case Study in Senegal

#### **General information**

Country of origin: Place of residence:

#### 1. Icebreaker

- a. What is your role in this institution?
- b. How long have you been working here?
- c. Have you held any other positions here? (Which ones?)

#### 2. Health sector financing

- a. What are the priority areas of [NAME OF ORGANISATION] for your support to the health sector in Senegal? (Why do you give priority to these areas?)
- b. In regard to the topic of health financing, or more broadly health economics, does [NAME OF ORGANISATION] have any specific projects/ programs in this area in Senegal? (Can you tell me about this?)
- c. What about universal health coverage and access to health services? Do you have any programs in this area? (Can you tell me about this?)
- d. Why has [NAME OF ORGANISATION] chosen to support the programs you just mentioned? (What are their qualities of these programs) ?
- e. How does [NAME OF ORGANISATION] want the balance between public and private health financing in Senegal to develop over time: Towards higher dependence on private or public funding? Why?
- f. Who does [NAME OF ORGANISATION] think should contribute the majority of health funding in Senegal? (Government, donors, patients)? Why ?

#### 3. Involvement of international stakeholders in determining the public health budget

- a. In your opinion, does the [NAME OF ORGANISATION] have a say when the overall size and distribution between the different expenditure items are decided for the health budget in Senegal? (Please explain)
- b. In your opinion, which stakeholders outside of the government have the most influence on these decisions? (Donors, creditors, civil society, voters)? (Why do you think so)?
- c. Under your impression, what are their main priorities or objectives in these processes?

- d. Do you see any conflicting priorities between these international actors' and the government's priorities with regard to Senegal's health budget? (Which ones?)
- e. Do you see any mutually contradictory priorities between the different international actors, including yours, with regard to the health budget? (Which ones?)

#### 4. Fungibility

- a. Does the [NAME OF ORGANISATION] have an opinion on how national resources for a sector should be spent when development assistance for that sector is received, i.e. should the government increase its funding, keep at same level, or lower it? Can you tell me about this?
- b. Have you ever seen a decrease in domestic government funding for the health sector after having received development assistance for this sector? Can you tell me about this ?
  - i. If yes: How has this affected the health sector?
  - ii. *If yes*: What happens when the aid is exhausted? Do you normally see domestic funding increased back to the initial level so that health services can continue?
  - iii. *If yes*: Do you think that this reduction in domestic funding, when development assistance is received, affects access to health services? If so, how? (For the poor, in rural areas?)
- c. (If applicable): The effects you mention, when have they mainly occurred?

#### Are they still happening today?

In your experience, does the [NAME OF ORGANISATION] have an influence on how the Senegalese government spends its own money when you provide development assistance for a sector? In what ways?

- d. (If applicable) : What is your opinion on this redistribution of funds? Why ?
- 5. Loans and debt repayments
  - a. In your experience, which sectors have development loans received from international financial institutions (International Monetary Fund (IMF), World Bank, African Development Bank) and bilateral creditors mainly benefited?
  - b. In your experience, have the development loans received from these agents benefited the health sector? (In what way?)
  - c. Do you think that the debt owed to external official creditors and the repayments made on this debt affect the Senegalese health budget?
    - i. *If yes*: In what ways?
    - ii. *If yes*: Do you think it affects the health sector more than other sectors? Why/ why not?
    - iii. *If yes*: And how do you think this may affect access to health services? (For the poor, in rural areas?)

- *If yes*: Do you think this is a growing, stable or diminishing problem? Why?
- 6. IMF / World Bank Programs and Ioan conditionalities
  - a. Today, Senegal is part of IMF and World Bank programmes and is subject to the policy conditionalities necessary to be able to lend from these institutions. Do you think that these programmes with their conditionalities have affected Senegal's health sector?
    - i. If yes: In what ways?
    - ii. If yes: Do you think these conditionalities have affected access to health services? (In what ways?) (For the poor, in rural areas?)
  - b. Are you aware of any conditionalities imposed by IMF/WB programs directly on the health sector?
    - i. If yes: How do you think these conditionalities have affected the health sector?
    - ii. If yes: Do you think these conditionalities have affected access to health services? (How?) ( For the poor, in rural areas?)
  - c. Some countries have experienced national wage ceilings [*explain*] for the health sector as part of these programs. Do you know if this has been the case for Senegal? (Can you tell me about them?)
  - d. What about the privatization of health facilities due to loan conditionalities from the IMF/WB? (Can you tell me about this?)
  - e. What about the introduction of user fees for health services? (Can you tell me about this ?)
    - i. (*If applicable*): How do you think these things have affected access to health services in Senegal? (For the poor, in rural areas?)
    - ii. (If applicable): The effects you mention, when have they mainly occurred?
      - i. Are they still happening today?

#### 6. Last question

a. Is there anyone else you would recommend I interview ?

#### Thank you

# 10.5.4. Interview analysis codebook

Name	Description
Access	Access to health services.
Balance public vs. private	The balance between public and private health services.
Coordination, alignment or conflict (priorities & activities)	Statements about coordination, alignment or conflict between stakeholders, both in terms of their different stated priorities and their actual activities. (This category pertains to all stakeholders).
Creditor financing	Statements about what or with how much money a creditor finances.
Debt	Anything pertaining to debt.
Debt constraining health financing	Debt having - or not having - a constraining effect on domestic health financing.
Debt sustainability	The sustainability of public debt.
Debt, general	Statements about debt that are not captured by the other sub-categories.
Domestic health financing	Financing of health services with internally derived funds, i.e. no external assistance.
Domestic health financing, other	Statements about domestic health financing that do not fit into any of the other categories
Government health financing (schemes)	Financing for health services from the government. When specific schemes are mentioned, these are coded under the sub-nodes.
Couverture Maladie Universelle (CMU)	Specific government health financing schemes providing free care for the following groups: - People over 60 (Plan Sesame) - Children under 5 - School children - Free cesarian sections and dialysis
Govt. health fin. schemes, other	Other government health financing schemes not captured under the above themes
Mutuelles	Specific node for the "Mutuelles", a government-subsidised community health insurance scheme. This also falls under the "Couverture Maladie Universelle (CMU)" but is separately coded due to its significance.
Out-of-pocket payments (OOP), user fees	Out-of-Pocket Payments (OOPs), user fees or other terms for direct contributions to the health system from the patient/caregiver/household at the point of care.
Private health insurance	Private Health Insurance (PHI) schemes, Voluntary Health Care Payment Schemes (VHPS) or other terms for private health care payment schemes without relation to government schemes.
Private sector	Private sector participation in the health sector, both in terms of supply of services (e.g. private clinics) and financial contributions to the health sector (e.g. through taxation or donations).
Donor financing	Statements about what or with how much money a donor finances.
Efficiency, effectiveness	Statements about efficiency or effectiveness, e.g. administrative efficiency, cost-effectiveness, etc.
Equity	
Equity in allocation (access)	Equity in allocation of resources. Statements relating to issues of equity in access to health services are also included under this node, as these often relate to issues of allocative equity.

Name	Description				
Equity in contributions	Equity in terms of "who contributes" financially to health services.				
Equity, general	Statements about equity that are not captured by the other sub-categories.				
Gender equity	Gender equity				
Geographical equity	Statements relating to equity between regions, urban vs. rural or other geographical delineations.				
Fungibility	Fungibility				
Governance, management, administration	Governance, management or administration				
Inadequate health financing	Statements that health financing is inadequate or should increase.				
Inadequate, domestic	Inadequate domestic health financing				
Inadequate, external	Inadequate external health financing				
Inadequate, unspecified	Inadequate health financing, specific source unspecified				
Inclusion, transparency	Inclusion by any stakeholder of other stakeholders in policy or decision-making processes. Transparency of processes in stakeholder organisations.				
Influence	The influence of stakeholders on other stakeholders, primarily on the government. (Aggregate)				
Academic influence	Academic influence on any stakeholder group or any domestic affairs.				
Creditor influence	Creditor influence on domestic affairs.				
CSO influence	CSO influence on any stakeholder group or any domestic affairs.				
Donor influence	Donor influence on domestic affairs.				
Government influence	Government influence on any non-governmental stakeholder group or influence by one ministry/department on another.				
Health System Agent influence	HSA influence on any stakeholder group or any domestic affairs.				
Influence, other	Influence by other stakeholder				
Loan & creditor policy conditionalities	Anything on loan and creditor policy conditionalities. Includes policy requirements in lending and fiscal policy programs, e.g. budget prescriptions (e.g. sectoral budget floors), wage ceilings, decentralisation / privatisation requirements, etc.				
National sovereignty, autonomy, ownership	Statements relating to notions of national sovereignty, autonomy or policy ownership (for Senegal).				
Official Development Assistance	Anything pertaining to Official Development Assistance (ODA), including Development Assistance for Health (DAH).				
Development Assistance for Health	Anything pertaining to Development Assistance for Health (DAH) (both loans and grants).				
Development Assistance for Health, general	Statements about Development Assistance for Health (DAH) that are not captured by the other sub-categories.				
Grants for health	Development assistance grants for health.				
Loans for health	Development assistance loans for the health sector.				
Loans vs. grants	Statements mentioning both development assistance				

Name	Description				
	loans and grants. This can be for the health sector or not (or not specified).				
ODA, general	Statements about ODA that are not captured by the other sub-categories.				
Other	Category for relevant statements that do not fit into any other categories.				
Politically motivated allocation of funds	Statements about funds being allocated due to political motives.				
Priorities, attitudes, values	Statements that explicitly describe stakeholder priorities, attitudes or values, i.e. what they "care" about (not just implicitly).				
Creditor priorities, attitudes, values	Statements that explicitly describe creditor priorities, attitudes or values (not just implicitly).				
Donor priorities, attitudes or values	Statements that explicitly describe donor priorities, attitudes or values (not just implicitly).				
Government priorities, attitudes or values	Statements that explicitly describe government priorities, attitudes or values (not just implicitly).				
Priorities, attitudes or values (other stakeholder)	Priorities, attitudes or values of other stakeholder				
Specific subject areas					
Communicable diseases	Communicable diseases				
HRH	Human Resources for Health (HRH)				
HSS	Health System Strengthening (HSS).				
NCDs	Non-Communicable Diseases (NCDs)				
PPPs	Public-Private Partnerships (PPPs)				
SRMNCAH	Sexual, Reproductive, Maternal, Neonatal, Child and Adolescent Health (SRMNCAH)				
UHC	Universal Health Coverage				
Technical support	Statements about technical support.				

# **10.6.** Appendix 6: Abbreviations and detailed results table for

# Chapter 8

Abbreviation	Full name	Explanation
PPG	Public and Publicly Guaranteed	Public and publicly guaranteed external debt comprises external debt obligations of public debtors, which includes the national government, publicly owned corporations and enterprises, development banks, autonomous public bodies and the like, as well as external debt obligations of private bodies that are guaranteed by a public entity (1).
SAP	Structural Adjustment Program	A set of economic reforms that a country had to implement to be eligible to borrow from the IMF/WB in the 1980's-1990's (2).
ESAF	Enhanced Structural Adjustment Facility	IMF facility established in 1987 that provides loans " <i>on concessional terms</i> " to LICs facing protracted balance of payments problems. This was changed to the PRGF in 1999 (3).
PRGF	Poverty Reduction and Growth Facility	IMF loan facility, established in 1999 as an extension of ESAF. Loans provided to address balance of payments problems and promote growth for "higher living standards and a reduction in poverty" (3).
PRSP	Poverty Reduction Strategy Paper	National 3-year strategy paper prepared by borrower countries, involving stakeholders including IMF/WB, describing macroeconomic, structural and social policies and programs to <i>"promote broad-based growth and reduce poverty"</i> (4).
PSI	Policy Support Instrument	IMF tool for providing economic advice without a borrowing arrangement (5).
ESF	Exogenous Shock Facility	IMF facility that provides financing " <i>at more appropriate terms</i> " to LICs facing sudden, exogenous shocks, without an existing PRGF (6).
PCI	Policy Coordination Instrument	Non-financial IMF tool that enables dialogue, monitoring and endorsement by the IMF, allowing countries to " <i>signal commitment to</i> <i>reforms and catalyze financing from other sources</i> " (7).
RCF	Rapid Credit Facility	IMF facility that provides "rapid concessional financial assistance with limited conditionality to low-income countries (LICs) facing an urgent balance of payments need" (8).
RFI	Rapid Financing Instrument	IMF instrument that "provides rapid financial assistance, which is available to all member countries facing an urgent balance of payments need" (9).
SBA	Stand-By Arrangement	IMF decision assuring a country that it will be able borrow a given amount during a given period, contingent upon observing the terms in the supporting arrangement (3).
SCF	Standby Credit Facility	IMF facility that "provides financial assistance to low-income countries (LICs) with short-term balance of payments needs" (10).
ECF	Extended Credit Facility	IMF facility that "provides medium-term financial assistance to low- income countries (LICs) with protracted balance of payments problems" under the Poverty Reduction and Growth Trust (PRGT) (11).
EFF	Extended Fund Facility	IMF facility that "provides financial assistance to countries facing serious medium-term balance of payments problems because of structural weaknesses that require time to address" (12).
RSF	Resilience and Sustainability Facility	IMF facility that "provides affordable long-term financing to countries undertaking reforms to reduce risks to prospective balance of payments stability, including those related to climate change and pandemic preparedness" (13).
SDR	Special Drawing Rights	The currency or financial unit used by the IMF, originally equivalent to one US dollar, but closer to 1.4-1.5 US dollars in the past two decades (3, 14).

Table 10.3: Explanation of IMF/WB terms and abbreviations used in this paper.

Time period	Organisa tion	Instrument/ Arrangeme nt/ Paper	Loans disbursed (SDR if not otherwise specified)*	Loan duration	Loan purpose	Main policy content	Policies, targets and objectives (health financing sources)	Binding Ioan conditionaliti es (health financing sources)	Non-binding loan conditionaliti es (health financing sources)
2023 (15)	WB	Second Equitable and Resilient Recovery in Senegal Developme nt Policy Financing (15)	€276 million (15)	30 years	Support development programme =>	Increase resilience and productivity of households; strengthen productive resources and management of fiscal accounts	Improve access to health services for poorest and strengthen financial soundness of UHC (15)	-	-
2023- (16)	IMF	Extended Fund Facility (EFF); Extended Credit Facility (ECF) and Resilience and Sustainabilit y Facility (RSF) (16, 17)	372 million (17, 18)	3 years	ECF/EFF: Address balance of payment needs. RSF: Address challenges related to climate change and implementation of climate policies. (16).	Fiscal consolidation; stabilize wage bill; cuts in current spending; tax reform; increase domestic revenue mobilisation; cut fuel subsidies; improved public financial management; limit new borrowing; enhance debt management; strengthen fiscal governance; climate change adaptation and mitigation; expand social protection; enhance private sector role in wealth and job creation; diminish gender-based disparities; improve social welfare system (16, 17).	-	-	IT: Social spending floor: 40% of total government spending (16, 17).
2020- 24 (19)	WB	Country Partnership Framework	Strategy paper	-	-	Build human capital; boost competitiveness and job creation through private sector-led growth; increase resilience and sustainability in context of growing risks; digitalisation; modernisation; gender equity; climate change effect mitigation; contain fiscal vulnerabilities; high inclusive growth; contain fiscal deficit to 3% of GDP; strengthen revenue mobilisation; increase capital investments; energy sector subsidy reform; improved data on living conditions; reduce poverty; oil and gas revenue management; improvements to WB collaboration and projects. Several sector- and SDG-specific goals (19).	Improve progressiveness of social protection programs including Universal Health Coverage ( <i>Couverture Maladie</i> <i>Universelle</i> , CMU) (19).	-	-
PCI: 2020- 2023	IMF	Policy Coordinatio n Instrument	PCI: Non- financing instrument.	Details not identified	Support the overall fiscal and macroeconomic	Temporarily increased fiscal deficit during Covid-19 response then return to 3% of GDP; sale of government buildings; investment cuts; limit debt stock; transient	Increased health spending for Covid response in 2021; increased wage bill for health	-	IT: Social spending floor: 40% of total

(16, 20, 21); SBA and SBCF 2021 (18, 22)		(PCI); Stand-By Arrangemen t (SBA); Stand-By Credit Facility (SBCF)	SBA: 388 million. SBCF: 194 million (18)	in document (s)	country programme (20- 25)	energy subsidies to be phased out; adjust energy prices; improve revenue management; including new gas and oil management framework; food subsidies and lower food taxes; strengthen domestic revenue mobilization; new import levy; strengthened debt management; limit new public borrowing; improved spending efficiency; higher education sector investment; promote private-sector led growth and job creation; expand social protection (20-25).	staff; strengthen UHC program (20-26).		government spending (increased from 35% in 2022) (e.g. (23)).
2021 (27)	WB	Additional Financing for the Senegal COVID-19 Response Project	Loan: 57 million. Grant: 47 million (27)	38 years	Emergency health system and social response to Covid-19	Supporting emergency Covid-19 response; Community Engagement and Risk Communication; Project Management and Monitoring and Evaluation	-	-	-
2020 (28)	WB	Investing in Maternal, Child, and Adolescent Health Project (28)	€123 million (28)	37 years	Improve utilization of essential reproductive, maternal, neonatal, child and adolescent health and nutrition (RMNCAH-N) services (28)	Improving availability of RMNCAH-N services of adequate quality; promoting adolescent health and women's empowerment; supporting reforms to strengthen governance, equity, and financing sustainability in the health sector; contingency emergency response	Expand coverage of universal health insurance; CBHI grants for poor and vulnerable; expand maternal health vouchers (28)	-	-
2020 (29)	WB	Third Multi- Sectoral Structural Reforms Developme nt Policy Operation: Supplement al Financing	Loan: €46 million (Grant: SDR 36.6 million)	38 years	Development programme support	Details not identified in document(s)	-	-	-
2020 (18, 30)	IMF	Rapid Credit Facility (RCF); Rapid Financing Instrument (RFI)	RCF: 108 million; RFI: 216 million (18)	Details not identified in document (s)	Address urgent balance of payment needs due to Covid-19 (30)	Temporarily increased fiscal deficit during Covid-19 response then return to 3% of GDP; support vulnerable households and firms; increase liquidity to banks to ensure credit supply (30).	Increased health spending for Covid-19 response; expand social protection to one million vulnerable households (30).	-	-
2015- 2019 (21, 31)	IMF	Policy Support Instrument (PSI)-III	Non- financing instrument	-	-	Increased domestic revenue mobilization via tax reform; simplify tax administration; more equitable tax collection; create fiscal space; closure/merger of government agencies; strengthen public financial management; target	Freeze public sector hiring except for ( <i>other groups and</i> ) frontline health workers (e.g. (31))	-	IT: Social spending floor: 35% of total government

						fiscal deficit of 3% of GDP and contain current expenditure; contain wage bill; enhance efficiency of investment; fiscal consolidation; sustain growth; ensure debt sustainability; strengthen debt management; facilitate private investment and competitiveness; invest in human capital and infrastructure; attract foreign direct investment; tackle rent seeking; develop fiscal framework for oil and gas revenue; facilitate small- and medium-sized enterprise access to credit; improve business environment; address gender and inequality issues; reduce energy subsidies; promote financial sector development and stability (31-38).			spending (e.g. (32))
2017 (39)	WB	First, second and third governance and growth support credits	SDR 91.4 million	Details not identified in document (s)	Support government reforms =>	Improving economic governance; increasing public sector performance; promoting growth through private sector development.	Facilitate girls' and women's access to health services; save resources for better financing of primary health clinics	-	-
2013- 17 (40, 41)	IMF, WB	Poverty Reduction Strategy Paper (PRSP)-III (National Strategy for Economic and Social Developme nt, NSESD)	Strategy paper	-	-	High growth rate; develop production sectors; contain inflation; grow revenue; improve business climate; facilitate access to finance; stabilise public finance; reduce poverty and hunger; control public spending; strengthen exports and foreign direct investment; reduce public deficit; stabilize debt levels; labour market reform; create jobs; numerous sectoral aims; reinforce social protection; promote human rights; justice; gender equity and equality; decentralize government agencies; improve public financial management and efficiency (40, 41).	Numerous aims for health sector including establishing UHC; including through mutual health organizations; extend free healthcare to vulnerable groups; set up health solidarity fund; technical health structure upgrades; target 95% health budget execution; increase the health mutual scheme coverage rate to 66% by 2017 (40, 41).	-	-
2013- 17 (42)	WB	Country Partnership Framework	Strategy paper	-	-	Strengthen public governance, accountability and performance; enhance economic resilience; accelerate inclusive growth and create employment; improve service delivery, particularly regarding governance, access and equity in social sectors; energy sector reform; improve education and close skill gap; gender equity; several sectoral goals; improve targeting of social protection programs (42).	-	-	-
2010- 15 (43- 45)	IMF	Policy Support Instrument (PSI)-II	Non- financing instrument	-	-	Target fiscal deficit of 3% of GDP (about 5-7% in earlier years); invest in human capital and infrastructure; attract foreign direct investment; issue Eurobonds; boost inclusive growth; inclusiveness and poverty reduction; improve business climate; improve access to credit for small- and medium-sized enterprises; improved public financial management; increased public expenditure efficiency; strengthen and improve efficiency of social safety nets; limit	Increase fiscal space for social spending; Front-line health workers exempt from wage bill freeze; decrease current expenditure to allow for progress towards universal health coverage; rationalization of health	-	IT: Social spending floor: 35% of total government spending (e.g. (45))
						public debt; closure/merger of government ministries and agencies; reduce energy subsidies; restructuring of SENELEC (national electricity company); export diversification; job creation; labour market reform including easier layoffs with support for job search and skill training; tax reform; increase revenue; freeze public consumption / decrease current expenditure; freeze wage bill; improve fiscal transparency (43, 45-51).	expenditures with support from WB (43, 45-51).		
------------------	----	---	---	----------	---	---	--	---	--
2014 (52)	WB	Health and Nutrition Financing Project (52)	Grant: 3 million Loan: 10 million (52)	40 years	Health and nutrition project =>	Results-based financing for health and nutrition services and capacity building; improvement of accessibility to maternal nutrition and child health services; institutional strengthening and project implementation (52)	Establish and implement a universal health insurance system and an equity fund (52)	-	-
2011 (53, 54)	WB	Fifth Poverty Reduction Support Credit (PRSC) (53)	SDR 26.5 million (55)	40 years	Public financial management, (40%), education (10%), health, (20%), infrastructure (20%), environment (10%), and statistics (10%) (54)	Strengthen government policy making and institutional capabilities towards: Improving environment for private investment; private sector development; improving access to basic services; improve efficiency of public expenditure on human capital formation; human resource management in education sector; public hospital performance contracts; nutrition reinforcement; mitigate social and environmental vulnerabilities; improved public financial management, procurement, debt management and statistical systems reforms (53-55).	To increase the population covered by health mutuals (54). Target: 50 mutual schemes over baseline by 2011, 50 more in 2012 (54). Increase share of national budget in financing of new vaccines (40% by 2020) (54).	-	Trigger: Implement state financial support mechanism for health mutuals. Trigger: Support at least 50 new health mutuals covering rural populations and informal workers. Trigger: Ensure adequate budget appropriations for vaccines and related consumables (54).
2010 (56)	WB	Fourth Poverty Reduction Support Credit (PRSC)	SDR 28.4 million (56, 57)	40 years	Public financial management, including budget management, procurement, and debt management (20%), education (20%), health, (20%), social protection	Support implementation of PRSP-II (see below), including wealth creation and private sector development; human development and provision of basic social services, improving recruitment system for teachers; protecting vulnerable groups; improving good governance and participation (56, 57).	To increase the population covered by health mutuals (56). Target: 50 mutual schemes over baseline by 2011 (56).	PA: Initiate preparation of state financial support mechanism for health mutuals. PA: Adoption of the National Strategy for Extending	Trigger: Implement PA (left cell) Trigger: Support at least 50 new health mutuals covering rural populations

					(15%), environment (15%), statistics (5%), and business environment (5%) (56)			Health Risk Coverage.	and informal workers (56).
2008- 2010 (58-60)	IMF	Exogenous Shock Facility (ESF)	121 million (18, 60)	Details not identified in document (s)	To cope with repercussions of global financial crisis, including balance of payments issues	Allow Senegal to continue program under the PSI (below); settle payment delays; infrastructure investment; temporary easing of fiscal deficit to 4.5% of GDP; Reform public financial management and tax administration; energy sector reform; improving ease of "doing business"; increase monitoring of financial sector (59)	Finance priority social sectors	-	-
2007- 10 (61, 62)	IMF	Policy Support Instrument (PSI)-I	Non- financing instrument	-	-	High growth rate; reduce poverty; improve debt management, preserve debt sustainability; increase government revenue; stabilize the wage bill; invest in infrastructure; improve public financial management / fiscal governance; tax reform; improve tax administration; support private sector development; improve business climate; reduce financial sector vulnerabilities; improve financial sector oversight and their contribution to growth; energy sector reform; reform of SENELEC (national electricity company); reduce fiscal deficit to 4% of GDP; settle payment delays with private sector; reduce current expenditure including cutting untargeted subsidies in 2008, then countercyclical fiscal stimulus in 2009/10 (44, 62-66).	Protect social spending and progress toward Millennium Development Goals (MDGs), planned government investments in health were "welcomed" by IMF staff; prioritize health, education, environment and judicial sectors (44, 62-66).	-	IT: Social spending floor: 35% of total government spending from 2010 (e.g. (44))
2007- 10 (67)	WB	Country Assistance Strategy	Strategy paper	-	-	Improve transparency and efficiency of public resources; improve public sector accountability and governance; strengthen and modernize the judicial system; enhance private sector development and governance. Wealth creation; infrastructure investment; credit access for small and medium enterprises; develop labour and technology. A number of sectoral goals. Human development/shared growth for poor and vulnerable groups, improved equity, including gender equity; labour market reform and improved productivity; protection of vulnerable groups; prudent fiscal policy; expenditure efficiency; civil service reforms (67).	Improved access for vulnerable groups to health insurance schemes (under the PRSP) (67).	-	-
2008 (68)	WB	Completion report for PRSC I, II and III	Completion report	see below	see below	Strengthening budgetary and financial procedures; decentralization process; health services: improved financial and human resources management; wealth creation; Improving living conditions (68).	increase access to risk management instruments and social protection measures (outputs/outcomes were pilots to extend health insurance, and introduction of exemption scheme for elderly)	see below	see below

2007 (69)	WB	Third Poverty Reduction Support Credit (PRSC)	SDR 13.5 million (69, 70)	40 years	Support poverty reduction strategy (69)	Strengthening budgetary and financial procedures; decentralization; Health services, including performance contracts: financial and human resources management; wealth creation; national social protection strategy, fighting child labour (69, 70).	Increase state subsidy of essential drugs and services, government enforced stronger controls on drug tariffs, notably in remote regions (69)	-	IT: Extend pilot program to provide health insurance to uninsured workers in transport and agricultural sectors (69).
2006 (71)	IMF	Article IV consultation	Consultatio n conducted outside of an existing instrument	-	-	Resolve energy sector problems; prudent fiscal policy; improve fiscal governance and transparency; encourage private sector activity; banking reforms; increase growth; Reduce fiscal deficit to 4% of GDP; phase out subsidies; contain wage bill; eliminate private sector arrears; effective debt management; strengthen tax administration; privatize SENELEC; raise financial sector contribution to the economy; export diversification; improve infrastructure; further trade openness (71)	-	-	-
2006- 10 (44, 72)	IMF, WB	Poverty Reduction Strategy Paper (PRSP)-II	Strategy paper	-	-	Main pillars: Wealth creation for pro-poor growth; access to basic social services; improve living conditions of vulnerable groups; social protection; risk and disaster prevention and management; good governance; decentralised participatory development. Also included: Generate fiscal surplus over time; promote private sector participation; preserve debt sustainability; limit fiscal deficit to 4% of GDP; keep wage bill below 30% of tax revenue; strengthen public financial management; decentralize decision-making; improve public investment efficiency; eliminate arrears to private sector (72-75).	Increase social spending to 40% of total spending by 2010; improved access of the poor to medical services through sufficient funding and quality infrastructures + community- based health services; relieving the burden of expenses for health care for the poor; strengthen health mutuals; improve health insurance system to cover all employed labour force and dependents; establish health risk protection system for vulnerable persons; increase health insurance coverage to 50% of population by 2015, compared to 7.8% in 2005. Increased health operating expenditure between 2006-10; Allocate debt cancellation resources to health and other priority sectors. Extend pilot program providing health insurance to uninsured workers in the transport and agricultural sectors, enhance contracting with the private	-	

2005 (88)	WB	Second Poverty Reduction Support Credit (PRSC)	US\$30 million	Details not identified in document (s)	Central government administration (50%); Other social services (40%); Sub- national government administration (10%) (89)	Support the implementation of PRSP and strengthen government's capacity to design and carry out its own development and poverty reduction programs; improve effectiveness and transparency of national budget process and expenditure management; promote development of a conducive environment for private investment; improve living conditions of the poor, with focus on delivery of better health services. (88, 89)	Details not identified in document(s)	Details not identified in document(s)	Details not identified in document(s)
2002- 06 (80)	IMF, WB	Poverty Reduction Strategy Paper (PRSP)-I	Strategy paper	-	-	Main pillars: Wealth creation; capacity building and provision of basic social services; improve living conditions of vulnerable groups; decentralized monitoring and evaluation. Numerous development strategies for different sectors. Also included: Accelerated growth; reduce poverty; fiscal decentralization and improved transparency; expand social protection; target wage bill relative decrease to 27% of revenue; broaden tax base; privatization; reduced corporate income tax from 35 to 33%; some tax exemptions abolished; improve business climate for investments; promote private sector; increase economic productivity; lower energy costs; strengthen debt management capacity; improved public expenditure management, governance, and investment spending efficiency; (80-87).	Access to subsidy for chronically ill health mutual members; Provide infrastructure/ improve equipment at health centres and stations; part of HIPC funds allocated to the health sector; Increase government health spending from 9.2% to 10% of total budget; subsidized care for people living with a disability; enhance access of poorest groups to health services; reducing health expenditures for the poor; access to health mutuals; funding for health care insurance associations, decentralize health budget execution, increase private health service delivery (69, 76, 81, 84, 85)	-	-
2003- 06 (77)	IMF	Poverty Reduction and Growth Facility (PRGF)	24 million (60, 77)	Details not identified in document (s)	Details not identified in document(s)	High growth; contain inflation; Fiscal deficit contained to 1- 4% of GDP; increase wage bill to 29% of revenue, with wage bill ceiling in place; floor on tax revenue; improved tax administration; privatize SONACOS (vegetable oil refining company); rehabilitate SENELEC and ICS (chemical company); improve business environment; simplify private sector regulations; reinforce external competitiveness; trade liberalization; improve public expenditure management and fiscal transparency; contain public investment; cut corporate income tax; improve debt management; enhance bank soundness; improve credit availability (77-79).	sector for health care (56, 69, 72-76). Increased health capital expenditure; increased budget allocations to health in 2006 (77-79).	-	-
				Details		High growth; contain inflation; Fiscal deficit contained to 1- 4% of GDP; increase wage bill to 29% of revenue, with wage bill ceiling in place; floor on tax revenue; improved tax administration; privatize SONACOS (vegetable oil	sector for health care (56, 69, 72-76).		

2004 (90)	WB	First Poverty Reduction Support Credit (PRSC)	SDR 20.5 million	40 years	Support poverty reduction strategy (90)	Support implementation of poverty reduction strategy. Strengthening budgetary and financial; decentralization; improving financial and human resources management in health sector, accelerating access to basic health (90).	Enhance the role of the private sector in health service delivery. Enhance foreign and local financing to the private health sector. Contracting policy for government to provide coverage for selected private health services. Reducing financial barriers to health service access for poor and vulnerable groups. Developing pooling systems and mitigating effects of cost recovery system on poor and vulnerable groups. Harmonized fee system for drugs and cost recovery. Exemption policy and better management of government subsidies to drugs and other services used by poor, pregnant women and children under 5 years old. Support development of mutual health insurance associations and enabling government to pay premiums for poor and guarantee funds for their sustainability. Increase public health expenditures and pro-poor emphasis (90).	PA: Start decentralizatio n of budgetary execution in Ministry of Health (90).	-
2000- 2004 (91)	IMF, WB	Heavily Indebted Poor Countries (HIPC) Initiative	<u>Debt relief</u> <u>amount</u> : US \$850 million (91)	-	-	HIPC Completion criteria: PRGF and PRSP arrangements in place; privatize 11 public sector enterprises (3/4 reduction of public ownership portfolio); simplify business income taxation; reduce hurdles and "policy distortions" for private sector; liberalization of petroleum sector including oil exploration; reinforce domestic resource mobilization; targets for teacher recruitment and primary education; targets for immunization, prenatal care; primary health centre utilization (91, 92)	-	-	-
2002 (93)	WB	HIV/AIDS Prevention and Control Project	23.6 million (93)	40 years	HIV/AIDS Prevention and Control	Within HIV/AIDS prevention and control: strengthening capacity of executing and implementing agencies and project coordination structures; support to civil society and community based initiatives; support to multi-sector programs of government agencies; support to project management, administration and monitoring (93)	-	-	-

1998- 2002 (94)	IMF	Poverty Reduction and Growth Facility (PRGF)	96 million (18)	Details not identified in document (s)	Details not identified in document(s)	Strengthen public resource management; target fiscal balance of -1% of GDP deficit to 1% surplus; remove impediments to private sector development; privatization of SONACOS; privatization/reform of SENELEC, strengthen oversight with postal service and avert losses; improve public health infrastructure; tax reform; broaden tax base; streamline corporate taxation; strengthen tax and customs administration; improve external competitiveness; floor on tax revenue; contain wage bill within wage bill ceiling; introduction of single-rate VAT; strengthen revenue collection (94-97).	Increase public spending on basic social services; Enhance the pro-poor emphasis of public spending on primary health care (94-96).	-	-
1998- 2001 (98, 99)	IMF	Enhanced Structural Adjustment Facility (ESAF), Extended Credit Facility (ECF)	87 million (100)	Details not identified in document (s)	Details not identified in document(s)	High growth rate; low inflation; restrict external current account deficit; government budget balance between -2 to +2% of GDP; decrease public spending per GDP by one percentage point; improve social indicators; reduce poverty and unemployment; fiscal consolidation; reduce relative share of wage bill with education and health having highest priority for recruitment; strengthen revenue collection via tariff reform and single-rate VAT; broaden tax base, eliminate tax exemptions, reduce import tax; increased capital expenditure; privatization/liquidation of public enterprises; develop the private sector; administrative and judicial reform; sectoral reforms including several health sector targets; debt sustainability; decentralization (98, 99).	Increase social sector spending; increase health expenditure share out of total budget to 9% in 2002; improve access of vulnerable groups to health services; decentralized local government public health cost sharing; revitalize formal sector health insurance (IPMs); creation of Mutual Association Support Unit and formal sector health insurance (IPMs); promotion of community based health insurance; rehabilitate private sector health insurance companies (98, 99).	-	-
1998- 2000 (101).	WB	Country Assistance Strategy	Strategy paper	-	-	Reduce poverty; create jobs; high growth rate; "unshackle" and develop private sector; development of social and human resources; broaden VAT base; improved public expenditure management; financial reform; trade liberalization; debt sustainability; reduce size of public sector; improve infrastructure; decentralization; environmental sustainability of growth; ensure social sustainability; several sectoral projects and goals (101).	Increase current budget share allocation to health from 6% in 1996 to 9% in 2002; greater use of beneficiaries and NGOs in the financing of basic health care services (101).	-	-
1995- 98 (102)	WB	Country Assistance Strategy	Strategy paper	-	-	Economic growth with equity; poverty reduction; promote competitiveness; trade and price liberalization; sectoral reforms including privatization and liberalization reforms of agricultural sector for increased efficiency and exports, restructuring/privatization of parastatals; improved infrastructure; improve composition of public spending; increased private sector investment; slowing population growth; promoting status of women; natural resource management (102)	Decentralized health cost recovery (102)	-	-

1997 (103)	WB	Endemic Disease Control Project	11 million (103)	40 years	Control endemic and epidemic diseases	Malaria, schistosomiasis and onchocerciasis control; developing and integrating a health management information system; rehabilitation of buildings	-	-	-
1997 (104)	WB	Integrated Health Sector Developme nt Program	36 million (104)	40 years	Health sector development	Constructing/rehabilitating health posts, health centres, tertiary hospitals, social and health-related buildings and centres; provision of equipment for above; information, education, communication, technical advice in public health/health systems (104)	-	-	-
1993 (105, 106)	WB	Macroecono mic update report; Public expenditure review	Economic reports	-	-	Reduce and rationalize public expenditure; increase domestic resource mobilisation by customs and trade tariffs; broaden tax base; become less dependent on external support; privatize/reform parastatals; depreciate real exchange rate to increase external competitiveness and exports; human resources development; several sectoral recommendations (106); Several recommendations for public expenditure reallocation within and between sectors; domestic deflation to improve competitiveness (105)	Protect expenditure on health to ensure minimum efficiency of public services (106); Increase current health expenditure out of total government expenditure from 4.1% in 1989/90 to 6.7% in 1995; develop and establish health insurance; drug cost recovery (105).	-	-
1983- 93 (107- 109)	IMF, WB	Structural Adjustment Facility (SAF); Enhanced Structural Adjustment Facility (ESAF); Extended Credit Facility (ECF) (including 1985-92 Structural Adjustment Program (SAP))	IMF: 335 million (100)	Details not identified in document (s)	Details not identified in document(s)	Reduce fiscal deficit; cut government expenditures; credit restraint and interest rate close to or above French levels; liberalization of prices, trade and markets; eliminate subsidies and trade regulations; increase private sector participation in economic activities; reduce public sector's involvement in the economy, promotion of private sector initiative; productive sector investment; improved efficiency of public resource management; reform/privatize/liquidate public enterprises; reduce production costs; widen tax base; improve customs duty collection and reduce tax exemptions; sectoral reforms including financial sector reform; wage bill reductions but with health workers excluded from recruitment ceilings (106-111).	-	-	-
1991 (112)	WB	Human Resources Developme nt Project	25 million (112)	40 years	Control fertility, reduce population growth and restructure health sector (112)	Strengthening of the national family planning program; promotion of the status of women; promotion of youth through family life education; institutional support for the promotion of the national population program; development of the district health system; promotion of essential drugs; institutional strengthening of ministry of health (112)	Restructure the health sector to enable it to provide improved basic health services, including through private sector provision of family planning services (112)	-	-

1987- 90 (110)	IMF, WB	Policy Framework Paper (PFP)	Strategy paper	-	-	High growth rate; curb inflation; reduce external current account deficit; promotion of private sector initiative; greater efficiency in public resource management (110).	-		
1984 (111)	WB	Country Economic Memorandu m	Economic report	-	-	Improve public finances and public sector management; reduce fiscal deficit; intensify domestic revenue mobilisation; improve tax administration; ; develop growth points; streamline government investment planning; restrain domestic demand; promote exports; improve customs duty administration; fees and user charges for government services; more liberal employment policies for private and public enterprises; openness to foreign trade; reduced government intervention in industry; reduce civil service numbers; improve public debt management; parapublic sector reform; several sectoral recommendations (111).	Careful consideration before embarking on projects to improve or to expand the coverage of public health services, because government cannot expect to finance the recurrent costs of such projects; Introduce user charges for health services; community participation in cost of dispensaries (111).	-	-
1983 (113)	WB	Rural Health Project	14 million (113)	40 years	Health systems strengthening	Constructing and equipping new health centres; renovating and re-equipping existing health centres; training health personnel; improving supply and utilisation of basic drugs; strengthening health education; ministry of health capacity building; nutrition (113)	-	-	-
1979- 85 (111, 114)	IMF, WB	Medium- Term Plan for Economic and Financial Recovery (Structural Adjustment Program (SAP)-1); IMF Stand- By Arrangemen t (SBA); IMF Extended Fund Facility (EFF); WB Structural Adjustment Loan (SAL)	EFF: 185 million (partly cancelled); SBA: 126 million; SAL: 53 million (115)	Details not identified in document (s)	Support medium-term adjustment program (115)	Reform of the parapublic sector; reform of agricultural policies; expand rural production; improved production incentives; economic stabilization; raise public savings through austere fiscal policies; reduce external current account deficit; reduce fiscal deficit to 4.8% of GDP; limited and selective growth of investments focusing on productive sectors; improve monetary policies; credit restriction; improve balance of payments; encourage exports; improve foreign debt management; liberate prices; liberalise trade; limit wage growth; reduce subsidies; reduce state participation in the economy (111, 115, 116).	-	-	-
1979 (117)	WB	The Economic Trends and	Economic report	-	-	Eliminate housing subsidies for established workers; public investment in slum upgrading, land preparation and basic collective facilities; increase pricing of public services; population policy; higher economic growth rate; Several	-	-	-

		Prospects of Senegal				sectoral recommendations; continue promotion of rainfed agriculture; accelerate irrigation; develop a labour- intensive, export-oriented industry in Dakar; currency devaluation; export subsidies; restrict commercial borrowing (117)			
1973 (118, 119)	WB	Senegal - The economy	Economic report	-	-	Rationalise public investment; improved public financial management and planning; build ministerial human resource capacity (118); agricultural diversification; develop water resources; expand fishery and manufacturing; several other sectoral recommendations; stimulate exports; improve import competition; promote local enterprise; reduce unemployment; grow and diversify economy; invest in productive sectors (119)	Increase government expenditures for materials and supplies in health sector (119); faster increase in health sector investment allocations (119)	-	-
1970 (120)	WB	Senegal - The current economic situation and prospects of Senegal	Economic report	-	-	Increase economic competitiveness; diversify production; develop industrial exports; several sectoral recommendations (120)	-	-	-
1968 (121)	WB	Current economic position and prospects of Senegal	Economic report	-	-	-	-	-	-
1966 (122)	WB	The Economy of Senegal	Economic report	-	-	-	-	-	-

Table 10.4: Overview of IMF and World Bank instruments and arrangements in Senegal. Main policy content in associated documents, including policy recommendations, targets, objectives and loan conditionalities directly relevant to domestic health financing sources, were extracted. \*One Special Drawing Rights (SDR) has historically been worth approximately 1.3-1.5 US\$, depending on the daily exchange rate (123). We have only included IMF disbursements made specifically under the listed arrangements/instruments, not the running smaller charges on Senegal's General Resource Account (GRA). Similarly, WB documents include major development policy and loan agreements, and not smaller project financing agreements. IT: Indicative Target. PA: Prior Action.

## References

World Bank. World Bank Open Data. 2024 [cited 2024 March 28]. Available from: 1. https://data.worldbank.org/. Halton C. Structural Adjustment. 2021 [cited 2022 February 9]. Available from: 2 https://www.investopedia.com/terms/s/structural-adjustment.asp. IMF. IMF Glossary. 2022 [cited 2022 February 9]. Available from: 3. https://www.imf.org/en/About/Glossary. IMF. Poverty Reduction Strategy Papers (PRSP). 2016 [cited 2022 February 9]. Available from: 4. https://www.imf.org/external/np/prsp/prsp.aspx. IMF. Policy Support Instrument (PSI). 2020 [cited 2022 February 9]. Available from: 5. https://www.imf.org/en/About/Factsheets/Sheets/2016/08/02/21/21/Policy-Support-Instrument. IMF. Guidance Note on the Exogenous Shocks Facility. 2006 [cited 2022 February 10]. Available from: 6. https://www.imf.org/en/Publications/Policy-Papers/Issues/2016/12/31/Guidance-Note-on-the-Exogenous-Shocks-Facility-PP534. IMF. IMF Policy Coordination Instrument (PCI). 2021 [cited 2022 February 10]. Available from: 7. https://www.imf.org/en/About/Factsheets/Sheets/2017/07/25/policy-coordination-instrument. 8. IMF. IMF Rapid Credit Facility (RCF). 2022 [cited 2022 February 10]. Available from: https://www.imf.org/en/About/Factsheets/Sheets/2016/08/02/21/08/Rapid-Credit-Facility. IMF. The IMF's Rapid Financing Instrument (RFI). 2021 [cited 2022 February 10]. Available from: 9 https://www.imf.org/en/About/Factsheets/Sheets/2016/08/02/19/55/Rapid-Financing-Instrument. IMF. IMF Standby Credit Facility (SCF). 2022 [cited 2022 February 10]. Available from: 10. https://www.imf.org/en/About/Factsheets/Sheets/2016/08/02/21/10/Standby-Credit-Facility. IMF. The Extended Credit Facility (ECF). 2023 [cited 2023 August 25]. Available from: 11. https://www.imf.org/en/About/Factsheets/Sheets/2023/Extended-Credit-Facility-ECF. IMF. The Extended Fund Facility (EFF). 2023 [cited 2023 August 25]. Available from: 12. https://www.imf.org/en/About/Factsheets/Sheets/2023/Extended-Fund-Facility-EFF. IMF. The Resilience and Sustainability Facility (RSF). 2023 [cited 2023 August 25]. Available from: 13. https://www.imf.org/en/About/Factsheets/Sheets/2023/Resilience-Sustainability-Facility-RSF CUEX. Special Drawing Rights (SDR) (XDR) / US Dollar (USD). 2022 [cited 2022 February 10]. 14. Available from: https://cuex.com/en/xdr-usd. World Bank. Second Equitable and Resilient Recovery in Senegal Development Policy Financing. 2023 15 [cited 2024 July 15]. Available from: https://documents1.worldbank.org/curated/en/099062023122065708/pdf/P17529301be541076098d00fab699502 811.pdf 16. IMF. Senegal: Requests for an Extended Arrangement Under the Extended Fund Facility, an Arrangement Under the Extended Credit Facility, and an Arrangement Under the Resilience and Sustainability Facility-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2023 [cited 2023 August 22]. Available from: https://www.imf.org/en/Publications/CR/Issues/2023/07/07/Senegal-Requests-for-an-Extended-Arrangement-Under-the-Extended-Fund-Facility-an-535844. 17. IMF. First reviews under the extended fund facility, the extended credit facility, and the resilience and sustainability facility arrangements, requests for modification of the quantitative performance criteria rephasing of access—press release; staff report; and statement by the executive director for senegal. 2023 [cited 2024 July 15]. Available from: https://www.imf.org/en/Publications/CR/Issues/2023/12/19/Senegal-First-Reviews-Under-the-Extended-Fund-Facility-the-Extended-Credit-Facility-and-the-542659. IMF. Flows from May 01, 1984 to August 31, 2023. 2023 [cited 2023 September 11]. Available from: 18 https://www.imf.org/external/np/fin/tad/gueryoutput.aspx. 19. World Bank. Country partnership framework for the Republic of Senegal for the period FY20-FY24. 2020 [cited 2023 September 18]. Available from: https://documents.worldbank.org/en/publication/documentsreports/documentdetail/608781583719225540/senegal-country-partnership-framework-for-the-period-fy20-fy24. IMF. First review under the policy coordination instrument and request for modification of quantitative 20 targets-press release; staff report; and statement by the executive director for Senegal. 2020 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2020/07/24/Senegal-First-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-49608. IMF. Request for a three-year policy coordination instrument-press release; staff report; and statement 21. by the executive director for senegal. 2020 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2020/01/18/Senegal-Request-for-a-Three-Year-Policy-Coordination-Instrument-Press-Release-Staff-Report-48958. 22. IMF. Senegal: Third Review Under the Policy Coordination Instrument and Request for Modification of Quantitative Targets, and Requests for a Stand-By Arrangement and an Arrangement Under the Standby Credit Facility-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2021 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2021/06/17/Senegal-Third-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-460831. 23. IMF. Senegal: Sixth Review Under the Policy Coordination Instrument and Third Reviews Under the

23. IMF. Senegal: Sixth Review Under the Policy Coordination Instrument and Third Reviews Under the Stand-By Arrangement and the Arrangement Under the Stand-By Credit Facility-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2023 [cited 2023 August 22]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2023/01/18/Senegal-Sixth-Review-Under-the-Policy-Coordination-Instrument-and-Third-Reviews-Under-the-528312.

24. IMF. Fifth review under the policy coordination instrument, second reviews under the standby arrangement and the arrangement under the standby credit facility, and requests for augmentation of access, waiver of the nonobservance of a performance criterion, and modification of a performance criterion and quantitative targets—press release; staff report; and statement by the executive director for senegal. 2023 [cited 2023 August 23]. Available from: <a href="https://www.imf.org/en/Publications/CR/lssues/2022/06/27/Senegal-Fifth-Review-Under-the-Policy-Coordination-Instrument-Second-Reviews-Under-the-520104">https://www.imf.org/en/Publications/CR/lssues/2022/06/27/Senegal-Fifth-Review-Under-the-Policy-Coordination-Instrument-Second-Reviews-Under-the-520104</a>.

25. IMF. 2021 article IV consultation, fourth review under the policy coordination instrument, first reviews under the stand-by arrangement and the arrangement under the standby credit facility, and request for modification of performance criteria and quantitative targets. 2021 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2022/01/14/Senegal-2021-Article-IV-Consultation-Fourth-Review-Under-the-Policy-Coordination-Instrument-511932.

26. IMF. Second review under the policy coordination instrument and request for modification of quantitative targets—press release; and staff report. 2021 [cited 2023 August 23]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2021/01/19/Senegal-Second-Review-Under-the-Policy-Coordination-Instrument-and-Request-for-Modification-50026.

27. World Bank. Financing Agreement: Additional Financing for the Senegal COVID-19 Response Project. 2021 [cited 2024 July 15]. Available from:

https://documents1.worldbank.org/curated/en/745311624664202803/pdf/Official-Documents-Financing-Agreement-for-Additional-Financing-Credit-No-6964-SN-and-Grant-No-D8890-SN.pdf.

28. World Bank. Financing agreement: Investing in Maternal, Child, and Adolescent Health Project). 2020 [cited 2024 July 15]. Available from:

https://documents1.worldbank.org/curated/en/549231586273646943/pdf/Official-Documents-Financing-Agreement-for-Credit-6495-SN.pdf

29. Association RoSalD. Financing Agreement (Third Multi-Sectoral Structural Reforms Development Policy Operation: Supplemental Financing). 2020 [cited 2024 June 12]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/860601592574172413/official-documents-financing-agreement-for-credit-6681-sn-and-grant-d651-sn.

30. IMF. IMF Executive Board Approves for Senegal a US\$442 Million Disbursement Under the Rapid Credit Facility and Purchase Under the Rapid Financing Instrument. 2020 [cited 2023 August 23]. Available from: https://www.imf.org/en/Publications/CR/Issues/2020/04/16/Senegal-Request-for-Disbursement-Under-the-Rapid-Credit-Facility-and-Purchase-Under-the-49335.

31. IMF. Request for a three-year policy support instrument—press release; staff report; and statement by the executive director for Senegal. 2015 [cited 2023 August 28]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Request-for-a-Three-Year-Policy-Support-Instrument-Press-Release-Staff-Report-and-43308

32. IMF. Senegal: Staff Report for the 2018 Article IV Consultation and Seventh Review Under the Policy Support Instrument and Request for Modification of Assessment Criteria--Debt Sustainability Analysis-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2019 [cited 2023 August 24]. Available from: <a href="https://www.imf.org/en/Publications/CR/lssues/2019/01/28/Senegal-Staff-Report-for-the-2018-Article-IV-Consultation-and-Seventh-Review-Under-the-46553">https://www.imf.org/en/Publications/CR/lssues/2019/01/28/Senegal-Staff-Report-for-the-2018-Article-IV-Consultation-and-Seventh-Review-Under-the-46553</a>.

33. IMF. Senegal: Sixth Review Under the Policy Support Instrument-Press Release; Staff Report; and Statement by the Executive Director for Senegal. 2018 [cited 2023 August 24]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2018/07/09/Senegal-Sixth-Review-Under-the-Policy-Support-Instrument-Press-Release-Staff-Report-and-46057.</u>

34. IMF. Senegal: Fifth Review Under the Policy Support Instrument and Request for Modification of Assessment Criterion-Press Release; Staff Report. 2018 [cited 2023 August 24]. Available from: https://www.imf.org/en/Publications/CR/Issues/2018/01/12/Senegal-Fifth-Review-Under-the-Policy-Support-Instrument-and-Request-for-Modification-of-45558.

35. IMF. Fourth review under the policy support instrument and request for an extension of the policy support instrument—press release; staff report; and statement by the executive director for Senegal. 2023 [cited 2023 August 25]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2017/07/24/Senegal-Fourth-Review-Under-the-Policy-Support-Instrument-and-Request-for-an-Extension-of-45115">https://www.imf.org/en/Publications/CR/Issues/2017/07/24/Senegal-Fourth-Review-Under-the-Policy-Support-Instrument-and-Request-for-an-Extension-of-45115</a>.

36. IMF. Staff report for the article IV consultation and third review under the policy support instrument press release, and staff report. 2017 [cited 2023 August 25]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2017/01/10/Senegal-Staff-Report-for-the-Article-IV-Consultation-">https://www.imf.org/en/Publications/CR/Issues/2017/01/10/Senegal-Staff-Report-for-the-Article-IV-Consultation-</a>

https://www.imf.org/en/Publications/CR/Issues/2017/01/10/Senegal-Staff-Report-for-the-Article-IV-Consultationand-Third-Review-Under-the-Policy-44505.

37. IMF. Second review under the policy support instrument and request for modification of an assessment criterion—press release; and staff report. 2016 [cited 2023 August 25]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Second-Review-Under-the-Policy-Support-Instrument-and-Request-for-Modification-of-an-43940</u>.

38. IMF. First review under the policy support instrument and request for modification of assessment criteria—press release; and staff report

2016 [cited 2023 August 28]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-First-Review-Under-the-Policy-Support-Instrument-and-Request-for-Modification-of-43544.

39. World Bank. Implementation completion and results report. 2017 [cited 2024 July 2]. Available from: https://documents1.worldbank.org/curated/en/664321502112241626/pdf/ICR00004100-07262017.pdf.

40. IMF. Senegal: Poverty Reduction Strategy Paper. 2013 [cited 2023 August 28]. Available from: https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/pubs/ft/scr/2013/cr13194.ashx.

41. IMF, World Bank. Poverty reduction strategy paper—joint staff advisory note on the national strategy for economic and social development. 2013 [cited 2023 August 29]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Joint-Staff-Advisory-Note-on-the-National-Strategy-40740.

42. World Bank. Country Partnership Strategy (FY2013-2017) for the Republic of Senegal. 2013 [cited 2023 September 18]. Available from:

https://documents1.worldbank.org/curated/en/527551468103763222/pdf/NonAsciiFileName0.pdf.

43. IMF. Senegal: First Review Under the Policy Support Instrument: Staff Report; Staff Supplement; Press Release; and Statement by the Executive Director for Senegal. 2011 [cited 2023 September 6]. Available from: https://www.elibrary.imf.org/view/journals/002/2011/139/002.2011.issue-139-en.xml?rskey=XFITLh&result=1.

44. IMF. Senegal: Sixth Review Under the Policy Support Instrument, Request for a Three-Year Policy Support Instrument and Cancellation of Current Policy Support Instrument—Staff Report; Debt Sustainability Analysis; Press Release; Executive Director Statement. 2010 [cited 2023 September 6]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Sixth-Review-Under-the-Policy-Support-Instrument-Request-for-a-Three-Year-Policy-24502.

45. IMF. 2014 article IV consultation and eighth review under the policy support instrument—staff report; press release; and statement by the executive director for Senegal. 2015 [cited 2023 August 28]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2014-Article-IV-Consultation-and-Eighth-Review-Under-the-Policy-42578.

46. IMF. Senegal: Seventh Review under the Policy Support Instrument and Request for Modification of Assessment Criteria. 2014 [cited 2023 August 28]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Seventh-Review-under-the-Policy-Support-Instrument-and-Request-for-Modification-of-41687.

47. IMF. Senegal: Sixth Review Under the Policy Support Instrument and Request for Modification of an Assessment Criterion—Staff Report; Informational Annex; Press Release and Executive Director's Statement. 2014 [cited 2023 August 28]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Sixth-Review-Under-the-Policy-Support-Instrument-and-Request-for-Modification-of-an-41186.</u>

48. IMF. Fifth Review Under the Policy Support Instrument and Request for Program Extension and Modification of Assessment Criteria—Staff Report; Debt Sustainability Analysis; Informational Annex; and Press Release. 2013 [cited 2023 August 30]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Fifth-Review-Under-the-Policy-Support-Instrument-and-Request-for-Program-Extension-40700

49. IMF. Staff Report for the 2012 Article IV Consultation, Fourth Review Under the Policy Support Instrument, and Request for Modification of an Assessment Criterion—Staff Report; Staff Supplements; Public Information Notice and Press Release on the Executive Board Discussion; and Statement by the Executive Director for Senegal. 2013 [cited 2023 August 31]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2012-Article-IV-Consultation-Fourth-Review-Under-the-Policy-40177

50. IMF. Senegal: Third Review Under the Policy Support Instrument and Request for Modification of Assessment Criteria: Staff Report; Informational Annex; Press Release; and Statement by the Executive Director for Senegal. 2012 [cited 2023 September 4]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Third-Review-Under-the-Policy-Support-Instrument-and-Request-for-Modification-of-26158.

51. IMF. Senegal: Second Review Under the Policy Support Instrument and Request for Modification of Assessment Criteria—Staff Report; Informational Annex; Press Release. 2011 [cited 2023 September 6]. Available from: <u>https://www.elibrary.imf.org/view/journals/002/2011/373/002.2011.issue-373-en.xml?rskey=XtDJUF&result=7.</u>

52. World Bank. Health and Nutrition Financing Project. 2014 [cited 2024 July 15]. Available from: https://documents1.worldbank.org/curated/en/435611468168568721/pdf/RAD2073336869.pdf.

53. World Bank. Senegal - Fifth Poverty Reduction Support Credit. 2011 [cited 2024 July 3]. Available from: https://www.worldbank.org/en/news/loans-credits/2011/04/26/senegal-fifth-poverty-reduction-support-credit.

54. World Bank. International Development Association program document for a proposed fifth poverty reduction support credit. 2011 [cited 2024 July 3]. Available from:

https://documents1.worldbank.org/curated/en/820181468104332374/pdf/603730PGD0RepI0y0900same0box0nu mber.pdf

World Bank. Fifth Poverty Reduction Support Development Policy Financing. 2011 [cited 2024 July 16].
 Available from: <u>https://documents1.worldbank.org/curated/en/656411468057321845/pdf/FA00Conformed0.pdf</u>.
 World Bank. Fourth Poverty Reduction Support Credit. 2010 [cited 2024 July 3]. Available from:

56. World Bank. Fourth Poverty Reduction Support Credit. 2010 [cited 2024 July 3]. Available from: <u>https://documents1.worldbank.org/curated/en/241821468303664169/pdf/527480PGD0Box31ly10IDA1R2010101</u> 2511.pdf.

World Bank. Fourth Poverty Reduction Support Development Policy Financing. 2010 [cited 2024 July 57. 161. Available from:

https://documents1.worldbank.org/curated/en/226771468107670407/pdf/FA047380SN0conformed.pdf.

IMF. Senegal: Letter of Intent, Memorandum of Economic and Financial Policies, and Technical 58 Memorandum of Understanding, May 7, 2010. 2010 [cited 2023 September 6]. Available from: https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/np/loi/2010/sen/050710.ashx.

IMF. Senegal: Third Review Under the Policy Support Instrument and First Review and Augmentation of 59. Access Under the Exogenous Shocks Facility: Staff Report: Staff Supplement: Staff Statement: Press Release on the Executive Board Discussion; and Statement by the Executive Director for Senegal. 2009 [cited 2023 September 6]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Third-Review-Under-the-Policy-Support-Instrument-and-First-Review-and-Augmentation-23071.

IMF. Senegal: Transactions with the Fund from May 01, 1984 to August 31, 2023 2023 [cited 2023 60. September 11]. Available from:

https://www.imf.org/external/np/fin/tad/extrans1.aspx?memberKey1=840&endDate=2023-08-31.

IMF. Senegal: Letter of Intent, Memorandum of Economic and Financial Policies, and Technical 61. Memorandum of Understanding. 2010 [cited 2023 September 6]. Available from: https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/np/loi/2010/sen/111010.ashx.

IMF. Senegal: Request for a Three-Year Policy Support Instrument—Staff Report; Staff Statement; 62. Press Release on the Executive Board Discussion; and Statement by the Executive Director for Senegal. 2007 [cited 2023 September 7]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Request-for-a-Three-Year-Policy-Support-Instrument-Staff-Report-Staff-Statement-21439.

63. IMF. Senegal: Staff Report for the 2010 Article IV Consultation, Fifth Review Under the Policy Support Instrument, Third Review Under the Exogenous Shocks Facility, Request for Waiver of Nonobservance of Performance Criterion, and Modification of Assessment Criterion-Staff Report; Debt Sustainability Analysis; Press Release; Executive Director Statement; Public Information Notice. 2010 [cited 2023 September 6]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2010-Article-IV-Consultation-Fifth-Review-Under-the-Policy-23944.

IMF. Senegal: Fourth Review Under the Policy Support Instrument and Second Review Under the 64 Exogenous Shocks Facility-Staff Report; Press Release. 2010 [cited 2023 September 6]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Fourth-Review-Under-the-Policy-Support-Instrument-and-Second-Review-Under-the-23534.

IMF. Senegal: Second Review Under the Policy Support Instrument, Request for a Twelve-Month 65. Arrangement Under the Exogenous Shocks Facility, and Request for Waivers and Modification of Assessment Criteria -Staff Report; Staff Statement; Press Release on the Executive Board Discussion; and Statement by the Executive Director for Senegal. 2009 [cited 2023 September 7]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Second-Review-Under-the-Policy-Support-Instrument-Request-for-a-Twelve-Month-22596.

IMF. Senegal: Staff Report for the 2008 Article IV Consultation, First Review Under the Policy Support 66. Instrument, and Request for Waiver of Assessment Criterion and Modification of Assessment Criteria: Staff Report; Staff Supplement; Staff Statement; Public Information Notice and Press Release on the Executive Board Discussion; and Statement by the Executive Director for Senegal. 2009 [cited 2023 September 7]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2008-Article-IV-Consultation-First-Review-Under-the-Policy-22125.

67. World Bank. <u>https://documents.worldbank.org/en/publication/documents-</u> reports/documentdetail/979831468308065665/senegal-country-assistance-strategy. 2007 [cited 2023 September 19]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/979831468308065665/senegal-country-assistance-strategy.

World Bank. Implementation completion and results report. 2008 [cited 2024 July 2]. Available from: 68. https://documents1.worldbank.org/curated/en/106551468305075940/text/ICR6690Multipl1IC010disclosed081251 .txt.

69. World Bank. Senegal - Third Poverty Reduction Support Credit. 2007 [cited 2024 June 13]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/100271468103766953/senegal-third-poverty-reduction-support-credit.

70. World Bank. Third Poverty Reduction Support Development Policy Financing. 2007 [cited 2024 July 16]. Available from: https://documents1.worldbank.org/curated/en/456611468305069129/pdf/FA0Conformed0.pdf.

IMF. Senegal: 2006 Article IV Consultation: Staff Report; Public Information Notice on the Executive 71 Board Discussion; and Statement by the Executive Director for Senegal. 2007 [cited 2023 Septembber 7]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-2006-Article-IV-Consultation-Staff-Report-Public-Information-Notice-on-the-Executive-21359.

72. Republic of Senegal. Senegal: Poverty Reduction Strategy Paper. 2006 [cited 2023 September 7]. Available from: https://www.imf.org/-/media/files/publications/loi/imported-cpidpdfs/external/pubs/ft/scr/2007/cr07316.ashx.

IMF. Senegal: Poverty Reduction Strategy Paper Annual Progress Report. 2010 [cited 2023 September 73. 6]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Annual-Progress-Report-24516.

74. IMF, World Bank. Senegal: Poverty Reduction Strategy Paper—Joint Staff Advisory Note. 2007 [cited 2023 September 8]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Joint-Staff-Advisory-Note-21334</u>.

75. IMF, World Bank. Senegal - Joint IDA-IMF staff advisory note and the second poverty reduction strategy paper progress report. 2008 [cited 2023 September 19]. Available from:

https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/842141468106140771/senegal-joint-ida-imf-staff-advisory-note-and-the-second-povertyreduction-strategy-paper-progress-report.

76. World Bank. Senegal - First, Second, and Third Poverty Reduction Support Credits. 2008 [cited 2024 June 13]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/106551468305075940/senegal-first-second-and-third-poverty-reduction-support-credits. 77. IMF. Third and Fourth Reviews Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility and Request for Waiver of Performance Criteria. 2005 [cited 2023 September 8]. Available from: https://www.imf.org/-/media/Websites/IMF/imported-full-text-pdf/external/pubs/ft/scr/2006/ cr06127.ashx.

78. IMF. Senegal: 2004 Article IV Consultation and Second Review Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility and Requests for Waise of New Second Review of New Second Review of the Second Review of New S

Waiver of Nonobservance of Performance Criteria and Rephasing of the

Arrangement—Staff Report; Press Releases on the Executive Board Discussion; and

Statement by the Executive Director for Senegal. 2006 [cited 2023 September 8]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Staff-Report-for-the-2004-Article-IV-Consultation-and-Second-Review-Under-the-Three-19483.

79. IMF. Senegal: First Review Under the Three-Year Arrangement Under the Poverty Reduction and Growth Facility, and Requests for Waiver of Performance Criteria and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2004 [cited 2023 September 11]. Available from: https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-First-Review-Under-the-Three-Year-Arrangement-Under-the-Poverty-Reduction-and-Growth-17381.

80. Republic of Senegal. Senegal: Poverty Reduction Strategy Paper Progress Report. 2004 [cited 2023 September 11]. Available from: <u>https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/pubs/ft/scr/2005/cr05114.ashx</u>.

81. Republic of Senegal. Senegal: Poverty Reduction Strategy Paper—Second Annual Progress Report. 2004 [cited 2023 September 8]. Available from: <u>https://www.imf.org/-/media/files/publications/loi/imported-cpid-pdfs/external/pubs/ft/scr/2006/cr0668.ashx</u>.

82. IMF. Senegal: Selected Issues and Statistical Appendix. 2005 [cited 2023 September 11]. Available from: <a href="https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Selected-Issues-and-Statistical-Appendix-18246">https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Selected-Issues-and-Statistical-Appendix-18246</a>.

83. IMF. Senegal: Poverty Reduction Strategy Paper-Annual Progress Report—Joint Staff Advisory Note. 2004 [cited 2023 September 11]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/31/Senegal-Poverty-Reduction-Strategy-Paper-Annual-Progress-Report-Joint-Staff-Advisory-Note-18165.

84. IMF, World Bank. Poverty Reduction Strategy Paper Joint Staff Assessment. 2002 [cited 2023 September 11]. Available from: <u>https://www.imf.org/external/np/jsa/2002/sen/eng/120202.pdf</u>.

85. Republic of Senegal. Poverty Reduction Strategy Paper. 2002 [cited 2023 September 12]. Available from: <u>https://www.imf.org/external/np/prsp/2002/sen/01/100502.pdf</u>.

86. World Bank. Senegal - Poverty Reduction Strategy Paper (PRSP) second annual progress report. 2005 [cited 2023 September 19]. Available from: <u>https://documents.worldbank.org/en/publication/documents-reports/documentdetail/458201468103463175/senegal-poverty-reduction-strategy-paper-prsp-second-annual-progress-report.</u>

87. World Bank. Senegal - Poverty Reduction Strategy Paper Annual Progress Report and Joint IDA-IMF Staff Assessment of the PRSP. 2004 [cited 2023 September 19]. Available from:

https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/826341468107946235/senegal-poverty-reduction-strategy-paper-annual-progress-reportand-joint-ida-imf-staff-assessment-of-the-prsp.

88. World Bank. Program Information Document - Concept Stage: Senegal - Poverty Reduction Support Credit (PRSC 2). 2005 [cited 2024 July 3]. Available from:

https://documents1.worldbank.org/curated/en/888621468777602940/text/PID010Concept0Stage.txt. 89. World Bank. Program Information Document (PID) Appraisal Stage. 2005 [cited 2024 July 7]. Available

from: <a href="https://documents1.worldbank.org/curated/en/134311468304193996/pdf/PID010Appraisal0Stage.pdf">https://documents1.worldbank.org/curated/en/134311468304193996/pdf/PID010Appraisal0Stage.pdf</a>. 90. World Bank. First Poverty Reduction Support Credit. 2004 [cited 2024 July 2]. Available from: <a href="https://documents1.worldbank.org/curated/en/449131468760175785/text/283320SN0original.txt">https://documents1.worldbank.org/curated/en/134311468304193996/pdf/PID010Appraisal0Stage.pdf</a>. 90. World Bank. First Poverty Reduction Support Credit. 2004 [cited 2024 July 2]. Available from: <a href="https://documents1.worldbank.org/curated/en/449131468760175785/text/283320SN0original.txt">https://documents1.worldbank.org/curated/en/449131468760175785/text/283320SN0original.txt</a>.

91. IMF, World Bank. Senegal: Enhanced Heavily Indebted Poor Countries (HIPC) Initiative Completion Point Document. 2004 [cited 2023 September 11]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Enhanced-Heavily-Indebted-Poor-Countries-HIPC-Initiative-Completion-Point-Document-17380.

92. IMF, World Bank. Decision Point Document under the Enhanced Heavily Indebted Poor Countries(HIPC) Initiative. 2000 [cited 2024 July 7]. Available from:

https://documents1.worldbank.org/curated/en/433661583532267858/pdf/Senegal-Enhanced-Heavily-Indebted-Poor-Countries-HIPC-Debt-Initiative.pdf. 93. World Bank. Development Credit Agreement (HIV/AIDS Prevention and Control Project). 2002 [cited 2024 July 17]. Available from:

https://documents1.worldbank.org/curated/en/151421468106481346/pdf/B74DF895D60239A785256F03001850 14.pdf

94. IMF. Senegal: 2002 Article IV Consultation and Requests for a Three-Year Arrangement Under the Poverty Reduction and Growth Facility and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2003 [cited 2023 September 11]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-2002-Article-IV-Consultation-and-Requestsfor-a-Three-Year-Arrangement-Under-the-16631.

95. IMF. Senegal: Second Review Under the Third Annual Arrangement Under the Poverty Reduction and Growth Facility and Requests for Waiver of Performance Criteria, and for Additional Interim Assistance Under the Enhanced Initiative for Heavily Indebted Poor Countries. 2002 [cited 2023 September 12]. Available from: <u>https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Second-Review-Under-the-Third-Annual-Arrangement-Under-the-Poverty-Reduction-and-15777.</u>

96. IMF. Senegal: Staff Report for the 2001 Article IV Consultation, First Review Under the Third Annual Arrangement Under the Poverty Reduction and Growth Facility, and Request for Waiver of Performance Criteria. 2001 [cited 2023 September 12]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Staff-Report-for-the-2001-Article-IV-Consultation-First-Review-Under-the-Third-15430.

97. IMF. Republic of Senegal—Letter of Intent, Memorandum on Economic and Financial Policies for 2001, Technical Memorandum of Understanding. 2001 [cited 2023 September 12]. Available from:

https://www.imf.org/en/Publications/CR/Issues/2016/12/30/Senegal-Staff-Report-for-the-2001-Article-IV-Consultation-First-Review-Under-the-Third-15430.

98. Republic of Senegal. Senegal: Enhanced Structural Adjustment Facility Economic and Financial Policy Framework Paper (1999-2001). 1999 [cited 2023 September 13]. Available from: https://www.imf.org/external/np/pfp/1999/senegal/index.htm.

99. Republic of Senegal, IMF, World Bank. Senegal: Enhanced Structural Adjustment Facility Policy Framework Paper, 1998-2000. 1998 [cited 2023 September 14]. Available from:

https://www.imf.org/external/np/pfp/senegal/senge00.htm

100. IMF. IMF Financial Data Query Tool. 2022 [cited 2022 February 8]. Available from: https://www.imf.org/external/np/fin/tad/query.aspx.

101. World Bank. Memorandum of the president of the International Development Association to the executive directors on a Country Assistance Strategy of the World Bank Group for the Republic of Senegal. 1997 [cited 2023 September 20]. Available from: <u>https://documents.worldbank.org/en/publication/documents-reports/documentdetail/624001468307173324/</u>.

102. World Bank. Memorandum of the president of the International Development Association to the executive directors on a country assistance strategy of the World Bank Group for the Republic of Senegal.1995 [cited 2023 September 20]. Available from:

https://documents1.worldbank.org/curated/en/972171468169158047/pdf/multi-page.pdf.

103. World Bank. Development Credit Agreement (Endemic Disease Control Project). 1997 [cited 2024 July
17]. Available from: <u>https://documents.worldbank.org/pt/publication/documents-</u>

reports/documentdetail/902761468104967553/conformed-copy-c2951-endemic-disease-control-projectdevelopment-credit-agreement.

104. World Bank. Development Credit Agreement (Integrated Health Sector Development Program). 1997 [cited 2024 July 17]. Available from:

https://documents1.worldbank.org/curated/en/532951468304777075/pdf/3A5F3840AED9BD9C85256F0200824F 92.pdf

105. World Bank. Senegal - Public expenditure review. 1993 [cited 2023 September 20]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/825081468103462556/senegal-public-expenditure-review.

106. World Bank. Senegal - Macroeconomic update report. 1993 [cited 2023 September 20]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/145871468306282080/senegal-macroeconomic-update-report.

Tahari A, Vrijer Jd, Fouad M. Adjustment for Growth: The African Experience. 1996 [cited 2022 February 3]. Available from: <u>https://www.elibrary.imf.org/view/books/084/00149-9781557755667-en/C4.xml</u>.
 IMF. The Composition of Fiscal Adjustment and Growth: Lessons from Fiscal Reforms in Eight Economies. 1997 [cited September 15 2023]. Available from:

https://www.elibrary.imf.org/display/book/9781557756299/9781557756299.xml.

109. IMF. Economic Adjustment and Reform in Low-Income Countries. 1997 [cited 2023 September 15]. Available from: <a href="https://www.elibrary.imf.org/display/book/9781557757159/ch08.xml">https://www.elibrary.imf.org/display/book/9781557757159/ch08.xml</a>.

110. World Bank. Senegal - An economy under adjustment (Vol. 2) : Main report. 1987 [cited 2023 September 21]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/744271468113087188/main-report

111. World Bank. Senegal - Country economic memorandum. 1984 [cited 2023 September 21]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/901191468335438265/senegal-country-economic-memorandum.

112. World Bank. Human Resources Development Project. 1991 [cited 2024 July 16]. Available from: <u>https://documents1.worldbank.org/curated/en/933341468107951172/pdf/10B0ADAAC91EB7F685256F02008245</u> 19.pdf.

113. World Bank. Development Credit Agreement (Rural Health Project). 1983 [cited 2024 July 17]. Available from: https://documents1.worldbank.org/curated/en/754991468058791015/pdf/Credit-1310-Senegal-Rural-Health-Project-Credit-Agreement.pdf.

114. IMF. Senegal: Background Papers and Statistical Appendix. 1995 [cited 2023 September 14]. Available from: <u>https://www.elibrary.imf.org/view/journals/002/1995/071/article-A001-en.xml</u>.

115. World Bank. Senegal - Second Structural Adjustment Program Project. 1986 [cited 2023 September 21]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/867321468105873397/senegal-second-structural-adjustment-program-project. 116. World Bank. Loan Agreement (Structural Adjustment Loan) between Republic of Senegal and International Bank for Reconstruction and Development. 1981 [cited 2023 September 22]. Available from: https://documents1.worldbank.org/curated/en/845781468058508059/pdf/Loan-1931-Senegal-Structural-Adjustment-Loan-And-Credit-Loan-Agreement-1.pdf.

117. World Bank. Senegal - Economic trends and prospects : The main report (English). 1979 [cited 2023 September 22]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>reports/documentdetail/309431468107088736/the-main-report.

118. World Bank. Senegal - The economy (Vol. 4) : Planning methods and plan implementation. 1973 [cited 2023 September 22]. Available from: https://documents.worldbank.org/en/publication/documents-

reports/documentdetail/159681468105849319/planning-methods-and-plan-implementation

119. World Bank. Senegal - The economy : The main report. 1973 [cited 2023 September 22]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u>

reports/documentdetail/675231468307171165/the-main-report.

120. World Bank. Senegal - The current economic situation and prospects of Senegal : Main report. 1970 [cited 2023 September 22]. Available from: <u>https://documents.worldbank.org/en/publication/documents-</u> reports/documentdetail/636701468104941050/main-report.

121. World Bank. Current economic position and prospects of Senegal. 1968 [cited 2023 September 25]. Available from: <u>https://documents1.worldbank.org/curated/en/720391468335437727/pdf/multi0page.pdf</u>.

122. World Bank. The Economy of Senegal. 1966 [cited 2023 September 25]. Available from: https://documents1.worldbank.org/curated/en/767641468107090623/pdf/multi0page.pdf.

123. xe.com. XDR to USD Chart. 2023 [cited 2023 September 11]. Available from: https://www.xe.com/currencycharts/?from=XDR&to=USD&view=10Y.